## STUDIES ON THE REGULATION OF APOLIPOPROTEIN E GENE EXPRESSION IN MACROPHAGES

BY

KIRSTY RHIAN GREENOW B.Sc. (HONS)



A THESIS PRESENTED FOR THE DEGREE OF DOCTOR OF PHILOSOPHY AT CARDIFF UNIVERSITY

SEPTEMBER 2004

CARDIFF SCHOOL OF BIOSCIENCES BIOMEDICAL SCIENCES BUILDING CARDIFF UNIVERSITY MUSEUM AVENUE CARDIFF CF10 3US UMI Number: U487422

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



UMI U487422

Published by ProQuest LLC 2013. Copyright in the Dissertation held by the Author. Microform Edition © ProQuest LLC. All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code.



ProQuest LLC 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106-1346

## **CONTENTS**

DECLARATION	I
Contents	11
Acknowledgements	VII
Abstract	VIII
Abbreviations	IX
CHAPTER ONE - GENERAL INTRODUCTION	1
1.1. Atherosclerosis	2
1.1.1. LIPOPROTEINS AND ATHEROSCLEROSIS	3
1.1.2. MACROPHAGES AND ATHEROSCLEROSIS	6
	10
1.2. APOLIPOPROTEIN E	10 11
1.2.1. Reverse cholesterol transport 1.2.2. Apolipoprotein E and atherosclerosis	15
	15
1.2.3. OTHER ANTI-ATHEROGENIC FUNCTIONS OF APOLIPOPROTEIN E	10
1.2.4. BIOCHEMICAL CHARACTERISTICS OF APOLIPOPROTEIN E	-
1.2.4.1. THE APOLIPOPROTEIN E GENE	19
1.2.4.2. TRANSCRIPTIONAL REGULATION OF APOLIPOPROTEIN E	19
1.2.4.2. <i>i</i> . Proximal apoE regulatory elements and factors	20
1.2.4.2. II. Distal regulatory elements and factors	21
1.2.4.3. POLYMORPHISMS OF APOLIPOPROTEIN E	22
1.2.4.4. PROTEIN SYNTHESIS AND SECRETION	23
1.2.4.5. STRUCTURE OF APOLIPOPROTEIN E	25
1.2.4.5.i. N-terminal domain of apoE	25
1.2.4.5. <i>II. C-terminal domain of apoE</i>	26
1.2.4.6. MODULATION OF APOLIPOPROTEIN E EXPRESSION	27
1.2.4.6. <i>i</i> . Role of cytokines on macrophage apoE	27
1.2.4.6.II. Effect of sterol loading on macrophage apoE	28
1.2.4.6.III. Regulation of apoE by cholesterol acceptors	29
1.3. Liver X receptors	31
1.3.1. LXR agonists	31
1.3.2. LXR antagonists	33
1.3.3. LXR PROTEIN STRUCTURE	33
1.3.4. LXR AND RXR	35
1.3.5. LXR AND COREGULATORS	36
1.3.6. LXR and target genes	38
1.3.6.1. LXRS AND THE CONTROL OF CHOLESTEROL HOMEOSTASIS	39
1.3.6.1.1. Cholesterol synthesis	40
1.3.6.1.II. Dietary cholesterol absorption	40
1.3.6.1.III. Reverse cholesterol transport	41
The ABC transporters	41
Apolipoprotein E	42
Lipoprotein remodelling enzymes	43
1.3.6.1.iv. Bile acid synthesis	45
1.3.6.2. LXRs and the control of fatty acid synthesis	46
1.3.7. REGULATION OF LXR EXPRESSION	47
1.3.7.1. HUMAN LXRa AUTOREGULATION	48
1.3.7.2. LXRa regulation by PPARs	49

	CONTENTS
1.3.8. LXRs and atherosclerosis	50
1.3.9. LXRs and inflammation	52
<b>1.4.</b> Aims of the study	53
CHAPTER TWO- MATERIALS AND METHODS	56
2.1. Materials	57
2.1.1. Plasmids	60
2.1.2. BACTERIAL CULTURE MEDIA	60
2.1.3. Stock solutions	60
2.1.4. PREPARATION OF CERAMICS, PLASTICS AND GLASSWARE	64
2.2. Isolation and manipulation of DNA	65
2.2.1. PREPARATION OF COMPETENT CELLS	65
2.2.2. TRANSFORMATION OF COMPETENT CELLS	65
2.2.3. SMALL-SCALE DNA PREPARATION (MINIPREP PROTOCOL)	65
2.2.4. LARGE-SCALE DNA PREPARATION (MAXIPREP PROTOCOL)	66
2.2.5. QUANTIFICATION OF DNA	66
2.2.6. RESTRICTION ENDONUCLEASE DIGESTION OF DNA	66
2.2.7. GEL ELECTROPHORESIS OF DNA	67
2.3. TISSUE CULTURE AND RELATED TECHNIQUES	67
2.3.1. MAINTENANCE OF CELL LINES IN CULTURE	67
2.3.2. SUB-CULTURING OF CELLS	67
2.3.3. Human primary monocyte-derived macrophages	68
2.3.3.1. Isolation of primary monocyte-derived macrophages	68
2.3.3.2. CULTURE ROUTINE OF HUMAN PRIMARY MONOCYTE-DERIVED MACROPHAGES	68
2.3.4. DELIPIDATION OF HI-FCS	68
2.3.5. TRANSIENT TRANSFECTION OF THE THP-1 AND U937 CELL LINES	69
2.3.5.1. TRANSIENT TRANSFECTION OF CELLS USING SUPERFECT <sup>TM</sup>	69
2.3.5.2. Transient transfection of cells using Effectene™	69
2.3.5.3. Transient transfection of cells using the DEAE-dextran mammalian	69
TRANSFECTION KIT	70
2.3.6. DIFFERENTIATION OF MONOCYTIC CELL LINES	70
2.3.7. TREATMENT OF CELLS WITH EXTRACELLULAR MEDIATORS	70
2.3.8. TREATMENT OF CELLS WITH OXYSTEROL LIGANDS	70
2.3.9. TREATMENT OF CELLS WITH INHIBITORS	
2.3.10. PREPARATION OF CELL EXTRACTS FOR THE DETERMINATION OF REPORTER GENE	71
ACTIVITY	71
2.3.11. MEASUREMENT OF LUCIFERASE ACTIVITY	71
2.3.12. Measurement of $\beta$ -galactosidase activity	71
2.3.13. TRYPAN BLUE EXCLUSION ASSAY	
	72
2.4. RNA-RELATED TECHNIQUES	72
2.4.1. RNA ISOLATION	72
2.4.2. ANALYSIS OF RNA ON DENATURING AGAROSE GELS	73
2.4.3. QUANTIFICATION OF RNA	73
2.4.4. REVERSE TRANSCRIPTASE- POLYMERASE CHAIN REACTION (RT-PCR)	73
2.4.4.1. CDNA SYNTHESIS FROM TOTAL RNA	73
2.4.4.2. POLYMERASE CHAIN REACTION	75
2.4.5. TAQMAN REAL-TIME QUANTITATIVE RT-PCR	

	CONTENTS
	75
2.5. PROTEIN ANALYSIS	75
2.5.1. PREPARATION OF WHOLE CELL EXTRACTS	77
2.5.2. DETERMINATION OF PROTEIN CONCENTRATION	77
2.5.3. IMMUNOPRECIPITATION OF PROTEINS	77
2.5.4. SDS-POLYACRYLAMIDE GEL ELECTROPHORESIS (SDS-PAGE)	78
2.5.5. WESTERN BLOTTING	78
2.5.6. IMMUNODETECTION OF PROTEINS	79
2.5.7. Enzyme-linked immunosorbent assay (ELISA)	
	80
2.6. PROTEIN KINASE ACTIVITY ASSAYS	80
2.6.1. CASEIN KINASE 2 ASSAY	80
2.6.2. AKT KINASE ASSAY	81
2.6.3. JNK kinase assay	81
CHAPTER THREE - STUDIES ON APOLIPOPROTEIN E PROMOTER ACTIVITY IN RESPONSE TO PMA	82
3.1. INTRODUCTION	83
3.2. Results	85
3.2.1. MORPHOLOGICAL ANALYSIS OF THP-1 CELLS DURING PMA-INDUCED DIFFERENTIATION	
INTO MACROPHAGES	85
3.2.2. Analysis of ApoE mRNA and protein expression during PMA-induced	
DIFFERENTIATION OF THP-1 MONOCYTES INTO MACROPHAGES	87
3.2.3. TRANSFECTION ANALYSIS OF APOE PROMOTER ACTIVITY DURING PMA-INDUCED	
DIFFERENTIATION OF MONOCYTES INTO MACROPHAGES	89
3.3. Discussion	100
<u>CHAPTER FOUR - EFFECT OF PHARMALOGICAL INHIBITORS ON THE LXR-MEDIATED INDUCTION</u> OF APOE	103
4.1. INTRODUCTION	104
4.2. Results	108
4.2.1. EFFECT OF LXR LIGAND, 22(R)-HYDROXYCHOLESTEROL ON APOE MRNA EXPRESSION IN	
THP-1 macrophages	108
4.2.2. EFFECT OF LXR LIGAND, 22(R)-HYDROXYCHOLESTEROL ON APOE PROTEIN EXPRESSION	
IN THP-1 MACROPHAGES	111
4.2.3. EFFECT OF LXR LIGAND, 22(R)-HYDROXYCHOLESTEROL ON APOE SECRETION IN THP-1	440
	112
4.2.4. EFFECT OF INHIBITORS ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF APOE MRNA EXPRESSION IN THP-1 MACROPHAGES	113
4.2.5. EFFECT OF INHIBITORS ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF	115
APOE PROTEIN EXPRESSION IN THP-1 MACROPHAGES	116
4.2.6. EFFECT OF INHIBITORS ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF	117
APOE SECRETION IN THP-1 MACROPHAGES	119
4.2.7. EFFECT OF 22(R)-HYDROXYCHOLESTEROL AND CELL SIGNALLING INHIBITORS ON APOE	
PROTEIN EXPRESSION IN HUMAN PRIMARY MONOCYTE-DERIVED MACROPHAGES	120
4.2.8. EFFECT OF 22(R)-HYDROXYCHOLESTEROL AND CELL SIGNALLING INHIBITORS ON ABCA1	
AND ABCG1 MRNA EXPRESSION IN THP-1 MACROPHAGES	122

	CONTENTS
4.2.9. Effect of 22(R)-hydroxycholesterol and cell signalling inhibitors on LXR $\alpha$ mRNA expression in THP-1 macrophages	122
4.3. Discussion	126
4.5. DISCUSSION <u>CHAPTER FIVE ROLE OF JNK/SAPK MAP KINASE SIGNALLING IN THE LXR-MEDIATED</u> REGULATION OF APOE	130
	131
5.1. INTRODUCTION	135
<b>5.2. Results</b> 5.2.1. Effect of dominant negative expression plasmids on the LXR-mediated induction of apoE in THP-1 macrophages	135
5.2.2. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON JNK/SAPK PHOSPHORYLATION IN THP-1 MACROPHAGES	137
5.2.3. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON JNK/SAPK ACTIVITY IN THP-1 MACROPHAGES	139
5.2.4. EFFECT OF CURCUMIN AND SP600125 ON THE LXR-MEDIATED INDUCTION OF JNK/SAPK ACTIVITY IN THP-1 MACROPAHGES	<b>140</b>
5.2.5. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED INDUCTION OF JNK/SAPK ACTIVITY IN THP-1 MACROPHAGES	141
5.2.6. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON C-JUN PHOSPHORYLATION IN THP-1 MACROPHAGES	144
5.2.7. EFFECT OF CURCUMIN AND SP600125 ON THE LXR-MEDIATED PHOSPHORYLATION OF C-JUN IN THP-1 MACROPHAGES	145
5.2.8. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED PHOSPHORYLATION OF C-JUN IN THP-1 MACROPHAGES	147
5.2.9. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON ATF-2 PHOSPHORYLATION IN THP-1 MACROPHAGES	148
5.3. Discussion	149
CHAPTER SIX - ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION	154
OF APOE	155
6.1. INTRODUCTION 6.2. RESULTS	158
6.2.1. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT SER473 PHOSPHORYLATION IN THP-1 MACROPHAGES	158
6.2.2. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED PHOSPHORYLATION OF AKT SER473 IN THP-1 MACROPHAGES	160
6.2.3. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT THR308 PHOSPHORYLATION IN THP-1 MACROPHAGES	161 162
6.2.4. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT ACTIVITY IN THP-1 MACROPHAGES 6.2.5. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED INDUCTION OF AKT ACTIVITY IN THP-1 MACROPHAGES	<sup>7</sup> 163
6.2.6. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON PDK1 PHOSPHORYLATION IN THP-1 MACROPHAGES	164
6.2.7. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON GSK-3 PHOSPHORYLATION IN THP-1 MACROPHAGES	166
6.2.8. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON FKHR PHOSPHORYLATION IN THP-1 MACROPHAGES	167

	CONTENTS
6.2.9. EFFECT OF CK2 DOMINANT NEGATIVE EXPRESSION PLASMIDS ON THE LXR-MEDIATED INDUCTION OF APOE IN THP-1 MACROPHAGES 6.2.10. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON CK2 ACTIVITY IN THP-1 MACROPHAGES	168 171
6.2.11. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON PTEN PHOSPHORYLATION IN THP-1 MACROPHAGES	173
6.3. Discussion	174 177
Chapter Seven – General Discussion	178
7.1. Overview of the results presented in this thesis	182
7.2. WIDER PERSPECTIVES OF THE NOVEL FINDINGS IN THIS THESIS	183
7.3. Future studies	185
7.4. CONCLUDING REMARKS	186
References	235
Appendices	253
PUBLICATIONS ARISING AS A RESULT OF THIS THESIS	

#### **ACKNOWLEDGEMENTS**

The completion of this thesis would not have been possible without the help and support of so many people. I would firstly like to thank my supervisors Dr Dipak P. Ramji and Dr Nigel Pearce, for their constant guidance and advice throughout all stages of this PhD.

I would also like to thank Dr John Yates for all his help during my time at Glaxo and especially for his help with the Taqman and ELISA work.

I also want to thank Dr Timothy Hughes, who always made me look on the brighter side, and taught me to question everything. Without his patience, help and conversation I would never have made it this far and would probably be doing something exciting instead! Cheers! Thank you for everything you've taught me.

I'd also like to thank the other members of the DPR group that I've had the pleasure to work with the last four years. Firstly, thanks to Scott for always helping me when the computers or any other technical piece of equipment would not cooperate and always having an explanation for everything. I'd also like to thank Peli, both you and greek ways have made me laugh and scream the last couple of years, thank you for all your questions, I can see that Tim has trained you well. A big thank you must also go to the other fabulous, and not to mention glamorous ladies of the lab. Saira, Liz, and Sanj whose friendship the last couple of years has got me through a variety of bad experiments and has made lab life not only bearable but enjoyable. Thank you Saira for the constant girly-ness you've supplied the last couple of years. A special thanks must also go to my bench buddy, Sanj, who has had to put up with my constant complaints and non-science questions, day in and day out for the last three years. I'm amazed you're still sane! Thank you also to Nishi, my fellow apoE survivor, Sarah, Mimi and all the other members of the DPR group past and present. I'd also like to thank lant, an honorary member of the DPR group, I've thoroughly enjoyed our chats the last couple of years. A big thanks also has to go to the other members of the W201 study area and anybody else that has had to put up with my complaints about science the last couple of years.

Finally, last but not least, I would like to thank the men in my life, Matthew and Thomas for putting up with me the last four years. Thank you for always being so understanding, never complaining and always making me smile.

#### ABSTRACT

Apolipoprotein E (apoE) plays a vital role in lipid metabolism and reverse cholesterol transport, and is highly expressed in the atherosclerotic plaque, where it is mainly associated with macrophage-derived foam cells. Macrophage apoE has several anti-atherogenic functions including key roles in the maintenance of macrophage lipid homeostasis and cellular cholesterol efflux. ApoE may therefore be considered as a potential therapeutic target in the prevention or treatment of atherosclerosis and it is therefore essential that a detailed understanding is obtained of the regulation of apoE in macrophages.

The initial aim of the project was to determine, in detail, the mechanism(s) involved in the induction of apoE expression during macrophage differentiation and to identify the *cis*-acting elements in the apoE promoter that were responsible for this upregulation. These aims required the optimisation of a macrophage-based transfection system, which allowed the analysis of several DNA constructs containing regions of the apoE promoter and the enhancer regions, ME1 and ME2. However, whilst this system was able to demonstrate the previously reported TNF- $\alpha$  induction of the apoE promoter, no such induction was observed during PMA-induced differentiation, indicating that the expression of apoE during macrophage differentiation may require other sequences in addition to the promoter and enhancer sequences.

Consequently, it was decided to investigate the regulation of apoE by the LXR subfamily of nuclear receptors. LXRs have atheroprotective properties due to their ability to upregulate the expression of genes involved in the reverse cholesterol transport process. ApoE is a direct target of these ligand-mediated transcription factors and a clearer understanding of the LXR-mediated transcription of apoE would aid the development of potential therapeutic agents for atherosclerosis. Therefore, the involvement of signal transduction pathways in the LXR-mediated regulation of apoE in macrophages was investigated.

Through the use of commercially available inhibitors, we firstly identified a novel role for the JNK/SAPK MAPK, PI3K and CK2 pathways in the LXR-mediated induction of apoE mRNA, protein and secretion in human THP-1 macrophages. This inhibition of apoE induction was also shown to occur in human primary monocyte-derived macrophages and further investigations demonstrated the potential role of these cell signalling pathways in the LXR-mediated regulation of ABCA1, ABCG1 and LXRα. It was also found, for the first time, that treatment of THP-1 macrophages with the oxysterol LXR ligand, 22(R)-hydroxycholesterol, induced JNK phosphorylation and kinase activity, and the subsequent phosphorylation of the c-jun transcription factor. Treatment of THP-1 macrophages with the LXR ligand also resulted in the phosphorylation and activation of Akt, a downstream component of PI3K signalling. In addition CK2 activity was found to be increased in THP-1 macrophages treated with 22(R)-hydroxycholesterol.

In conclusion, the studies presented in this thesis demonstrated, for the first time, an important role for the JNK/SAPK MAPK, PI3K and CK2 pathways in the activation of macrophage apoE gene expression by the LXR subfamily of nuclear receptors. A potential role for these cell signalling pathways was also implicated in the LXR-mediated regulation of ABCA1, ABCG1 and LXRa.

## **ABBREVIATIONS**

ABC	ATP binding cassette
ACAT-1	acyl-coA: cholesterol acyltransferase-1
ACAT-T AF-2	activation function-2
AP-1	
	activator protein-1
apigenin	4', 5, 7-trihydroxyflavone
apo APS	apolipoprotein
AR	ammonium persulphate
ASC-2	androgen receptor
ASC-2 ATF-2	activating signal cointegrator-2
ATP	activating transcription factor-2
	adenosine triphosphate
bFGF	basic fibroblast growth factor
bp	base pairs
BSA	bovine serum albumin
cAMP	cyclic adenosine monophosphate
CBP	CREB-binding protein
cdk	cyclin dependent kinase
cDNA C/EBP	complementary deoxyribonucleic acid
CETP	CCAAT/enhancer binding protein
Ci	cholesteryl ester transfer protein
CK2	Curie(s) Casein kinase 2
CMV	
COX-2	cytomegalovirus
CREB	cyclooxygenase
C-terminal	cAMP response element binding protein carboxyl terminal
curcumin	1, 7-bis(4-hydroxy-3-methoxyphenyl)-1, 6-heptadiene-3, 5-dione
CYP7A1	cholesterol 7α-hydroxylase
Da	Daltons
DAG	diacylglycerol
dATP	deoxyadenosine triphosphate
DBD	DNA binding domain
dCTP	deoxycytidine triphosphate
DEAE	
dGTP	diethylaminoethyl deoxyguanosine triphosphate
DIPE	di-isopropyl ether
DMSO	dimethyl sulphoxide
DN	dominant negative
DNA	deoxyribonucleic acid
DNase	deoxyribonuclease
dNTP	deoxynucleoside triphosphate
DRIP	vitamin D3 receptor interacting proteins
DTT	dithiothreitol
ECL	enhanced chemi-luminescence
EDTA	ethylenediaminetetracetic acid
EGF	epidermal growth factor
EMSA	electrophoretic mobility shift assay
ER	estrogen receptor
ERK	extracellular signal-regulated kinase
FCS	foetal calf serum
	toetal cait serum

FGF	fibroblast growth factor
g	grams
g	gravity
GAPDH	glyceraldehyde-3-phosphate dehydrogenase
GM-CSF	granulocyte/macrophage-colony stimulating factor
GR	glucocorticoid receptor
GRIP1	glucocorticoid receptor-interacting protein-1
GSK-3	glycogen synthase kinase-3
h	hour(s)
	.,
HAT	histone acetyltransferase
HCR	hepatic control region
HDL	high density lipoprotein
HEPES	N-2-hydroxyethylpiperazine-N'-2-ethanesulphonic acid
HI-FCS	heat-inactivated foetal calf serum
HL	hepatic lipase
HRP	horse radish peroxidase
HSPG	heparin sulphate proteoglycans
ICAM	intracellular adhesion molecule
IDL	intermediate density lipoprotein
IFN	interferon
IL	interleukin
inos	inducible nitric oxide synthase
JNK	c-jun N-terminal kinase
Kb	kilobase(s)
kDa	kilo Dalton(s)
LB	Luria Bertani
LBD	ligand binding domain
LCAT	Lecithin: cholesterol acyltransferase
LDL	low density lipoprotein
LDL-R	low density lipoprotein receptor
LPL	lipoprotein lipase
LPS	lipopolysaccharide
LRP	low density lipoprotein receptor-related protein
LXR	liver X receptors
LXRE	LXR response element
LY294002	•
	2-(4-morpholinyl)-8-phenyl-4H-1-benzopyran-4-one
MAPK	mitogen-activated kinase
MCP	monocyte chemoattractant protein
M-CSF	macrophage-colony stimulating factor
ME	multienhancer
min	minutes
MIP	macrophage inflammatory protein
MMP	matrix metalloproteinase
MMLV	Maloney Murine Leukemia Virus
MOPS	3-(N-morpholino)-propanesulphonic acid
mRNA	messenger ribonucleic acid
mTOR	mammalian target of rapamycin
NCoA	nuclear receptor co-activator
NCoR	nuclear receptor co-repressor
NF-ĸB	• •
NG	nuclear factor for kappa light chain in B cells
	nitric oxide
N-terminal	amino-terminal

OD	optical density
ONPG	o-nitrophenyl β-D-galactopyranoside
oxLDL	oxidised LDL
PAGE	polyacrylamide gel electrophoresis
PBS	phosphate buffered saline
P/CAF	
	p300/CBP-associated factor
PCR	polymerase chain reaction
PD98059	2'-amino-3'-methoxyflavone
PDGF	platelet-derived growth factor
PEG	polethyleneglycol
PI3K	phosphoinositide-3-kinase
PIP	phosphatidylinositol phosphate
PK	protein kinase
PLTP	phospholipids transfer protein
PMA	phorbol 12-myristate 13-acetate
PMSF	phenylmethylsulphonyl fluoride
PPAR	peroxisome proliferator-activated receptor
PR	progesterone receptor
PVDF	polyvinylidene fluoride
RAR	retinoic acid receptor
RNA	ribonucleic acid
RNase	ribonuclease
RSV	Rous sarcoma virus
RT	reverse transcriptase
RXR	retinoid X receptor
SAPK	stress-activated protein kinase
SB202190	4-(4-fluorophenyl)-2-(4-methylsulphinylphenyl)-5-(4-pyridyl)1H-
с <b>л</b>	imidazole
SD SDS	standard deviation
	sodium dodecyl sulphate
sec SEK	second(s) SAPK/ERK kinase
SMC	smooth muscle cell(s)
SMRT	silencing mediator for RXR and TR
SP600125	
SR	anthra[1,9-cd]pyrazol-6(2H)-one scavenger receptor
SRC	steroid receptor co-activator
SREBP	sterol response element binding protein
SWI/SNF	switch/sucrose non-fermenting
TAF	TBP associated factor
TBE	Tris-borate-ethylenediaminetetracetic acid
TBP	TATA-binding protein
TE	Tris-ethylenediaminetetracetic acid
TEMED	N, N, N, N-tetramethylethylenediamine
TG	triglycerides
TGF	transforming growth factor
TGRP	triglyceride-rich particles
TNF	tumour necrosis factor
TR	thyroid receptor
TRAP	thyroid hormone receptor associated protein
U	unit
V	volts

v/v	volume to volume
VCAM	vascular cell adhesion molecule
VEGF	vascular endothelial growth factor
VLDL	very low density lipoprotein
WCE	whole cell extract
w/v	weight to volume

1

# **CHAPTER ONE – GENERAL INTRODUCTION**

#### **1.1. ATHEROSCLEROSIS**

Atherosclerosis is a slow progressive disease of the arteries and is the major underlying cause of heart attack and stroke, which accounts for 50% of all deaths in Western societies. The disease is characterised by the accumulation of lipids, fibrous elements and blood constituents in the intima of large arteries. Epidemiological studies have revealed several important environmental and genetic risk factors related to the increased incidence of this disease, including smoking, hypertension, diabetes mellitus, stress, physical inactivity and hypercholesterolaemia. The development of atherosclerotic lesions involves a complicated sequence of events in which various cell types, including monocyte-derived macrophages, T cells and the normal cellular elements of the arterial wall, such as smooth muscle cells and collagen, interact to form a complicated atherosclerotic plaque (Ross, 1999).

Atherosclerosis can be considered to be a form of chronic inflammation caused by a primarily protective mechanism that is triggered after injury to the endothelial lining of the arterial wall. Injury to the endothelium may be induced by several possible risk factors, including hypertension and raised plasma lipid levels, and results in changes in the permeability of the arterial wall, the increased expression of several cell surface adhesion molecules, and the production of cytokines (Ross, 1999; Glass and Witzum, 2001). These changes in endothelial function result in an increased migration of monocytes and T-lymphocytes from the circulation into the intima of the arterial wall. Once monocytes enter the intima, they differentiate into macrophages and begin to accumulate cell debris, lipoproteins and oxidatively modified lipoproteins, resulting in the development of lipid-laden foam cells (Lusis, 2000; Ross, 1999). These activated foam cells then amplify their cellular invasion by secreting their own growth factors and cytokines. This early atherosclerotic lesion consisting of cholesterol-engorged macrophages in the intima of the arterial wall is termed the fatty streak. The fatty streak progresses to an intermediate fibrofatty lesion, which consists of several layers of foam cells and smooth muscle cells (SMCs) with T-lymphocytes, surrounded by a relatively poorly developed matrix of connective tissue (Ross, 1999). The transition from the fatty streak to the more complex lesion is characterised by the migration of smooth muscle cells from the medial layer of the artery wall into the intima. Intimal smooth muscle cells proliferate and take up modified lipopoproteins, contributing to foam cell formation, and synthesise extracellular matrix proteins that lead to the development of the fibrous cap (Ross, 1999). Further increase of the lesion severity subsequently results in the formation of an advanced fibrous lesion that is characterised by a dense fibrous cap, which protrudes into the arterial lumen impeding the flow of blood, and covers a core of macrophages, smooth muscle cells, Tlymphocytes, extracellular lipid and necrotic material (Ross, 1999; Lusis, 2000). Instability of the

atherosclerotic lesion may then lead to rupture of the plaque and the induction of secondary haemorrhage and thrombosis, which may ultimately lead to occlusion of the artery.

#### **1.1.1. LIPOPROTEINS AND ATHEROSCLEROSIS**

Among the various genetic and environmental risk factors that have been identified, elevated levels of cholesterol in the blood seems to be unique due to its ability to drive the development of atherosclerosis in humans and experimental animals, even in the absence of other known risk factors (Glass and Witzum, 2001). Serum cholesterol is carried by several lipoprotein particles, which are responsible for the transport of dietary and endogenously produced lipids. Lipoproteins are spherical macromolecular particles that comprise of a core of hydrophobic cholesteryl esters and triglycerides surrounded by a shell of phospholipids and unesterified free cholesterol. The phospholipids in the lipoprotein shell are polarized and the charged head groups face the hydrophilic blood compartment, while the hydrophobic fatty acid chains are in contact with the core lipids. In addition to the phospholipids and the unesterified free cholesterol, the lipoprotein shell also contains a number of specific proteins, called apolipoproteins, that play a major role in the regulation of lipoprotein metabolism and the stabilisation of the lipoprotein particles.

Five major classes of lipoproteins can be distinguished, including chylomicrons, very low density lipoproteins (VLDL), intermediate density lipoproteins (IDL), low density lipoproteins (LDL) and high density lipoproteins (HDL). These different classes of lipoproteins differ in properties such as their size, electrophoretic mobility, composition of lipids and apolipoproteins, all of which effect their interaction with different tissues and their metabolism. Each class also vary in their function; chylomicrons are primarily responsible for the transport of dietary lipids, while VLDL, LDL and HDL function to transport endogenous lipids.

VLDL particles are synthesised by the liver from cholesterol and triglycerides derived from either *de novo* synthesis or lipoprotein uptake (Gibbons, 1990). VLDL particles contain apolipoprotein B-100 (apoB-100), apolipoprotein CI, II and III, and apolipoprotein E (apoE). These particles are responsible for the transport of fatty acids to peripheral tissues such as adipose tissue and muscle. In the plasma, the triglycerides in VLDL are subject to lipolysis by lipoprotein lipase (LPL), which is bound to the capillary endothelium of skeletal muscle, cardiac muscle and adipose tissue (Goldberg, 1996). The free fatty acids are taken up, mainly by adipocytes where they are stored in triglyceride droplets. Progressive hydrolysis of core triglycerides of VLDL, is also accompanied by transfer of phospholipids by phospholipid transfer protein (PLTP) and apolipoproteins to HDL. After triglyceride removal in the peripheral tissues, a proportion of the remaining VLDL remnants are rapidly removed from the circulation by the high-affinity binding sites on parenchymal liver cells, where the cholesterol is either used in membrane or lipoprotein

biosynthesis or excreted as free cholesterol or bile acids. VLDL remnant particles that are not endocytosed by the liver are further processed by the enzyme hepatic lipase and converted into LDL particles. The conversion of VLDL remnants into LDL is accompanied by further removal of core triglycerides and dissociation of all apolipoproteins apart from apoB-100.

Although 70% of the formed LDL is taken up by hepatic LDL receptors, it is mainly LDL particles in humans that are responsible for the delivery of cholesterol to the peripheral tissues. Therefore, high LDL cholesterol levels are associated with increased risk for cardiovascular disease. The uptake of LDL by cells occurs via LDL receptors that recognise an N-terminal domain of apoB-100, its sole lipoprotein. The circulating level of LDL is determined in large by its rate of uptake through the hepatic LDL receptor pathway, as evidenced by the fact that lack of functional LDL receptors is responsible for the massive accumulation of LDL in patients with homozygous familial hypercholesterolaemia (Goldstein and Brown, 1977). Control of LDL receptor levels through transcription of the LDL receptor gene is regulated by a family of membrane-bound transcription factors called sterol regulatory element binding proteins (SREBPs), which are in turn regulated by the cholesterol lead to activation of SREBP-2, which stimulates the transcription of the LDL receptor gene and other genes involved in cholesterol biosynthesis (Brown and Goldstein, 1999).

The recruitment of macrophages to the intima of the arterial wall and their subsequent uptake of LDL-derived cholesterol are the major cellular events contributing to fatty streak formation (Figure 1.1). Native LDL is not taken up by macrophages rapidly enough to generate foam cells and several studies have suggested that oxidative modification of the lipid and apoB-100 component of LDL may drive the initial formation of fatty streaks (Navab et al., 1996). Oxidation of LDL by free radicals and lipoxygenases occurs in the arterial wall and becomes prevalant when levels of circulating LDL are raised. LDL diffuses passively through endothelial cell junctions to the subendothelial matrix, and its retention in the vessel wall involves the interaction of the apoB component of LDL with matrix proteoglycans (Schwenke and Carew, 1989; Boren et al., 1998). Studies have demonstrated that mice lacking 12/15-lipoxygenase have considerably diminished atherosclerosis, suggesting that this enzyme may be an important source of reactive oxygen species in LDL oxidation (Cyrus et al., 1999). Also evidence that inducible nitric oxide synthase (iNOS) contributes to LDL oxidation *in vivo* has recently been provided by studies demonstrating that inhibitors of iNOS decrease atherosclerosis in rabbits (Behr-Roussel et al., 2000; Detmers et al., 2000). The specific properties of oxidized LDL, depends on the extent of modification, which can range from minimal to extensive. Minimal modification (mmLDL), allows the LDL particle to be recognised by the LDL receptor as normal (Navab et al., 1996), whereas extensive modification of LDL (oxLDL) results in the particle being bound by scavenger receptors expressed on the surface of macrophages and smooth muscle cells. The inability of the extensively modified LDL particle to bind to the LDL receptor is due to the apoB-100 component of LDL being fragmented and the lysine residues of the particle becoming covalently modified with reactive breakdown products of oxidized lipids (Steinberg *et al.*, 1989).



#### FIGURE 1. 1. MECHANISM CONTRIBUTING TO FOAM-CELL FORMATION.

LDL penetrates into the artery wall where it binds to proteoglycans. These interactions are thought to trap the LDL particles and increase their susceptibility to oxidation. Enzymes contributing to LDL oxidation include lipoxygenases, myeloperoxidase and iNOS. VLDL particles are subject to modification by LPL and the resulting remnants are subject to trapping by the proteoglycans, oxidative modification and uptake by macrophages. Macrophage uptake of modified lipoproteins occurs via the scavenger receptors CD36 and SR-A. VLDL, very low density lipoproteins; LDL, low density lipoproteins; mmLDL, minimally modified LDL; oxLDL, oxidised LDL; 15-LO, 15lipoxygenase; iNOS, inducible nitric oxide synthase; MPO, myeloperoxidase; NADPHox, NADPH oxidase; LPL, lipoprotein lipase; SRA, scavenger receptor A. See text for further explanation. Figure taken from Li and Glass (2002).

LDL also undergoes other types of modifications such as non-enzymatic glycation, enzymatic degradation and aggregation, all of which, consequently, generates a wide range of oxidation-specific neo-epitopes, which renders the modified LDL immunogenic and leads not only to a cellular immune response but a humoral response also (Horkko *et al.*, 2000). Also lipoprotein accumulation in the artery wall is a strong stimulus to the activation of several inflammatory pathways and VLDL and other triglyceride-containing particles are able to activate the transcription factor, nuclear factor- $\kappa$ B (NF $\kappa$ B) [Dichtl *et al.*, 1999], which increases the expression of several inflammatory genes, including cyclooxygenase-2 (COX-2) [Belt *et al.*, 1999].

#### **1.1.2. MACROPHAGES AND ATHEROSCLEROSIS**

Monocytes and macrophages are present at all stages of atherosclerotic lesion development, from the fatty streak to the more complicated atherosclerotic plague. The turnover of macrophages in the atherosclerotic plaque is the net effect of monocyte infiltration, local proliferation, efflux, and cell death. Monocytes are derived from pluripotent haematopoietic stem cells in the bone marrow and migrate into the blood circulation 24h after production (van Furth, 1989). Once they enter the blood circulation they either remain in the blood stream or migrate into tissues and body cavities where they differentiate into macrophages. Macrophages play an important role in protecting the vascular wall from injury and in ordinary inflammatory reactions macrophages scavenge debris, such as cytotoxic and proinflammatory oxLDL particles or apoptotic cells and subsequently remove the debris by migrating away from the site of injury into the blood circulation. Although the recruitment of monocytes to the arterial wall and their subsequent differentiation into macrophages, initially serves a protective function, the progressive accumulation of macrophages and their uptake of oxLDL, ultimately results in a prolonged inflammatory response and an increase in the severity of the atherosclerotic lesion. The importance of macrophages in the development of atherosclerosis has been illustrated by the osteopetrotic mouse, which carries a naturally occurring mutation in the gene encoding macrophage colony stimulating factor (M-CSF), an important factor involved in the proliferation and survival of monocytes. These mice exhibit a near complete absence of macrophages, and when bred to apoE-deficient mice they are extremely resistant to the development of atherosclerosis, despite an increase in circulating cholesterol levels (Smith et al., 1995a; Qiao et al., 1997).

The development of atherosclerosis is initiated by the adhesion of monocytes to the endothelial cell layer of the vessel wall. The recruitment of monocytes to lesion-prone sites of large arteries is regulated by several cell adhesion molecules that are expressed on the surface of the endothelial cells in response to inflammatory stimuli. Initial adhesion involves selectins, which mediates a rolling interaction, which is followed by a firmer attachment by means of integrins (Figure 1.2). Vascular cell adhesion molecule-1 (VCAM-1) was the first cell adhesion molecule to be implicated in the initial recruitment of monocytes, due to its increased expression on endothelial cells over lesion-prone areas in response to cholesterol feeding and also its pattern of regulation by pro-inflammatory cytokines such as interleukin-1 $\beta$  (IL-1 $\beta$ ) and tumour necrosis factor- $\alpha$  (TNF- $\alpha$ ) [Cybulsky and Gimbrone, 1991]. Although targeted deletion of VCAM-1 in mice causes embryonic lethality, studies where hypomorphic variants of VCAM-1 have been introduced into atherosclerosis-susceptible mice (apoe-/-) have demonstrated a reduction in atherosclerotic lesion formation (Cybulsky *et al.*, 2001). In addition to VCAM-1, both E-selectin and P-selectin appear to play quantitative roles in monocyte adhesion, as apoE-deficient mice lacking both genes show a

#### CHAPTER ONE - GENERAL INTRODUCTION

40-60% decrease in atherosclerosis (Dong *et al.*, 1998). Similarly, a reduction in monocyte recruitment to atherosclerotic lesions can be seen in apoE-deficient mice with a deletion of the gene encoding intracellular adhesion molecule-1 (ICAM-1) [Collins *et al.*, 2000].



#### FIGURE 1.2. MONOCYTE RECRUITMENT AND DIFFERENTIATION.

Circulating monocytes attach to endothelial cells by cell adhesion molecules that are induced in response to inflammatory signals. Selectins mediate low-affinity interactions that permit rolling, whereas integrins mediate firm adhesion. Monocytes migrate through the endothelial layer into the intima, where they differentiate into macrophages in response to locally produced factors such as M-CSF. This process of differentiation is accompanied by the upregulation of scavenger receptors. VCAM-1, vascular cell adhesion molecule-1; ICAM-1, intercellular adhesion molecule-1; MCP-1, monocyte chemotactic protein-1; CCR-2, MCP-1 receptor; oxLDL, oxidised LDL; M-CSF, macrophage colony stimulating factor; SR-A, scavenger receptor A. See text for further explanation. Figure taken from Li and Glass (2002).

Adherent monocytes then migrate through endothelial cell junctions into the subendothelial space under the influence of chemoattractant molecules. The migration of monocytes into the intima of the arterial wall is likely to be stimulated in part by oxLDL, which is able to directly attract monocytes (Steinberg *et al.*, 1989) and induce the expression of chemotactic molecules by endothelial cells, such as monocyte chemotactic protein-1 (MCP-1) [Navab *et al.*, 1996]. Recent work using compound mutant mice lacking MCP-1 or its receptor CCR2, which are susceptible to atherosclerosis (apoE-/- or LDLR-/- mice), have shown striking decreases in monocyte accumulation and local lipid levels (Gu *et al.*, 1998; Boring *et al.*, 1998). IL-8, which is present in atherosclerotic lesions, may also play a role in the migration of cells into the intima, as mice lacking CXCR2, a high affinity receptor for IL-8 results in significantly reduced risk of atherosclerosis (Boisvert *et al.*, 1998).

Once monocytes have entered the intima, they differentiate into macrophages and start to accumulate massive amounts of cholesteryl esters resulting in the development of cholesterol-filled foam cells. This accumulation of cholesterol by macrophages is thought to be mediated primarily by the uptake of modified forms of LDL via scavenger receptors (Yamada et al., 1998). Two proteins shown to significantly contribute to this process, are scavenger receptor A (SR-A) and CD36 as apoE-deficient mice lacking SR-A or CD36 develop significantly less atherosclerosis than control apoE-deficient mice (Suzuki et al., 1997). OxLDL taken up by macrophage scavenger receptors is delivered to lysosomes, where its cholesteryl ester content is hydrolysed to free cholesterol and fatty acids. This free cholesterol has a number of potential metabolic fates, including esterification and storage in the lipid droplets that characterise foam cells. The cholesteryl ester stores of macrophages have been shown to undergo a continuous cycle of hydrolysis (to free cholesterol) and re-esterification (Brown et al., 1980). The hydrolytic step of this cycle is mediated by a neutral cytoplasmic cholesteryl ester hydrolase, which is responsive to cyclic-AMP (Khoo et al., 1981) and the re-esterification step is mediated by acyl CoA: cholesterol acyltransferase-1 (ACAT-1) [Brown et al., 1980]. As intracellular cholesterol levels increase, the proteolytic activation of SREBP transcription factors required for cholesterol biosynthesis and LDL receptor expression (Brown and Goldstein, 1999) are inhibited. Although this prevents the further accumulation of cholesterol via these pathways, cholesterol is still taken up via scavenger receptors or phagocytic mechanisms and cholesterol homeostasis can not be maintained. As mammalian cells do not have the ability to degrade the sterol ring the majority of the excess sterols can only be eliminated from the body by biliary excretion. Macrophages and other cell types must export cholesterol to extracellular acceptors for transport to the liver. Therefore, mechanisms mediating cholesterol efflux are critical for the maintenance of cholesterol homeostasis in the macrophage.

The macrophage has two potential mechanisms for disposing of excess cholesterol: enzymatic modification to the more soluble forms and efflux via membrane transporters. The enzyme 27-hydroxylase is expressed in macrophages at relatively high levels and may play a role in cholesterol excretion by converting it to the more soluble 27-oxygenated steroids that can be readily accepted by albumin (Babiker *et al.*, 1997). The major mechanism for cholesterol efflux however, is reverse cholesterol transport, where HDL serves as the primary extracellular acceptor. There is also evidence that macrophages may contribute directly to the availability of extracellular cholesterol levels through the secretion of apolipoprotein E (apoE) [Linton *et al.*, 1995], which contributes to the formation of HDL particles.

#### **CHAPTER ONE - GENERAL INTRODUCTION**

During lesion development, significant cross-talk occurs among the cellular elements of the plaque. Macrophages, endothelial cells, T cells and smooth muscle cells become active during atherogenesis and secrete a variety of proteins. A wide range of the activated macrophage products are involved in the initiation and progression of the lesion (Table 1.1) and although all of the macrophage secretory products indicate the importance of macrophages in atherogenesis, discussion of all these products is beyond the scope of this thesis, therefore the remainder of this chapter will discuss in detail the role of apolipoprotein E in the development of atherosclerosis.

#### TABLE 1.1. MONOCYTE/MACROPHAGE PRODUCTS AND THEIR POTENTIAL ROLE IN ATHEROGENESIS

Product	FUNCTION IN ATHEROGENESIS	
SECRETED PROTEINS		
Apolipoprotein E Interleukin-1	Lipoprotein metabolism and cholesterol efflux Adhesion molecules and SMC migration and proliferation	
Interleukin-6 Interleukin-8	Activation of coagulation Recruitment of lymphocytes and endothelial cells	
M-CSF	Monocyte proliferation, survival, and modulation of gene expression	
TNF-α	▲ Adhesion molecules, monocyte attraction, SMC proliferation	
MCP-1	Monocyte chemoattraction	
ENZYMES AND INHIBITORS	eres lever south and he would be been as the	
Collagenase	Matrix degradation, cell migration and plaque rupture	
Complement C3b	Prothrombotic	
Factor XIIIa	Prothrombotic	
Gelatinases (MMP-2/MMP-9) Lipoprotein lipase	Matrix degradation, cell migration and plaque rupture Alteration of lipoprotein metabolism	
15-Lipoxygenase	Lipid oxidation and foam cell formation	
Metalloproteinases	Matrix degradation and plaque rupture	
Myeloperoxidase	Formation of cytotoxins and reactive oxygen species	
CELL-ASSOCIATED PROTEINS		
VLA-4	Monocyte/macrophage activation and retention	
LDL-receptor-related protein	Lipoprotein uptake and protease clearance	
LDL receptor	Lipoprotein uptake	
VLDL receptor	Lipoprotein uptake	
Scavenger receptors	Lipoprotein uptake, monocyte recruitment and adhesion	
Thrombin receptor Tissue factor	Monocyte/macrophage activation Coagulation and thrombosis	
13300 1000	Coaguiation and thrombosis	

M-CSF, macrophage colony stimulating factor; TNF- $\alpha$ , tumour necrosis factor- $\alpha$ ; MCP-1; monocytic chemotactic protein-1; MMP, matrix metalloproteinase; VLA-4, alpha 4 beta 1 integrin; LDL, low density lipoprotein; VLDL, very low density lipoprotein;  $\blacktriangle$ , increase. Table derived from van Eck (1999).

#### **<u>1.2. APOLIPOPROTEIN E</u>**

Apolipoprotein E (apoE) is a 34kDa protein with a mature form comprising of 299 amino acids (Rall et al., 1982). Initially apoE was termed the "arginine-rich apoprotein" and was first identified in 1973 by Shore and Shore as a lipoprotein constituent of triglyceride-rich VLDL (Shore and Shore, 1973). ApoE has since been found to be a major protein constituent of several plasma lipoproteins that carry dietary- and liver-derived cholesterol, including very low density lipoproteins (VLDL), chylomicrons and a sub-population of high density lipoproteins (HDL) (Mahley et al., 1984). Most of the apoE found circulating in the body is synthesised by the liver, although 20-40% is derived from extra-hepatic sources (Newman et al., 1985), such as macrophages. One important role of apoE is in lipoprotein metabolism, where it is the primary determinant for the receptor-mediated catabolism of all triglyceride-rich lipoproteins, apart from LDL. ApoE mediates the uptake and degradation of lipoproteins via its high affinity binding to the low-density lipoprotein receptor (LDL-R) on parenchymal liver cells. ApoE is an extremely efficient ligand for the receptor, as it has a 25-fold greater affinity for the receptor than LDL itself (Innerarity and Mahley, 1978), in which apoB-100 is the ligand. The importance of the interaction of apoE with the LDL receptor is underscored by the massive accumulation of lipoproteins and lipoprotein remnants in the plasma of patients with type III hyperlipoproteinaemia, where an apoE variant is expressed, which is defective in LDL receptor binding (Weisgraber et al., 1982).

ApoE also binds to a second hepatic receptor involved in lipoprotein metabolism, the low density lipoprotein receptor-related protein (LRP) which has been implicated in the clearance of chylomicron remnants (Herz and Willnow, 1995), a process which has been shown to be dependant on apoE (Linton *et al.*, 1998). Evidence for the existence of the remnant receptor comes from the ability of individuals, with familial hypercholesterolaemia, that have defective or absent LDL receptors to clear chylomicron remnants from their plasma (Mahley, 1988). The clearance of chylomicron remnants by apoE binding to the LRP, involves the ability of apoE to bind to heparan sulfate proteoglycans (HSPG). This binding of apoE to HSPGs serves a secretion-capture role in lipoprotein remnant metabolism as apoE-enriched remnants appear to bind to HSPG before it interacts with LRP on the hepatic surface (Ji *et al.*, 1994a). The requirement for HSPG binding by apoE has been confirmed in studies using heparinase treated cells and the infusion of heparinase in mice (Ji *et al.*, 1993; Ji *et al.*, 1994b; Ji *et al.*, 1995). In addition, lipoproteins are required to be artificially enriched in apoE in order to be active *in vitro* (Kowal *et al.*, 1989).

The HSPG-LRP pathway involves a three-step process, involving sequestration, processing, and uptake (Mahley and Ji, 1999). Initially, lipoprotein particles are sequestered in the hepatic space of Disse, which is rich in apoE; there the particles become enriched with apoE, which interacts with the HSPG molecules anchored on the hepatic surface (Ji *et al.*, 1994a). The

retained lipoproteins then undergo lipolytic events involving either lipoprotein-bound LPL or cellsurface bound hepatic lipase (Ji *et al.*, 1994c; Ji *et al.*, 1997). Finally, the "secretion-capture" role of apoE facilitates the interaction of lipoproteins with the LDL receptor, the LRP receptor, or HSPG acting as a receptor. It appears that lipoproteins containing apoE are either transferred from HSPG to the LRP and then taken up as a ligand-receptor complex by endocytosis or are taken up as an HSPG-LRP-lipoprotein complex together by the hepatocytes (Ji *et al.*, 1997). The HSPG-LRP system has been shown to function *in vivo* by several studies (Ji *et al.*, 1995; Willnow *et al.*, 1994; Hussain *et al.*, 1991) and although the LDL receptor-mediated pathway is sufficient for clearing lipoprotein remnants during fasting, the LRP/HSPG-mediated pathway is thought to be required for efficient clearance of post-prandial lipoprotein remnants (Mahley and Ji, 1999).

#### **1.2.1. REVERSE CHOLESTEROL TRANSPORT**

In addition to the role of apoE in the metabolism of lipoproteins, apoE plays a vital role in cholesterol efflux. The elimination of cholesterol from the body occurs mainly through the liver via bile formation. Therefore, it is essential that excess cholesterol in peripheral tissues is transported back to the liver for excretion, via a process involving HDL, called "reverse cholesterol transport" (Glomset, 1968). Nascent preB-HDL, a discoidal apoA-I phospholipid particle, which is secreted by the intestine or the liver, is a potent acceptor of excess cholesterol from peripheral tissues. Once HDL has taken up free cholesterol from peripheral cells, the cholesterol becomes esterified to cholesteryl esters by the enzyme lecithin-cholesterol acyltransfrease (LCAT) and nascent HDL is converted to small, spherical HDL<sub>3</sub> (Figure 1.3). Subsequently HDL<sub>3</sub> is converted into HDL<sub>2</sub> by acquirement of phospholipids and apolipoproteins that are released into the interstitial fluid during the lipolysis of triglycerides in chylomicrons or VLDL by LPL. ApoE, which is present in the interstitial fluid due to secretion by various cell types, has avidity for HDL as it becomes cholesterolenriched and studies have also shown that the presence of apoE actually facilitates HDL uptake of cholesterol. For example, macrophages derived from apoE-/- mice exhibit a decreased capacity to efflux cholesterol and other lipids to HDL<sub>3</sub>, lipid-free apoAI or lipid-free apoE3 compared with those from wild-type mice (Langer et al., 2000; Havek et al., 1994).



#### FIGURE 1.3. SCHEMATIC ILLUSTRATION OF REVERSE CHOLESTEROL TRANSPORT.

HDL, high density lipoprotein; IDL, intermediate density lipoprotein; VLDL, very low density lipoprotein; LDL, low density lipoprotein; HL, hepatic lipase; LCAT, lecithin-cholesterol acyltransfrease; CETP, cholesteryl ester transfer protein; SR-BI, scavenger receptor BI; LDL-R, low density lipoprotein receptor. See text for further details.

The return of peripheral cholesterol to the liver can occur via three different pathways. The first is by the direct uptake of the whole HDL particle by the LDL receptors of the liver, via apoE binding. The second pathway is indirect but appears to be more important in humans than the direct ligand role of apoE in HDL (Tall, 1993). This pathway occurs via the transfer of the cholesteryl esters in HDL to other lipoproteins, such as VLDL, IDL or LDL by the cholesteryl ester transfer protein (CETP), these lipoproteins are then subsequently taken up by the liver, where apoE behaves as a ligand. The anti-atherogenic properties of this CETP/HDL pathway is confirmed by recent data that demonstrate that CETP-deficiency is associated with increased atherosclerosis in humans, despite increased HDL levels (Bruce et al., 1998). Also over-expression of CETP in mice that normally lack CETP activity is anti-atherogenic (Bruce et al., 1998). The third pathway of reverse cholesterol transport is the selective delivery of cholesteryl esters in HDL to the liver via scavenger receptor class B1 (SR-B1) [Acton et al., 1996]. SR-B1, which is expressed mainly in the liver and steroidogenic tissues, binds HDL with great affinity, and mediates the selective delivery of cholesteryl esters to the liver without internalisation and degradation of the HDL particle. The direct uptake of cholesteryl ester from HDL is facilitated by apoE-mediated binding of apoE-containing HDL to cell surface HSPG (Arai et al., 1999). Subsequently, this apoE-HSPG interaction enhances the ability of the cell surface protein SR-BI to mediate the delivery of HDL-cholesteryl esters to cells. A direct interaction of cell surface apoE with these receptors may not be required; only the close proximity of apoE as a sterol acceptor may be needed although

recent studies have shown that apoE contains an SR-BI binding sequence (Li *et al.*, 2002a). The importance of SR-BI in the reverse cholesterol transport process has been demonstrated by the inactivation of this gene in mice, which was associated with a dramatic increase in serum cholesterol levels and the size of HDL, without influencing the HDL protein concentration (Rigotti *et al.*, 1997). Also liver-specific overexpression of this protein in mice resulted in reduced HDL levels and an increase in reverse cholesterol transport (Wang *et al.*, 1998).

The process of cholesterol efflux is particularly important in maintaining cholesterol homeostasis in cells that are incapable of limiting their uptake of lipids, such as macrophages. This removal of free cholesterol from macrophages by specific cholesterol acceptors is vital in the prevention of foam cell formation. Reverse cholesterol transport is an important pathway in maintaining the cholesterol balance of macrophages and the antiatherogenic properties of apoE are largely attributed to its ability to deplete foam cell macrophages of cholesterol. Various studies have confirmed the involvement of apoE in macrophage cholesterol efflux. It has been extensively reported that secretion of apoE and cholesterol from macrophages occurs via independent but concurrent mechanisms and it was thought that apoE facilitates cholesterol efflux by improving the cholesterol accepting properties of HDL (Mazzone, 1996; Basu et al., 1983; Dory, 1989; Mazzone and Reardon, 1994). Work by Zhu et al. (1998) has determined that reduced atherosclerosis in human apoE transgenic mice, compared to apoE deficient mice was due to the ability of the transgenic apoE to associate with  $\alpha$ -migrating HDL, increasing its capacity to accept cholesterol. Also the capacity of apoE-depleted HDL from humans or mice to promote cholesterol efflux from mouse peritoneal macrophages is decreased and can be restored to normal by the addition of exogenous apoE (Basu et al., 1982; Hayek et al., 1994). The expression of endogenous apoE also enhances cholesterol efflux in macrophages (Zhang et al., 1996; Mazzone and Reardon, 1994; Smith et al., 1996; Kruth et al. 1994), and the experimental characteristics of cholesterol efflux in such cases can be distinguished from that produced by the exogenous addition of lipid free apoE (Lin et al., 1998). The precise mechanisms by which endogenous apoE stimulates cholesterol efflux currently remains unclear although its retention at the cell surface via association with proteoglycans makes a contribution (Lin et al., 2001). ApoE arrives at the cell surface in a relatively lipid-poor state and conformation changes attained during lipidation may cause its release (Lin et al., 2001). ApoE held in proximity to the plasma membrane by proteoglycan binding could facilitate passive desorption of lipid from the plasma membrane by a process that may involve SR-BI or ATP-binding cassette transporter A1 (ABCA1) (Krieger, 1999; Orso et al., 2000). Indeed, it has previously been shown that apoE is an acceptor for free cholesterol and phospholipid released from macrophages by ABCA1 (Smith et al., 1996) and work by Krimbou et al., (2004) has demonstrated that lipid-free apoE is able to interact with ABCA1 in vitro. It may be possible that the

interactions of apoE with ABCA1 leads to the lipidation of apoE through a membrane microsolubilisation process whereby there is simultaneous release of phospholipids and free cholesterol to apoE, as recently suggested for the lipidation of apoAI (Liu *et al.*, 2003). This is in agreement with the idea that the transfer of phospholipids and cholesterol from the active site of ABCA1 transporter to apoE molecule weakens the interaction of apoE/ABCA1 and causes disocciation of the lipidated apoE product. However, it is worth noting that apoE may be the primary driver of cholesterol efflux in macrophages through a pathway independent of ABCA1 expression or activity (Huang *et al.*, 2001).

In addition to the ability of apoE to facilitate HDL cholesterol efflux, apoE is able to directly stimulate hepatic VLDL and triglyceride production as demonstrated by studies using apoE-/-mouse hepatocytes that have impaired secretion of VLDL triglycerides (Kuipers *et al.*, 1997). Also hepatocytes stably transfected with human apoE showed a two-fold increase in VLDL triglyceride secretion with increasing levels of human apoE (Huang *et al.*, 1998) and apoE transgenic mice lacking murine apoE with inhibited lipolysis showed increased VLDL production of 50% compared with that of non-transgenic controls (Huang *et al.*, 1998). Plasma apoE concentration is also an important determinant for VLDL metabolism which indirectly affects HDL levels. Studies using apoE-expressing transgenic mice have shown that the apoE:apoCII ratio is a major factor in determining whether VLDL is a good substrate for lipoprotein lipase (Huang *et al.*, 1998) and increased levels of apoE may increase triglycerides by displacing the lipoprotein lipase cofactor apoCII and inhibiting lipolysis. This increase in apoE and triglyceride levels reduces HDL levels, which is probably due to the inhibition of VLDL lipolysis, which reduces the availability of surface components for HDL production.

Another role of apoE in reverse cholesterol transport, separate to its role as a high-affinity ligand for the various lipoproteins, is in the activation of enzymes involved in lipoprotein metabolism such as HL, CETP and LCAT. The hydrolysis of phospholipid monolayers by HL in apoE-enriched HDL was found to be enhanced as compared to apoE-poor HDL, indicating that apoE activates HL (Thuren *et al.*, 1992). ApoE also enhances the cholesteryl and triglyceride transfer between VLDL and HDL via CETP by enhancing the affinity of CETP for VLDL (Kinoshita *et al.*, 1993), and the C-terminal domain of apoE has been reported to play a role in the activation of LCAT (De Pauw *et al.*, 1995). ApoE is also thought to indirectly facilitate LPL-mediated hydrolysis of triglycerides in VLDL as apoE-mediates binding to the VLDL receptor, which co-localizes with LPL (Takahashi *et al.*, 1995; Niemeier *et al.*, 1996).

#### **1.2.2. Apolipoprotein E and atherosclerosis.**

Interest in apoE has mainly arisen from the observation that apoE is an atheroprotective protein. Depressed expression of apoE in humans leads to a pro-atherogenic lipoprotein profile associated with increased risk for atherosclerosis (Schaefer et al., 1986; Ghiselli et al., 1981). In addition, apoE knockout (apoE-/-) mice are severely hypercholesterolemic compared to the wild-type counterparts (500mg/dl compared with 60-90 mg/dl), and the increase in cholesterol is largely distributed in the lower density lipoproteins due to their impaired clearance (Plump et al., 1992; Zhang et al., 1992 and 1994a; Nakashima et al., 1994). ApoE deficient mice also loose their normal resistance to cholesterol feeding and their cholesterol levels can rise to greater than 2000mg/dl (Plump *et al.*, 1992; Zhang *et al.*, 1992 and 1994a; Nakashima *et al.*, 1994). This plasma accumulation of remnant lipoproteins results in the development of complex atherosclerotic lesions, which develop even when mice are fed a low fat diet (Plump et al., 1992; Zhang et al., 1992 and 1994a; Nakashima et al., 1994; Reddick et al., 1994; Piedrahita et al., 1992). In addition, atherosclerosis in apoE-null mice has been shown to be prevented by increasing circulating apoE levels through recombinant adenovirus-mediated apoE gene transfer to the liver (Kashyap et al., 1995). The decrease in atherosclerosis seen in this model was accompanied by decreased total cholesterol and VLDL/IDL levels (Kashyap et al., 1995) suggesting that apoE prevents atherosclerosis by lowering plasma cholesterol levels.

ApoE expression is absent in normal vessels, but during atherosclerosis apoE is expressed in the atherosclerotic plaque and although it can enter the artery wall from the periphery, it is mainly thought to be synthesised locally by resident macrophages (Rosenfeld *et al.*, 1993; O'Brien *et al.*, 1994). Macrophage apoE plays a direct role in preventing or delaying foam cell formation, and lack of macrophage apoE leads to induction and progression of atherosclerosis despite the abundance of apoE in plasma and on lipoproteins (Fazio *et al.*, 1997). Studies by Boisvert and Curtiss (1999) confirm that the majority of apoE in the lesion is synthesised locally by lesion macrophages and established that the expression of this macrophage-derived apoE in the lesion can retard the development of atherosclerosis in male mice fed a high fat diet. Several other *in vivo* studies have demonstrated that the secretion of apoE by macrophages exerts an important protective effect. Selective expression of apoE in macrophages of apoE-/- mice through bone marrow transplantation or transgenic expression decreases hypercholesterolaemia and reduces lesion development (Linton *et al.*, 1995). Conversely, the transplantation of apoE-/- bone marrow into apoE +/+ mice confers increased susceptibility to atherosclerosis (Fazio *et al.*, 1997).

Several studies have shown that macrophage-derived apoE exerts anti-atherogenic properties largely independent of its effects on plasma lipid levels (Fazio *et al.*, 1997; Boisvert and Curtiss, 1999; Van Eck *et al.*, 2000). For example, work by Zhang *et al.* (1994a) demonstrated that

atherosclerosis was found to be more severe in cholesterol-fed apoE-heterozygous(+/-) mice than in cholesterol-fed apoE(+/+) mice, despite the relative similar plasma cholesterol level in the two groups. It has also been shown that the lack of macrophage apoE in C57BL/6 mice leads to a 10fold increase in diet-induced atherosclerosis in the absence of any changes in serum cholesterol levels or lipoprotein profiles (Fazio *et al.*, 1997; Boisvert and Curtiss, 1999). Moreover, transgenic mice expressing human apoE in the vessel wall show reduced atherosclerotic lesions in the absence of any changes in plasma cholesterol and lipoprotein profile (Shimano *et al.*, 1995). In addition, the expression of low levels of apoE in the artery wall of apoE-deficient mice using a retroviral transduction system leads to reduced early foam cell lesion formation (Hasty *et al.*, 1999). Therefore, although the ability of apoE to influence plasma cholesterol homeostasis and facilitate cellular cholesterol efflux are central to its atheroprotective role, apoE is also thought to protect from the development of atherosclerosis by other independent mechanisms, which will be discussed in more detail below.

#### **1.2.3. OTHER ANTI-ATHEROGENIC FUNCTIONS OF APOLIPOPROTEIN E**

As described above, a number of studies have shown that low expression of apoE in the vessel wall is able to limit the development of atherosclerosis without having any effect on plasma cholesterol levels or lipoprotein profile (Fazio *et al.*, 1997; Boisvert and Curtiss, 1999; Van Eck *et al.*, 2000; Shimano *et al.*, 1995; Bellosta *et al.*, 1995; Hasty *et al.*, 1999; Gough and Raines, 2003). This suggests a potent anti-atherogenic role of apoE that is independent of its action on cholesterol metabolism and transport. Further evidence in support of an alternative anti-atherogenic role for apoE comes from recent work by Thorngate *et al.* (2003) who demonstrated that low-level apoE expression, is able to inhibit atherogenesis using a mechanism separate to its ability to enhance cholesterol efflux via the ABCA1 pathway or by facilitating reverse cholesterol transport via apoAI or apoAI-containing HDL. They also demonstrated that other HDL-dependent cholesterol efflux pathways do not play a role in mediating the atheroprotective effects of low level apoE. Therefore it must be considered that apoE may block atherogenesis by mechanisms that are independent of its ability to normalise plasma cholesterol concentrations.

ApoE has been shown to exhibit local cytokine- and hormonal-like effects on the surrounding cells of the vessel wall, which also contributes to its atheroprotective role. It was originally thought that HDL had anti-platelet activities (Aviram and Brook, 1983) but further research revealed that the anti-aggregatory action of HDL was due to a minor subset of HDL which contained apoE (Riddell *et al.*, 1996). For example, apoE inhibits agonist-induced platelet aggregation, which has been proposed to occur through apoE binding to specific receptors on the platelet surface, which in turn initiates cell signalling events that results in the stimulation of nitric

oxide-synthase (Riddell *et al.,* 1999). The cytoplasmic tail of this cell surface protein, apoE receptor 2 (apoER2) is thought to contain domains important in cellular signalling, including potential targets for cyclic-GMP- and cyclic-AMP-dependant protein kinases and consensus sequences for Src Homology 3 (SH3) recognition. The binding of apoE to apoER2 stimulates endothelial nitric oxide release in an isoform-dependant manner, and it has been proposed that tyrosine phosphorylation of apoER2 initiates phosphoinositide-3-kinase (PI3K) signalling and activation of nitric oxide synthase (Sacre *et al.,* 2003). Nitric oxide (NO) is a potent inhibitor of platelet activation and aggregation and acts through the activation of platelet soluble guanylate cyclase to increase levels of cGMP (Lincoln *et al.,* 1997).

ApoE can also directly modify the T lymphocyte- and smooth muscle cell-mediated inflammatory responses of atherosclerosis. T-lymphocytes have been shown to accumulate locally in complicated human atherosclerotic plaques (Jonasson et al., 1986), and apoE-deficient mice crossed with mice defective in both T and B lymphocyte function show a 42% reduction in atherosclerosis when fed a chow diet (Dansky et al., 1997). However when these same mice are fed cholesterol, lesion formation was unaffected, which indicates that atherosclerosis-prone mice display a prominent immune system component within their lesions and that once early lesions develop, immune responses modulate lesion progression (Dansky et al., 1997). Various studies have shown that apoE can suppress the activation and proliferation of peripheral blood mononuclear cells as well as purified T cells stimulated in vitro with polyclonal mitogens and antigens (Avila et al., 1982; Pepe and Curtiss, 1986; Okano et al., 1985; Cardin et al., 1988; Hui and Harmony, 1980a; Dyer et al., 1991). This antiproliferative effect of apoE is thought to be due to inhibition of critical events such as intracellular calcium accumulation and phosphatidylinositol turnover (Hui and Harmony, 1980a and 1980b), which results in the reduction of IL-2 expression (Kelly et al., 1994). Also apoE was shown to be capable of inhibiting the high affinity IL-2 receptor (IL-2R) on primary IL-2 dependant T cells, which consequently arrests lymphocytes at the G1A/G1B boundary in the cell cycle (Mistry et al., 1995).

Smooth muscle cells (SMCs) also participate in inflammatory responses within the intima, and it was recently reported that apoE inhibits SMC migration and proliferation induced by the inflammatory agonists platelet-derived growth factor (PDGF) and oxLDL (Ishigami *et al.*, 2000). This inhibition of SMC proliferation is thought to be mediated by apoE inhibition of mitogen-activated protein (MAP) kinase activity and a corresponding decrease in cyclin D1 expression (Ishigami *et al.*, 1998). Recent work has also demonstrated that the inhibition of PDGF-stimulated cell migration is mediated via suppression of signal transduction through the LRP. Tyrosine phosphorylation of the cytoplasmic tail of LRP is stimulated by PDGF and this activity is inhibited by apoE (Boucher *et al.*, 2002). There is also evidence that this inhibition of SMC proliferation is partly

due to an apoE-dependant increase in heparan sulfate proteoglycan synthesis (inhibitors of smooth muscle cell proliferation) from endothelial cells within the lesion (Paka *et al.*, 1988).

Another anti-atherogenic property of apoE is its ability to inhibit lipid oxidation (Miyata *et al.*, 1996), this antioxidative property of apoE may influence early events of atherogenesis by preventing the accumulation of oxLDL. Evidence of enhanced oxidative stress has been found in apoE-deficient mice and LDL oxidation-specific epitopes have been identified in aortic lesions of apoE-null mice by immunohistochemical techniques (Palinski *et al.*, 1994). According to Hayek *et al.* (1994) circulating lipoproteins in apoE-deficient mice are more oxidized and more susceptible to oxidation *in vitro* than lipoproteins from the wild-type animals. Although the mechanism is not completely understood, it has been shown that apoE can bind metal ions such as copper and iron, possibly sequestering them and preventing their participation in the oxidation process (Miyata and Smith, 1996).

A fraction of macrophage-secreted apoE is sequestered within the pericellular proteoglycan matrix, where apoE, in addition to influencing the uptake and retention of atherogenic lipoproteins into the vessel wall can modulate the availability of cytokines and growth factors retained within it. ApoE is an effective inhibitor of endothelial cell proliferation and is able to inhibit the chemotactic response of endothelial cells to basic fibroblast growth factor (bFGF) [Vogel *et al.*, 1994], which was partly due to the competition of apoE with growth factors for interactions with cell surface HSPG. ApoE has also been shown to inhibit VCAM-1 expression by endothelial cells (Stannard *et al.*, 2001). Another potential anti-atherogenic role of apoE is its ability to increase triglyceride-rich particle (TGRP) accumulation in the endothelial layer, these particles become bound by the arterial wall matrix which seems to effectively block deep penetration of TGRPs into the arterial wall (Mullick *et al.*, 2002).

Therefore, the above properties of apoE demonstrate how this protein may potentially inhibit atherogenesis without affecting plasma cholesterol levels. Also these characteristics of apoE may explain how low levels of apoE can protect from atherosclerosis. Work by Thorngate *et al.* (2000) using apoE-deficient mice that express only adrenal apoE have revealed that low levels of apoE (< 1% to 2% of wild-type plasma apoE) can effect the progression of atherosclerosis without affecting plasma cholesterol levels. These studies also demonstrated that apoE plasma levels of up to 3% of wild-type levels can be achieved by adrenal expression and that this level is sufficient to normalise total plasma cholesterol levels, indicating that although in wild-type mice the liver is the major source of plasma apoE, apoE protein from any source, when present in the plasma compartment, can contribute to the hepatic uptake of the cholesterol-containing lipoproteins. Also heterozygous apoE+/- mice do not have elevated plasma cholesterol (Zhang *et al.*, 1992) and plasma apoE concentrations of only 40ug/dL generated solely by bone marrow-transplanted

macrophages can effectively lower plasma cholesterol levels (Hasty *et al.*, 1999). Conversely, overexpression of human apoE in apoE-/- mice at levels of more than 30mg/dl leads to hypertriglyceridaemia (Huang *et al.*, 1998) which is due to both an apoE-dependant increase in hepatic VLDL triglyceride production (Mensenkamp *et al.*, 1999) and interference with apoC-IIdependant VLDL lipolysis by LPL (Huang *et al.*, 1998), indicating that an optimal range of apoE exists, in which it can prevent atherogenesis.

#### **1.2.4. BIOCHEMICAL CHARACTERISATION OF APOLIPOPROTEIN E**

### 1.2.4.1. THE APOLIPOPROTEIN E GENE

The apoE gene consists of 3.7 kb of DNA and has been mapped to chromosome 19 where it is located at the 5' end of a 50 kb gene-cluster with apoCl, an apoCl pseudogene, apoCll and apoClV (Allan *et al.*, 1995). The mRNA consists of 1163 bases and the coding region contains four exons separated by three introns. ApoE is produced in most organs and significant quantities of apoE mRNA have been detected in the liver, brain, spleen, lung, ovary, adrenal gland, kidney and muscle, in several different species (Driscoll and Getz, 1984; Elshourbagy *et al.*, 1985). Tissue specific expression of apoE is regulated by multiple independent positive and negative regulatory elements, many of which are located between the apoE and apoC-I gene (Smith *et al.*, 1988; Simonet *et al.*, 1991). The highest expression of apoE is found in the parenchymal liver cells, where its expression is directed by two distal regulatory elements, the hepatic control regions, located downstream of the apoE promoter (Simonet *et al.*, 1993). ApoE mRNA expression has also been demonstrated in the spleen and lungs, where macrophages are most likely to be responsible for its synthesis (Srivastava *et al.*, 1996). The expression of apoE in macrophages is thought to be controlled by two homologous regulatory regions, designated multi-enhancer 1 and 2, both of which are located downstream of the apoE promoter (Shih *et al.*, 2000).

#### 1.2.4.2. TRANSCRIPTIONAL REGULATION OF APOLIPOPROTEIN E

ApoE gene expression is extremely complex due to interactions from a number of proteins that bind to the proximal promoter region as well as downstream elements (Paik *et al.*, 1988; Smith *et al.*, 1988; Simonet *et al.*, 1991,1993; Berg *et al.*, 1996). Expression of the apoE gene varies between different cell types and several potential regulatory regions have been identified that may be involved in basal and tissue-specific control of the gene, although much less is known about the transcription factors that regulate expression.

#### **1.2.4.2.I. PROXIMAL APOE REGULATORY ELEMENTS AND FACTORS**

Little is known about the regulatory elements controlling apoE gene expression in macrophages, although one study by Basheeruddin *et al.* (1994), showed that the –623 to –447 promoter region was necessary for transcriptional activation during 12-O-tetradecanoylphorbol-13-acetate (TPA)-induced differentiation of human THP-1 monocytes into macrophages. This study suggests that AP-1-like proteins were involved, possibly due to the phorbol ester activating the protein kinase C (PKC) pathway which in turn modulates the binding activity of AP-1 transcription factors. A study by Andreani-Mangeney *et al.* (1996) demonstrated that cyclic-AMP (cAMP) inhibits apoE transcription in HepG2 cells, which is mediated by elements located between –614 to –804 in the apoE promoter region. Interestingly, this study also provided evidence of a minimal promoter region, –200/-1, which contains elements that respond positively to cAMP, thereby indicating a more complex regulation by this factor. Indeed, work by Garcia *et al.* (1996) has identified the transcription factor AP-2, which interacts with the –48 to –74 and –107 to –135 regions of the human apoE promoter, as a mediator of synergistic activation by cAMP and retinoic acid in astrocytic cells. It was also suggested that protein kinase A (PKA)-mediated phosphorylation of AP-2 may be important in this regulation.

Promoter deletions and footprinting analysis of the apoE promoter have also identified several other proximal promoter regulatory elements, many of which are thought to be recognised by Sp1. A strong non-specific enhancer, termed upstream regulatory element (URE) 1, has been identified at nucleotides –193 to –124 upstream of the first exon (Paik *et al.*, 1988; Smith *et al.*, 1988). Also another non-specific enhancer was identified at nucleotides –366 to –246, and a GC-box motif that binds the Sp1 transcription factor was identified at nucleotides –59 to –45, both of which promote apoE gene expression (Paik *et al.*, 1995; Paik *et al.*, 1988; Smith *et al.*, 1988; Chang *et al.*, 1990). Also by using nuclear extracts from various cell lines, a DNase I footprint was identified at nucleotides -103 to –87, termed URE3 (Jo *et al.*, 1995). This positive transcriptional element may have a dual role in apoE gene expression depending on the cell type, as a part of the URE3 region has been shown to be a binding site for the nuclear repressor factor BEF-1, which becomes phosphorylated for activity via a PKC pathway in HepG2 cells (Berg *et al.*, 1995; Berg *et al.*, 1996)

In addition to the promoter elements described above, several consensus sequences for various other transcription factors that interact with the proximal promoter region have been identified through the use of various techniques, including transfection assays, DNase-I and *in vivo* footprinting, cell free transcription assays and the yeast one-hybrid system. A summary of these transcription factors are shown in Table 1.2.

CONSENSUS SEQUENCE	CELL TYPE	REFERENCES
SP-1	HepG2, mouse liver nuclei	Paik <i>et al.,</i> 1995; Smith <i>et al.,</i> 1988; Dang and Taylor, 1996
AP-2	U87	Garcia et al., 1996
AP-1	THP-1	Basheeruddin et al., 1994
HNF3	Mouse liver nuclei	Dang and Taylor, 1996
HNF4	Mouse liver nuclei	Dang and Taylor, 1996
TF-LF2	Mouse liver nuclei	Dang and Taylor, 1996
GATA-1	Mouse liver nuclei	Dang and Taylor, 1996
C/EBP	Mouse liver nuclei	Dang and Taylor, 1996
BEF-1	Hep G2	Berg <i>et al.</i> , 1995
Aa	Hep G2	Smith <i>et al.</i> , 1988
B1b	Hep G2	Smith <i>et al.</i> , 1988
B2b	Hep G2	Smith <i>et al.</i> , 1988
Zic1, Zic2	U87	Salero et al.,2001

TABLE 1.2. TRANSCRIPTION FACTOR CONSENSUS SEQUENCES IDENTIFIED AS AFFECTING APOE GENE EXPRESSION

#### 1.2.4.2.II. DISTAL REGULATORY ELEMENTS AND FACTORS

Although various promoter elements have been identified, studies have shown that promoters lack the ability to direct apoE gene transcription *in vivo*, in any cell in the absence of distal enhancers (Simonet *et al.*, 1991). Various such distal regulatory elements that control the transcription of the apoE gene in liver and other tissues have been determined using transgenic mice and it has been demonstrated that a number of sequences in the intergenic region between apoE and apoCl are required for the expression of both these genes in a variety of tissues including liver, testis, spleen, skin, submaxillary gland, kidney, brain, small intestine, heart, stomach, and the pancreas (Simonet *et al.*, 1993). In addition, two hepatocyte-specific enhancers, hepatic control regions (HCR) 1 and 2, have been reported and have been found to be essential for the hepatic expression of apoE/apoCl gene locus (Simonet *et al.*, 1991). Both of these enhancers are located 15kb (HCR1) and 26kb (HCR2) 3' to the apoE gene and share 87% sequence identity, which is thought to have arisen as a result of an evolutionary gene duplication that yielded the apoCl and apoCl' genes (Allan *et al.*, 1995).

Two enhancer regions which direct macrophage- and adipose-specific expression in transgenic mice were also identified recently (Shih *et al.*, 2000). These two homologous enhancers,

designated multi-enhancer 1 and 2 (ME1 and ME2) are located 3.3kb and 15.9kb downstream of the apoE gene and contain 620 and 619 nucleotides, respectively. Several binding motifs for the glucocorticoid receptor (GR) and C/EBP- $\alpha$  and C/EBP- $\beta$  have been located in these enhancer regions although the precise mechanism of activation is unknown. Another recent study has demonstrated that the nuclear liver-X-receptor (LXR) - $\alpha$  and - $\beta$  regulate apoE expression in macrophages and adipocytes. Laffitte *et al.* (2001b) identified that both the promoter and the ME1 and ME2 enhancers of the apoE gene contain a conserved LXR-responsive element (LXRE) that activated gene transcription in response to oxysterol ligands in macrophages, but not monocytes. Also work by Galetto *et al.* (2001) has demonstrated a peroxisome proliferator-activated receptor (PPAR)- $\gamma$  response element within the apoE/apoCI intergenic sequence, which may play a vital role in apoE up-regulation during monocyte differentiation.

Although recent studies are uncovering binding motifs important in apoE expression, further research is essential in order to determine precisely what regions and what transcription factors are important for the regulation of macrophage-derived apoE expression. In addition, the signalling pathways associated with inducible apoE expression are poorly understood and need to be elucidated in detail.

#### 1.2.4.3. POLYMORPHISMS OF APOLIPOPROTEIN E

Three common isoforms of apoE exist in humans, E2, E3 and E4 and are products of three alleles  $\epsilon 2$ ,  $\epsilon 3$ , and  $\epsilon 4$  at a single gene locus (Zannis and Breslow, 1981). The three apoE isoforms can be distinguished from each other by isoelectric focusing due to charge differences, which are the result of cysteine-arginine amino acid substitutions at residues 112 and 158. Although the isoforms differ from one another only by a single amino acid substitution, these changes have profound functional consequences at the cellular and molecular level. The most common phenotype in humans is E3/3 (60%) and seems to be the normal isoform in all known functions; therefore apoE3 (Cys<sup>112</sup>, Arg<sup>158</sup>) is considered to be the parent form of the protein, and apoE4 (Arg<sup>112</sup>, Arg<sup>158</sup>) and E2 (Cys<sup>112</sup>, Cys<sup>158</sup>) are variants (Rall *et al.*, 1982). The frequency of the different alleles in a normolipidemic population is 8%, 77%, and 15% for apoE2, apoE3 and apoE4 respectively (Rall *et al.*, 1982).

Polymorphism of apoE is one of the major determinants of the initiation and progression of atherosclerosis (Davignon *et al.*, 1988; De Andrade *et al.*, 1995) and each of the three isoforms differ with respect to their binding affinity for the LDL receptor, lipoprotein preference and their affinity for heparan sulfate proteoglycans (Davignon *et al.*, 1988). *In vitro* LDL receptor binding studies have demonstrated that the binding affinity of the apoE2 phenotype for the LDL receptor is less than 2% of the affinity of apoE3 and apoE4 (Weisgraber *et al.*, 1982). Due to this impaired

receptor binding, the E2 phenotype is associated with the development of type III hyperlipidemia, a disorder caused by impaired metabolism of apoE containing lipoproteins, which results in a higher proportion of VLDL and IDL in the plasma, a risk factor in atherosclerosis (Gregg *et al.*, 1981). Patients with this form of hyperlipidemia are homozygous for the apoE2 allele, however, other genetic or environmental factors must influence the phenotype because only 1% to 2% of E2 homozygotes have symptomatic hyperlipidemia (Mahley *et al.*, 1999). In fact, most E2 homozygotes have either normal or low cholesterol levels (Rall and Weisgraber, 1982; Davignon *et al.*, 1988). Although the apoE4 phenotype displays normal receptor binding it is still associated with an increased risk for atherosclerosis due to elevated levels of plasma LDL cholesterol (Davignon *et al.*, 1988). ApoE4 is also associated with increased risk for Alzheimer's disease, impaired cognitive function, and reduced neurite outgrowth; isoform-specific differences in cellular signaling events may also exist (Huang *et al.*, 2004; Mahley and Huang, 1999).

The apoE phenotype is also responsible for the preference of apoE variants for the different lipoproteins in the circulation. Human apoE4 preferentially associates with VLDL, whereas apoE3 and apoE2 preferentially associate with HDL (Dong and Weisgraber, 1996). In addition, apoE2 has an even greater affinity for HDL as compared to apoE3 (Luc *et al.*, 1997). Therefore the apoE genotype has a profound effect on the efficiency of cholesterol efflux (Zhang *et al.*, 1996; Cullen *et al.*, 1998; Huang *et al.*, 1998).

Several other naturally-occuring rare variants of apoE have been described, most of which are characterised by substitution of a single amino acid within the LDL receptor binding domain, thereby directly reducing interactions with the LDL receptor. However, one apoE variant with a defective LDL-receptor binding capability, apoE3-Leiden, consists of a 7-amino acid tandem repeat of residues 121-127, which is located outside the LDL receptor binding domain of apoE (Wardell *et al.*, 1989).

A secondary form of apoE polymorphism is explained by posttranslational glycosylation (Mahley, 1988). Several glycosylated isoforms arise from either the presence or the absence of a mono- or disialyted carbohydrate moiety at a single site in apoE, O-linked glycosylation of threonine 194 (Wernette-Hammond *et al.*, 1989).

#### 1.2.4.4. PROTEIN SYNTHESIS AND SECRETION.

ApoE is a 299-amino acid secretory protein, which follows the classical pathway for secretory protein synthesis and release. The primary translation product is 317 amino acids and contains 18 amino acid extension at the N-terminus, which acts as a signal peptide, directing the nascent chain to the endoplasmic reticulum (Zannis *et al.*, 1984). The protein is transported to the Golgi prior to secretion and the protein is glycosylated with carbohydrate chains containing sialic acid (Mazzone
*et al.*, 1986). The glycosylation is O-linked and newly synthesised apoE differ by -2, -4 and -6 charges from the corresponding asialo form that is detected in plasma. Sialation is thought to be a requirement for apoE secretion as 92% of secreted apoE is sialyted in comparison to 42% of intracellular apoE and 24% of plasma apoE (Zannis *et al.*, 1984).

The secretion of apoE is a complex process and a significant proportion of newly synthesised apoE undergoes proteasomal degradation prior to secretion (Mazzone *et al.*, 1992; Duan *et al.*, 1997), as has been shown for apoAl and apoB in liver (White *et al.*, 1999; Yeung *et al.*, 1996). For example, the rate of apoE secretion from mouse peritoneal macrophages is only 44% of the measured rate of apoE synthesis (Dory, 1991). In addition, newly secreted apoE may remain tightly bound to proteoglycans on the cell surface (Lucas and Mazzone, 1996), and this source of the protein can serve as a precursor for the secreted apoE. ApoE arrives at the cell surface in a relatively lipid-poor state and whilst retained at the cell surface by proteoglycans, acquires lipid, this lipidation of the protein results in changes in apoE conformation and subsequent release from the cell surface (Zhao and Mazzone, 2000). This efflux mechanism may explain why endogenous apoE expression is more efficient at producing cholesterol efflux from macrophages compared with exogenous apoE (Lin *et al.*, 1999). Endogenously synthesised apoE is more efficiently sequestered at the cell surface due to an increased exposure to low affinity-binding sites at the cell surface during passage through the pericellular space.

ApoE retained at the cell surface may also undergo re-uptake and then be degraded in a lysosomal compartment (Deng *et al.*, 1995), or recycled back to the cell surface/extracellular medium (Lucas and Mazzone, 1996: Zhao and Mazzone, 2000). This recycling of proteoglycanbound apoE through intracellular membrane compartments including the Golgi could facilitate lipid acquisition by apoE (Zhao and Mazzone, 2000) and also enable cells to maintain their apoE levels at an optimal level. Work by Heeren *et al.* (1999) showed by the use of radiolabeled triglyceride-rich lipoproteins, that 60% of the labelled apoE was released back into the medium, where it again became part of lipoproteins. In addition to hepatocytes, evidence for apoE recycling has been provided in macrophages where a substantial amount of the internalised lipoprotein-associated apoE is retained inside the cell and then reappears in the secretion medium (Takahashi and Smith, 1999). This recent discovery that internalised apoE may escape the lysosomal degradation pathway (Fazio *et al.*, 1999) suggests the presence of an active cellular mechanism which dissociates apoE from the lipoprotein particle and carries apoE back to the secretory apparatus.

# 1.2.4.5. STRUCTURE OF APOLIPOPROTEIN E

The predicted secondary structure of apoE is thought to consist of  $\alpha$ -helices (62%),  $\beta$ -sheet structures (9%) and  $\beta$ -turns (11%) and the remaining 18% having a random structure (Mahley, 1988). Within the apoE structure two distinct independantly folded structural domains are thought to exist and are joined by a randomly arranged protease-susceptible hinge region (Mahley, 1988). Digestion of apoE with thrombin, that cleaves at residues 191 and 215, generates a 22kDa N-terminal fragment (residues 1-191) and a 10kDa C-terminal fragment (residues 216-299) [Aggerbeck *et al.*, 1988]. The molecule is predicted to have a highly ordered structure throughout the amino-terminal half of the molecule. Then, beginning at residue 165, a long stretch of predicted random structure extends for about 35 residues. This random structure is then followed by a highly ordered structure representing approximately the carboxyl-terminal third of the molecule.

#### 1.2.4.5.I. N-TERMINAL DOMAIN OF APOE

The structure of the N-terminal domain of apoE has been revealed by crystallization and X-ray diffraction analysis. The domain is a compact structure consisting of 5 helices, in which 4 of the 5 helices are arranged in an antiparallel helix bundle (Wilson et al., 1991). This region of apoE has been implicated as the LDL receptor and LRP binding domain and a basic arginine- and lysine-rich segment of helix 4 (residues 136-158) is thought to be essential for receptor binding. In vitro studies using apoE that has been chemically modified at the lysine or arginine residues within this LDL receptor binding domain, have demonstrated the complete inhibition of receptor binding (Mahley et al., 1977). ApoE variants associated with type III hyperlipoproteinaemia show a reduced binding activity, the most common variant is apoE2 (Arg<sup>158</sup> to Cys), where the normally occurring arginine at position 158 is replaced with cysteine. This reduced binding capability of the apoE2 variant is due to the absence of a positive charge at position 158, which dramatically changes salt bridges within helix 4 and between helices 3 and 4 (Innerarity et al., 1984). This absence of salt bridges alters the conformation of the LDL receptor-binding region and greatly lowers the positive ion potential of this region (Dong et al., 1996; Wilson et al., 1994). Also, a region spanning residues 171-183 contains critical elements necessary for the stabilisation or the alignment of the LDL receptor-binding domain (Lalazar and Mahley, 1989). In addition, lipid association of the N-terminal domain is thought to be essential for high-affinity binding to the LDL receptor, as the four helix bundle undergoes a structural reorganization upon lipid binding, where the helical segments of the bundle realign, presenting a convex receptor-active surface (Raussens et al., 1998).

A segment of helix 4 (residues 142-147) also contains a high-affinity HSPG binding site that is highly complementary to heparan sulfate rich in N- and O-sulfo groups such as that found in the liver and brain (Libeu *et al.*, 2001). This binding of apoE to proteoglycans involves an

interaction between the positively charged arginine- and lysine-rich region of helix 4 with the negatively charged carboxylate and sulfate groups of proteoglycans and is the initial step of the HSPG-LRP pathway. Rare variants of apoE with substitutions for the normally occurring basic amino acids in this region are moderately defective in binding to the LDL receptor but tend to be severly defective in binding to HSPG (Ji *et al.*, 1994b).

Recent work has also suggested that the N-terminal domain of apoE contains an SR-BI binding sequence (Li *et al.*, 2002a). However, work by Thuahnai *et al.* (2003) has also demonstrated that the C-terminal region of apoE is also required for SR-BI binding in the lipid-free state.

#### 1.2.4.5.II. C-TERMINAL DOMAIN OF APOE

The C-terminal region of apoE contains 3 amphipathic helices spanning the regions 203-222, 225-266, and 268-289 (Nolte and Atkinson, 1992), all of which are thought to be important for lipid binding. In the absence of lipids, apoE self-associates as a tetramer due to protein-protein interactions of the C-terminal domains (Aggerbeck *et al.*, 1988). In the presence of lipid, no selfassociation occurs as the C-terminal domain is required for the association of apoE with lipoproteins. Studies with synthetic peptides from the carboxyl terminus (Sparrow *et al.*, 1992) and apoE3 C-terminus truncation studies (Westerlund and Weisgraber, 1993) revealed that residues 263-286 are critical for apoE tetramerization and lipoprotein association. The C-terminal domain of apoE also contains a glycosaminoglycan binding site that is located between residues 243-272 (Weisgraber *et al.*, 1986). It has also been proposed that the C-terminal region of apoE plays a role in the activation of the enzyme LCAT, which is the enzyme responsible for transforming cholesterol into cholesteryl esters (De Pauw *et al.*, 1995).

The structure of apoE is also responsible for the lipoprotein preferences of the apoE isoforms as domain interaction between the amino- and carboxyl- segments of apoE distinguish the lipoprotein preferences of apoE4 for VLDL versus apoE3 and apoE2 for HDL. This domain interaction in apoE4 is due to an arginine residue replacing a cysteine at position 112, which causes a reorientation in the sidechain of arginine 61 (Dong and Weisgraber, 1996). This reorientation allows arginine 61 to adopt a different exposed position on helix 2, which leads to the formation of a new salt bridge between glutamic acid 109 and residue 112 in helix 3, and more importantly causes the side chain of arg 61 to interact with the carboxyl terminus causing an alteration in conformation which distinguishes it from apoE3 and apoE2 (Dong and Weisgraber, 1996).

#### **1.2.4.6. MODULATION OF APOLIPOPROTEIN E EXPRESSION**

The synthesis and secretion of macrophage apoE is controlled at multiple loci, and is thought to be regulated via transcriptional, posttranscriptional and posttranslational mechanisms. Altering the functional state and metabolism of macrophages such that they mimic situations in atherogenesis results in altered apoE expression. For example, cholesterol loading of macrophages is accompanied by a 10-fold induction of apoE gene transcription (Mazzone et al., 1989; Mazzone et al., 1987). In addition to cholesterol loading, macrophage apoE production can be altered by extracellular cholesterol acceptors and cytokines and also the very act of monocyte differentiation induces apoE expression (Werb and Chin, 1983; Auwerx et al., 1988; Menju et al., 1989; Tajima et al., 1985). For example, bone marrow cells cultured in the presence of colony stimulating factor-1 for 9 days mature into macrophages and secrete apoE (Werb and Chin, 1983). Also, studies in THP-1 cells, which require activation with phorbol ester in order to acquire the macrophage phenotype have revealed that differentiation is associated with increased apoE expression mediated at the level of mRNA, cellular protein and secreted protein (Auwerx et al., 1988; Menju et al., 1989; Tajima et al., 1985). Basheeruddin et al. (1992) postulated that this induction in apoE was due to apoE mRNA being more slowly degraded in macrophages compared to monocytes due to inhibition of a specific nuclease, eventually leading to increased levels of apoE. Another mode of apoE regulation is degradation, as a substantial amount of newly synthesised apoE in macrophages is degraded prior to secretion due to either direct targeting of apoE to lysosomal pathways (Deng et al., 1995) or ubiquitinated-mediated proteasomal degradation (Duan et al., 1997: Wenner et al., 2001), or association of secreted apoE to an extracellular matrix that is then subject to rapid degradation (Lucas and Mazzone, 1996).

# 1.2.4.6.I. ROLE OF CYTOKINES ON MACROPHAGE APOE

Inflammatory cytokines are considered to be important in the pathogenesis of atherosclerosis and other vascular diseases. Various cytokines are produced by multiple cell types during atherogenesis and many of these cytokines regulate apoE expression in macrophage foam cells. For example, the inflammatory agent, bacterial lipopolysaccharide (LPS) decreases apoE secretion from mouse macrophages when added exogenously or injected *in vivo* (Werb and Chin, 1983), which may be due to LPS stimulation of TNF- $\alpha$  secretion (Zuckerman and O'Neal, 1994; Zuckerman *et al.*, 1992; Starck *et al.*, 2000). This indirect form of apoE regulation, through stimulation of TNF- $\alpha$  may also be the case for granulocyte-macrophage colony-stimulating factor (GM-CSF) down-regulation of apoE (Zuckerman and O'Neal, 1994). ApoE production by macrophages is also inhibited by interferon- $\gamma$  (IFN- $\gamma$ ), which is due to increased intracellular degradation of newly synthesised apoE (Brand *et al.*, 1993). Interleukin-1 (IL-1) also affects apoE

expression in HepG2 cells through phosphorylation of the trans-repressor BEF-1, which, in turn inhibits apoE gene expression (Berg *et al.*, 1996). Conversely, the anti-inflammatory agent TGF- $\beta$  has been associated with an increase in apoE mRNA expression and secretion (Zuckerman *et al.*, 1992; Duan *et al.*, 1995). Various other cellular factors also modulate apoE expression, including the thyroid hormone (Vandenbrouck *et al.*, 1994), insulin (Ogbonna *et al.*, 1993), oestrogen (Srivastava *et al.*, 1996), and cAMP (Andreani-Mangeney, 1996).

#### **1.2.4.6.II. EFFECT OF STEROL LOADING ON MACROPHAGE APOE**

A number of studies have investigated the effects of cholesterol loading on apoE expression as cholesterol loaded macrophages are prominent within the atherosclerotic lesion. Although these studies have shown an enhancement of apoE secretion and synthesis, and stimulation of apoE mRNA expression due to increased gene transcription, the mechanisms of regulation have not yet been elucidated (Dory, 1989; Rees *et al.*, 1999; Cader *et al.*, 1997; Mazzone and Basheeruddin, 1991). Work by Paik *et al.* (1988) demonstrated that the sterol-mediated upregulation of apoE required a positive element for apoE gene transcription (PET) [-169/-140], a core sequence of the URE1 region of the human apoE gene. This sequence has previously been shown to have a high homology with the SRE42 of the human LDL receptor gene (Paik *et al.*, 1988) and has been renamed as SRE4 based on its new role as a sterol sensor.

The segregation of cell-surface apoE between secretion and degradation is modulated by the presence of extracellular lipid, such as phospholipid vesicles or lipoproteins (Mazzone *et al.*, 1992). Work by Duan *et al.* (1997) has demonstrated that pre-incubation of macrophages with sterol increases secretion of apoE, which is due to the suppression of apoE intracellular degradation. This effect of sterol on apoE degradation requires the presence of the C-terminal domain of apoE (Duan et al., 2000). This domain is thought to to be necessary for apoE to interact with membrane lipids, which results in an alteration in apoE conformation, which leads to a decrease in susceptibility to proteolysis (Duan *et al.*, 2000). This reduction in apoE degradation may also be due to sequestration of apoE away from proteases (Duan et al., 2000).

In human monocyte-derived macrophages (HMDM) the release of cholesterol and apoE are kinetically parallel, although there is substantial evidence that the two processes are independent of each other (Dory, 1991; Rees *et al.*, 1999; Basu *et al.*, 1983). Work by Rees *et al.* (1999) revealed that upon cholesterol loading there was greater spontaneous release of cholesterol from apoE secreting macrophages from normal mice than those of apoE knockout mice. However there was no difference in cholesterol efflux to apoAI between the two cell sources. Also, when cholesterol-loaded HMDM were incubated with β-cyclodextrins, which are known to promote

cholesterol efflux without associated phospholipid release, cholesterol efflux was observed without increased apoE secretion (Rees et al., 1999).

Oxidised forms of cholesterol, such as those present in foam cells, also modulate apoE expression. Incubation of mouse peritoneal macrophages with 25-hydroxycholesterol increased synthesis (Mazzone et al., 1987) and also increased apoE mRNA levels in differentiated THP-1 cells (Mazzone and Basheeruddin, 1991). However the relevance of these studies to the in vivo situation is limited as 25-hydroxycholesterol is a minor constituent in the atherosclerotic plaque. When differentiated THP-1 cells were incubated with copper oxidised LDL, which has 7ketocholesterol as its major oxysterol [the most abundant oxysterol in the atherosclerotic plaque (Brown and Jessup, 1999)] an increase in apoE mRNA abundance and protein was detected (Cader et al., 1997), which was due to an increase in gene transcription. The action of oxysterols is mediated through the nuclear receptors LXR $\alpha$  and LXR $\beta$ . A recent study by Laffitte *et al.* (2001b) has demonstrated that, in addition to controlling the tissue-specific expression of apoE in macrophages and adjocytes, the ME1 and ME2 enhancers are also likely to facilitate gene induction in response to cellular lipid loading. Recent work in knockout mice has also shown that synthetic LXR ligands are able to inhibit the development of atherosclerosis in mice (Joseph et al., 2002b) and various studies have established a role for these nuclear receptors in dietary cholesterol absorption and in the control of cholesterol efflux in peripheral cells (Repa and Mangelsdorf, 2000; Venkateswaran et al., 2000a). In addition, a recent study by Perez et al. (2003) demonstrated the sequential and co-ordinate regulation of apoE and nuclear receptor transcription, providing further evidence for the roles of nuclear receptors in apoE induction.

#### **1.2.4.6.III. REGULATION OF APOE BY CHOLESTEROL ACCEPTORS**

ApoE expression can also be regulated by HDL and apoAI, the protein constituent of HDL, both of which are extracellular acceptors of the reverse cholesterol transport pathway (Rees *et al.*, 1999; Bielicki *et al.*, 1999). Studies show that incubation of apoAI with cholesterol loaded macrophages from human monocyte derived macrophages, mouse peritoneal macrophages and THP-1 macrophages results in enhanced cholesterol efflux as well as enhanced apoE secretion (Rees *et al.*, 1999; Bielicki *et al.*, 1999). This regulation is likely to occur at the post-transcriptional level as incubation of apoAI treated cholesterol-loaded human monocyte-derived macrophages with actinomycin D, an inhibitor of transcription did not block apoE secretion (Rees *et al.*, 1999). This regulation of apoAI is of physiological relevance as this protein is small and can easily penetrate the arterial intima and access foam cells within the lesion. Also, the incubation of cholesterol-loaded mouse peritoneal macrophages (MPM) with HDL did not alter apoE mRNA levels or rates of protein synthesis although an increase in apoE secretion was observed (Dory,

1991). Work by Lin *et al.* (1998) has demonstrated that the binding affinity of HDL to macrophages increases in cells expressing apoE. Treatment with heparinase significantly reduced this HDL binding, suggesting that apoE secreted was from a cell surface pool, although cycloheximide decreased apoE secretion induced by apoAI, indicating that ongoing protein synthesis is required for apoE secretion in addition to a preformed pool (Rees *et al.*, 1999). A possible explanation for post-translational regulation of apoE is re-direction from a degradatory pathway to a secretory pathway, which may occur via a receptor-mediated mechanism, whereby apoAI or HDL are internalised and come into contact with vesicles containing new apoE.

The ATP binding cassette transporter ABCA1, and to a lesser extent ABCG1, both of which are important regulators of cholesterol efflux (Oram and Vaughn, 2000) have also recently been shown to modulate apoE secretion from human monocyte-derived macrophages (von Eckardstein *et al.*, 2001). Cholesterol loading and cAMP treatment, both of which increase apoE expression, also up-regulate ABCA1 expression in macrophages (Bortnick *et al.*, 2000; Langmann *et al.*, 1999; Takahashi *et al.*, 2000; Oram *et al.*, 2000). ABCA1 may play a role in the intracellular trafficking of apoE as inhibition of ABCA1 leads to the disappearance of apoE containing granular structures from the plasma membrane (von Eckardstein *et al.*, 2001). It is also possible that ABCA1 may serve either as a regulator of vesicular transport between the trans-Golgi network and the plasma membrane (Orso *et al.*, 2000) or as a channel protein within the plasma membrane (Orso *et al.*, 2000; Oram and Vaughn, 2000), both of which would allow a role of ABCA1 in re-direction of apoE from a degradatory pathway.

Although the previous section has given a brief overview of the various factors that are able to modulate apoE expression in macrophages, the major focus of this thesis is the regulation of apoE by LXRs. Therefore, the next section will discuss these members of the nuclear receptor family in more detail.

30

# **1.3. LIVER X RECEPTORS**

The liver X receptors (LXRs) are a family of transcription factors that were first identified as orphan members of the nuclear receptor superfamily. Nuclear receptors, includes both non-steroidal and steroid receptors, which regulate every major metabolic and developmental process by regulating gene transcription in response to ligand binding. Identified ligands for these receptors are small, lipophilic molecules such as steroids, retinoids, thyroid hormone, and vitamin D. In addition, >30 different orphan nuclear members of this receptor family have been identified for which ligands and/or biological functions are yet to be determined (Mangelsdorf and Evans, 1995).

The LXRs were originally isolated in 1995 using low-stringency hybridization screening of liver cDNA libraries (Willy et al., 1995). Two members of the LXR family have been identified. LXRa (NR1H3, also known as RLD-1) and LXRB (NR1H2, also known as UR, NER, OR-1 and RIP15) and sequence analysis shows that human LXR $\alpha$  and  $\beta$  share 77% amino acid identity in both their DNA- and ligand-binding domains (Teboul et al., 1995). The expression patterns of the two LXR receptors differ significantly, with LXR $\alpha$  expression being restricted predominantly to tissues involved in lipid metabolism, with highest levels being found in the liver and also in the kidney, small intestine, spleen, adipose tissue, macrophages, pituitary and adrenals (Apfel et al., 1994; Willy et al., 1995; Auboeuf et al., 1997). In contrast the expression of LXRB is much more widespread and it is found in almost every tissue examined, including liver and brain (Song et al., 1995; Kainu et al., 1996). Recent work has shown that the LXRs are important mediators of the effects of oxLDL on macrophage gene expression, and are also central components of the cellular response to lipid loading. To date more than a dozen LXR target genes have been identified and the majority of which appear to have one of two biological functions; firstly, removal of excess cholesterol through efflux, catabolism or decreased absorption; and secondly, synthesis of fatty acids.

# 1.3.1. LXR AGONISTS

The identification of a specific class of oxidised derivatives of cholesterol as ligands for the LXRs has been crucial to helping understand the function of these receptors *in vivo* and first suggested their role in the regulation of lipid metabolism. Cholesterol neither activates nor binds LXR, although the introduction of an epoxide, hydroxyl, or keto group on the side chain renders this compound biologically active (Janowski *et al.*, 1996; Lehmann *et al.*, 1997). The original identification of oxysterols as natural ligands for LXR was reported almost simultaneously by three laboratories and initial ligand screening experiments employing organic extracts isolated from several tissues identified the first activators as a unique class of meiosis-activating sterols (Forman *et al.*, 1997; Janowski *et al.*, 1996; Lehmann *et al.*, 1997). Subsequent screening of numerous

compounds in the cholesterol metabolic pathway identified a specific group of mono-oxidised derivatives of cholesterol as the most potent ligands (Janowski *et al.*, 1996; Lehmann *et al.*, 1997; Forman *et al.*, 1997). These studies also demonstrated a structure-activity relationship for these compounds, and a requirement for the  $3\beta$ -hydroxyl group of cholesterol and a single, stereoselective oxygen on the sterol side chain, which functions as a hydrogen bond acceptor.

The natural ligands of LXRs are oxysterols and include: 24(S), 25-epoxycholesterol, which is formed from squalene in a shunt pathway of cholesterol biosynthesis and is abundant in the liver; 22(R)-hydroxycholesterol, which is a transient intermediate in steroid hormone synthesis and abundant in the adrenal gland; 24(S)-hydroxycholesterol, which is abundant in the central nervous system; 20(S)-hydroxycholesterol;  $6\alpha$ -hydroxy bile acids; and mevalonate, which is an intermediate of cholesterol biosynthesis. Importantly, these oxysterols elicit an LXR response at concentrations consistent with those found in tissue extracts (Lehmann et al., 1997; Janowski et al., 1999) and the naturally occurring enantiomers are consistently more active than the synthetic enantiomers. Studies have shown that these compounds bind directly to the LXRs with K<sub>d</sub> values ranging from 0.1 to 0. 4µM (Janowski et al., 1999; Spencer et al., 2001). Recently, it was found that 27hydroxycholesterol, formed via the action of the enzyme 27-hydroxylase, is also an endogenous ligand for LXRs in cholesterol-loaded human macrophage cells, with Kd values of 85 and 71nM for LXRa and LXRB respectively (Fu et al., 2001). However it is not known whether 27hydroxycholesterol binds to LXR or whether a metabolite is produced, which subsequently binds and activates the nuclear receptor, although recent work has demonstrated that cholestenoic acid. a metabolite of 27-hydroxycholesterol is able to activate LXR at physiological concentrations (Song et al., 2000).

No significant differences in the ability of LXR $\alpha$  or LXR $\beta$  to respond to oxysterols have been reported, although recently the oxysterol 5,6-24(S), 25-diepoxycholesterol has been shown to be a relatively selective LXR $\alpha$  ligand, suggesting the possibility of developing subtype-selective LXR ligands for pharmacological applications (Janowski *et al.*, 1999). Also, natural and synthetic 6 $\alpha$ -hydroxy bile acids appear to be more selective activators of LXR $\alpha$  than LXR $\beta$  (Song *et al.*, 2000) and cholestenoic acid, a metabolite of 27-hydroxycholesterol, appears to be a naturally occurring ligand for LXR $\alpha$ , but not for LXR $\beta$  (EC<sub>50</sub> for LXR $\alpha$ = 200nM, EC<sub>50</sub> for LXR $\beta \ge 5\mu$ M).

A number of synthetic pharmacological LXR ligands have also been developed, which are structurally unrelated to oxysterols. These compounds are high-affinity LXR-specific agonists and their increased potency compared to physiological ligands makes them useful tools for the study of LXR function. The first such ligands were the compound T0314407 and its derivative T0901317, which showed EC<sub>50</sub> values for LXR $\alpha$  of 100nM and 20nM, respectively, compared to 300nM for

24(S), 25-epoxycholesterol (Schultz *et al.*, 2000). Another compound, acetyl-podocarpic (APD) was shown to be 1000-fold more potent than 22(R)-hydroxycholesterol for the activation of LXR $\alpha$  and LXR $\beta$ , and the maximal stimulation achieved was more than 6-fold higher than for the endogenous ligand (Sparrow *et al.*, 2002). More importantly, APD was approximately 1000-fold more potent than endogenous ligands for the induction of LXR target genes (Sparrow *et al.*, 2002). Most recently, a tertiary amine, GW683965, has been identified as an orally active LXR agonist (Collins *et al.*, 2002).

## 1.3.2. LXR ANTAGONISTS

Antagonists are compounds that compete with agonists for binding and prevent activation of receptors. Although it was initially assumed that the binding sites for agonists and antagonists were the same, there is now evidence that they can be different, although overlapping is required. It is thought that the conformation of the antagonist-bound receptor blocks productive interaction with one or more proteins required for transcription, or prevents efficient DNA binding, thereby reducing the ability of the receptor to activate target genes.

In contrast to the oxysterols, another product of the cholesterol metabolic pathway, geranylgeranyl-PP, negatively regulates LXR $\alpha$  and LXR $\beta$  transcriptional activity by inhibiting interaction of the LXRs with nuclear coactivators (Forman *et al.*, 1997; Gan *et al.*, 2001). Unsaturated fatty acids also antagonise LXR activity by acting as competitive inhibitors of LXR ligands (Ou *et al.*, 2001; Yoshikawa *et al.*, 2002). Additionally, human plasma contains natural antagonists for LXR $\alpha$  and LXR $\beta$  (Song *et al.*, 2001). These antagonists are specific 3-sulfate derivatives of oxysterols, and may be formed by the oxidation of cholesterol 3-sulfate, which is present in elevated plasma concentrations in hypercholesterolaemia (Tamasawa *et al.*, 1993).

Interestingly, the bile acid-responsive farnesol X receptor (FXR) dramatically represses LXR $\alpha$  activation in the absence of LXR ligands (Wang *et al.*, 1999). Since FXR and LXR $\alpha$  show overlapping expression patterns (liver, intestine, kidney), it may be hypothesised that their co-regulation by oxysterol and bile acid derivatives of cholesterol enables the organism to coordinately regulate cholesterol homeostatic pathways in several tissues.

# **1.3.3. LXR PROTEIN STRUCTURE**

The LXRs, like other transcriptional regulators, exhibit a modular structure with different regions of the protein corresponding to autonomous functional domains that can be interchanged between related proteins without loss of function. The modular structure of nuclear receptors is similar throughout, despite the wide variation in ligand sensitivity between the receptor sub-types. LXRs, like all the nuclear receptors, contains a variable NH<sub>2</sub>-terminal domain, also called the A/B region,

#### CHAPTER ONE - GENERAL INTRODUCTION

which contains a ligand-independent transcriptional activation function (AF-1) [Figure 1.4]. The A/B domain also shows promoter- and cell-specific activity, which suggests that it may contribute to the specificity of action among receptor isoforms and may interact with cell type-specific factors (Aranda and Pascual, 2001). This domain also contains several phosphorylation sites that may be the target of various different signaling pathways (Shao and Lazar, 1999). Although no phosphorylation of LXR has yet been shown, several studies, using other members of the nuclear receptor family have shown that this phosphorylation is able to affect transcriptional activity (Rochette-Egly *et al.*, 1997; Kato *et al.*, 1995; Patrone *et al.*, 1996; Juge-Aubry *et al.*, 1999).

The DNA-binding domain (DBD) is the most conserved domain of nuclear receptors and contains two highly conserved zinc finger motifs that target the receptor to specific DNA sequences (Luisi *et al.*, 1991; Schwabe *et al.*, 1993). These zinc finger motifs are located within two  $\alpha$ -helices that cross at right angles to form the core of the DBD that recognises a hemi-site of the response element. Between the DBD and the ligand-binding domain (LBD) is a hinge region (D domain) [Figure 1.4] that allows protein flexibility and rotation of the DBD for simultaneous receptor dimerisation and DNA binding. This D domain in many nuclear receptors contains the nuclear localisation signals and contains residues whose mutation abolishes interaction with nuclear receptor corepressors (Aranda and Pascual, 2001).

The ligand-binding domain (LBD) is a large multifunctional domain located at the COOHterminal of the receptor (Figure 1.4). This domain encompasses the ligand-binding domain, dimerisation interface, and contains a ligand-dependant activation function (AF-2) [Wurtz *et al.*, 1996]. The LBD is formed by twelve  $\alpha$ -helices, with a conserved  $\beta$ -turn situated between helix 5 and 6. The helices are folded into a three-layered, antiparallel helical sandwich that contains a cavity, the ligand-binding pocket, which accomodates the ligand (Aranda and Pascual, 2001). This cavity is mainly hydrophobic and is buried within the bottom half of the LBD.



#### FIGURE 1.4. SCHEMATIC REPRESENTATION OF A NUCLEAR RECEPTOR

A typical nuclear receptor is composed of several functional domains. See text for further detail. Figure adapted from Aranda and Pascual, 2001.

The structure of ligand-bound receptors is much more compact than the unliganded receptor, demonstrating that upon ligand binding the receptor undergoes a conformational change (Aranda and Pascual, 2001). This conformational change is thought to enable transcriptional activation by the receptor as it co-ordinately dissociates corepressors and facilitates recruitment of coactivator proteins, necessary for ligand activation (Svensson *et al.*, 2003). Recruitment of coactivators appears to depend on specific interactions between the receptor and coactivator, and involves a common, highly conserved, LXXLL interaction domain found in the nuclear receptor interaction domain of many coactivators (McInerney *et al.*, 1998). Ligand-dependent recruitment of coactivators is dependent on AF-2, which consists of a short conserved helical sequence within the LBD (Bourguet *et al.*, 2000), although a distinct set of coactivators can also associate with the AF-1 domain. Point mutations in this AF-2 region had little or no effect on the binding of regulatory ligands, dimerisation or DNA-binding, but did abolish transcriptional activation (Svensson *et al.*, 2003). This region is therefore predicted to serve as an adapter surface for interactions with other molecules necessary for transcriptional activation (Danielian *et al.*, 1992; Barettino *et al.*, 1994).

# 1.3.4. LXR AND RXR

LXRs belong to the nonsteroidal subclass of nuclear receptors, which include the farnesol X receptor (FXR), peroxisome-proliferator activated receptors (PPAR), and the vitamin D receptor (VDR). As with other non-steroidal nuclear receptors, the majority of LXR proteins are localized in the nucleus and form obligate heterodimers with the retinoid X receptor (RXR) for function (Willy *et al.*, 1995). Target genes of this heterodimer are characterised by the presence of specific *cis* elements in their proximal promoter, distal enhancers or introns. The heterodimers bind to an LXR response element (LXRE) that contains a direct repeat of two similar hexanucleotide half-sites (consensus-AGGTCA) separated by 4 nucleotides, also known as a direct repeat-4 (DR-4) element (Edwards *et al.*, 2002; Mangelsdorf and Evans, 1995; Zhang *et al.*, 2001). Analysis of the functional LXREs that have been characterised in the known LXR target genes indicates that the consensus LXRE contains a number of invariant nucleotides in each half-site, and that the nucleotides at other positions can vary considerably. Although LXR $\alpha$  and LXR $\beta$  both bind with similar affinities to consensus LXR response elements, it is likely that these two nuclear receptors have differential effects on certain target genes.

The LXR/RXR heterodimer is a so-called permissive heterodimer, which is characterised by the ability to be activated independently by either the RXR ligand (i.e. 9-cis retinoic acid), the LXR ligand, or synergistically by both receptors' ligands together. This mechanism of dual-ligand activation contrasts with that exhibited by the second, well-characterised subclass of RXR heterodimers that include the vitamin D receptor (VDR), thyroid hormone receptor (TR), and retinoic acid receptor (RAR) as partners. In these heterodimers, the RXR ligand cannot activate the heterodimer complex (as in the case of RXR/VDR and RXR/TR) or can do so only after the partner's ligand has first bound the receptor (RXR/RAR). One particularly novel aspect of the RXR/LXR heterodimer is that it only requires the carboxy-terminal activation function (the AF-2 domain) of LXR to become transcriptionally active (Willy and Mangelsdorf, 1997; Wiebel and Gustaffson, 1997), demonstrating that binding of the RXR ligand results in a conformational change in LXR that leads to transcriptional activation.

# 1.3.5. LXR AND COREGULATORS

Transcriptional regulation by the LXR/RXR heterodimer is achieved when the LXRE-bound dimer is activated by its ligand. As with other transcriptional regulatory proteins, the primary mechanism by which LXR heterodimers affect the rate of RNA polymerase II-directed transcription likely involves the interaction of receptors with components of the transcription preinitiation complex. This interaction may be direct or it may occur indirectly through the action of coregulators, which act as bridging factors. Evidence for the recruitment by activated receptors of coregulators came initially from yeast experiments based on transcriptional interference (Gill and Ptashne, 1988) and subsequently from squelching noted between cotransfected receptors in mammalian cell-based reporter assays (Meyer et al., 1989). In general, coregulators play one or more of the following roles in nuclear receptor transcriptional activity: (1) they function as bridging factors to recruit cofactors to DNA-bound nuclear receptors, e.g. the steroid receptor coactivator (SRC) proteins, which can recruit p300/CREB-binding protein (CBP) to DNA-bound receptors; (2) they acetylate nucleosomal histones and various transcription factors at the promoters of hormone target genes, e.g. p300/CBP and p300/CBP-associated factor (P/CAF), which have potent nucleosomal histone acetyltransferase (HAT) and factor acetyltransferase activities; and (3) they function as bridging factors between the DNA-bound receptors and the basal transcriptional machinery e.g. the human mediator-like thyroid hormone receptor associated protein (TRAP)/vitamin D3 receptor interacting proteins (DRIP) multipolypeptide complex. Although the activation mechanisms and coregulators used by LXR have so far been unclear, recent work has identified a potential coregulator model for the activation of LXR transcription.

In this model, the non-induced transcriptional state of LXR is thought to occur when corepressors, such as silencing mediator for retinoic acid and thyroid hormone receptor (SMRT) and nuclear receptor co-repressor (N-CoR), are bound to LXR/RXR in the absence of ligand (Wagner *et al.*, 2003; Hu *et al.*, 2003). Upon ligand binding, a basal rate of transcription occurs when the corepressors start to dissociate from the LXR/RXR heterodimer and transcription of the target gene becomes fully activated when coactivators are recruited to the ligand-activated

LXR/RXR heterodimer. This recruitment of co-activator proteins occurs via a two-step mechanism. Firstly, corepressors dissociate and the p160 coactivators, steroid receptor coactivator-1 (SRC-1) or glucocorticoid receptor-interacting protein 1 (GRIP1), appear to be directly recruited to the liganded receptors (Huuskonen et al., 2004). These coactivators then serve as a platform to recruit p300 or CBP (Huuskonen et al., 2004). The next step in this activation process is currently unclear for the LXR heterodimer. Other nuclear receptors appear to recruit associated proteins such as P/CAF, and by using their HAT activities, they remodel the nucleosomal structures so that the mediator complex TRAP/DRIP can replace SRC-1/CBP and bind the liganded receptors. Subsequent recruitment of RNA polymerase II complex to TRAP/DRIP completes the second step in nuclear receptor transactivation. Unfortunately, how the activation signal is transmitted from LXR-p160 coactivator p300/CBP to the basal transcription machinery is currently unknown and recent work has ruled out a role for P/CAF in LXR transcriptional activation (Huuskonen et al., 2004). It is possible that p300/CBP directly interacts with the basal transcription machinery (Felzien et al., 1999) or that p300/CBP can acetylate histones and recruit SWI/SNF chromatin remodelling complexes to the nuclear receptors thereby facilitating the interaction with basal machinery (Huang et al., 2003). Therefore, these interactions among LXR, chromatin remodeling complexes and coactivators at oxysterol-activated promoters lead to the stimulation of gene transcription.

LXR transactivation can also be activated by at least two other mechanisms. Activating signal co-integrator 2 (ASC-2) has been shown to interact specifically with LXR and activate transcription (Kim *et al.*, 2003). The precise mechanism isn't known although ASC-2 exists in a complex containing methyltransferases. Futhermore, LXR can interact with multi-protein bridging family 1 (MBF-1) [Brendel *et al.*, 2002], which directly interacts with transcription factor IID (TFIID), making it a possible bridging molecule between LXR and basal transcription machinery.

In addition to the ability of the unliganded LXR/RXR heterodimers to actively repress target genes by recruiting corepressors (NcoR and SMRT) [Wagner *et al.*, 2003], LXR is capable of inhibiting the activities of other transcription factors, such as AP-1 and NFkB, in a ligand dependant manner. This effect does not require DNA binding by the nuclear receptor and is termed transrepression and is thought to be the primary mechanism by which nuclear receptors inhibit pro-inflammatory genes in macrophages (Joseph *et al.*, 2003; Schule *et al.*, 1991; Ray and Prefontaine, 1994; Ricote *et al.*, 1998).

Another function of nuclear receptors that does not require DNA binding is a recently characterised mechanism, termed non-genomic or extranuclear signalling, whereby nuclear receptor agonists also appear to participate in rapid signalling. Although no such action has yet been described for the LXRs, it is becoming increasingly obvious that non-genomic signalling of nuclear receptor agonists is an important aspect of nuclear receptor-regulated activation of genes.

#### CHAPTER ONE - GENERAL INTRODUCTION

Whilst the traditional mechanism of target gene transactivation by nuclear receptors and their agonists is characterised by the delayed onset of action, and its sensitivity to blockers of transcription and protein synthesis, the non-genomic action of nuclear receptors and their agonists is rapid, and inhibitors of transcription and protein synthesis have no effect (Schmidt *et al.*, 2000). This non-genomic function is thought to be the mechanism responsible for the recently observed effects of nuclear receptor ligands on intracellular signalling pathways and whilst it is still not clear the exact mode of mechanism, recent studies have suggested the existence of a sub-population of nuclear receptors that exist in the cytoplasm and at the plasma membrane, to which agonists bind and activate cell signalling cascades (Schmidt et al., 2000; Boonyaratanakornkit and Edwards, 2004).

# 1.3.6. LXR TARGET GENES

The characterisation of LXR target genes in the macrophage, as well as in liver and intestine, and the function of these target genes in reverse cholesterol transport and atherosclerosis are briefly described in the following section. The use of LXR null mice and LXR-selective agonists have been crucial in the identification of these genes and in establishing the important role of LXRs as sterol sensors that govern the absorption, transport, and catabolism of cholesterol.

TARGET GENE	Target Tissues	FUNCTION	LXRE	REFERENCES
<b>↑</b> СУР7А1	Liver	Bile acid synthesis	TGGTCActcaAGTTCA rat TGGTCAcccaAGTTCA mouse TGGTCAtccaAGTTCA hamster TTGTCaccaAGCTCA human	Repa and Mangelsdorf, 2000; Lehmann <i>et al.</i> , 1997; Peet <i>et al.</i> , 1998.
<b>↑</b> ABCA1	Macrophage Intestine	Cholesterol efflux	AGGTTActatCGGTCA human GGATCAcctgAGGTCA human AGATCActtgAGGTCA human AGGTTActgaAGGCCA human	Repa <i>et al.</i> , 2000b; Costet <i>et al.</i> , 2000; Schwartz <i>et al.</i> , 2000; Singaraja <i>et al.</i> , 2001.
↑ABCG5/G8	Liver Intestine	Sterol transport	Not identified	Berge <i>et al.,</i> 2000; Repa <i>et al.,</i> 2002.
∱ABCG1	Macrophage	Cholesterol efflux	TGGTCActcaAGTTCA human AGTTTAtaatAGTTCA human	Venkateswaran et al., 2000b; Kennedy et al., 2001.

↑АроE	Macrophage Adipocyte	Cholesterol efflux	GGGTCActggCGGTCA human GAATCActtaAGGTCA mouse	Laffitte et <i>al.,</i> 2001b.
<b>↑</b> LPL	Liver Macrophage	Triglyceride hydrolysis	A/T?GGTCAccacCGGTCA mouse AGGTGActgcAGGTCA human	Zhang <i>et al.,</i> 2001.
<b>↑</b> CETP	-	Cholesteryl ester transfer	GGGTCAttgtCGGGCA human	Luo and Tall, 2000.
<b>∱</b> SREBP-1c	Liver Intestine	Fatty acid synthesis	GGGTTActggCGGTCA mouse GGGTTActagCGGGCG human	Repa et al., 2000a; Schultz et al., 2000; Yoshikawa et al., 2001.
<b>↑</b> FAS	Liver	Fatty acid synthesis	GGGTTActgcCGGTCA human GGGTTActacCGGTCA rat	Joseph <i>et al.,</i> 2002a.
<b>↑</b> LXRa	Macrophage	Cholesterol sensor	AGGTTActgcTGGTCA human	Laffitte <i>et al.,</i> 2001a; Whitney <i>et al.,</i> 2001.

CYP7A1, cholesterol 7 $\alpha$ -hydroxylase; ABCA1, ATP-binding cassette A1; ApoE, apolipoprotein E; LPL, lipoprotein lipase; CETP, cholesteryl ester transfer protein; SREBP-1c, sterol response element-binding protein-1; FAS, fatty acid synthase; LXR, liver X receptor. Table derived from Zhang and Mangelsdorf, 2002.

# 1.3.6.1. LXRs AND THE CONTROL OF CHOLESTEROL HOMEOSTASIS

Cholesterol homeostasis is maintained via dietary intake, intestinal absorption and the regulated pathways of *de novo* synthesis and catabolism. Cholesterol biosynthesis is controlled by the accumulation of sterols in the liver, which feedback and inhibit further synthesis through a protease-sensitive mechanism involving the SREBP family of transcription factors (Brown and Goldstein, 1997). Excess cholesterol also activates a feed-forward pathway that leads to the catabolic elimination of cholesterol as bile acids. Bile acid production from cholesterol and the solubilization of cholesterol by bile acids are the final steps in reverse cholesterol transport, facilitating cholesterol excretion from the body. The LXRs are important in the regulation of each of these processes and the following section will discuss the role of these ligand-mediated transcription factors and the function of its target gene in more detail.

#### 1.3.6.1.I. CHOLESTEROL SYNTHESIS

A role for the LXRs in the negative control of cholesterol synthesis was first suggested by the observation that the expression of several important cholesterogenic genes is altered in the liver of LXR $\alpha$ -/- mice (Peet *et al.*, 1998). The expression of the cholesterol synthesis transcriptional regulator SREBP-2, as well as the cholesterogenic enzymes 3-hydroxy-3-methylglutaryl coenzyme A (HMG-CoA) reductase and synthase, farnesyl diphosphate synthase (FPP synthase), and squalene synthase (SQS) were significantly increased in these mice compared to wild-type mice (Peet *et al.*, 1998). This action of LXRs on cholesterol synthesis has since been confirmed by the ability of oxysterols to suppress the nuclear levels of SREBP-2 (Yoshikawa *et al.*, 2001; DeBose-Boyd *et al.*, 2001). The regulation of cholesterol synthesis is thought to differ between the two LXR isoforms as only minor increases in the hepatic expression of HMG-CoA reductase, FPP synthase, and SQS were observed in LXR $\alpha$ -/- mice (Alberti *et al.*, 2001), and SREBP-2 expression was unchanged, indicating that LXR $\alpha$  is more important as a regulator of cholesterol synthesis than LXR $\beta$ . This role of LXRs in cholesterol synthesis has been further supported by studies demonstrating that the liver expression of SQS and HMG-CoA synthase is markedly reduced following the administration of the LXR agonist T0901317 to wild-type mice (Schultz *et al.*, 2000).

# 1.3.6.1.II. DIETARY CHOLESTEROL ABSORPTION

Dietary and secreted biliary cholesterol enters the intestinal lumen and is absorbed by proximal enterocytes. In the small intestine, at least three ABC transporters are transcriptionally regulated by LXRs (Repa et al, 2000b; Berge *et al.*, 2000) and limit cholesterol absorption by mediating free cholesterol export from inside the enterocyte into the intestinal lumen. ABCA1-/- mice show a marked increase in cholesterol absorption, confirming the importance of ABCA1 in this process (McNeish *et al.*, 2000). Ligands of LXR and RXR increase ABCA1 expression and decrease cholesterol absorption in the distal duodenum and jejenum; however these responses are absent in LXR $\alpha\beta$ -/- mice (Repa *et al.*, 2000b). Recently it has been suggested that ABCA1 may reside in the abluminal membrane of the enterocyte, and its role may be to redirect absorbed cholesterol into a secretion pathway involving HDL-like particles, rather than the typical chylomicrons that enter the lymph (Murthy *et al.*, 2002; Wellington *et al.*, 2002).

Two additional targets for LXR regulation are ABCG5 and ABCG8. ABCG5 and ABCG8 are half-transporters, each composed of a single transmembrane domain and ATP-binding cassette (Berge *et al.*, 2000) and are co-expressed exclusively in the liver and intestine. Mutations in either gene cause a rare autosomal recessive disorder called sitosterolemia, which is characterised by increased absorption of cholesterol and toxic plant sterols (sitosterols), and decreased biliary sterol secretion, resulting in the development of premature cardiovascular

disease (Lee et al., 2001; Berge et al., 2000). In vivo and in vitro experiments indicate that ABCG5 and ABCG8 are direct targets of LXRs (Berge et al., 2000; Repa et al., 2002). In mice treated with LXR agonists, ABCG5 and ABCG8 expression is enhanced (Repa et al., 2002), biliary cholesterol is increased and cholesterol absorption is decreased (Repa et al., 2000b; Plosch et al., 2002). Similar changes occur in transgenic mice overexpressing the human ABCG5 and ABCG8 genes (Plosch et al., 2002; Yu et al., 2002b). Additional evidence that ABCG5 and ABCG8 are important in this system is the finding that LXR activity still affects biliary secretion and cholesterol absorption in the ABCA1-/- mouse (Plosch et al., 2002).

#### **1.3.6.III. REVERSE CHOLESTEROL TRANSPORT**

The elimination of cholesterol through bile acid production occurs exclusively in the liver, so that excess cholesterol from peripheral tissues must be returned to the liver, in a process termed reverse cholesterol transport for catabolic elimination. As described previously in this chapter, this process is particularly important in macrophages. LXRs are implicated as key sensors of sterol metabolism, maintaining normal cholesterol balance by promoting sterol efflux from peripheral cells, increasing circulating HDL-cholesterol, and ultimately increasing hepatic sterol catabolism and excretion. LXR regulation of reverse cholesterol transport also suggests a role for LXRs in protecting against atherogenesis, a hypothesis that has been confirmed by studies of apoE-/- mice in which activation of LXR reduced atherosclerosis (Claudel *et al.*, 2001; Joseph et al., 2002b). Recent studies suggest that LXRs control lipid efflux from macrophages through the coordinate regulation of a set of genes involved in cholesterol efflux, including ABCA1 and apoE (Laffitte *et al.*, 2001; Claudel *et al.*, 2001; Repa *et al.*, 2000b; Venkateswaran *et al.*, 2000a).

# THE ABC TRANSPORTERS

ABCA1 facilitates the transport of cholesterol and phospholipids to apoAI, apoE or HDL, the first step in reverse cholesterol transport. The role of LXR in the regulation of ABCA1 expression and cholesterol efflux has been supported by multiple studies (Repa *et al.*, 2000b; Costet *et al.*, 2000; Schwartz *et al.*, 2000; Venkateswaran *et al.*, 2000a). Venkateswaran *et al.* (2000a) and Schwartz *et al.* (2000) have demonstrated that ABCA1 mRNA expression is induced in macrophages in response to lipid loading and that expression and activation of LXR stimulates apoAI mediated efflux of cholesterol. This ability of LXR ligands to stimulate cholesterol efflux is lost in fibroblasts from Tangier patients, whose disease is characterised by mutations in the ABCA1 gene, therefore, confirming the importance of LXR induction of ABCA1 in this process (Venkateswaran *et al.*, 2000a). The ability of LXR ligands to raise HDL-cholesterol levels in mice is also consistent with the established function of ABCA1 in cholesterol efflux (Schultz *et al.*, 2000). Also LXR $\alpha\beta$ -/- mice

develop splenomegaly and accumulate foam cells in multiple peripheral tissues, a phenotype remarkably similar to that of the ABCA1-/- mice (Schuster *et al.*, 2002; Tangirala *et al.*, 2002). Furthermore, oxysterol-induced ABCA1 expression can be attenuated in cells expressing a dominant-negative form of LXR $\alpha$  (Venkateswaran *et al.*, 2000a). Work by Costet *et al.* (2000) has identified a DR-4 element in a region of the ABCA1 promoter that is responsible for the sterol-mediated activation of ABCA1 promoter. In addition, a study addressing the role of ABCA1 in endothelial cells showed that native LDL acts via LXR activation to up-regulate ABCA1 expression and cholesterol efflux, indicating that ABCA1 may additionally play an important role in cholesterol homeostasis in the vascular endothelium (Liao *et al.*, 2002).

In accordance with the above evidence confirming the role of LXRs in ABCA1 expression, the synthetic LXR agonist APD is able to stimulate macrophage ABCA1 expression and cholesterol efflux, this induction is also much greater in comparison to the induction observed in response to cholesterol loading (Sparrow *et al.*, 2002). Also, the LXR agonist GW3965 was shown to induce the expression of ABCA1 and ABCG1 in the aortas of hyperlipidemic mice, and to concurrently cause a dramatic 50% reduction in lesion development in two mouse models of atherosclerosis (Joseph *et al.*, 2002b).

Similar to ABCA1, expression of ABCG1 is also induced in macrophages in response to cholesterol loading and specific oxysterol LXR ligands (Venkateswaran *et al.*, 2000b) and the gene also contains two functional LXREs (Kennedy *et al.*, 2001). The function of ABCG1 is currently unknown, but it has been proposed to play a role in cholesterol efflux, perhaps by working in concert with ABCA1 (Klucken *et al.*, 2000). Peritoneal macrophages isolated from mice deficient for both LXR $\alpha$  and LXR $\beta$  failed to show an increase in ABCG1 expression in response to oxysterol treatment (Venkateswaran *et al.*, 2000b). A novel ABC half-transporter, ABCG4, has also been characterised (Costet *et al.*, 2000; Dean *et al.*, 2001; Engel *et al.*, 2001) and it shares high sequence identity with ABCG1. ABCG4 expression is also believed to be upregulated in response to oxysterol to oxysterols and retinoids in human macrophages, suggesting a possible role for this transporter in macrophage lipid homeostasis (Engel *et al.*, 2001).

#### Apolipoprotein E

As described earlier in this chapter, another protein involved in the cholesterol efflux pathway is apoE. In macrophages, apoE transcription is increased by cholesterol loading, indicating the existence of a molecular sensor that regulates apoE expression in response to lipid. The LXRs represent this molecular sensor, since LXR ligands increase apoE expression in human macrophages, an effect mediated via two LXREs present in the apoE enhancer regions (Laffitte *et al.*, 2001). ApoE was the first gene shown to be regulated by LXR/RXR heterodimers in a tissue-

specific manner as LXR mediates lipid-inducible expression of the apoE gene in adipose tissue and macrophages but not in liver (Laffitte *et al.*, 2001b). This induction of apoE expression by LXR ligand is significantly reduced in macrophages from either LXR $\alpha$ -/- or LXR $\beta$ -/- mice, even in the presence of high concentrations of ligand (Laffitte *et al.*, 2001b).

ApoE is present in a gene cluster that contains apoCI, apoCII, and apoCIV, and recent work has demonstrated that these apolipoprotein genes are also LXR responsive, in both human and murine macrophages (Mak *et al.*, 2002a). Like apoE the other members of this gene cluster have been shown to serve as acceptors in ABCA1-mediated cholesterol efflux (Bortnick *et al.*, 2000) and the induction of these acceptors in macrophages within the arterial wall would be expected to promote cholesterol efflux and reverse cholesterol transport. The induction of the members of this gene cluster by LXR ligand was reduced in LXR $\alpha\beta$ -/- mice, but was generally unchanged in LXR $\alpha$ -/- or LXR $\beta$ -/- mice, indicating that both LXR isoforms are functional regulators of lipid-inducible apolipoprotein expression (Mak *et al.*, 2002a).

#### LIPOPROTEIN REMODELING ENZYMES

LXR has also been shown to influence the expression of several enzymes that act on lipoproteins, including lipoprotein lipase (LPL), cholesteryl ester transfer protein (CETP) and the phospholipid transfer protein (PLTP) (Cao et al., 2002; Laffitte et al., 2003; Luo et al., 2000; Mak et al., 2002b). Both CETP and PLTP are lipoprotein-remodelling enzymes that transfer lipids between lipoproteins. CETP is secreted by the liver and circulates in plasma principally bound to HDL (Barter, 2000). CETP mediates the transfer of HDL cholesteryl esters to apoB-containing lipoproteins for return to the liver, and in exchange HDL receives trialycerides (Oliveira et al., 1997). This modification of HDL by CETP makes HDL more susceptible to hydrolysis by hepatic lipase at the hepatocyte surface, which is an important component of the regeneration of small preβ-HDL-like particles and free apoAI that can re-circulate in the reverse cholesterol transport pathway. The contribution of CETP activity to atherosclerosis is unclear, although the generation of pre-β-HDL-like particles, suggests an antiatherogenic role. However, over-expression of human CETP in atherosclerotic mice (apoE-/- or LDLR-/-) lowers plasma HDL levels and increases lesion formation (Plump et al., 1999). The LXR/RXR heterodimer acts via a DR-4 element in the promoter of the human CETP gene to mediate sterol induction of the expression of CETP in response to a high fat diet (Luo and Tall, 2000). In transient transfection experiments, the CETP gene could be transactivated by both LXR $\alpha$  and LXR $\beta$ , suggesting that both LXR isoforms play a role in the control of CETP expression (Luo and Tall, 2000).

PLTP has been identified as a key modulator of HDL metabolism, and may also be involved in reverse cholesterol transport (van Tol, 2002). In addition, PLTP has recently been shown to regulate VLDL secretion from the liver as PLTP-/- mice exhibit decreased levels of VLDL and LDL in an apoE-deficient or apoB-transgenic background (Jiang et al., 1999). During the lipolysis of serum VLDL by LPL, surface remnants containing phospholipids and apolipoproteins are transferred by PLTP and contribute to pre- $\beta$ -HDL and large  $\alpha$ -HDL production. PLTP can also mediate lipid transfer between HDL particles to produce a small pre- $\beta$ -HDL and a large  $\alpha$ -HDL and the expression of a human PLTP transgene in mice increases production of pre- $\beta$ -HDL and enhances hepatic uptake and clearance of cholesteryl ester. Recent work has demonstrated that LXR ligands induce the expression of PLTP in cultured HepG2 cells and mouse liver in vivo and also in macrophages (Laffitte et al., 2003; Cao et al., 2002; Mak et al., 2002b). Also PLTP activity is increased in mice treated with the LXR ligands and this ability of synthetic and oxysterol ligands to regulate PLTP mRNA was lost in LXR $\alpha\beta$ -/- mice (Laffitte et al., 2003). These studies also demonstrated that the PLTP promoter contains a high-affinity LXRE that is bound by LXR/RXR heterodimers in vitro and is activated by LXR/RXR in transient-transfection studies (Mak et al., 2002; Laffitte et al., 2003). Finally, immunohistochemistry studies reveal that PLTP is highly expressed by macrophages within human atherosclerotic lesions, suggesting a potential role for this enzyme in lipid-loaded macrophages (Laffitte et al., 2003).

LPL catalyses the hydrolysis of lipoprotein triglycerides and is highly expressed in adipose tissue and muscle, and is also produced by macrophages (Mead et al., 2002). The contribution of LPL expression to atherogenesis is complex and seems to be context dependent. On the antiatherogenic side, non-macrophage LPL is able to aid the clearance of serum cholesterol-rich lipopoproteins by remodelling triglyceride-rich chylomicrons and VLDL into chylomicron remnants and LDL for uptake and this hydrolysis of lipoproteins also results in the concomitant generation of material for HDL formation. Indeed systemic overproduction of human LPL in atherogenic animal models appears to protect against atherosclerosis and diet-induced atherogenesis (Shimada et al., 1996; Yagyu et al., 1999). In addition to its enzymatic activity, LPL can also facilitate HDLassociated cholesterol uptake into liver cells by acting as a bridging factor (Merkel et al., 2002; Rinninger et al., 1998). Hepatic LPL expression also promotes the maturation of HDL in mice (Strauss et al., 2001) and liver-specific expression of human LPL by adenovirus-mediated gene delivery in apoE-/- and LDLR-/- mice improves the hyperlipidemic serum profile seen in these mice (Zsigmond et al., 1997). On the other hand, evidence suggests that macrophage expression of LPL in the artery wall may be pro-atherogenic, as studies in which macrophage-specific overproduction of LPL in apoE-/- and LDLR-/- mice was shown to accelerate atherosclerosis (Babaev et al., 2000; Wilson et al., 2001). Also a study showing that macrophage-selective elimination of LPL led to a

decrease in lesion formation in LDLR-/- mice (Babaev *et al.*, 2000). LPL would promote the conversion, and perhaps retention, of triglyceride rich lipoproteins to cholesterol-rich lipoproteins such as LDL, thereby creating an environment conductive to foam cell development (Pentikainen *et al.*, 2002). In addition, recent evidence indicates that LPL may serve as a bridging factor that links lipoprotein particles to the arterial extracellular matrix, thus increasing their residence time in the arterial wall and potentiating their conversion into more atherogenic forms (Saxena *et al.*, 1992; Eisenberg *et al.*, 1992).

Similar to apoE, the regulation of LPL gene expression by LXR is tissue specific as LXR agonists induce expression of LPL in liver and macrophages but not in adipose tissue (Zhang *et al.*, 2001). Liver LPL expression is greatly increased in wild-type mice fed a high cholesterol diet or treated with a synthetic LXR agonist, while no change in LPL expression occurs in LXR $\alpha\beta$ -/- mice (Zhang *et al.*, 2001). LXR $\alpha$  is thought to be a more selective regulator of LPL than LXR $\beta$  due to a higher affinity of LXR $\alpha$  for a DR-4-like LXRE that has been identified within the murine LPL gene and conserved in the human LPL gene (Zhang *et al.*, 2001).

# 1.3.6.1.IV. BILE ACID SYNTHESIS

Conversion of cholesterol to bile acids in the liver is a quantitatively important pathway for elimination of cholesterol from the body. Bile acids are physiological agents important for the absorption and transport of lipid-soluble vitamins, steroids and xenobiotics. Bile acid synthesis occurs exclusively in the liver and is initiated by cholesterol  $7\alpha$ -hydroxylase (CYP7A1), a cytochrome P450 isozyme of the CYP7A family (Russell and Setchell, 1992; Chiang, 1998). CYP7A1 is the rate-limiting enzyme in the classic bile acid synthesis pathway and transcription of this gene is inhibited by bile acid feedback and is stimulated by dietary factors such as cholesterol. This regulation of the CYP7A1 gene by cholesterol appears to be mediated by LXR, and a functional LXRE has been identified in the proximal promoter region of the rat CYP7A1 gene, and has been shown to be activated by RXR/LXR heterodimers in an oxysterol and retinoid-dependent manner (Lehmann et al., 1997; Peet et al., 1998). Evidence supporting LXR regulation of CYP7A1 came from work demonstrating that on a high cholesterol diet, LXR $\alpha$ -/- mice fail to up-regulate CYP7A1 expression and bile acid production, resulting in increased plasma LDL cholesterol, decreased HDL cholesterol levels and massive cholesterol accumulation in the liver (Peet et al., 1998). This study also observed that the knockout of the LXR $\alpha$  gene did not effect the basal activity of CYP7A1, indicating that LXRs may not play a significant role in CYP7A1 expression in hepatocytes under normal dietary conditions. Despite the strong evidence of a role for LXR $\alpha$  in the control of CYP7A1 expression in rats and mice, it appears that this function of LXR $\alpha$  is not present

in humans as co-transfection with LXR $\alpha$  and RXR expression plasmids strongly stimulates the activity of the rat CYP7A1 promoter but has no significant effect on the human CYP7A1 promoter (Chiang *et al.*, 2001). This difference is due to the LXR element not being conserved in the promoter of the human CYP7A1 gene (Agellon *et al.*, 2002).

The function of LXR $\beta$  in CYP7A1 regulation is still unclear. Allthough LXR $\beta$  is also expressed in the liver and appears to respond to the same ligands as LXR $\alpha$ , it is unable to rescue the altered bile acid phenotype in LXR $\alpha$ -/- mice. This suggests that the two LXRs may subserve distinct biological roles. Accordingly, LXR $\beta$ -/- mice maintain normal hepatic cholesterol metabolism and are thus resistant to dietary cholesterol (Alberti *et al.*, 2001); however LXR $\alpha\beta$ -/- mice have an exacerbated liver phenotype (Zhang and Mangelsdorf, 2002). Co-transfection assays have also shown that the CYP7A1 LXRE is a much stronger response element for LXR $\alpha$  than LXR $\beta$  (Lehmann *et al.*, 1997; Peet *et al.*, 1998), which may explain the difference between the regulation of CYP7A1 by the two LXR isoforms.

# 1.3.6.2. LXRs and the control of Fatty acid Synthesis

Although all of the target genes described so far demonstrate that LXRs have the ability to play an antiatherogenic role, some of the earliest studies on LXR pointed to an important role for these receptors in the control of fatty acid synthesis. Mice carrying a targeted disruption in the LXRa gene were noted to be deficient in the hepatic expression of SREBP-1c. FAS, steroyl coenzyme A desaturase 1 (SCD-1), acyl coenzyme A carboxylase (ACC) [Peet et al., 1998]. Consistent with these observations, oral administration of synthetic ligands to mice triggers induction of the lipogenic pathway and elevates plasma and hepatic triglyceride levels (Schultz et al., 2000; Joseph et al., 2002a). These effects were also confirmed by the finding that the treatment of LXR $\alpha\beta$ -/mice with the synthetic ligand. T0901317, induced only a minor increase in plasma trialyceride levels, and had no effect on liver expression of SREBP-1, SCD-1, FAS or ACC (Schultz et al., 2000). The primary mechanism through which LXR agonists stimulate lipogenesis appears to be through direct activation of the SREBP-1c promoter (Repa et al., 2000a; Yoshikawa et al., 2001). SREBP-1c is a membrane-bound transcription factor that governs the expression of several enzymes in the biosynthesis and esterification of unsaturated fatty acids (Shimano et al., 1999; Kim and Spiegelman, 1996; Horton and Shimomura, 1999). Such SREBP-1c-activated genes include acetyl-CoA synthetase, acetyl-CoA carboxylase, FAS, SCD-1, glycerol phosphate acyltransferase and CTP:phosphocholine cytidylyltransferase. Dietary cholesterol or the administration of LXR or RXR agonists specifically induces the expression of the SREBP-1c isoform in the tissues of wildtype, but not LXRab-/- mice (Repa et al., 2000a). Interestingly it has been shown that the basal

expression of SREBP-1c in a rat hepatoma cell line is dependant on the activation of LXR by an endogenous oxysterol (DeBose-Boyd *et al.*, 2001). *In vitro* studies have shown that LXR and RXR agonists, as well as LXR $\alpha$  or LXR $\beta$  over-expression, induced increased transcriptional activity of the SREBP-1c proximal promoter via two inverted LXREs of the DR-4 type, thus demonstrating that SREBP-1c is a direct target gene of the oxysterol-activated LXR/RXR heterodimer (Repa *et al.*, 2000a; Yoshikawa *et al.*, 2001). Differential regulation by the two LXR isoforms has also been demonstrated for SREBP-1c as LXR $\alpha$  appears to be a more potent regulator of SREBP-1c expression in the liver than LXR $\beta$ , since LXR $\beta$  was unable to compensate for the absence of LXR $\alpha$  in LXR $\alpha$ -/- mice (Peet *et al.*, 1998). An additional mechanism for LXR-mediated up-regulation of fatty acid synthesis was identified with the finding that the LXRs can up-regulate FAS expression directly through interaction with an LXRE in the FAS promoter, as well as indirectly via SREBP-1c (Joseph *et al.*, 2002a). This LXR/RXR-mediated upregulation of SREBP-1c and FAS expression has been shown to occur in the human hepatoma cell line HepG2, indicating that LXR activation may increase fatty acid synthesis in humans (Yoshikawa *et al.*, 2001).

Although the ability of LXRs to regulate fatty acid synthesis is widely considered as a negative aspect of LXR activation, it is also important to note that this regulation of lipogenesis by the LXRs may also contribute to its role in cholesterol homeostasis. Regulation of lipogenesis may function to coordinate the levels of fatty acids and phospholipids, both of which are components essential for the mobilisation of free cholesterol. For example, oleoyl-CoA, a product of SCD-1 catalysis, is the preferred substrate for cholesterol esterification. Therefore, by increasing SCD-1 activity via SREBP-1c, the LXRs would ultimately promote the esterification and storage of free cholesterol within the cell and protect the organism from toxic cholesterol accumulation. In addition, phospholipids facilitate bile acid flow, lipoprotein transport of excess cholesterol, and help maintain the appropriate ratio of cholesterol to other lipids in plasma membranes (Repa *et al.*, 2000a; Yoshikawa *et al.*, 2001). Therefore, it is possible to appreciate why a transcription factor that upregulates cholesterol efflux and downregulates cholesterol synthesis, would also upregulate fatty acid synthesis. Unfortunately, pharmacological activation of LXRs increases lipogenesis to potentially harmful levels, causing hypertriglyceridaemia (Yoshikawa *et al.*, 2001).

#### **1.3.7. REGULATION OF LXR EXPRESSION**

Although more and more mediators that effect the gene expression of the LXRs are being identified (Tobin *et al.*, 2000; Tobin *et al.*, 2002; Lundholm *et al.*, 2004), the mechanisms that control expression of these genes are not well understood. In mice LXR $\alpha$  is expressed primarily in the liver, intestine, adipose tissue and macrophages, whereas LXR $\beta$  is ubiquitously expressed,

indicating that distinct *trans*-acting factors must be involved in the regulation of these genes as expression of the LXRs is tissue specific and regulated in response to certain metabolic signals. Recent studies attempting to characterise the promoter region of the mouse LXR $\alpha$  gene have identified a role for the transcription factors Zta, Ets, and Hes1 in LXR gene regulation (Steffensen *et al.*, 2003). These factors have been linked to cell cycle and differentiation processes (Sinclair, 2003; Oikawa and Yamada, 2003; Kageyama *et al.*, 2000) suggesting that expression of LXR $\alpha$  might be under control of signalling mechanisms regulating cell proliferation. Also several putative binding sites of the glucocorticoid receptor (GR) were also identified in the LXR $\alpha$  promoter, and transient co-transfections of the GR and LXR $\alpha$  promoter constructs induced reporter gene activity, suggesting cross-talk between GR and LXR signalling (Steffensen *et al.*, 2003). Furthermore, members of the nuclear factor 1 (NF1) and CCAAT/enhancer binding protein (C/EBP) families have also been shown to affect the transcriptional potential of the mouse LXR $\alpha$  promoter (Steffensen *et al.*, 2001 and 2002).

Although information on the transcriptional control of the LXR genes is limited, studies have demonstrated that macrophages have established mechanisms whereby LXR gene expression can be amplified during cholesterol loading.

# <u>1.3.7.1. HUMAN LXR $\alpha$ AUTOREGULATION</u>

Several studies (Whitney *et al.*, 2001; Laffitte *et al.*, 2001a; Li *et al.*, 2002b) have demonstrated the existence of an auto-regulatory loop controlling the expression of LXR $\alpha$ , but not LXR $\beta$ . Studies have shown that natural and synthetic LXR ligands induce the expression of LXR $\alpha$  in multiple human cell types, including primary human macrophages. This induction of LXR $\alpha$  occurs via three functional LXREs identified in the human LXR $\alpha$  promoter (Whitney *et al.*, 2001; Laffitte *et al.*, 2001a; Li *et al.*, 2002b). One of these LXREs in the distal portion of the LXR $\alpha$  promoter was strongly activated by both LXR $\alpha$  and LXR $\beta$ , while the other two LXREs were selectively but more weakly activated by LXR $\alpha$  (Li *et al.*, 2002). Although this region is conserved in both the mouse and human promoter, autoregulation is only observed in human cells, suggesting that humans may be more responsive to LXR $\alpha$  agonists than mice. As a result, the induction of LXR target genes, by LXR ligands may be significantly greater in human than in murine macrophages, allowing amplification of the cholesterol catabolic cascade and maximise the effects of LXR (Laffitte *et al.*, 2001a; Whitney *et al.*, 2001).

# 1.3.7.2. LXRa REGULATION BY PPARS

In addition to autoregulation the macrophage has evolved an additional mechanism to further amplify the LXR signalling pathway. Certain lipid components of oxLDL can activate another nuclear receptor subfamily expressed in human macrophages, the peroxisome proliferatoractivated receptors (PPARs) [Delerive et al., 2000; Han et al., 2000]. This activation of the PPAR family in turn causes a marked increase in LXRa expression (Shiffman et al., 2000). Two subtypes of the PPARs, PPAR $\alpha$  and PPAR $\gamma$ , are particularly interesting as they are well characterised and are involved in lipid metabolism and have also been found in macrophages that are associated with atherosclerotic plaques. Work in rat hepatocytes has demonstrated that unsaturated fatty acids are able to increase the mRNA and protein expression of LXR $\alpha$ , but caused no change in the expression of LXRB (Tobin et al., 2000). This upregulated expression of LXR $\alpha$  was mainly due to an increase in transcription, which was due to the binding of PPAR $\alpha$  to a PPAR response element (PPRE) in the promoter region of the murine LXR $\alpha$  gene (Tobin et al., 2000). Subsequently, a PPRE was also identified in the human LXRα promoter region (Laffitte et al., 2001a) and was found to be responsible for the induction of the LXR $\alpha$  gene by PPARy agonists (Chawla et al., 2001a). Notably, in this study only ligands for the PPAR $\gamma$  isoform were able to induce LXR $\alpha$  gene expression, whilst ligands for the other isoforms had no effect. PPARy agonists have also been shown to induce LXRα expression in human primary macrophages (Chinetti et al., 2001). Interestingly, no regulation of LXRB has been observed thus far in any studies, implicating distinct regulation of LXR $\alpha$  and LXR $\beta$  genes.

This transcriptional regulation of LXR $\alpha$  by the PPAR family suggests an overlap between LXR and PPAR signalling pathways, which allows the PPARs to affect LXR target genes involved in cholesterol efflux. Exposure of macrophages to PPAR $\gamma$  ligands increases the expression of LXR target genes such as ABCA1 and ABCG1 and promotes cholesterol efflux in human macrophages through a transcriptional cascade mediated by LXR $\alpha$  (Chawla *et al.*, 2001b; Chinetti *et al.*, 2001). In addition to the ABC transporters it has been determined that the apoE gene is regulated by the PPAR $\gamma$ -LXR axis, and the macrophage specific deletion of PPAR $\gamma$  reduces levels of apoE and drastically decreases basal cholesterol efflux (Akiyama *et al.*, 2002).

Another way in which the LXR and PPAR pathways may overlap is through direct interaction of the receptors. It has been suggested that LXR $\alpha$  can bind directly to PPAR $\alpha$ , as well as to the common heterodimerization partner RXR $\alpha$  (Miyata *et al.*, 1996). Although LXR $\alpha$  did not form a DNA binding complex with PPAR $\alpha$  on naturally occurring PPREs or LXREs, LXR $\alpha$  was able to inhibit the binding of PPAR $\alpha$ /RXR $\alpha$  heterodimers to PPREs, and coexpression of LXR $\alpha$  in

mammalian cells antagonized peroxisome proliferator signalling mediated by PPAR $\alpha$ /RXR $\alpha$  *in vivo*. These findings suggested that LXR $\alpha$  plays a role in modulating PPAR-signalling pathways in the cell (Miyata *et al.*, 1996). Further evidence for LXR-PPAR overlap has come from recent studies identifying a role for the LXRs in adipocyte differentiation (Seo *et al.*, 2004). In adipocytes, LXR activation with T0901317 primarily enhanced the expression of lipogenic genes such as the SREBP-1c and FAS genes and substantially increased the expression of PPAR $\gamma$ . This study showed that the PPAR $\gamma$  promoter contains a conserved LXRE and was transactivated by the expression of LXR $\alpha$  (Seo *et al.*, 2004). Furthermore, the suppression of LXR $\alpha$  by small interfering RNA attenuated adipocyte differentiation. Taken together, these results suggest that LXR plays a role in the execution of adipocyte differentiation by regulation of lipogenesis and adipocyte-specific gene expression.

#### 1.3.8. LXRs AND ATHEROSCLEROSIS

As described previously in this chapter, oxidised LDL uptake in macrophages of the arterial wall is central to the pathogenesis of atherosclerosis. In addition to serving as a source of lipids to the macrophage and other cells of the atherosclerotic plaque during lesion development, oxLDL regulates gene expression in macrophages, smooth muscle cells and endothelial cells. Recent data suggests that LXRs may mediate many of the effects of oxLDL on gene expression in vascular cells. LXRs are expressed in one or more of the cell types present in the atherosclerotic lesion. LXR $\alpha$  is expressed at a low level in circulating monocytes and at a much higher level following differentiation into macrophages (Laffitte et al., 2001a; Nagy et al., 1998) and the exposure of macrophages to oxLDL also induces the expression of LXR $\alpha$  mRNA (Laffitte et al., 2001a). The uptake of oxLDL by macrophages during atherogenesis, supplies the cells with ligand for the activation of LXR. Therefore oxLDL induces both the expression and the transcriptional activity of LXRs in macrophages. Oxysterol ligands of LXR can also be derived intracellularly through enzymatic modification of cholesterol. Consequently, loading macrophages with acetylated LDL, which does not contain oxysterols, also activates LXR (Laffitte et al., 2001a; Laffitte et al., 2001b; Venkateswaran et al., 2000b). Therefore LXRs are likely to be transcriptionally active within atherosclerotic lesions. As described earlier, the activation of these receptors and their target genes stimulates cholesterol efflux in macrophages, promotes bile acid synthesis in liver, and inhibits intestinal cholesterol absorption, actions that would collectively be expected to reduce atherosclerotic risk.

Recent studies have directly addressed the role of the LXR signalling pathway in atherosclerosis using mouse models. Tangirala et al. (2002) were able to address the importance

of macrophage LXR signalling using bone marrow transplantation studies into apoE-/- and LDLR-/mice. They showed that selective elimination of macrophage LXRs by lethal irradiation, and transplant with LXRαβ-/- bone marrow, led to a significant increase in atherosclerotic lesion formation in both mouse models. Also, foam cell accumulation occurred in several other tissues including lung, spleen and brain (Joseph et al., 2002b; Tangirala et al., 2002; Schuster et al., 2002). Additionally, macrophage-specific rescue of LXRs significantly decreased atherosclerosis development in these mice, despite no significant changes in circulating lipid levels (Tangirala et al., 2002). Direct evidence for the potential use of LXR activators in atherosclerosis has also come from studies in murine models. The LXR agonist GW3965 was shown to decrease lesion area approximately 50% in both apoE-/- and LDLR-/- mice (Joseph et al., 2002b; Terasaka et al., 2003). This reduction in atherosclerosis was thought to be due to increased ABCA1 and ABCG1 expression, as demonstrated in atherosclerotic aortas of apoE-/- mice treated with LXR agonists. which probably caused an increase in cholesterol efflux, causing net lipid loss from macrophage foam cells, and antagonism of fatty streak formation (Joseph et al., 2002b; Claudel et al., 2001). Chronic ligand administration only moderately affected the lipoprotein profile of these mice, confirming that the antiatherogenic ability of LXR ligands is probably through a direct effect on the cells of the artery wall. A similar reduction in atherosclerosis was observed with the RXR agonist, LG268 (Claudel et al., 2001). Therefore, despite the undesirable effects of LXR agonists on lipogenesis, the net effect of whole body LXR activation can be considered to be anti-atherogenic.

Two other nuclear receptors, RXR and PPAR<sub>Y</sub>, have also been associated with the pathogenesis of atherosclerosis and treatment of atherogenic mouse models with RXR or PPAR<sub>Y</sub> agonists results in significant decreases in lesion development (Claudel *et al.*, 2001; Li *et al.*, 2000). Interestingly both RXR and PPAR<sub>Y</sub> impact LXR-regulated pathways and LXR $\alpha$  may be a particularly important downstream target with regards to the antiatherogenic effects of PPAR<sub>Y</sub> ligands. The LXR/RXR heterodimers are permissive heterodimers that respond to agonists for both receptors, therefore it is not surprising that RXR agonists mimic many of the effects of LXR activators including induction of ABCA1 and reverse cholesterol transport in macrophages (Repa *et al.*, 2000b; Claudel *et al.*, 2001). Also studies have demonstrated strong and synergistic induction of ABCA1, ABCG1 and apoE by LXR and RXR agonists in LXR $\alpha\beta$ +/+ but not LXR $\alpha\beta$ -/-macrophages indicating that any effects of RXR agonists on these genes are indeed mediated by LXR (Tangirala *et al.*, 2002). As mentioned in the above section, the expression of the LXR $\alpha$  gene is responsive to PPAR<sub>Y</sub> and as a result of this crossregulation, PPAR<sub>Y</sub> and LXR agonists have additive effects on ABCA1 expression and reverse cholesterol transport in macrophages (Chawla *et al.*, 2001a; Laffitte *et al.*, 2001b; Chinetti *et al.*, 2001). Transplantation of PPAR<sub>Y</sub> -bone marrow

into LDLR -/- mice also increases atherosclerosis (Chawla *et al.*, 2001a), suggesting that PPAR $\gamma$ -LXR cross-talk may be important for atherosclerosis susceptibility. Consistent with this, PPAR $\gamma$ deficiency results in a marked reduction of LXR $\alpha$  expression and macrophage-cholesterol efflux (Chawla *et al.*, 2001a; Moore *et al.*, 2001; Akiyama *et al.*, 2002). Also PPAR $\gamma$  agonists have been reported to reduce atherosclerosis in mouse models (Li *et al.*, 2000; Claudel *et al.*, 2001) and clinical trials (Minamikawa *et al.*, 1998) which is most likely a consequence of indirect LXRmediated effects. In combination, this *in vivo* data highlights the potential usefulness of LXR, PPAR or RXR agonists in the management of atherosclerosis.

#### **1.3.9. LXRs and inflammation**

Although the role of LXRs in the regulation of lipid metabolism is well established and it is clear from mouse model studies that LXRs exert an important atheroprotective role in macrophages, recent work has been aimed at identifying other potentially atheroprotective roles for these receptors in macrophages, distinct from reverse cholesterol transport. Atherosclerosis has long been recognised as a chronic inflammatory disease as well as a disorder of lipid metabolism and recent observations have identified LXRs as regulators of immune functions in macrophages. LXR agonists have been shown to inhibit the macrophage response to bacterial pathogens and prevented the induction of several pro-inflammatory genes, such as IL-1β, IL-6, MMP-9, iNOS, COX-2, MCP-1 and MCP-3, macrophage inflammatory protein (MIP)-1β and IP-10 (Joseph *et al.*, 2003; Castrillo *et al.*, 2003a). Consistent with these *in vitro* effects, LXR null mice exhibit enhanced responses to inflammatory stimuli and LXR ligands reduce inflammation in a murine model of contact dermatitis (Joseph *et al.*, 2003).

Work by Castrillo *et al.* (2003a) demonstrated that LXR $\alpha$  and LXR $\beta$  inhibit the basal and inducible expression of MMP-9. MMP-9 is highly expressed by macrophages and smooth muscle cells in atherosclerotic plaques and participates in the degradation of ECM components, which leads to plaque instability and ultimately plaque rupture (Mason *et al.*, 1999; Prescott *et al.*, 1999; Brown *et al.*, 1995). Therefore mechanisms by which excessive MMP production can be limited are of therapeutic interest. Treatment of murine peritoneal macrophages with synthetic LXR ligands reduces MMP-9 mRNA expression and decreased its induction by pro-inflammatory stimuli including LPS, IL-1 $\beta$  and TNF $\alpha$ . This regulation was strictly receptor-dependant and absent in macrophages from LXR $\alpha\beta$ -/- mice. In addition, treatment of apoE-/- mice with LXR agonists reduced the expression of the inflammatory mediator MMP-9 in atherosclerotic aortas (Joseph *et al.*, 2003). These observations suggest that LXR agonists may exert their antiatherogenic effects not only by promoting cholesterol efflux, but also by acting to limit the production of inflammatory

mediators in the artery wall. Both studies demonstrated that this regulation was LXR-dependent and involved repression of NF-kB signalling.

Conversely, inflammatory pathways might alter the LXR-mediated metabolism of cholesterol. Recent work has demonstrated that viral and bacterial pathogens antagonise LXR transcriptional activity and cholesterol efflux. Some viral and bacterial pathogens activate Toll-like receptor 3 (TLR3) and TLR4 in macrophages, which triggers a rapid antimicrobial response through stimulation of proinflammatory pathways. This in turn inhibits LXR-induced expression of ABCA1 and apoE, and therefore decreases macrophage cholesterol efflux (Castrillo *et al.*, 2003b). This effect is NF-κB-independent and appears to involve IRF3, a transcriptional regulator of TLR signalling that inhibits the transcriptional activity of LXR on its target promoters. This crosstalk between signaling pathways controlling innate immunity and macrophage cholesterol metabolism may contribute to the ability of bacterial and viral pathogens to modulate cardiovascular disease.

Unfortunately, LXR has also been shown to play a role in the production of the proinflammatory cytokine tumor necrosis factor- $\alpha$  (TNF $\alpha$ ) [Landis *et al.*, 2002]. Co-administration of 22(R)-hydroxycholesterol and the RXR-specific ligand 9-cis-retinoic acid to human monocytes resulted in a dramatic increase in TNF $\alpha$  production. LXR activation induced TNF $\alpha$  gene transcription and translation via an LXRE in the TNF $\alpha$  promoter, while RXR activation induced the release of TNF $\alpha$  protein from cells co-stimulated with an LXR activator (Landis *et al.*, 2002). It may be possible that specific LXR-mediated stimulation of TNF $\alpha$  expression may reduce atherosclerotic lesion size by the induction of apoptosis (Landis *et al.*, 2002). LXR also induces vascular endothelial growth factor (VEGF), a cytokine produced by macrophages and is a primary inducer of angiogenesis and neovascularization. VEGF mRNA was induced in response to synthetic LXR agonists in murine and human primary macrophages as well as in murine adipose tissue *in vivo* and LXR heterodimers bound directly to a conserved LXRE in the promoter of the murine and human VEGF genes (Walczak *et al.*, 2004).

# **<u>1.4. AIMS OF THE STUDY</u>**

The evidence presented in this chapter clearly demonstrates a protective role for apoE in the development of atherosclerosis. ApoE is expressed at high levels in the atherosclerotic lesions, where it is mainly associated with macrophage-derived foam cells (Rosenfeld *et al.*, 1993; O'Brien *et al.*, 1994). This macrophage-specific expression of apoE is potentially of great importance and several studies have demonstrated that macrophage-derived apoE has potent anti-atherogenic properties (Linton *et al.*, 1995; Boisvert and Curtiss, 1999; Fazio *et al.*, 1997). Therefore macrophage apoE can be considered as an excellent target for therapy against atherosclerosis and studies investigating the regulation of this protein would be of great therapeutic potential.

Regulation of apoE expression is complex and involves multiple levels of control, including gene transcription, mRNA stability, synthesis, degradation and secretion (Duan *et al.*, 1995; Zuckerman *et al.*, 1992; Menju *et al.*, 1989; Basheeruddin *et al.*, 1992; Mazzone *et al.*, 1989). Although it has been well established through numerous studies that differentiation of monocytes into macrophages is associated with increased apoE expression (Werb and Chin, 1983; Auwerx *et al.*, 1988; Menju *et al.*, 1989; Tajima *et al.*, 1985), relatively little is known about the mechanisms involved in this activation of apoE. Currently, investigations into the differentiation-dependent activation of apoE in macrophages, is currently restricted to only a single study, which was rather limited and partly inconclusive (Basheeruddin *et al.*, 1994). Therefore the initial aims of this project were:

- To delineate fully the regulatory elements in the apoE promoter responsible for apoE activation during macrophage differentiation, and;
- (ii) Identify the nuclear factors involved in this differentiation-dependent activation of apoE.

Also, during this project, studies in transgenic mice suggested that two distal enhancers, ME1 and ME2 were important in the macrophage- and adipocyte-specific expression of the apoE gene (Shih *et al.*, 2000; Laffitte *et al.*, 2001b). However, the precise regulatory sequences had not been identified, although multiple C/EBP sites were suggested as potential targets. Therefore, as an extension to the initial aims of the study, the role of ME1 and ME2 in apoE expression during macrophage differentiation was investigated.

The outcome of these initial aims are presented in Chapter 3 and show that, similar to the case in hepatocytes, the regulation of macrophage apoE expression is extremely complex. More specifically, in contrast to the endogenous gene, neither the promoter nor the enhancer were activated during macrophage differentiation. Therefore, as a result of this outcome, the overall aims of the project were modified slightly to understanding the regulation of apoE transcription by the Liver X receptor (LXR) family of nuclear receptors.

As described previously in this chapter, the activation of LXRs *in vivo* by ligands leads to a number of favourable changes in lipid metabolism, including promotion of reverse cholesterol transport, elevation of plasma HDL cholesterol, inhibition of cholesterol absorption, and the antagonism of inflammatory signalling. The relevance of these effects for the development of cardiovascular disease is clear from studies showing that synthetic ligands inhibit atherosclerosis in animal models (Joseph *et al.*, 2002b). Unfortunately, at the same time, the potent lipogenic activity of the current generation of LXR agonists is a significant limitation. From a theapeutic standpoint,

the most desirable LXR agonist would be one that was a strong inducer of genes involved in cholesterol efflux, such as ABCA1, apoE, ABCG1, and a strong suppressor of inflammatory gene expression yet lacked activity on the SREBP-1c and FAS promoters. This requirement to dissociate fatty acid synthesis from the beneficial effects on cholesterol metabolism requires a much clearer understanding of LXR-regulated transcription.

LXR-mediated regulation of target gene transcription has recently been shown to be more complicated than first anticipated. The requirement of coregulators for ligand activation of target genes has revealed that LXR transcription is regulated by mechanisms, other than just the availability of the ligand. In addition, recent studies have demonstrated an essential role for cell signalling pathways in nuclear receptor signalling (Rochette-Egly, 2003). These pathways may be responsible for the nuclear receptor recruitment of coregulators by affecting the availability and activity of these proteins and therefore affecting transcription. Also, recent studies have demonstrated that nuclear receptors are phosphoproteins, whose transcriptional activation may be modified directly through phosphorylation (Shao and Lazar, 1999). As very little work has so far been done on the effect of cell signalling on the ligand-induced activation of LXR target genes, the revised aims for this project were:

- Investigate the mechanism of oxysterol action on the expression of apoE
- (ii) Identify signalling pathways that may be involved in this regulation of apoE

This has led to the identification of a novel involvement of cell signalling pathways such as the c-jun N-terminal kinase (JNK)/stress-activated protein kinase (SAPK), casein kinase 2 (CK2), and phosphoinositide-3-kinase (PI3K) pathways in the ligand-mediated induction of apoE in macrophages (see Chapters 5 and 6).

# **CHAPTER TWO - MATERIALS AND METHODS**

# 2.1. MATERIALS

The chemical reagents and materials required for general use in this study are listed alongside their suppliers in Table 2.1.

MATERIALS	SUPPLIER	
Acrylamide:bisacrylamide (37.5:1)	Anachem, Luton, UK	
[γ- <sup>32</sup> P]-ATP, X-OMAT Kodak X-ray film, ECL- Western blotting detection reagent, Rainbow protein size markers, Streptavidin- HRP complex	Amersham Int., Buckinghamshire, UK	
MicroAmp® 96-well reaction tubes, TaqMan® Universal PCR mastermix, TaqMan® primers, TaqMan® probes	Applied Biosystems, Warrington, UK	
RPMI-1640 with stabilix	Autogen Bioclear, Wiltshire, UK	
Goat polyclonal ApoE antibody	Biogenesis Ltd, Poole, UK	
Curcumin, LY294002, PD98059, SB202190, SP600125	Calbiochem, Nottingham, UK	
Agarose, BIOTAQ DNA polymerase, Magnesium Chloride, 10x NH₄ Buffer	BioLine, London, UK	
Cell lines: THP-1 and U937	European Collection of Animal Cell Culture, Salisbury, UK	
Butan-1-ol, EDTA, ethanol, formaldehyde, formamide, hydrochloric acid, isopropanol, methanol, SDS, sodium chloride, sodium fluoride, sodium hydroxide, sodium orthovanadate, sodium phosphate, sodium pyrophosphate, sulphuric acid, tri-sodium citrate, Tris-buffer and other general chemicals	Fisher, Loughborough, UK	

# TABLE 2.1 MATERIALS/ CHEMICAL REAGENTS AND THEIR SUPPLIERS

Saran Wrap	Genetic Research Instrumentation, Essex, UK
Concert <sup>™</sup> plasmid purification kit, FCS, 1Kb DNA marker, Trypsin/EDTA	Gibco BRL Life Technologies Ltd, Paisley, UK
Tissue culture plastics, 96-well plates, 6-well plates, Falcon 15ml and 50ml polypropylene tubes	Greiner, Gloucestershire, UK
Human serum apoE, SB biotinylated secondary antibody, goat serum	GlaxoSmithKline, Stevenage, UK
Goat polyclonal ATAB atlantic ApoE antibody	Inc Star, Minnesota, USA
Immobilin™_P <sup>sq</sup> PVDF membrane	Millipore Corporation, Massachusetts, USA
10x TBE	National Diagnostics, Atlanta, USA
Akt antibody, Phospho-Akt (Ser473) antibody, Phospho-Akt (Thr308) antibody, Akt kinase assay kit, Phospho ATF-2 (Thr71) antibody, DNA markers, Phospho-FKHR (Ser256) antibody, Phospho-GSK-3β (Ser9) antibody, JNK kinase assay kit, JNK total antibody, Phospho-JNK (Thr183/Tyr185) antibody, c-jun antibody, Phospho c-jun (Ser63) antibody, Phospho-PDK1 (Ser241) antibody, Phospho- PTEN (Ser 380) antibody, Restriction endonucleases and buffers	New England Biolabs, Herts, UK
Lymphoprep™	Nycomed Pharma, Oslo, Norway
TGF-β, TNF-α	PeproTech EC Ltd, London, UK
Immunopure®TMB substrate kit	Perbio Science UK Ltd, Cheshire, UK

Deoxyribonucleoside triphosphate, Random hexanucleotides	Pharmacia, Hertfordshire, UK
Micro BCA™ protein assay kit	Pierce, Chester, UK
Purified CK2 enzyme, 5x Passive Lysis Buffer, Wizard® <i>Plus</i> SV miniprep kit, MMLV reverse transcriptase, 5x reverse transcriptase buffer, Firefly Luciferase assay reagent, RNasin® Ribonuclease inhibitor	Promega, Southampton, UK
Effectene™ transfection reagent, RNeasy Total RNA isolation kit, Superfect™ transfection reagent	Qiagen Ltd, West Sussex, UK
Casein kinase 2 ( $\alpha$ -subunit) antibody, Protein A/G agarose	Santa Cruz Biotechnology Inc., California, USA
Accuspin <sup>™</sup> tubes, β-Actin antibody, Alkaline Protease, Ampicillin, Apigenin, Aprotonin, APS, Benzamidine, Bromophenol blue, BSA, β-Casein, DTT, DMSO, Ethidium Bromide, Gamma-globulin, Glycerol, HEPES, 22-(R) hydroxycholesterol, 22-(S) hydroxycholesterol, LB-Agar capsules, LB-Medium, Leupeptin, β-mercaptoethanol, Mineral oil, MOPS, ONPG, Penicillin/Streptomycin, Pepstatin A, PMA, PMSF, Ponceau S solution, TEMED, Tissue culture water, Triton X-100, Trypan blue solution, TWEEN-20, TWEEN-40, zinc chloride	Sigma-Aldrich Company Ltd, Dorset, UK
RT-PCR primers	Sigma-Genosys, Cambridgeshire, UK
Mammalian transfection kit (DEAE-dextran)	Stratagene, California, USA
Buffy coat, human serum	Welsh blood service, Talbot Green, UK
# 2.1.1. PLASMIDS

During this study several DNA constructs were used to analyse promoter activity and to investigate various signalling pathways. All constructs were kind gifts from various laboratories and the details of the cloning process for each construct can be found in the appropriate reference (Table 2.2).



PLASMID	REFERENCE
ApoE –3000/+86 pGL3	Basheeruddin et al., 1994
pGL-apoE-890	Laffitte <i>et al.</i> , 2001b
pGL-apoE-890-ME.1.	Laffitte <i>et al.</i> , 2001b
pGL-apoE-890-ME.2.	Laffitte <i>et al.</i> , 2001b
LPL-1824/+188	Gimble <i>et al.</i> , 1989
pSG-alpha (K68A)	Heriche et al., 1997
SAPKa-VPF	Moriguchi et al., 1995
SEK1D/N	Sanchez <i>et al</i> ., 1994

# 2.1.2. BACTERIAL CULTURE MEDIA

Both the LB-medium and LB-agar were supplied in capsule form and made up according to the manufacturer's instructions.

# 2.1.3. STOCK SOLUTIONS

All stock solutions used in the procedures described in this chapter are shown in Tables 2.3, 2.4, 2.5, 2.6, 2.7, 2.8 and 2.9. The solutions were prepared using double distilled water and were of molecular biology grade unless otherwise stated.

# TABLE 2.3. SOLUTIONS USED FOR THE ISOLATION OF DNA

SOLUTIONS	COMPOSITION
Miniprep Resuspension solution	50mM Tris-HCI (pH 7.5), 0.1mM EDTA, 100µg/ml RNase A
Miniprep Lysis solution	0.2M sodium hydroxide, 1% (w/v) SDS
Miniprep Neutralisation solution	2.55M potassium acetate (pH 4.8)
Miniprep Column Wash solution	200mM NaCl, 20mM Tris-HCl (pH 7.5), 5mM EDTA
Miniprep Resin	1.5% (w/v) celite resin in 7M guanidine hydrochloride (pH 5.5)
Maxiprep Cell Suspension buffer	50mM Tris-HCI (pH 8.0), 10mM EDTA, 0.2mg/ml RNase A
Maxiprep Cell Lysis solution	200mM NaOH, 1% (w/v) SDS
Maxiprep Neutralization buffer	3.1M potassium acetate (pH 5.5)
Maxiprep Equilibration buffer	600mM NaCl, 100mM sodium acetate (pH 5.0), 0.15% (v/v) Triton® X-100
Maxiprep Wash buffer	800mM NaCl, 100mM sodium acetate (pH 5.0)
Maxiprep Elution buffer	1.25M NaCl, 100mM Tris-HCl (pH 8.5)
TE buffer	10mM Tris-HCI (pH 8.0), 0.1mM EDTA

Solution	COMPOSITION
10x TBE	890mM Tris-HCI, 890mM boric acid, 20mM EDTA
10x DNA loading dye	1xTBE, 50% (v/v) glycerol, 0.25% (w/v) bromophenol blue
RNA loading dye stock	50% (v/v) glycerol, 1M EDTA, 0.25% (w/v) bromophenol blue, pH 8.0
2x RNA loading buffer	RNA loading dye stock: 10x MOPS: formaldehyde: formamide [4:4:7:20(v/v)]
20x MOPS	400mM MOPS, 100mM sodium acetate, 100mM EDTA, pH 7.0

# TABLE 2.4. SOLUTIONS FOR ELECTROPHORESIS OF DNA AND RNA

# TABLE 2.5. SOLUTIONS FOR CELL TRANSFECTIONS

SOLUTION	COMPOSITION
2x β-galactosidase assay buffer	200mM sodium phosphate buffer (120mM Na <sub>2</sub> HPO <sub>4</sub> , 80mM NaH <sub>2</sub> PO <sub>4</sub> ) [pH 7.3], 2mM MgCl <sub>2</sub> , 100mM $\beta$ -mercaptoethanol, 1.33mg/m ONPG
2x HBS	10g/l Hepes, 16g/l NaCl, pH 7.1
100x PO₄ buffer	70mM Na <sub>2</sub> HPO <sub>4</sub> , 70mM NaH <sub>2</sub> PO <sub>4</sub> in a 1:1 ratio

SOLUTION	COMPOSITION
PBS	137mM NaCl, 2.7mM KCl, 8.1mM Na <sub>2</sub> HPO <sub>4</sub> , 1.5mM KH <sub>2</sub> PO <sub>4</sub>
Whole Cell Extraction buffer	10mM HEPES (pH 7.9), 400mM NaCl, 0.5mM DTT, 2μg/ml Aprotinin, 0.5mM PMSF, 5% (v/v) glycerol, 2mM Benzamidine, 0.1mM EDTA
Phosphatase-free whole cell extraction buffer	10mM Tris-HCI (pH 7.05), 50mM NaCI, 50mM NaF, 1% (v/v) Triton X-100, 30mM tetra- sodium pyrophosphate (Na <sub>4</sub> P <sub>2</sub> O <sub>7</sub> ), 5μM ZnCl <sub>2</sub> , 100μM sodium orthovanadate (Na <sub>3</sub> VO <sub>4</sub> ), 1mM DTT, 2.8μg/ml aprotinin, 2.5μg/ml leupeptin, 2.5μg/ml pepstatin, 0.5mM benzamidine, 0.5mM PMSF

# TABLE 2.6. SOLUTIONS FOR THE PREPARATION OF CELL EXTRACTS

# TABLE 2.7. SOLUTIONS FOR WESTERN BLOT ANALYSIS OF PROTEINS

SOLUTION	
SDS-PAGE upper buffer	1M Tris-HCI (pH 6.8), 10% (w/v) SDS
SDS-PAGE lower buffer	1.5M Tris-HCI (pH 8.8), 10% (w/v) SDS
SDS-PAGE 10x running buffer	25mM Tris-HCI (pH 8.3), 250mM glycine, 0.1% (w/v) SDS
SDS-PAGE gel loading buffer	50mM Tris-HCl (pH 6.8), 20% (w/v) sucrose, 2% (w/v) SDS, 0.1% (w/v) bromophenol blue, 10% (v/v) glycerol, 5%(v/v)β-mercaptoethanol
Western blot transfer buffer	25mM Tris-HCI (pH 8.3), 192mM glycine, 20% (v/v) methanol
1x TBS	10mM Tris-HCl, 200mM NaCl, pH 7.4

# TABLE 2.8. SOLUTIONS FOR ELISA

SOLUTION	Composition
ELISA assay buffer	0.5% (w/v) BSA, 0.05% (w/v) Gamma- globulin, 50mM Tris-HCI, 150mM NaCl, 0.01% (v/v) TWEEN-40, pH 7.4
Blocking Buffer	1% BSA (w/v), 50mM Tris-HCl, 150mM NaCl, 1mM MgCl <sub>2</sub> , pH 7.4
Washing Buffer	10mM Tris-HCl, 150mM NaCl, 0.05% (v/v) TWEEN-20, pH7.4

# TABLE 2.9. SOLUTIONS FOR PROTEIN KINASE ASSAYS

SOLUTION	Composition
CK2 kinase buffer	100mM Tris-HCI (pH 8.0), 100mM NaCl, 20mM MgCl₂, 50mM KCl, 100µM Na₃VO₄, 5mg/ml β-casein
1x Akt kinase buffer	25mM Tris-HCl (pH 7.5), 5mM β- glycerophosphate, 2mM DTT, 0.1mM Na <sub>3</sub> VO <sub>4</sub> , 10mM MgCl <sub>2</sub>
1x JNK kinase buffer	25mM Tris-HCl (pH 7.5), 5mM β- glycerophosphate, 2mM DTT, 0.1mM Na <sub>3</sub> VO <sub>4</sub> , 10mM MgCl <sub>2</sub>

# 2.1.4. PREPARATION OF CERAMICS, PLASTICS AND GLASSWARE

All glassware, ceramics and plastics (including pipette tips and Eppendorf tubes) were autoclaved for 20min at 121°C and at a pressure of 975 kPa.

#### 2.2. ISOLATION AND MANIPULATION OF DNA

# 2.2.1. PREPARATION OF COMPETENT CELLS

For the preparation of competent bacterial cells, 10ml of LB-medium was inoculated with a single bacterial colony of the DH5 $\alpha$  *E.coli* strain [*supE44*,  $\Delta$  *lacU169*, ( $\Phi$ 80 *lac Z* $\Delta$ *M15*), *hsdR17*, *recA1*, *endA1*, *gyrA96*, *thi-1*, *relA1*] and incubated overnight at 37°C, with moderate shaking. Following incubation, 100 $\mu$ l of the culture was used to inoculate 100ml of fresh LB-medium and incubated with moderate shaking at 37°C, until an OD<sub>550</sub> of between 0.5 and 0.6 was reached. The cells were then pelleted by centrifugation at 3,000g at 4°C for 5min and resuspended in 50ml of ice-cold CaCl<sub>2</sub> solution (50mM). The suspension was left on ice for 25min and the cells were again pelleted by centrifugation (3,000g for 5min at 4°C). The cells were then resuspended in 10ml of ice-cold CaCl<sub>2</sub> solution (50mM) and mixed with 40% (v/v) sterile glycerol. The suspension was aliquoted into chilled Eppendorf tubes and stored at -70°C.

# 2.2.2. TRANSFORMATION OF COMPETENT CELLS

For each transformation, 10ng of recombinant plasmid DNA was added to  $200\mu$ l of competent cells (Section 2.2.1) and subsequently incubated on ice for 40min. The cells were then heat-shocked for 90sec at 42°C. Following heat shock treatment, the cells were placed on ice for 2min and 800µl of pre-warmed (37°C) LB-medium was added. The mixture was then left to incubate for 1h at 37°C with moderate shaking. After incubation, 200µl aliquots were spread on agar plates containing ampicillin (100µg/ml) for selection of transformed bacteria and left to incubate overnight at 37°C. An ampicillin plate spread with untransformed cells was used as a control for each transformation.

# 2.2.3. SMALL-SCALE DNA PREPARATION (MINIPREP PROTOCOL)

Small-scale preparation of recombinant plasmid DNA was carried out using the Wizard® *Plus* SV miniprep kit (Promega) according to the manufacturer's protocol. A single colony of transformed bacteria was used to inoculate 10ml of LB-medium containing 100µg/ml of ampicillin and incubated overnight at 37°C, with moderate shaking. Cells were pelleted by centrifugation at 10,000g, at room temperature for 5min and resuspended in 250µl Cell Resuspension solution (Table 2.3). The suspension was then transferred to a microcentrifuge tube and 250µl of Cell Lysis solution was added (Table 2.3). The suspension was mixed by inversion and clearing was observed confirming that lysis was complete. Alkaline Protease (10µl) was added to the solution and after inversion the mixture was incubated at room temperature for 5min. Once the incubation period was over, 350µl of Neutralisation solution (Table 2.3) was added, mixed by inversion and centrifuged at 14,000g for 10min. The clear supernatant was transferred to a spin column inserted into a collection tube and

subjected to centrifugation at 14,000g for 1min. The flow through was discarded and the spin column was washed twice with Column Wash solution (Table 2.3). The column was then dried by centrifugation at 14,000g for 2min. The DNA was eluted from the column by adding 50µl of Nuclease-Free water and centrifuging at 14,000g for 1min.

# 2.2.4. LARGE-SCALE DNA PREPARATION (MAXIPREP PROTOCOL)

Large-scale preparation of plasmid DNA was carried out using the Concert<sup>™</sup> plasmid purification kit (Gibco) according to the manufacturer's protocol. A single colony of transformed bacteria was used to inoculate 5ml of LB-medium containing 100µg/ml of ampicillin and incubated for 4-6h at 37°C with moderate shaking. This mini-culture was then added to 200ml of LB-ampicillin medium and left to grow overnight with shaking at 37°C. The cells were then pelleted by centrifugation at 6,000g for 20min, at a temperature of 4°C. All medium was thoroughly removed and the resulting pellet was resuspended, using 10ml of Cell Suspension buffer (Table 2.3). The mixture was lysed using 10ml of Cell Lysis solution (Table 2.3) and mixed gently by inversion. Following incubation at room temperature for 5min, 10ml of Neutralization solution (Table 2.3) was added. The solution was mixed by inversion and centrifuged at 15,000g at room temperature for 10min. The supernatant was then transferred to a column, which had previously been equilibrated by allowing 30ml of Equilibration buffer (Table 2.3) to drain through the column by gravity flow. The resulting flow-through was discarded and the column was washed with 60ml of Wash buffer (Table 2.3). The DNA was then eluted by adding 15ml of Elution buffer (Table 2.3) to the column. Once all the solution had drained from the column by gravity flow, 10.5ml of isopropanol was added to precipitate the DNA. The solution was mixed and centrifuged at 15,000g for 30min, at a temperature of 4°C. The pellet was washed with 5ml of 70% (v/v) ethanol by centrifugation at 15,000g for 5min at 4°C. The ethanol was removed and the DNA pellet was left to air-dry for 10min and then dissolved in 500µl of TE buffer (Table 2.3).

# 2.2.5. QUANTIFICATION OF DNA

The concentration and purity of recombinant plasmid DNA was determined by measuring the ratio of absorbance at 260nm and 280nm, percentage purity and yield in  $\mu g/\mu l$  of each sample using the GeneQuant spectrophotometer (Pharmacia).

### 2.2.6. RESTRICTION ENDONUCLEASE DIGESTION OF DNA

Restriction endonuclease digestion of DNA was carried out using buffers and conditions recommended by the supplier (New England Biolabs). The incubation times varied depending on the enzymes with the majority of the reactions being incubated for at least 2h at 37°C. The digests

were typically carried out using 10 Units of each enzyme. The only exception was in the case of digests involving KpnI where 20 Units were used due to the enzyme's low activity. All double digests were performed sequentially, where the first digest step was followed by the addition of a single volume of sterile water, the second enzyme and the recommended buffer. For each digest performed in this study, the reaction mixture was supplemented with 100µg/mI BSA. All reactions were stopped by incubation at 65°C for 10min.

# 2.2.7. GEL ELECTROPHORESIS OF DNA

Size-fractionation of DNA was carried out using agarose gel electrophoresis. The gels used in the present study were 0.7% to 1.5% (w/v) and the sizes of the various fragments were determined by comparison of their migration with standard molecular weight markers (Appendix I). Both the electrophoresis buffer and gel contained ethidium bromide in 0.25x TBE at a final concentration of  $0.5\mu$ g/ml. Before loading, all samples were mixed with 10x DNA loading buffer (Table 2.4) where bromophenol blue was used as the tracking dye. Electrophoresis was routinely performed in a horizontal gel apparatus (GNA-100 and GNA-200, Pharmacia) at 100V for 1.5-2h. Following electrophoresis, the DNA was visualised using an ultraviolet transilluminator and the gel photographed using Genetools (Syngene).

# 2.3. TISSUE CULTURE AND RELATED TECHNIQUES

#### 2.3.1. MAINTENANCE OF CELL LINES IN CULTURE

The human monocyte THP-1 and U937 cell lines were both grown in RPMI-1640 with Stabilix supplemented with 10% (v/v) heat-inactivated (30min, 56°C) foetal calf serum (HI-FCS) in the presence of penicillin (100U/ml) and streptomycin (100ug/ml). Before addition to the media, all solutions were filter-sterilised by passing them through a 0.2µm sterile filter. The cells were maintained at 37°C in a humidified atmosphere containing 5% (v/v) CO<sub>2</sub>. The cell culture medium for each cell line was changed every three days.

# 2.3.2. SUB-CULTURING OF CELLS

Both monocyte cell lines (THP-1 and U937) were sub-cultured when they reached approximately 70% confluency (1x10<sup>6</sup> cells/ml). The U937 cell line was centrifuged at 1,000g for 5min and the pellet resuspended in medium containing 10% (v/v) HI-FCS. The cells were then plated into fresh tissue culture flasks at a ratio of 1:4 (i.e. cells from 1 tissue culture flask was sub-cultured into 4 new flasks) and incubated as above. The THP-1 cell line was also sub-cultured at a ratio of 1:4, which was carried out by dilution of the appropriate amount of cells into fresh medium containing 10% (v/v) HI-FCS.

#### 2.3.3. HUMAN PRIMARY MONOCYTE-DERIVED MACROPHAGE CELL CULTURE

#### 2.3.3.1. ISOLATION OF PRIMARY MONOCYTE-DERIVED MACROPHAGES

The isolation of human monocytes was carried out using Accuspin<sup>™</sup> centrifuge tubes (Sigma-Aldrich) and the separation medium used was Lymphoprep<sup>™</sup> (Nycomed Pharma). To each Accuspin<sup>™</sup> centrifuge tube, 15ml of Lymphoprep<sup>™</sup> (warmed to room temperature), was layered on top of the polyethylene barrier ("frit") and centrifuged at 800g for 30sec in order to place the Lymphoprep<sup>™</sup> in the chamber below the "frit". After this brief centrifugation step, 30ml of buffy coat (Welsh blood service) was layered over the "frit" and centrifuged at 800g for 30min at room temperature. The resultant mononuclear band was collected and transferred to a new centrifuge tube, to which an equal volume of ice-cold PBS-0.4% (w/v) tri-sodium citrate was added. Cells were pelleted at 800g for 5min at 4°C, after which a 0.2% saline solution (10ml) was added to the pellet (on ice) for 30sec followed by a 1.6% saline solution (10ml), and centrifuged immediately at 800g for 5min at 4°C. The resultant pellet was then washed 6-8 times with ice-cold PBS-0.4% (w/v) tri-sodium citrate to remove contaminating platelets. Monocytes were plated out in 75cm<sup>2</sup> flasks (8x10<sup>6</sup>cells/10ml/flask) in RPMI-1640 with Stabilix containing antibiotics (100U/ml penicillin and 100µg/ml streptomycin) and supplemented with 5% (v/v) human serum. The cells were allowed to adhere for 3h in an incubator at 37°C, 5% (v/v) CO<sub>2</sub> and then the medium was replaced with fresh culture medium. The monocytes were then left to differentiate for 7-10 days before use in experiments.

# 2.3.3.2. CULTURE ROUTINE OF HUMAN PRIMARY MONOCYTE-DERIVED MACROPHAGES

Every day, the medium in which the cells were growing was removed and replaced with fresh culture medium pre-warmed to 37°C, containing the appropriate antibiotics (100U/ml penicillin and 100µg/ml streptomycin) and 5% (v/v) human serum.

#### 2.3.4. DELIPIDATION OF HI-FCS

Lipids were removed from HI-FCS according to the method of Cham and Knowles (1976) with minor modification. To 10ml of HI-FCS, 20ml of 2:3 (v/v) butanol:di-isopropyl ether (DIPE) was added and mixed with rotation for 30min at room temperature. The phases were then separated by centrifugation at 2,000g for 2min and the serum removed. In order to remove any residual solvent, nitrogen was bubbled through the serum for 2-3h.

# 2.3.5. TRANSIENT TRANSFECTION OF THE THP-1 AND U937 CELL LINES

Prior to each transfection (about 24h), the cells were split 1:2 and put into fresh flasks. Approximately  $1 \times 10^6$  cells were seeded into each well of a 6-well plate (Greiner) four hours prior to transfection. All cells used for transfection experiments were between passages 3 and 10.

# 2.3.5.1. TRANSIENT TRANSFECTION OF CELLS USING SUPERFECT™

Transient transfection with Superfect<sup>™</sup> was carried out as described in the manufacturer's protocol (Qiagen). For each transfection, 2µg of the appropriate reporter DNA construct and 0.5µg of the internal control plasmid were made up to 100µl with serum-free and antibiotic-free culture medium. Each transfection DNA mix was then mixed by vortexing with the Superfect<sup>™</sup> transfection reagent (a charge ratio of 1:4 was used for all transfections carried out in this study i.e. 4µl of Superfect<sup>™</sup> was used per 1µg of DNA). The mixture was then left to incubate at room temperature for 10min in order to allow the formation of the DNA:Superfect<sup>™</sup> complexes. Complete medium (containing HI-FCS and penicillin/streptomycin) was then added to each transfection mixture (400µl) and the transfection complexes were added drop-wise onto the cells. The 6-well plates were swirled to allow even distribution of the complexes and then left to incubate (37°C, 5% (v/v) CO<sub>2</sub>) for the appropriate length of time for gene expression.

# 2.3.5.2. TRANSIENT TRANSFECTION OF CELLS USING EFFECTENE™

Transient transfection with Effectene<sup>TM</sup> was carried out as described in the manufacturer's protocol (Qiagen). For each transfection,  $0.4\mu g$  of the appropriate plasmid construct and  $0.2\mu g$  of the internal control plasmid were made up to  $100\mu l$  with buffer EC (Qiagen), along with  $3.2\mu l$  of the Enhancer reagent (Qiagen). The mixture was mixed by vortexing and incubated at room temperature for 5min. The DNA-Enhancer solution was then centrifuged for a few seconds and  $10\mu l$  of the Effectene<sup>TM</sup> reagent was added and mixed by pipetting. The solution was left to incubate at room temperature for 10min and  $600\mu l$  of complete cell culture medium was added to the transfection mixture. The transfection complexes were then added drop-wise to the cells and the plates were swirled to ensure uniform distribution of the complexes. The cells were incubated at  $37^{\circ}C$  with 5% (v/v) CO<sub>2</sub> for the appropriate length of time.

# 2.3.5.3. TRANSIENT TRANSFECTION OF CELLS USING THE DEAE-DEXTRAN MAMMALIAN TRANSFECTION KIT

Transient transfection with the DEAE-dextran mammalian transfection kit was carried out as described in the manufacturer's protocol (Stratagene) with only minor modification. In order for suspension cells to be transfected using this system it was necessary to bulk transfect the cells in

15ml polypropylene tubes (Falcon). Therefore, cells at a concentration of 1x10<sup>6</sup>cells/ml were washed twice with PBS and re-suspended in serum-free, antibiotic-free medium to the same concentration. Before transfection, the transfection solutions-A (2µg of DNA made up to 170µl with PBS) and -B (85µl of solution no.3 [2mg/ml DEAE-dextran in 0.9% NaCl] with 85µl PBS) were prepared. The solutions (A and B) were combined and added drop-wise to the cells. The transfection mix was left at room temperature for 15min and then removed from the cells. The cells were washed twice with PBS and resupended in complete medium. The cells were then aliquoted into 6-well plates and the final volume for each well was made up to 2ml with fresh medium. The plates were then left to incubate for 36h at 37°C with 5% (v/v) CO<sub>2</sub>.

# 2.3.6. DIFFERENTIATION OF MONOCYTIC CELL LINES

Differentiation of monocytes was routinely carried out by the addition of an appropriate concentration (depending on the experimental conditions) of the differentiation agent PMA. Prior to each experiment (about 24h), the cells were split 1:2 and put into fresh flasks. Before each experiment, the cells were plated out at the appropriate density and left to incubate for 4h, following which, PMA was added to the cells. For all transfection experiments, PMA was added once the transfection of cells was complete. After the addition of PMA, the cells were left to incubate as normal (Section 2.3.3.1) for the requisite time period.

#### 2.3.7. TREATMENT OF CELLS WITH EXTRACELLULAR MEDIATORS

For certain transfection experiments it was necessary to treat the transfected cells with the extracellular mediator TNF- $\alpha$ , in order to monitor the effect of this cytokine on promoter activity. Therefore, where necessary, the transient transfection protocol (Sections 2.3.5.1, 2, 3 and 4) was modified so that 16h after transfection, TNF- $\alpha$  was added to provide a final concentration of 1000U/ml. The cells were then left to incubate for a further 36h at 37°C with 5% (v/v) CO<sub>2</sub>.

# 2.3.8. TREATMENT OF CELLS WITH OXYSTEROL LIGANDS

For certain experiments the addition of the LXR ligands 22(R)- and 22(S)-hydroxycholesterol was necessary. The ligands were added 24h after PMA treatment of cells, at a concentration of  $2\mu g/ml$ . The cells were then left to incubate as normal (Section 2.3.3.1) for the requisite time period.

# 2.3.9. TREATMENT OF CELLS WITH INHIBITORS

Investigation of the signalling pathways involved in LXR regulation of apoE required the addition of various inhibitors 1h prior to the addition of the above ligands. Once the inhibitors were added in

their appropriate concentrations (depending on the experimental conditions) the cells were left to incubate as normal (Section 2.3.3.1) for the requisite incubation period of the study.

# 2.3.10. PREPARATION OF CELL EXTRACTS FOR THE DETERMINATION OF REPORTER GENE ACTIVITY

Following the completion of transient transfection (Sections 2.3.5.1, 2, 3), cells were detached from the plate using a cell scraper and the mixture was transferred to a microcentrifuge tube. The sample was centrifuged for 1min at 12,000g and the pellet was washed twice with PBS. The pellet was then re-suspended in 200µl of 1x Passive Lysis buffer (Promega) and left for 10min at room temperature. The cell lysate was centrifuged for 2min at 15,000g and the supernatant was retained and used for the determination of the reporter gene activity.

# 2.3.11. MEASUREMENT OF LUCIFERASE ACTIVITY

During this study, the apoE promoter and enhancer activity was monitored using the luciferase gene as the reporter. The assay was carried out essentially as described in the manufacturer's protocol (Promega). For each assay,  $40\mu$ I of cell extract (Section 2.3.9) was mixed with  $100\mu$ I of the Luciferase Assay Reagent (Promega) in a luminometer tube, and the light output for each sample was determined in duplicate using a Turner Designs 50/50 Luminometer. The luminometer was set at a sensitivity value of 70% with a 2sec-delay period and 20sec integrate period.

# **2.3.12. Measurement of \beta-galactosidase activity**

In some of the transfections carried out,  $\beta$ -galactosidase was used as an internal control for transfection efficiency and the luciferase activity was normalised to this value. The assay for  $\beta$ -galactosidase was carried out in 96-well plates (Greiner). In each well, 50µl of cell extract (Section 2.3.10) was mixed with 50µl of 2x  $\beta$ -galactosidase assay buffer (Table 2.5). The assay mixture was then incubated at 37°C for 1h. The reaction was terminated by the addition of 100µl of 1M sodium carbonate and the absorbance of the samples were measured at 405nm, using a Titertek Multiscan MCC/340 microplate reader.

For certain transfection assays it was necessary to perform a protein assay in order to normalise the luciferase activity of the cell extracts, as described in Section 2.5.2.

# 2.3.13. TRYPAN BLUE EXCLUSION ASSAY

Cell viability was assessed using trypan blue. Briefly, cells were adjusted to a density of 10<sup>6</sup>/ml and 20µl of this suspension was mixed with an equal volume of 0.4% (v/v) trypan blue and left for 5min. The cells were then placed on a haemocytometer and viewed using phase contrast microscopy. At

least 200 cells were counted and the percentage of dead cells (i.e. the cells that had absorbed the stain) was estimated.

### 2.4. RNA-RELATED TECHNIQUES

# 2.4.1. RNA ISOLATION

Total cellular RNA was purified from cells using the RNeasy Total RNA isolation kit (Qiagen) according to the manufacturer's protocol. The cells used for RNA extraction in this study were the THP-1 cell line, therefore the method used was that for cells grown in suspension. The cells (8x 10<sup>6</sup> per RNA extraction) and their media were transferred to a 15ml polypropylene tube (Falcon) and centrifuged at 1,000g for 5min. The resulting supernatant was discarded and the cell pellet was lysed by the addition of 600μl of Buffer RLT (Qiagen) [containing 10μl/ml of β-mercaptoethanol]. The solution was mixed by pipetting to remove all clumps and then homogenized by passing the lysate through a 0.9mm needle attached to a 1ml syringe for at least 10 times. Once the homogenization step was complete 600µl of 70% (v/v) ethanol was added and the lysate was mixed well by pipetting. The sample was transferred 700µl at a time to an RNeasy spin column inserted into a 2ml collection tube and centrifuged at 8,000g for 15sec. The flow-through was discarded and 700µl of buffer RW1 (Qiagen) was used to wash the column by centrifuging the sample for 15sec at 8,000g. Once again the flow-through was discarded and the RNeasy column was transferred to a new collection tube. The column was washed again by adding 500µl of buffer RPE (Qiagen) and centrifuging the column for 15sec at 8,000g. The final wash was carried out using 500µl of buffer RPE (Qiagen) and centrifuging the column for 2min at 10,000g. The RNA was eluted from the column by placing the column into a new 1.5ml collection tube and adding 50µl of RNase-free water onto the column. The column was then centrifuged for 1 min at 8,000g and the purified RNA was stored at -70°C.

# 2.4.2. ANALYSIS OF RNA ON DENATURING AGAROSE GELS

Size-fractionation and determination of RNA quality was carried out using denaturing formaldehyde agarose gels. The gels routinely used in this study were 1% (w/v) agarose gels, and were prepared using sterile double distilled water. For a 100ml gel, 1g of agarose was dissolved in 77ml of water to which were then added 5ml of 20x MOPS (Table 2.4) and 18ml of 37% (v/v) formaldehyde. Each RNA sample (5µl) was prepared by adding an equal volume of RNA loading buffer (Table 2.4) and 1µl of 1mg/ml ethidium bromide. The mixture was heated at 65°C for 10min and then cooled on ice before loading. Electrophoresis was performed for approximately 2h at 75V in a GNA-100 horizontal gel apparatus (Pharmacia) using 1x MOPS as the "running" buffer. The

ethidium bromide stained RNA was then visualised and photographed as described previously (Section 2.2.7).

# 2.4.3. QUANTIFICATION OF RNA

All isolated RNA was quantified using a GeneQuant spectrophotometer (Pharmacia) as described previously for DNA (Section 2.2.5).

# 2.4.4. Reverse Transcription Polymerase Chain Reaction (RT-PCR)

RT-PCR was carried out in two stages; cDNA synthesis using reverse transcriptase and the subsequent amplification of the synthesised cDNA using the Polymerase Chain Reaction.

# 2.4.4.1. COMPLEMENTARY DNA (CDNA) SYNTHESIS FROM TOTAL RNA

For each RNA sample, 1µg of total RNA was used as a template to generate random primed cDNA. To 1µg of total RNA in 12.5µl of RNase-free water, 200pmole of random hexamer primers were added. The mixture was incubated at 70°C for 5min and then cooled on ice. Then, 4µl of 5x reverse transcriptase NH₄ buffer (Bioline), 1µl of dNTP mix (10mM each), 0.5µl of 40U/µl RNase inhibitor and 1µl of 200U/µl Molony murine Leukemia virus (MMLV) reverse transcriptase were added to the mixture and incubated at 42°C for 1h. The reaction was then terminated by heating the mixture to 94°C for 2min and the cDNA was diluted 5-fold by adding 80µl of sterile double distilled water. The cDNA was either used immediately for Polymerase Chain Reaction (Section 2.4.4.2.) or stored at -70°C.

# 2.4.4.2. POLYMERASE CHAIN REACTION

During this study all PCR reactions were carried out in a final volume of 50ul. Each standard PCR reaction contained Taq polymerase (1.25Units), 10x PCR buffer (Bioline) [50mM KCl, 10mM Tris-HCl (pH 9.0), 1% (v/v) Triton® X-100], 0.1mM of each dNTP, 50pmoles of both the 3' and 5' primers. The concentration of MgCl<sub>2</sub> was varied according to the template/primer set used, and in some cases DMSO (5%, v/v) was added to the reaction to increase amplification specificity. Each reaction mixture was overlaid with 50ul of mineral oil and the cycle programmes used were optimised for each template/primer set. All PCR reactions were carried out using a Biometra TRIO Thermoblock. The PCR amplification products were analysed by agarose gel electrophoresis using a 1.5% (w/v) agarose gel as described in Section 2.2.6. The sequences of primers used are shown in Table 2.10.

# TABLE 2.10. SEQUENCES OF PRIMERS USED IN PCR ANALYSIS

TARGET MRNA	FORWARD PRIMER	Reverse Primer
Human ApoE	5'-TTCCTGGCAGGATGCCAGGC-3'	5'-GGTCAGTTGTTCCTCCAGTTC-3'
Human ABCA1	5'-GTTGGAAAGATTCTCTATACACCTGA-3'	5'-CGTCAGCACTCTGATGATGGCCTG-3'
Human LXR $\alpha$	5'-ATGGTGGATGGAGACGTAG-3'	5'-ATGGACACCTACATGCGTC-3'
Human GAPDH	5'-CCCTTCATTGACCTCAACTACATG-3'	5'-AGTCTTCTGGGTGGCAGTGATGG-3'

The primer sequences were obtained from the following references: Human ApoE (Wang et al., 1989); human ABCA1 (Klucken et al., 2000); human LXR $\alpha$  (Kohro et al., 2000).

TABLE 2.11. OPTIMISED PCR	TEMPERATURE AND CYCLING CONDITIONS FOR EACH PRIMER SET <sup>1</sup>

TARGET MRNA	PCR PROGRAMME
Human ApoE	Initial treatment of 96°C for 5min, 24 cycles of [60°C(1min), 72°C(2min), 93°C(30sec)], and 72°C for 10min at end
Human ABCA1	Initial treatment of 92.5°C for 2min, 31 cycles of [60°C(44sec), 60°C(40sec), 71°C(46sec)], and 72°C for 10min at end
Human LXRα	Initial treatment of 96°C for 5min, 32 cycles of [60°C(1min), 72°C(2min), 93°C(30sec)], and 72°C for 10min at end
Human GAPDH	Initial treatment of 96°C for 5min, 18 cycles of [60°C(1min), 72°C(2min), 93°C(30sec)], and 72°C for 10min at end

<sup>1</sup>The optimised conditions for each primer set were within the exponential phase of amplification and, therefore, provided a direct correlation between the amount of amplification products and RNA template abundance in the sample.

# 2.4.5. TAQMAN REAL-TIME QUANTITATIVE RT-PCR

The cDNA synthesis step was carried out as described in section 2.4.4.1. The PCR reactions were carried out in 96 sample tubes in a final volume of 29 $\mu$ l. Each standard PCR reaction contained 2x Taqman Master mix (Applied Biosystems), 300nM of both the 3' and 5' primers and 300nM of the probe, and 25ng of cDNA. The mixture was then overlaid with 12 $\mu$ l of mineral oil. Reactions were cycled on an ABI PRISM 7700 Sequence detector (Applied Biosystems) using the following default programme: a 50°C incubation for 2min, 95°C for 10min, followed by 40 cycles of 95°C for 15sec for denaturation and 60°C for annealing and extension. Each set of reactions also contained a standard curve, where human liver cDNA was diluted to give a range of concentrations (60ng/ $\mu$ l, 30ng/ $\mu$ l, 10ng/ $\mu$ l, 3.34ng/ $\mu$ l, 1.11ng/ $\mu$ l, 0.38ng/ $\mu$ l, 0.124ng/ $\mu$ l, 0.05ng/ $\mu$ l, 0.014ng/ $\mu$ l, 0.0046ng/ $\mu$ l, 0.00152ng/ $\mu$ l, 0.0005ng/ $\mu$ l). Reactions were monitored in real time mode using Sequence Detection Systems, version 1.7 software. Samples were normalised using the  $\beta$ -microtubulin Taqman probe. The sequences of the primers and probes used are shown in Table 2.12.

# 2.5. PROTEIN ANALYSIS

# 2.5.1. PREPARATION OF WHOLE CELL EXTRACTS

The cells used for the isolation of whole cell extracts in this study were the THP-1 cell line. The cells ( $8x \ 10^6$  per isolation) and their media were transferred to a 15ml Polypropylene tube (Falcon) and centrifuged at 1000g for 5min. The pellet was then washed three times in cold PBS by centrifugation for 1min at 10,000g. The resulting pellet was placed on dry ice for 5min and then resuspended in 5 volumes of whole cell extract buffer (Table 2.6). The cells were lysed by pipetting and the mixture was then centrifuged at 10,000g for 5min at 4°C. The resulting lysate was transferred to a fresh Eppendorf tube and stored at  $-70^{\circ}$ C.

For certain experiments the maintenance of proteins in their phosphorylated state was vital, and it was therefore necessary to slightly modify the method for whole cell extract isolation. In these instances, whole cell extracts were made in a buffer containing phosphatase inhibitors as well as protease inhibitors, as described by Hipskind *et al.* (1994). The cells ( $8x10^6$  per isolation) and their media were centrifuged at 1000g for 5min and the resulting pellet was washed twice with ice-cold PBS containing 10mM NaF and 100µM sodium orthovanadate (Na<sub>3</sub>VO<sub>4</sub>). The pellet was then resuspended in 500µl of phosphatase-free whole cell extraction buffer (Table 2.6) and vortexed vigorously for 45sec at 4°C. The lysate was cleared by centrifugation at 10,000g for 10min at 4°C and stored at  $-70^{\circ}$ C.

TABLE 2.12. SEQU	TABLE 2.12. SEQUENCE OF PRIMERS AND PROBES USED IN TAQMAN REAL-TIME RT-PCR	AQMAN REAL-TIME RT-PCR	
TARGET MRNA	FORWARD PRIMER	Reverse PRIMER	PROBE
Human ApoE	5'-CGTTGCTGGTCACATTCCTG-3'	5'-GCTCTGTCTCCACCGCTTG-3'	5'-CAGGATGCCAGGCCAAGGTGGA-3'
Human ABCA1	5'-TCCAGGCCAGTACGGAATTC-3'	5'-TTCCTCGCCAAACCAGTAGG-3'	5'-AGGCCCTGGTATTTTCCTTGCACCAA-3'
Human ABCG1	5'-ATGGAAGGCTGCCACAGC-3'	5'-AGGTCCTCTTGAAGAGGATGCA-3'	5'-TCTCTGCCAGCTGCCTCACGCA-3'
Human LXRa	5'-AGCCCTGCATGCCTACGTC-3'	5'-GCATCCGTGGGAACATCAG-3'	5'-CCATCCACCATCCCCATGACCG-3'
Huamn PPARy	5'-CACAGGCCGAGAAGGAGAA-3'	5'-TCGATATCACTGGAGATCTCCGCCCAAC-3'	5'-TCAGCGGACTCGGATTCAG-3'
Human β- microtubulin	5'-CGCTCCGTGGCCTTAGC-3'	5'-GAGTACGCTGGATAGCCTCCA-3'	5'-TGCTCGCGCTACTCTCTCTTCTGGC-3'

CHAPTER TWO - MATERIALS AND METHODS

92

# 2.5.2. DETERMINATION OF PROTEIN CONCENTRATION

The concentration of total protein in the cell extracts was determined using the BCA<sup>™</sup> protein assay kit, according to the manufacturer's protocol (Pierce). A standard curve was produced for each assay using suitable dilutions of a 2mg/ml bovine serum albumin solution (BSA) in order to give concentrations of 5µg/ml, 10µg/ml, 15µg/ml, 20µg/ml and 25µg/ml. The standards along with 100µl of diluted cell extract were placed into the wells of a 96-well plate. Both the standards and the test samples were diluted in PBS. To each well, a 100µl of the BCA working reagent was added, and after mixing the plate, was left to incubate at 37°C for 2h. The plate was then left to cool to room temperature and the absorbance of each sample was read at 595nm using a Titertek Multiscan MCC/340 microplate reader. The protein concentration of each sample was then calculated from the standard curve.

# 2.5.3. IMMUNOPRECIPITATION OF PROTEINS

Immunoprecipitation of proteins was carried out using Protein A/G agarose (Santa Cruz Biotechnology). Whole cell extracts, prepared using the phosphatase-free extraction buffer (Section 2.5.1), were left to incubate overnight at 4°C with gentle rolling with an antibody raised against the desired protein ( $2\mu g/ml$ ). The resulting protein-antibody complex was then captured by the addition of Protein A/G agarose ( $20\mu l$ ) with gentle rolling for 2h at 4°C. The beads, containing the captured immunocomplex, were collected by centrifugation at 13,000g, for 3min at 4°C and washed twice with phosphatase-free extraction buffer (without Triton X-100). At this point, the protocol differed according to the desired use of the immunoprecipitated protein. For Western blot analysis, the bead/immunocomplex pellet were resuspended in 50 $\mu$ l of 0.1M Glycine, pH 2.5, vortexed, and incubated with rolling for 10min at 4°C. The sample was then centrifuged at 9,000g (4°C) for 2min and the supernatant kept. To each sample 5 $\mu$ l of 1M Tris (pH 8.0) was added along with 30 $\mu$ l of SDS-PAGE gel loading buffer. The samples were boiled for 5min and used immediately for analysis by SDS-PAGE (Section 2.5.4). For the CK2 kinase assay (Section 2.6.1), the bead/immunocomplex pellet, after the extraction buffer washes was immediately incubated with the substrate mix.

# 2.5.4. SDS-POLYACRYLAMIDE GEL ELECTROPHORESIS

During this study, SDS-PAGE was carried out using separating gels and stacking gels containing 10% (w/v) and 5% (w/v) acrylamide, respectively. For a 10% separating gel (10ml), 2.5ml of acrylamide:bisacrylamide (37.5:1) was added to 2.5ml of lower buffer (Table 2.7), whilst the 5% stacking gel (10ml) consisted of 1.25ml of acrylamide:bisacrylamide (37.5:1) in 2.5ml of upper

buffer (Table 2.7). Both gels also contained  $100\mu$ I of 10% (w/v) APS and  $10\mu$ I of TEMED and the volume of each was made up to 10ml with double distilled water.

Electrophoresis was carried out using a Mini-PROTEAN II slab electrophoresis Cell (Bio-Rad Laboratories) and was assembled according to the manufacturer's instructions. The separating gel was poured first, allowing a space of 1.5cm at the top for the stacking gel. Butan-1ol was layered on top of the gel and the gel was left to polymerise for 30-40min. The butan-1-ol was then washed away using distilled water and the stacking gel was poured on top of the separating gel. The gel comb was inserted into the gel and the stacking gel was left to polymerise. After polymerisation of the gel, the comb was removed and the wells were washed with distilled water. The gel was then placed in the tank and the inner and outer chambers were filled with 1x running buffer (Table 2.7).

Prior to loading of the gel, 10-80µg of each whole cell extract sample was mixed with an equal volume of gel loading buffer (Table 2.7) and heated to 100°C for 2min. Once cooled the samples were loaded onto the gel. Onto the first lane of each gel, 10µl of the Amersham Rainbow protein size markers (Appendix II) were loaded to allow size estimation of the detected protein. Electrophoresis was carried out at 200V for 30-45min and the gel was then subjected to Western blotting (Section 2.5.4).

#### 2.5.5. Western BLOTTING

The electrophoretic transfer of proteins from the gel to a Immobilin<sup>TM</sup>-P<sup>SQ</sup> PVDF membrane was performed using a Bio-Rad Trans-Blot Electrophoretic Transfer Cell. Following SDS-PAGE (Section 2.5.3), the gel was removed and the stacking gel was carefully cut away. The gel was then left to soak for 15min in transfer buffer (Table 2.7). In addition to the gel, the PVDF membrane (which had been activated by soaking in methanol), Whatman 3MM paper (both pre-cut to the gel size) and the sponge pads of the transfer apparatus were also pre-soaked in transfer buffer. The PVDF membrane was then placed on the gel and sandwiched between the Whatman paper and the sponge pads, before being placed into the blotting cassette. The cassette was then subjected to electro-blotting in a tank containing transfer buffer (Table 2.7) at a temperature of 4°C. The gel was left to blot at a constant voltage of 100V for 1h or over-night at a constant voltage of 15V.

# 2.5.6. IMMUNODETECTION OF PROTEINS

Electrophoretic transfer of proteins was carried out as described in Section 2.5.4. Once the electroblotting process was complete, the Rainbow markers were cut from the PVDF membrane and stored in aluminium foil. The membrane was removed from the blotting cassette and incubated in a blocking solution [TBS (Table 2.5); 0.1% (v/v) TWEEN-20, containing 5% or 10% (w/v) non-fat dried milk] for 1h. The membrane was then washed three times, for 5min each, in TBS (Table 2.7) containing 0.1% (v/v) TWEEN-20 and left to incubate in blocking solution containing an appropriate dilution of the primary antibody for 1h. Certain primary antibodies required overnight incubation at 4°C, where the 5% (w/v) non-fat dried milk in the blocking solution was replaced by 5% (w/v) BSA. The membrane was then washed in TBS containing 0.1% (v/v) TWEEN-20 as before, and left to incubate in blocking solution containing the appropriate dilution of the second antibody-Horse Radish Peroxidase (HRP) conjugate. The membrane was then washed with TBS containing 0.1% (v/v) TWEEN-20 and detection of the membrane bound antigen-antibody complexes was carried out using Enhanced Chemi-luminescence (ECL) as specified by the ECL manufacturer's instructions (Amersham). The membrane was exposed to Kodak X-ray film in a light-proof X-ray cassette (Genetic Research Instrumentation) and developed using an automated developer (Gevamatic-60, AGFA). The expected protein size was then verified by aligning the Rainbow marker with the developed image.

# 2.5.7. ENZYME-LINKED IMMUNOSORBENT ASSAY (ELISA)

The cells used for the ELISA experiments in this study were the THP-1 cell line. Approximately 1x10<sup>6</sup> cells were seeded into each well of a 6-well plate (Greiner) and differentiated with PMA. After 24h, the cells were treated with the appropriate ligand and inhibitors and left for a further 24h. To each well, 200µl of ELISA assay buffer (Table 2.8) was added to the medium and left to incubate, with swirling, for 10min at room temperature. An aliquot of medium (500µl) was then removed from each well and analysed by ELISA.

Each 96 well plate was coated with primary antibody, goat polyclonal ATAB atlantic antiapoE (Inc Star) at a concentration of 250ng/100µl PBS/well and left to incubate over-night at 4°C. Unbound antibody was shaken away and 250µl of Blocking solution (Table 2.8) was added per well and left at 37°C for 1h, without shaking. The ELISA plate was washed three times with 300µl washing buffer (Table 2.8) and 50µl of SB biotinylated secondary antibody (GSK), prepared with 1% goat serum in assay buffer, was added to each well, followed by 50µl of apoE standards or samples. The standards used were human apoE serum (GSK) diluted in assay buffer to give a range of concentrations (160ng/ml, 80ng/ml, 40ng/ml, 20ng/ml, 10ng/ml, 5ng/ml). The plates were left at room temperature for 2h, with shaking and then washed four times with 300µl of washing buffer. To each well, 100µl of streptavidin-HRP complex (Amersham), prepared in assay buffer (1/2000) was added and left for 30min at room temperature with shaking. Each plate was washed three times with 300µl washing buffer and equal volumes of the Immunopure® TMB concentrate substrate solution (Pierce) and hydrogen peroxide solution were mixed and 100µl was added to each well. The reaction was left for 30min at room temperature, with shaking and stopped with  $100\mu$ I of 2M sulphuric acid. The absorbance was read at 450nm using a Titertek Multiscan MCC/340 microplate reader and the protein concentration of each sample was then calculated from the standard curve.

# 2.6. PROTEIN KINASE ACTIVITY ASSAYS

#### 2.6.1. CASEIN KINASE 2 ASSAY

This method was derived from those used by Sung *et al.* (2001) and Lodie *et al.* (1997) with minor modifications. Whole cell extracts were isolated as described in Section 2.5.1 using the phosphatase-free whole cell extraction buffer and immunoprecipitation was carried out as described in Section 2.5.3. using the CK2 ( $\alpha$ -subunit) antibody. The immunoprecipitates were recovered by centrifugation and the resulting pellet was washed twice using phosphatase-free whole cell extraction buffer, without the Triton X-100. The pellet was then resuspended in 25µl of kinase buffer (Table 2.9) to which 1µCi [ $\gamma$ - <sup>32</sup>P]-ATP was added. The kinase reaction was incubated for 15 min at 37°C and stopped by the addition of 10µl SDS-PAGE sample buffer. Samples were immediately subjected to SDS-PAGE using 15% (w/v) acrylamide gels. After electrophoresis, gels were fixed for 20 min in a solution containing 40% (v/v) methanol and 10% (v/v) acetic acid. These were washed with distilled water before being dried under vacuum and visualised by autoradiography.

# 2.6.2. AKT KINASE ASSAY

The Akt kinase assays were carried out using the Akt non-radioactive kinase assay kit (New England Biolabs) with only minor modification to the manufacturer's protocol. Whole cell extracts were isolated as described in Section 2.5.1 using the phosphatase-free whole cell extraction buffer. To 250µg of whole cell extract, 20µl of immobilised Akt antibody slurry was added and was left to incubate with gentle rocking for 3h, at 4°C. Following incubation, the mix was centrifuged for 30sec at 10,000g, and the resulting pellet was washed twice with 300µl of phosphatase-free whole cell extraction buffer, without Triton X-100 (Table 2.7) and twice with 1xAkt kinase buffer (Table 2.9). After the final wash, the pellet was resuspended in 40µl of 1xAkt kinase buffer supplemented with 200µM ATP and 1µg of GSK-3 fusion protein. This kinase mix was then incubated at 30°C for 30min. The kinase reaction was then terminated with 20µl SDS-PAGE sample buffer. The sample was vortexed and centrifuged at 10,000g for 2min. The supernatant was then boiled for 5min and loaded onto a SDS-PAGE gel. SDS-PAGE electrophoresis and Western Blotting were carried out

as described in sections 2.5.5 and 2.5.6, respectively and the immunodetection of proteins (Section 2.5.7) was carried out using the phospho GSK- $3\alpha/\beta$  (Ser21/9) antibody.

# 2.6.3. JNK KINASE ASSAY

The JNK kinase assays were carried out using the JNK non-radioactive kinase assay kit (New England Biolabs) with only minor modification to the manufacturer's protocol. Whole cell extracts were isolated as described in Section 2.5.1 using the phosphatase-free whole cell extraction buffer. To 250µg of whole cell extract, 20µl of c-jun fusion protein beads was added and was left to incubate overnight with gentle rocking, at 4°C. Following incubation, the mix was centrifuged for 30sec at 10,000g, and the resulting pellet was washed twice with 300µl of phosphatase-free whole cell extraction buffer, without Triton X-100 (Table 2.7) and twice with 1xJNK kinase buffer (Table 2.9). After the final wash the pellet was resuspended in 50µl 1xJNK kinase buffer supplemented with 100µM ATP. This kinase mix was then incubated at 30°C for 30min. The kinase reaction was then terminated with 25µl SDS-PAGE sample buffer and the supernatant was boiled for 5min and loaded onto a SDS-PAGE gel. SDS-PAGE electrophoresis and Western Blotting were carried out as described in sections 2.5.5 and 2.5.6, respectively and the immunodetection of proteins (Section 2.5.7) was carried out using the phospho c-jun (ser63) antibody.

# <u>CHAPTER THREE -</u> <u>STUDIES ON APOLIPOPROTEIN E PROMOTER ACTIVITY</u> <u>IN RESPONSE TO PMA</u>

# 3.1. INTRODUCTION

As described previously in Chapter one, apoE is found to be expressed at high levels in the atherosclerotic plaque, where it is mainly associated with macrophage-derived foam cells (Rosenfeld et al., 1993; O'Brien et al., 1994). Macrophage apoE has several anti-atherogenic functions including key roles in the maintenance of macrophage lipid homeostasis, cellular cholesterol efflux and reverse cholesterol transport. Therefore, macrophage apoE represents an excellent target for therapy against atherosclerosis. Although it is known that the expression of apoE is induced during differentiation of monocytes into macrophages (Werb and Chin, 1983; Auwerx et al., 1988; Menju et al., 1989; Tajima et al., 1995), the regulation of this process is poorly understood. A detailed understanding of how macrophage apoE expression is regulated would provide further information about factors involved in lesion development.

In order to investigate the regulation of apoE gene expression during the differentiation of monocytes into macrophages the human monocytic leukemia cell lines THP-1 (Tsuchiya et al., 1980) and U937 (Sundstrom and Nilsson, 1976) were used as the model systems during this study. Both cell lines have previously been widely used as in vitro models for studying the mechanisms involved in macrophage differentiation and atherogenesis (Olsson et al., 1983; Auwerx, 1991; Liu and Wu, 1992; Rival et al., 2004; Fu and Borensztain et al., 2002) and several studies have shown similarities in the response of these cells with those observed in primary cultures of human monocyte-derived macrophages (Li and Hui, 1997; Frostegard et al., 1990). Both cell lines can be considered as monoblasts as they represent a relatively immature cell of the monocytemacrophage cell lineage (Abrink et al., 1994). Although both these cell lines are blocked at certain steps in the differentiation process they can be induced to differentiate along the monocytic pathway by treatment with phorbol esters, such as phorbol 12-myristate 13-acetate (PMA) [Tsuchiya et al., 1982; Koren et al., 1979; Abrink et al., 1994; Auwerx, 1991; Ways et al., 1987]. Phorbol esters mimic the physiological activator diacylglycerol and are agonists of the  $\alpha$ -,  $\delta$ -,  $\epsilon$ -,  $\xi$ isoforms of protein kinase C (PKC) (Schwende et al., 1996), a family of 12 related serine/threonine kinases implicated in the regulation of cell proliferation, differentiation and other cellular functions (Parker et al., 1989; Ono et al., 1988). After PMA treatment, both cell lines differentiate into a macrophage-like cell, which resembles native monocyte-derived macrophages with regard to their morphological characteristics (Tsuchiya et al., 1982; Auwerx, 1991; Larrick et al., 1980), their membrane expression of antigens and receptors (Auwerx, 1991; Auwerx et al., 1992; Larrick et al., 1980), and their production of several secretory products, including apoE (Auwerx, 1991; Auwerx et al., 1988; Tajima et al., 1985).

ApoE expression can be regulated at a number of levels, with transcriptional control being responsible, at least in part, for the induction seen during macrophage differentiation (Auwerx et al., 1988), in response to oxysterol ligands (Laffitte et al., 2001b) and cholesterol loading (Mazzone et al., 1989). An understanding of the mechanisms involved in the transcriptional regulation of apoE during differentiation is lacking and there has only been a single, rather limited and partly inconclusive study (Basheeruddin et al., 1994) that aimed to identify the sequence elements that are involved in transcriptional activation during macrophage differentiation. Basheeruddin et al. (1994), demonstrated that the -623 to -447 region of the apoE promoter was necessary for the induction of apoE transcription during PMA-induced differentiation of monocytes into macrophages. This -623 to -447 region contained a consensus sequence for the AP-1 family of transcription factors. The binding of AP-1 members to this sequence was induced in response to exposure of THP-1 cells to PMA and mutation of this site produced a significant reduction in mediator-induced gene activity. Unfortunately, there are several limitations associated with this study; firstly the analysis of the regulatory elements was restricted solely to the AP-1 site with no experiments for sites of other transcription factors that may be present in this region. This point is particularly important as mutation of the AP-1 site produced only a slightly reduced response in apoE expression, suggesting that in the native gene, the AP-1 element functions in co-operation with other regulatory elements to induce the endogenous gene during PMA-induced differentiation. Secondly, the analysis of the interaction of nuclear proteins with the AP-1 response element was restricted to competition gel retardation experiments, with no information on the precise nature of the interacting factors (i.e. supershift experiments) and the mechanisms that are involved in their regulation. Also, gel retardation assays indicated that mutation of the AP-1 site caused the binding of an additional factor, the abundance of which was increased in differentiated cells, indicating that this additional DNA-protein complex, which could not be competed out by an AP-1 site oligonucleotide, was responsible for the reduced response of the mutated AP-1 construct. Another problem of this study was that all the experiments were carried out on THP-1 cells, therefore any results obtained may be a peculiar property of this cell line.

In light of the limitations of the study by Basheeruddin *et al.*, (1994), the aim of the work presented in this chapter was to systematically optimise a transient transfection method in order to re-evaluate the regulatory sequences in the apoE promoter that are required for increased apoE expression during macrophage differentiation. Also, whilst these studies were being carried out, work in other laboratories using transgenic mice showed that two distal enhancers, ME1 and ME2 were important in the macrophage- and adipocyte-specific expression of the apoE gene (Shih *et al.*, 2000). Therefore, further extension studies investigated the importance of these enhancer regions in macrophage differentiation.

# 3.2. RESULTS

# 3.2.1. MORPHOLOGICAL ANALYSIS OF THP-1 CELLS DURING PMA-INDUCED DIFFERENTIATION INTO MACROPHAGES

Previous studies have shown that apoE mRNA and protein expression are induced during PMAinduced differentiation of THP-1 monocytes into macrophages (Auwerx *et al.*, 1988; Tajima *et al.*, 1985; Menju *et al.*, 1989). It was therefore first decided to investigate whether this was also the case for the THP-1 cells present in our laboratory. Before investigating the apoE mRNA expression profile, the morphology of the cells was investigated during PMA-induced differentiation. PMA was used at a concentration of 0.16µM as this concentration has previously been shown to be the lowest concentration required to induce the complete differentiation of THP-1 cells (Auwerx *et al.*, 1988). As shown in Figure 3.1, after treatment with PMA, the THP-1 cells began to differentiate and after 48h, 95% of all cells present were adherent to the tissue culture plastic. In addition to cell adhesion and the arrest of proliferation, differentiation was accompanied by marked changes in the morphology of the cells as the cells become irregular and flattened in shape, and demonstrated a 'stellate' phenotype.



**FIGURE 3.1. EFFECT OF PMA ON THE MORPHOLOGY OF THP-1 CELLS** THP-1 cells were grown in 10% (v/v) HI-FCS and treated with PMA ( $0.16\mu$ M). The cells were observed using phase contrast microscopy and photographed at the times indicated.

# **3.2.2. ANALYSIS OF APOE MRNA AND PROTEIN EXPRESSION DURING PMA-INDUCED DIFFERENTIATION OF THP-1 MONOCYTES INTO MACROPHAGES**

Analysis of apoE mRNA expression was carried out in order to reproduce the previously noted mRNA induction during differentiation (Auwerx et al., 1988; Menju et al., 1989). Previous studies demonstrated that the increase in apoE mRNA was linear with time after the addition of PMA. reaching a maximum at 48h. RT-PCR was employed in this study to examine the expression of apoE mRNA during PMA-induced differentiation of THP-1 monocytes into macrophages. For this, THP-1 cells were grown in 10% (v/v) HI-FCS and treated for 24 and 48h with 0.16µM PMA. RNA was then isolated and the generation of cDNA from total cellular RNA was performed as described in Section 2.4.4. For comparative purposes, in addition to differentiated THP-1 cells (denoted 24h+ and 48h+), RNA was also isolated from monocytes (denoted 24h- and 48h-) at each time-point along with monocytes from the start of the experiment (0h). The quality of RNA was assessed by agarose gel electrophoresis (Figure 3.2a) and with all RNA used in this study the expected ratio of 2:1(28s:18s) was obtained, confirming that the RNA had not been degraded. PCR was carried out using conditions (Table 2.11) and primers (Table 2.10) that were based on previous studies in the laboratory. PCR using primers against GAPDH (Table 2.10) were also included in parallel in order to provide a control for the amount of RNA. The PCR conditions for GAPDH (Table 2.11) had also been established previously in the laboratory, where it was also shown that its mRNA expression remains constant through macrophage differentiation. As shown in Figure 3.2b, apoE mRNA expression was increased dramatically after PMA-induced differentiation, with the greatest induction occurring at 48h. This marked induction of apoE is consistent with previous studies (Auwerx et al., 1988; Menju et al., 1989).

Western blot analysis was carried out to evaluate whether the induction of apoE mRNA expression during differentiation of THP-1 cells was also shown at the level of protein. Previous studies have demonstrated that the time course of protein synthesis showed a similar profile to that of mRNA, with maximal induction occurring at 48h (Menju *et al.*, 1989). For this analysis, whole cell extracts were isolated from THP-1 cells treated with 0.16µM PMA and untreated cells as described in Section 2.5.1. Of each sample 20µg of total protein was subjected to SDS-PAGE (Section 2.5.4) and then transferred to a PVDF membrane (Section 2.5.5). The membrane was then incubated with an anti-apoE primary antibody (Biogenesis) and immunodetection was carried out using the appropriate secondary antibody and the ECL detection system (Amersham) as described in Section 2.5.6. The results of this experiment are presented in Figure 3.2c, and it can be seen that the level of apoE protein was also induced dramatically during macrophage differentiation.



FIGURE 3.2. EFFECT OF PMA-INDUCED DIFFERENTIATION ON APOE MRNA AND PROTEIN LEVELS IN THP-1 CELLS

THP-1 cells grown in 10% (v/v) HI-FCS were exposed to PMA ( $0.16\mu$ M) for the indicated times. For comparative purposes, sample isolation was carried out from both monocytes (denoted 24hand 48h-) and differentiated macrophages (denoted 24h+ and 48h+) at each time-point along with monocytes at the start of the experiment (0h). **a** Total RNA was extracted and the integrity was assessed by agarose gel electrophoresis (1% [w/v]). **b** RT-PCR was carried out using apoE- and GAPDH-specific primers. Each set of RT-PCR reactions included a negative control, where a reverse transcriptase reaction contained all the components except for the reverse transcriptase enzyme (denoted –RT). Following amplification, the products were subjected to electrophoresis using a 1.5% (w/v) agarose gel and product size was confirmed using the 100bp molecular weight marker (Appendix I). **c** Western blot analysis of whole cell extracts ( $20\mu$ g) isolated from THP-1 cells. Blotted membranes were incubated with anti-apoE antibody (goat polyclonal IgG) and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. All results are representative of three separate experiments.

# 3.2.3. TRANSFECTION ANALYSIS OF APOE PROMOTER ACTIVITY DURING PMA-INDUCED DIFFERENTIATION OF MONOCYTES INTO MACROPHAGES

Having established that both the apoE mRNA levels and protein content are induced during macrophage differentiation, the promoter activity of apoE was investigated using transient transfection assays. For this study, a DNA construct, containing the -3000/+86 promoter fragment of apoE linked to the luciferase gene in the pGL3 vector (a kind gift from Dr. T. Mazzone, of Rush Medical College, Chicago [Figure 3.3]), was transfected into THP-1 and U937 cell lines, both of which have been used extensively to investigate the regulation of genes during macrophage differentiation (Olsson *et al.*, 1983; Auwerx, 1991; Liu and Wu, 1992; Rival *et al.*, 2004; Fu and Borensztajn et al., 2002). Transfection experiments were carried out in both THP-1 and U937 cells to ensure that any results obtained were not a peculiar property of the cell line used.



# FIGURE 3.3. SCHEMATIC REPRESENTATION OF THE APOE-3000/+86 PGL3 PROMOTER CONSTRUCT



#### FIGURE 3.4. ANALYTICAL DIGEST OF THE APOE-3000/+86-PGL3 PLASMID.

The apoE-3000/+86 pGL3 promoter construct was digested with KpnI and Hind III restriction enzymes, which released the 3115bp apoE promoter fragment from pGL3 vector. The digest products were size-fractionated by electrophoresis on a 1% (w/v) agarose gel along with the 1kb molecular weight marker (Appendix I).

The activity of the apoE promoter was monitored by transient transfection of monocytes using the polycation transfection reagent, Superfect<sup>™</sup> (Qiagen). The initial analysis of promoter activity was assessed using the human monocytic cell line, THP-1. The apoE-3000/+86 pGL3 promoter construct (2µg) was added to the cells, along with the internal transfection efficiency control plasmid CMVβ-galactosidase (0.5μg) [Appendix III]. RSV-luciferase (De wet et al, 1987) was used as a positive control and pGL3-Basic vector (Appendix IV) was used as the negative control, All DNA used for transfection experiments was purified using the Concert<sup>TM</sup> plasmid purification kit (Gibco) and the purity and quality of the DNA preparation was checked by analytical digests using the appropriate restriction enzymes and subjected to fractionation using agarose gel electrophoresis (Figure 3.4). Following transfection, the cells were treated with 1µM PMA. This higher concentration of PMA was used in the initial transfection experiments as this concentration had been used in previous studies in the laboratory, involved in the detailed dissection of the LPL promoter in U937 cells (Hughes et al., 2002). After PMA treatment, cells were harvested at the indicated time-points (Figure 3.5) and the luciferase and β-galactosidase activity was determined as described in Section 2.3.11 and 2.3.12, respectively. Each transfection was carried out in triplicate and each time-point had a control where the cells were untreated.



#### FIGURE 3.5. ANALYSIS OF APOE PROMOTER ACTIVITY DURING MACROPHAGE DIFFERENTIATION.

Transfection was performed in THP-1 cells as described in Section 2.3.5.1. The cells were transfected with the apoE-3000/+86 reporter construct in medium containing 3% (v/v) HI-FCS, and incubated with  $1\mu$ M PMA. The luciferase activity was normalised to the  $\beta$ -galactosidase value and expressed as relative luciferase activity. Each experiment was carried out in triplicate and contained a control (untreated cells) for each time-point.

As shown in Figure 3.5, the relative luciferase activity was greater in the untreated cells at each time-point, indicating that apoE promoter activity did not increase during the differentiation process. These results were therefore contradictory to those obtained by Basheeruddin *et al.* (1994), which demonstrated that the apoE promoter activity increased with PMA treatment. One explanation for the higher apoE promoter activity in untreated THP-1 cells may be due to the use of the transfection reagent, Superfect<sup>TM</sup>. Therefore, the transfection procedure was repeated using Effectene<sup>TM</sup> (Qiagen), a lipid-based reagent that has also been extensively used in transfection-based studies in macrophages (Shanmugam *et al.*, 2003; Kikuchi *et al.*, 2002; Schwarz and Murphy, 2001; Nishiguchi *et al.*, 2001). All other transfection conditions were kept constant, with the exception of the amount of DNA used per experiment. According to the procedures supplied by the manufacturer, transfection using the Effectene<sup>TM</sup> reagent (Section 2.3.5.2) only required 0.4µg of the apoE-3000/+86-pGL3 construct and 0.2µg of the CMVβ-galactosidase control plasmid. The cells were treated with 1µM PMA after transfection and all assays were carried out as specified in Section 2.3.11 and 2.3.12.





Transfection was performed in THP-1 cells as described in Section 2.3.5.2. The cells were transfected with the apoE-3000/+86 reporter construct in medium containing 3% (v/v) HI-FCS, and incubated with 1 $\mu$ M PMA. The luciferase activity was normalised to the  $\beta$ -galactosidase value and expressed as relative luciferase activity. Each experiment was carried out in triplicate and contained a control (untreated cells) for each time-point.

The results obtained demonstrated that the transfection reagent was not responsible for the higher apoE promoter activity in monocytes compared to macrophages as transfection with the Effectene<sup>™</sup> reagent gave the same results as those obtained with the Superfect<sup>™</sup> (Figure 3.6). The transfection was then carried out using the U937 cell line to ensure that the results obtained were not due to the THP-1 cell line. The transfection of the U937 cell line was carried out using Superfect<sup>™</sup> as this reagent had been used successfully to dissect regulatory regions in the LPL gene promoter in this cell line (Hughes et al., 2002). All transfection conditions were identical to that of the THP-1 cell line, and each experiment was carried out in triplicate. The results (Figure 3.7) were consistent with those using THP-1 cells, as the relative luciferase activity was much greater in the control (untreated) cells than that in PMA-treated, differentiated cells.



#### FIGURE 3.7. ANALYSIS OF APOE PROMOTER ACTIVITY DURING MACROPHAGE DIFFERENTIATION.

Transfection of U937 cells was carried out as specified in Section 2.3.5.1. The cells were transfected with the apoE-3000/+86 reporter construct in medium containing 3% (v/v) HI-FCS, and incubated with  $1\mu$ M PMA. The luciferase activity was normalised to the  $\beta$ -galactosidase value and expressed as relative luciferase activity. Each experiment was carried out in triplicate and contained a control (untreated cells) for each time-point.

In order to ensure that the transfection results were not due to the concentration of PMA used, U937 cells transfected with the apoE-3000/+86 promoter construct were treated with 0, 0.5 and 1µM PMA for 24h. The cells were transfected using Superfect<sup>™</sup>, as specified above, and harvested as normal. The results demonstrated that varying the PMA concentration used to differentiate the cells had no effect on the apoE promoter activity, as the untreated cells still

showed a greater activity (Figure 3.8). Although the result of this experiment suggested that the observed results from the transfections were independent of the concentration of PMA used, all subsequent experiments were carried out using a PMA concentration of 0.32µM, as used by Basheeruddin *et al.* (1994), as this lower concentration of PMA was found to be sufficient to induce macrophage differentiation.



FIGURE 3.8. EFFECT OF PMA CONCENTRATION ON APOE PROMOTER ACTIVITY IN U937 CELLS

Transfection of U937 cells was carried out as specified in Section 2.3.5.1. The cells were transfected with the apoE-3000/+86 reporter construct in medium containing 3% (v/v) HI-FCS, and incubated with various concentrations of PMA. The luciferase activity was normalised to protein concentration and expressed as relative luciferase activity. Each experiment was carried out in triplicate and the control refers to untreated cells.

The concentration of serum in which transfection assays are carried out, 3% (v/v) HI-FCS, was based on previous extensive optimisation experiments in the laboratory (T.R. Hughes, personal communication). Factors present in serum are known to modulate the activity of several promoters, therefore, in order to ensure that the concentration of serum used was not the cause of the results obtained, THP-1 cells were transfected in medium containing 10% (v/v) and 3% (v/v) HI-FCS, respectively. As shown in Figure 3.9, the promoter activity was regulated marginally by serum concentration, with that from cells incubated in 3% (v/v) HI-FCS being approximately two-fold higher than that in 10% (v/v) HI-FCS. However, the previously noted higher apoE activity in undifferentiated cells was independent of serum concentration, being seen at both 3% (v/v) FCS and 10% (v/v) FCS. All subsequent transfections were carried out in medium containing 10% FCS

in order to limit the variables that may be affecting the activity of the apoE promoter during macrophage differentiation.



# FIGURE 3.9. EFFECT OF FCS ON THE PROMOTER ACTIVITY OF APOE DURING MACROPHAGE DIFFERENTIATION.

THP-1 cells were transfected with apoE-3000/+86-pGL3 as described in Section 2.3.3.2. The cells were transfected in medium containing either 3% (v/v) or 10% (v/v) HI-FCS. The luciferase activity was normalised to protein concentration, and the values are expressed as relative luciferase activity. Each experiment was carried out in triplicate and contained a control (untreated cells) for each time-point.

Initially the aim of this project was the analysis of the regulatory sequences in the apoE promoter that are required for activation during macrophage differentiation. The constant higher activity in undifferentiated cells in all of the transfection experiments in this project raised further doubts on the previously published study by Basheeruddin *et al.* (1994), on the promoter region of apoE. In addition, these results have raised questions about the ability of the apoE gene promoter to be activated during macrophage differentiation. Recent studies in transgenic mice have suggested that the apoE promoter lacks sufficient information for transcription during differentiation, and that the key regulatory elements for expression in macrophages in transgenic mice lie in the two enhancer regions, ME1 and ME2 (Laffitte *et al.*, 2001b). In order to compare the ability of the apoE-3000/+86 promoter construct to activate apoE expression with that of the ME1 and ME2 enhancers, three other apoE-luciferase DNA constructs were therefore analysed. In addition to the apoE-3000/+86 promoter construct, the THP-1 cell line was transfected with a reporter construct containing only the –890 to +93 promoter region (pGL-890) and constructs containing the ME1 or

ME2 upstream of this apoE promoter sequence (pGL-890-ME1 and pGL-890-ME2). All three DNA constructs (Figure 3.10) were kind gifts from Dr B. Laffitte, of University of California, Los Angeles and were transfected into the cells using Effectene™ (Qiagen). After transfection, the cells were treated with 0.32µM PMA and left to incubate for 12 and 24 hours. Cell extracts were prepared and assays were carried out as specified in Section 2.3.11 and 2.3.12. As shown in Figure 3.11, the activity of the apoE-3000/+86 construct was markedly less than that of pGL-890-ME1 and –ME2. However, surprisingly, the shorter, pGL-890 promoter construct which was transfected for comparative purposes showed similar activity to that of the enhancer constructs, which further raises the question of whether the higher activity of the pGL-890-ME1 and –ME2 constructs is actually due to the enhancer sequences or the promoter fragment present. Although these results demonstrate the difference in the activation levels of the different constructs, the previously noted increased activity in untreated cells remained consistent, especially at the 24h time-point. It is interesting to note that the level of apoE promoter activity in untreated and PMA-treated cells was similar at 12h.



FIGURE 3.10. SCHEMATIC REPRESENTATION OF THE APOE PROMOTER AND ENHANCER DNA CONSTRUCTS USED FOR THE TRANSFECTION OF THP-1 CELLS.




#### FIGURE 3.11. COMPARISON OF THE ACTIVITY OF THE DIFFERENT APOE PROMOTER AND ENHANCER SEQUENCES DURING MACROPHAGE DIFFERENTIATION

THP-1 cells were transfected as described in Section 2.3.5.2 and treated with  $0.32\mu$ M PMA. The cells were transfected in 10% (v/v) HI-FCS and harvested at the indicated time-points. The luciferase activity was normalised with protein concentration, and values are expressed as relative luciferase activity. Each experiment was carried out in triplicate and contained a control (untreated cells) for each time-point.

In order to ensure that the increased activity of the apoE promoter in untreated monocytes was specific to this gene and not due to the cellular system, a promoter construct for the LPL gene, -1824/+188 LPL (Figure 3.12) [Gimble *et al.*, 1989], whose activity is known to be induced during differentiation was transfected into the U937 cell line. In addition to the LPL construct, the apoE-3000/+86-pGL3 and pGL-890-ME2 constructs, were transfected. Transfection was carried out using the Superfect<sup>™</sup> transfection reagent and the cells were treated with 0.32µM PMA. The transfection was carried out in 3% (v/v) FCS because previous studies had shown this to be optimal for the LPL promoter (T.R. Hughes, personal communication).





#### CHAPTER THREE - STUDIES ON APOLIPOPROTEIN E PROMOTER ACTIVITY IN RESPONSE TO PMA

#### FIGURE 3.13. COMPARISON OF THE ACTIVITY OF THE LPL -1824/+188 PROMOTER CONSTRUCT WITH THE APOE PROMOTER AND ENHANCER SEQUENCES DURING MACROPHAGE DIFFERENTIATION

U937 cells were transfected as described in Section 2.3.5.1 and treated with  $0.32\mu$ M PMA. The cells were transfected in 3% (v/v) HI-FCS and harvested at the indicated time-points. The luciferase activity was normalised with protein concentration, and values are expressed as relative luciferase activity. Each experiment was carried out in triplicate and contained a control (untreated cells) for each time-point.

#### CHAPTER THREE - STUDIES ON APOLIPOPROTEIN E PROMOTER ACTIVITY IN RESPONSE TO PMA

The results of the transfection showed that the cellular system was working, as the activity of the LPL promoter was higher in cells treated with PMA (Figure 3.13b), although the apoE promoter activity was still greater in the untreated cells at the 24h time-point (Figure 3.13a). Once again, at the 12h time-point, the relative luciferase activity for the apoE constructs showed little or no difference between the treated and untreated cells. This similarity in apoE expression at 12 hours raised the question of whether transcriptional activation of apoE was an earlier event, as a previous study by Basheeruddin *et al.* (1992) demonstrated that transcriptional induction of apoE could be seen at 6h. Therefore a time-course experiment was carried out. The THP-1 cell line was transfected with the pGL-890 and pGL-890-ME1 constructs using Effectene<sup>TM</sup> (Qiagen) and harvested after 1, 3, 6, and 12 hours. The cells were transfected in 10% HI-FCS and treated with 0.32µM PMA. The results at 1 and 3 hours demonstrated that no apoE promoter activity was detectable (data not shown). After 6 hours there was a slight activity, with the highest activity seen at 12 hours (Figure 3.13), where the activity in the untreated cells was marginally higher than the differentiated cells.



FIGURE 3.13. TIME-COURSE RESPONSE OF APOE PROMOTER ACTIVITY DURING DIFFERENTIATION

THP-1 cells were transfected as described in Section 2.3.5.2 and treated with  $0.32\mu$ M PMA. The cells were transfected in 10% (v/v) HI-FCS and harvested at the indicated time-points. The luciferase activity was normalised with protein concentration and values are expressed as relative luciferase activity. Each experiment was carried out in triplicate and contained a control (untreated cells) for each time-point.

The final two transfections were carried out using the THP-1 cell line. The first transfection was carried out using a slightly modified transfection protocol. The cells were transfected using Effectene<sup>TM</sup> as described in Section 2.3.3.2 and then incubated for 16h with the DNA before the addition of  $0.32\mu$ M PMA. This alteration in the protocol was to ensure that PMA was not affecting the uptake of DNA by the cells. Although the results demonstrated a decrease in the difference of apoE promoter activity between treated and untreated cells (data not shown), the increase in apoE promoter activity seen by Basheeruddin *et al.* (1994) and by mRNA expression studies was not apparent.

The final transfection in the monocytic cell line THP-1, was carried out using the DEAEdextran method as described in Section 2.3.5.3. DEAE-dextran was employed as a transfection method as it has previously been used extensively for the transfection of macrophages (Wu *et al.*, 1990; Rupprecht and Coleman, 1991; Mack *et al.*, 1998). The cells were transfected with the apoE-3000/+86-pGL construct and treated with 0.32 $\mu$ M PMA. The action of TNF $\alpha$  (1000U/ml) on the apoE promoter was also investigated for comparative purposes as apoE mRNA expression has previously been shown to be induced by this cytokine (Duan *et al.*, 1995), therefore this experiment would also demonstrate whether other aspects of promoter regulation are functioning as normal. Altering the method of transfection made no difference to the activity of apoE in PMA-treated cells, as it was markedly less than that seen in the control (untreated) cells (Figure 3.14). As expected, the activity of apoE in cells treated with TNF $\alpha$  was increased nearly two-fold, which is similar to that shown previously by Duan *et al.* (1995), thereby indicating that the cellular system was suitable for the analysis of certain aspects of apoE promoter activity.

#### CHAPTER THREE – STUDIES ON APOLIPOPROTEIN E PROMOTER ACTIVITY IN RESPONSE TO PMA





THP-1 cells were transfected with  $2\mu g$  of the apoE-3000/+86 DNA construct using DEAE-dextran (Section 2.3.5.3). After 16h the cells were treated with either  $0.32\mu M$  PMA or 1000U/ml TNF $\alpha$ . After 36h, the cells were harvested and the assays were carried out as specified in Section 2.3.8. and 2.3.9. Luciferase activity was normalised to protein concentration and values are expressed as relative luciferase activity.

#### 3.3. DISCUSSION

The initial aim of the studies described in this chapter was to analyse the regulatory sequences in the apoE promoter that are required for activation during macrophage differentiation. Both RT-PCR and Western blot analysis (Figure 3.2) confirmed the previously noted induction of apoE mRNA and protein expression during macrophage differentiation. However, the apoE promoter activity was not increased during differentiation despite extensive transfection experiments in which a number of parameters were varied. Although the previously reported induction of apoE promoter activity by TNF- $\alpha$  (Duan *et al.*, 1995) was reproduced in the system, the results raise further doubts on the findings of Basheeruddin *et al.* (1994) regarding macrophage differentiation. As described in section 3.1, there were several problems associated with the studies described in this paper and these have now been confirmed by the results of the experiments carried out in this chapter. The results of this chapter also show that the recently identified ME1 and ME2 apoE enhancers are not sufficient for the activation of the gene seen during macrophage differentiation.

Several possibilities could account for the transfection results obtained in this chapter. Firstly, similar to the complexity seen for hepatocyte-specific expression, activation of apoE during macrophage differentiation may require other sequences in addition to the promoter and enhancers. With regards to the liver-specific expression of apoE, several studies have demonstrated that the proximal regulatory elements are not sufficient to direct this expression of the apoE gene in vivo (Smith et al., 1988; Simonet et al., 1991), and various regions in the intergenic sequence between apoE and apoCI have been shown to be required for the expression of these genes in a variety of tissues including liver, testis, spleen, skin, submaxillary gland, kidney, brain, small intestine, heart, stomach and pancreas (Simonet et al., 1993). Moreover, a region in the intergenic sequence between apoCI and the apoCI pseudogene has been shown to contain an element, termed the hepatic control region 1 (HCR1), which is vital for the hepatic expression of the human apoE and apoCI genes (Simonet et al., 1993; Taylor et al., 1987; Simonet et al., 1991). Also, a second hepatic control region designated HCR2 has been identified and is located in the intergenic sequence between the apoCI pseudogene and apoCIV gene and has also been shown to be essential in the liver-specific expression of apoE (Allan et al., 1995a; 1995b). Previous studies have also demonstrated that expression of the apoE gene is controlled by a complex interaction of several independent tissue-specific transcriptional enhancers and silencers and that the absence of more dominant regulatory sequences may result in cryptic elements exerting their silencing effects (Simonet et al., 1993). Therefore, it is a possibility that the important regulatory elements for the control of apoE expression during differentiation may reside outside the sequences contained in the constructs used for these studies, and a thorough understanding of the mechanisms controlling apoE expression during macrophage differentiation would be facilitated through an analysis of DNA constructs containing different combinations of regulatory elements. Indeed, recent work in macrophages has identified a sequence located in the apoE/apoCI intergenic region, separate to the ME1 enhancer element, which regulates apoE expression. This sequence, a PPARy response element was shown through transfection based studies using THP-1 cells to be the only sequence of the apoE/CI intergenic region that responds to the inducers of macrophage and adipocyte differentiation (Galetto et al., 2001).

An alternative explanation for the transfection results obtained in this chapter may be that the induction of apoE mRNA expression during macrophage differentiation is mediated predominately at the level of mRNA stability with transcriptional control making only a minor contribution. Indeed, the activation of the apoE promoter during macrophage differentiation seen in the studies by Basheeruddin *et al.* (1994) was approximately 3-fold. However, the activation of mRNA and protein expression seen in this study (Figure 3.3 and 3.4) along with those of other studies (Auwerx *et al.*, 1988; Basheeruddin et al., 1992; Werb and Chin, 1983; Menju *et al.*, 1989; Tajima *et al.*, 1985) is at least 10-15-fold. Evidence supporting this theory has previously been demonstrated by Basheeruddin *et al.* (1992), who demonstrated that an increase of up to 10-fold in steady state levels of apoE mRNA during differentiation was not accompanied by a corresponding

#### CHAPTER THREE - STUDIES ON APOLIPOPROTEIN E PROMOTER ACTIVITY IN RESPONSE TO PMA

increase in the rate of transcription initiation or elongation. They also demonstrated that this disparity between mRNA levels and transcription could be attributed to mRNA degradation, as a more rapid level of degradation of mature mRNA occurred in undifferentiated cells compared to differentiated (Basheeruddin *et al.*, 1992). Although no studies have elucidated the precise mechanism for this increase in mRNA stability, work by Wager and Assosian (1990) has demonstrated that PMA treatment of U937 cells leads to increased levels of TGF- $\beta$  mRNA, which is due to a PMA-mediated inhibition of the activity of a specific nuclease. This induction of mRNA stability in response to PMA treatment has also been shown for several other genes in macrophages (Kwon and Kim, 2003; Park *et al.*, 2001; Doyle *et al.*, 1997; Ohh *et al.*, 1994), indicating that increased mRNA stability may be a common mechanism for gene up-regulation during differentiation. Therefore, it is possible to speculate that the increase in apoE mRNA observed in PMA-treated cells is due to an increase in mRNA stability as a result of the inhibition of a nuclease responsible for degradation, and that transcriptional regulation makes only a minor contribution to this mode of apoE regulation during differentiation.

In conclusion, this Chapter has demonstrated that the regulation of apoE expression during macrophage differentiation is much more complex than first anticipated. Although apoE expression is induced by PMA at the level of mRNA and protein we were unfortunately unable to identify how transcriptional regulation relates to this induction, as our transfection system was not suitable for the analysis of the apoE promoter during differentiation. Therefore, in the light of these findings and the recent developments in the apoE field, it was decided to slightly modify the aims of the project, which are described in the next chapter.

#### 4.1. INTRODUCTION

The work presented in Chapter 3 has shown that the regulation of apoE during macrophage differentiation is a much more complex story than first anticipated. The results of the previous chapter have demonstrated that additional studies are required to further elucidate the factors which control apoE expression. As a result of this outcome and in light of the recent developments in the apoE field, the overall aims of the project were modified slightly to understanding the regulation of apoE transcription by the Liver X receptor (LXR) family of nuclear receptors. ApoE regulation by the LXRs occurs in response to lipid loading, whereby these receptors act as a molecular sensor, increasing apoE expression (Laffitte *et al.*, 2001b). Although it is known that LXRs mediate their effects through binding to two LXREs in the apoE enhancer regions and that this regulation is tissue-specific (Laffitte *et al.*, 2001b), very little else is known about LXR regulation of this gene, especially with regards to the involvement of cell signalling pathways. Therefore this chapter will address the role of cell signalling in nuclear receptor activation and identify the potential involvement of these pathways in LXR regulation of apoE.

More and more studies reveal that nuclear receptor activation of target genes is much more complicated than initially thought, and may involve the regulation of receptor function through cross-talk with other transcription factors and intracellular signalling pathways. One way in which cell signalling pathways affect nuclear receptor activity is through the modification of coregulator proteins (Figure 4.1) [Hermanson *et al.*, 2002; McKenna and O'Malley, 2002]. Coactivator recruitment by DNA-bound nuclear receptors occurs in response to ligand binding. The binding of ligand leads to a conformational change, which alters the affinity of the nuclear receptors for coregulator proteins, which results in the dissociation of coregulator recruitment by nuclear receptors, cell signalling pathways can regulate this function through direct modification of the coactivator and corepressor proteins. Post-translational modification of coregulators exerts considerable control over their availability and their recruitment to transcriptional complexes and also their enzymatic activity (Rowan *et al.*, 2000; Hong and Privalsky, 2000; Chawla *et al.*, 1998).

Cell signalling pathways can also regulate coregulator recruitment and various other receptor functions, through direct phosphorylation of the nuclear receptor (Figure 4.1). Both the ligand-binding domain and the DNA-binding domain, in addition to the N-terminal domain of nuclear receptors contain several, potential phosphorylation sites (Rochette-Egly, 2003; Shao and Lazar, 1999). Phosphorylation of the nuclear receptors by signalling pathways may occur in response to ligand or in the absence of ligand and individual phosphorylations can act either to enhance or to inhibit the transactivation potential of the receptor.



#### FIGURE 4.1. POSITIVE REGULATION OF NUCLEAR RECEPTOR TRANSACTIVATION THROUGH PHOSPHORYLATION

Nuclear receptor phosphorylation by kinases activated in response to a variety of signals, aids coactivator recruitment, thereby facilitating the recruitment of histone remodeling complexes, which decondense chromatin and promote a transcriptionally permissive environment at the promoter. Phosphorylation of the nuclear receptor also participates in the recruitment of components of the RNA polymerase II transcriptional machinery. In addition, coactivator modification also effects coregulator recruitment and enzymatic function thereby indirectly effecting nuclear receptors transactivation potential. P/CAF, p300/CBP-associated factor; CBP, p300/CREB-binding protein; P, phosphate residue; NR, nuclear receptor.

Phosphorylation has been shown to modulate the activity of many nuclear receptors, with a range of functions being affected, such as dimerisation, DNA binding and interactions with cofactors, all of which affect the transactivation function of these transcription factors (Chen *et al.*, 1999; Auricchio *et al.*, 1996; Hammer *et al.*, 1999). Although phosphorylation is usually secondary to ligand activation for most nuclear receptors, it is becoming apparent that phosphorylation may be the primary method for modulating certain receptors (Hammer *et al.*, 1999; Tremblay *et al.*, 1999; Arnold *et al.*, 1995; Lee *et al.*, 2000). The ability of kinase cascades to regulate nuclear receptor functions demonstrates that the activity of these receptors can be modified depending on the physiological state of the cell. Also, it is becoming increasingly clear that the phosphorylation of nuclear receptors and the modification of coregulators is necessary for the transactivation of certain genes, and that these modifications cooperate with the receptor ligand in order to achieve maximal transcriptional activity.

Although recent work has demonstrated that the LXRs require coactivators for effective transcriptional activation of certain genes (Huuskonen *et al.*, 2004; Oberkofler *et al.*, 2003) no involvement of any cell signalling pathways in the LXR-mediated regulation of genes have so far

been demonstrated, although its dimerisation partner RXR has been shown to be affected by several kinases, including the mitogen-activated protein kinases (MAPKs) and protein kinase A (PKA) [Solomon *et al.*, 1999; Dowhan and Muscat, 1996]

Therefore the aim of this chapter, was to determine whether cell signalling pathways play a role in the LXR-mediated activation of apoE transcription in macrophages through the use of commercially-available inhibitors against various cell signalling pathways. For these studies, it was decided to use several different inhibitors, as no previous work on the involvement of cell signalling pathways in the LXR-mediated regulation of genes has been published. All inhibitors chosen for this study have been used extensively in research examining gene regulation in macrophages (Garcia-Garcia *et al.*, 2002; Ma *et al.*, 2001; Zhao *et al.*, 2002; Abe *et al.*, 1999; Liang *et al.*, 1999; Wang *et al.*, 2002) and have been chosen to target pathways previously shown to be involved in nuclear receptor signalling. Below is a brief summary of the pathways to be investigated and their involvement so far in nuclear receptor signalling:

**MITOGEN-ACTIVATED PROTEIN KINASES (MAPKs)-** MAPK signalling pathways play an important role in regulating gene expression in eukaryotic cells and have been shown to participate in a variety of processes, such as cell differentiation, movement, division and cell death. Five distinctly regulated MAPK subgroups have been identified so far, all of which can signal independently from each other. These include the extracellular signal-regulated kinase 1 and 2 (ERK1/2) cascade, which preferentially regulates cell growth and differentiation, as well as the c-jun N-terminal kinase (JNK)/stress-activated protein kinase (SAPK) and p38 MAPK cascades, which function mainly in stress responses like inflammation and apoptosis.



FIGURE 4.2. SCHEMATIC SUMMARY OF THE THREE MAIN MAPK PATHWAYS

Each of these three branches of the MAPK pathway has been shown to affect nuclear receptor activity through phosphorylation of the nuclear receptor itself or through phosphorylation of coregulator proteins. The AF-1 domain of several nuclear receptors have been reported to be

substrates for the ERK or p38 MAPKs, including the progesterone receptor (PR) [Lange *et al.*, 2000; Shen *et al.*, 2001], estrogen receptor- $\alpha$  (ER $\alpha$ ) [Bunone *et al.*, 1996; Kato *et al.*, 1995], ER $\beta$  (Tremblay *et al.*, 1999; Driggers *et al.*, 2001), androgen receptor (AR) [Yeh *et al.*, 1999], PPARs (Hu *et al.*, 1996; Juge-Aubry *et al.*, 1999), and RAR $\gamma$  (Gianni *et al.*, 2002). Whilst the JNK/SAPK subfamily has been shown to target the AF-1 domain of RXR $\alpha$  (Adam-Stitah *et al.*, 1999; Lee *et al.*, 2000). The JNK/SAPK pathway is also thought to potentiate the activation function of ER through phosphorylation of ER-associated proteins, CBP and GRIP1 (Feng *et al.*, 2001). All three MAPK pathways are also able to modulate the activities of transcription coregulators, such as p300 and SMRT (Chen *et al.*, 2003).

**PHOSPHOINOSITIDE 3' KINASE (PI3K)**- PI3Ks modulate several physiological processes including metabolism, cell growth, proliferation and cell migration (Wymann *et al.*, 2003). PI3Ks phosphorylate inositol lipids at the 3' position of the inositol ring to generate the 3-phosphoinositides PI(3)P, PI(3,4)P<sub>2</sub> and PI(3,4,5)P<sub>3</sub>, which recruit 3-phosphoinositide-binding proteins to the plasma membrane and activate downstream signalling pathways (Leevers *et al.*, 1999). One downstream target of PI3K is Akt/PKB, a serine/threonine kinase that regulates a variety of downstream kinases (Datta *et al.*, 1999). Upon activation and subsequent translocation into the nucleus, Akt has been shown to phosphorylate certain nuclear receptors such as ER $\alpha$  (Campbell *et al.*, 2001) and AR (Lin *et al.*, 2001b) in their N-terminal region. The PI3K/Akt pathway also plays an important role in RAR signalling (Gianni *et al.*, 2002) and a downstream target of Akt, glycogen synthase kinase-3 (GSK-3), has also been shown to regulate the transcriptional activation function of the glucocorticoid receptor (GR) [Rogatsky *et al.*, 1998].



#### FIGURE 4.3. SCHEMATIC REPRESENTATION OF THE PI3K/AKT/GSK-3 PATHWAY

**CASEIN KINASE 2 (CK2)-** CK2 is a ubiquitous, constitutively active serine/threonine protein kinase and is typically found in tetrameric complexes consisting of two catalytic ( $\alpha$  and/or  $\alpha$ ') subunits and two regulatory  $\beta$  subunits. CK2 appears to reside in a variety of cellular compartments and participates in the phosphorylation and regulation of a broad array of substrates (Meggio and Pinna, 2003). CK2 has been shown to modulate the transactivation functions of several nuclear receptors through phosphorylation. Phosphorylation of the human Vitamin D receptor (VDR) at Ser208 by CK2, specifically modulates its transcriptional capacity, as phosphorylation alters the conformation of VDR and potentiates its interaction with the basal transcription machinery (Jurutka *et al.*, 1996). Other targets of CK2 phosphorylation include PR $\beta$  (Zhang *et al.*, 1994b), TR $\alpha$ 2 (Katz *et al.*, 1995), ER (Castano *et al.*, 1997) and the corepressor SMRT (Zhou *et al.*, 2001). SMRT phosphorylation occurs in the C-terminal domain of the corepressor, and stabilizes the ability of the SMRT protein to interact with nuclear hormone receptors (Zhou *et al.*, 2001).

### 4.2. RESULTS

## **4.2.1. EFFECT OF LXR LIGAND 22(R)-HYDROXYCHOLESTEROL ON APOE MRNA EXPRESSION IN THP-1** MACROPHAGES

Before addressing the overall aim of this chapter on the involvement of cell signalling pathways in LXR-mediated transactivation of apoE in macrophages, it was necessary to ensure that the earlier data produced by Laffitte et al. (2001b), which demonstrated that the LXR ligand 22(R)hydroxycholesterol induced apoE expression, could be reproduced. It was therefore decided to examine the ability of 22(R)-hydroxycholesterol to increase apoE mRNA expression in macrophages. For this experiment THP-1 monocytes were differentiated for 24h with 0.16µM PMA. After the 24h differentiation period 22(R)-hydroxycholesterol (2µg/ml) was added to the appropriate flasks and left to incubate for 12h (Figure 4.4). In addition to 22(R)-hydroxycholesterol, as a control for this experiment, THP-1 cells were treated with the ligand 22(S)-hydroxycholesterol (2µg/ml), an inactive enantiomer of 22(R)-hydroxycholesterol which binds but does not activate the LXR receptor. As both ligands were solubilised in DMSO, it was ensured that the final concentration was less than 0.1% (v/v) to ensure that it had no effect on cell viability. Also, as an additional control, all control samples for these experiments were treated with an equal volume of DMSO to ensure that any observed result was due to the ligand and not the DMSO. Total RNA was then extracted (Section 2.4.1) and RT-PCR was carried out (Section 2.4.4) using the conditions (Table 2.11) and primers for apoE and GAPDH (Table 2.10).



## FIGURE 4.4. SCHEMATIC REPRESENTATION OF THE EXPERIMENTAL STRATEGY USED TO INVESTIGATE THE EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON APOE MRNA EXPRESSION IN THP-1 MACROPHAGES.

For all experiments investigating the ability of oxysterol ligands to induce apoE gene expression it was necessary to use 10% (v/v) delipidated serum, in order to ensure that no other external source of sterols would effect the LXR-mediated induction of apoE. Delipidation of serum was carried out as described in Section 2.3.4 and was substituted for "normal" HI-FCS at the beginning of each experiment (Figure 4.4). This substitution of FCS had no effect on cell viability or rate of growth and did not affect PMA-induced adhesion of cells (data not shown). In addition to the use of delipidated serum all experiments investigating the LXR-mediated induction of apoE required a differentiation period of only 24h. Also, all PCR-based experiments required only a12h incubation with the LXR ligand. The reason for this was that the effect of the ligand on apoE expression in macrophages becomes hidden after this time-point by the increasing stimulation of apoE due to PMA.

From Figure 4.5 it can be seen that treatment of differentiated THP-1 cells with the oxysterol ligand 22(R)-hydroxycholesterol results in an increase in apoE mRNA expression as demonstrated previously by Laffitte *et al.* (2001b). This induction of apoE by 22(R)-hydroxycholesterol was not achieved in monocytes (data not shown), which also confirms the previous finding by Laffitte et al. (2001b) that LXR-mediated induction of apoE is differentiation dependent. THP-1 cells treated with the inactive ligand, 22(S)-hydroxycholesterol showed no change in apoE levels (Figure 4.5), which demonstrates that the ligand-mediated induction of apoE requires LXR activation.





#### FIGURE 4.5. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON APOE MRNA EXPRESSION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] and 22(S)-hydroxycholesterol (denoted 22S) [2µg/ml] for 12h. RT-PCR was carried out using apoE- and GAPDH-specific primers. Although not shown on this Figure, each set of RT-PCR reactions included a negative control, where no reverse transcriptase had been added during the cDNA synthesis step of the control sample. No signal was obtained from these samples, indicating that there was no genomic DNA contamination in the PCR. Following amplification, the products were subjected to electrophoresis using a 1.5% (w/v) agarose gel and product size was confirmed using the 100bp molecular weight marker (Appendix II). The apoE and GAPDH signals for each sample were determined by densitometric analysis and plotted on a bar chart. The apoE:GAPDH ratio in macrophages treated with DMSO only (denoted control) has been assigned as 1, with the ratio for the remaining samples being represented relative to this control. Results are representative of three separate experiments and the data shown in the histogram is the mean value ±SD.

## **4.2.2. EFFECT OF LXR LIGAND 22(R)-HYDROXYCHOLESTEROL ON APOE PROTEIN EXPRESSION IN THP-1** MACROPHAGES

Although the study by Laffitte *et al.*, (2001b) demonstrated an induction in apoE mRNA by the LXR ligand, 22(R)-hydroxycholesterol, the effect of this ligand on apoE protein levels was not shown. Therefore it was decided that the effect 22(R)-hydroxycholesterol on apoE protein levels should be investigated. Whole cell protein extracts were prepared (see Section 2.5.1) from differentiated THP-1 cells treated with 2µg/ml of 22(R)-hydroxycholesterol or 22(S)-hydroxycholesterol for 24h. Of each sample 20µg of total protein was subjected to SDS-PAGE (Section 2.5.4) and then transferred to a PVDF membrane (Section 2.5.5). The membrane was then incubated with an antiapoE primary antibody (Biogenesis) and immunodetection was carried out using the appropriate secondary antibody and the ECL detection system (Amersham) as described in Section 2.5.6. As shown in Figure 4.6, 22(R)-hydroxycholesterol treatment of differentiated THP-1 cells increases apoE protein levels, and this induction was shown to be time-dependent, with maximal induction occurring at 24h (Figure 4.7). Also, as shown at the mRNA level (Figure 4.5) the inactive enantiomer, 22(S)-hydroxycholesterol had no effect on apoE expression levels.



#### FIGURE 4.6. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON APOE PROTEIN EXPRESSION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] and 22(S)-hydroxycholesterol (denoted 22S) [2µg/ml] for 24h. In addition, differentiated cells were treated with vehicle (DMSO), in the place of ligand (denoted control). Western blot analysis was carried out using 20µg of whole cell extracts. Blotted membranes were incubated with anti-apoE primary antibody (goat polyclonal IgG) and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of three separate experiments.



FIGURE 4.7. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON APOE PROTEIN EXPRESSION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22R-hydroxycholesterol (denoted 22R) [2µg/mI] and 22(S)-hydroxycholesterol (denoted 22S) [2µg/mI] for various time-points. In addition, for each time-point, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Western blot analysis was carried out using 20µg of whole cell extracts. Blotted membranes were incubated with anti-apoE primary antibody (goat polyclonal IgG) and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of three separate experiments.

### 4.2.3. EFFECT OF LXR LIGAND 22(R)-HYDROXYCHOLESTEROL ON APOE SECRETION IN THP-1 MACROPHAGES

As apoE is a secretory protein and its secretion by macrophages in the atherosclerotic plaque is largely responsible for its atheroprotective role, the effect of 22(R)-hydroxycholesterol on apoE secretion in macrophages was investigated. THP-1 monocytes were differentiated for 24h with 0.16µM PMA and then treated with 2µg/ml of 22(R)-hydroxycholesterol or the control ligand, 22(S)-hydroxycholesterol for 24h. After treatment, the apoE content of the cell culture medium was determined by ELISA as described in Section 2.5.7. As shown in Figure 4.8, the level of apoE secretion increases with 22(R)-hydroxycholesterol treatment and remains unchanged in cells treated with 22(S)-hydroxycholesterol.



**FIGURE 4.8. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON APOE SECRETION IN THP-1 MACROPHAGES** THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] and 22(S)hydroxycholesterol (denoted 22S) [2µg/ml] for 24h. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Each ELISA was carried out as described in Section 2.5.7. The level of apoE protein is shown as a value above each bar and results are representative of three separate experiments.

## **4.2.4.** EFFECT OF INHIBITORS ON THE **22(R)**-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF APOE MRNA EXPRESSION IN THP-1 MACROPHAGES

The results from the previous sections have shown clearly that the oxysterol ligand 22(R)hydroxycholesterol increases apoE expression at both the mRNA and protein level, giving a strong basis from which we can confidently approach the primary aim of this chapter, which is to determine, through the use of inhibitors, whether cell signalling pathways play a role in the LXRmediated activation of apoE transcription in macrophages. For this study, THP-1 cells were differentiated in 10% (v/v) delipidated HI-FCS with PMA (0.16 $\mu$ M) for 24h. After the 24hdifferentiation period, inhibitors were added for 1h before treatment with LXR ligand (Figure 4.9), in order to ensure cellular uptake and inhibition of the target enzyme. The ligand 22(R)hydroxycholesterol (2 $\mu$ g/ml) was then added to the appropriate flasks and left to incubate for a further 12h. After incubation with the inhibitors and ligand, total RNA was extracted (Section 2.4.1). As all the inhibitors used in this study were solubilised in DMSO, it was ensured that the final concentration was less than 0.1% (v/v) to ensure that that it had no effect on cell viability, and as an additional control, all control samples for these experiments were treated with an equal volume of DMSO to ensure that the observed effect was due to the inhibitor and not the DMSO. In addition, trypan blue exclusion assays were carried out to ensure that the inhibition was not due to any cytotoxic effect of the compound.



FIGURE 4.9. SCHEMATIC REPRESENTATION OF THE EXPERIMENTAL STRATEGY USED TO INVESTIGATE THE EFFECT OF CELL SIGNALLING INHIBITORS ON THE LIGAND MEDIATED INDUCTION OF APOE MRNA EXPRESSION IN THP-1 MACROPHAGES

As the induction of apoE expression at the mRNA level is modest (2-2.5x), a more quantitative method of measuring mRNA expression was employed for these inhibitor studies. Taqman Real-Time quantitative RT-PCR was carried out using apoE-specific primers (Table 2.12) as described in Section 2.4.5 and reactions were monitored in real time mode using Sequence Detection Systems, version 1.7 software. In addition, Taqman reactions were carried out using primers specific for  $\beta$ -microtubulin (Table 2.12), which was used as the constitutive control as none of the inhibitors used in these experiments affected the expression of this gene. The concentration of the inhibitors used in the following experiments were based on previous work in our laboratory, where these inhibitors were used to elucidate key signalling pathways in macrophages and also previously published studies whereby these inhibitors were employed to investigate the regulation of macrophage gene expression (Liu et *al.*, 2001; Salomonsson *et al.*, 2002; Katsuyama *et al.*, 2001; Abe *et al.*, 1999; Liang *et al.*, 1999; Wang *et al.*, 2002). The inhibitors used and the pathways affected by them are shown in Table 4.1.

TABLE 4.1. INHIBITO	DRS USED TO	ELUCIDATE TH	E CELL	SIGNALLING	PATHWAYS	INVOLVED IN	THE LXR-
MEDIATED INDUCTION	OF APOE EX	PRESSION					

INHIBITOR	PATHWAY	MECHANISM	REFERENCE	
Apigenin	CK2	Reversible ATP/GTP-competitive inhibitor of CKII	Critchfield et al., 1997	
Curcumin	JNK/SAPK MAPK	Inhibits an upstream kinase of the JNK pathway, thought to be MEKK1	Chen and Tan, 1998; Jobin <i>et al.</i> , 1999	
LY294002	PI3K	Reversible ATP-competitive inhibitor of PI3K	Vlahos <i>et al.,</i> 1994	
PD98059	ERK MAPK	Inhibits MEK1 phosphorylation and activation of ERK	Dudley <i>et al.,</i> 1995	
SB202190	р38 МАРК	Bind the inactive form of p38 and inhibit its activation	Lee <i>et al.</i> , 1994; Frantz <i>et al.</i> , 1998	
SP600125	JNK/SAPK MAPK	Reversible ATP-competitive inhibitor of JNK1, -2 and -3	Bennett <i>et al.,</i> 2001	

The results of this experiment are presented in Figure 4.10 and it can be seen that inhibitors against three of the pathways investigated, had a significant effect upon the 22(R)-hydroxycholesetrol-mediated induction of apoE mRNA. Both the inhibitors, curcumin and SP600125, which target the JNK branch of the MAPK pathway, completely eliminated the 22(R)-hydroxycholesterol induced expression of apoE mRNA, indicating that the JNK/SAPK pathway is likely to play a key role in the LXR-mediated regulation of apoE mRNA by 22(R)-hydroxycholesterol, as did the CK2 inhibitor, apigenin. The inhibitors against the ERK (PD98059) and p38 (SB202190) branches of the MAPK cascade did not show any significant inhibition of apoE induction, indicating that these pathways are not involved in the LXR-mediated regulation of apoE.



#### FIGURE 4.10. EFFECT OF CELL SIGNALLING INHIBITORS ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF APOE MRNA EXPRESSION

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] for 12h. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: PD98059 (50µM), SB202190 (10µM), Curcumin (25µM), SP600125 (50µM), LY294002 (100µM), Apigenin (40µM). Taqman Real-Time quantitative RT-PCR was carried out using apoE- and β-microtubulin-specific primers and reactions were monitored in real time mode using Sequence Detection Systems, version 1.7 software. The apoE: β-microtubulin ratio in macrophages treated with vehicle (DMSO) only has been assigned as 1, with the ratio for the remaining samples being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant difference compared to control cells treated with 22(R)-hydroxycholesterol (\*p<0.05; \*\*p<0.005).

# **4.2.5.** EFFECT OF INHIBITORS ON THE **22(R)**-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF APOE PROTEIN EXPRESSION IN THP-1 MACROPHAGES

The effect of these inhibitors on apoE expression was also confirmed at the protein level. Whole cell extracts were prepared from differentiated THP-1 cells, which were subsequently treated with inhibitors for 1hr before 22(R)-hydroxycholesterol (2µg/ml) addition for 24h. Western blot analysis was carried out as described previously in this chapter (Section 4.2.2). As shown in Figure 4.11, the effect of the inhibitors at the protein level is consistent with that seen at the mRNA level, with inhibition of the ligand–mediated response occurring in cells pre-treated with curcumin (JNK/SAPK

inhibitor), LY294002 (PI3K inhibitor) and apigenin (CK2). Also, as was the case for apoE mRNA, no inhibition of the apoE induction by the ligand occurred in cells pre-treated with the ERK MAPK inhibitor, PD98059 or the p38 MAPK inhibitor SB202190.



#### FIGURE 4.11. EFFECT OF CELL SIGNALLING INHIBITORS ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF APOE PROTEIN EXPRESSION

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] for 24h. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: LY294002 ( $100\mu$ M), PD98059 ( $50\mu$ M), Curcumin ( $25\mu$ M), SB202190 ( $10\mu$ M), Apigenin ( $40\mu$ M). Western blot analysis was carried out using 20µg of whole cell extracts. Blotted membranes were incubated with anti-apoE primary antibody (goat polyclonal IgG) and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. The apoE protein level for each sample was determined by densitometric analysis and plotted on a bar chart. The control represents macrophages treated with vehicle (DMSO) only has been assigned as 1, with the apoE protein level for the remaining samples being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant difference compared to control cells treated with 22(R)-hydroxycholesterol (\*p<0.05).

The inhibition observed by pre-treatment with the JNK/SAPK, PI3K and CK2 inhibitors was also shown to be concentration-dependent (see Figure 4.12), as we next performed dose-response experiments with each inhibitor. Whole cell extracts were prepared from differentiated THP-1 cells, which were subsequently treated with varying concentration of inhibitors for 1hr before 22(R)-hydroxycholesterol (2µg/ml) addition for 24h.



#### FIGURE 4.12. EFFECT OF INHIBITOR CONCENTRATION ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF APOE PROTEIN EXPRESSION

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] for 24h. Inhibitors: (A) curcumin; (B) LY294002 and; (C) apigenin were added 1h before the ligand at the indicated concentrations. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Western blot analysis was carried out using 20µg of whole cell extracts. Blotted membranes were incubated with anti-apoE primary antibody (goat polyclonal IgG) and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of three separate experiments.

**4.2.6.** *EFFECT OF INHIBITORS ON 22R-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF APOE SECRETION* The effect of these inhibitors on the LXR-mediated induction of apoE was extended to apoE protein secretion. THP-1 monocytes were differentiated for 24h with 0.16μM PMA and subsequently treated with inhibitors for 1h before 22(R)-hydroxycholesterol (2μg/ml) addition for 24h. After incubation with the inhibitors and ligand, the apoE content of the cell culture medium was determined by ELISA as described in Section 2.5.7. Figure 4.13 shows that the inhibitors against the JNK/SAPK, PI3K and CK2 cell signalling pathways inhibit the induction seen in apoE secretion by ligand treatment. This result confirms further that these pathways are involved in the LXRmediated induction of apoE. Also from Figure 4.13 it can be seen that the result for the PD98059 (ERK MAPK) inhibitor is also consistent with that seen for the mRNA and protein levels, as it had no inhibitory effect. However, these experiments did show a conflicting result for SB202190 (p38 MAPK), as this inhibitor reduced apoE secretion by nearly 50% (Figure 4.13). As this inhibition only occurs at the level of protein secretion, it may be that the p38 MAPK pathway plays a role in the post-translational regulation of apoE. Also curcumin reduced the level of apoE to below that of the control, indicating that this pathway may also be involved in the basal regulation of apoE secretion.





THP-1 monocytes were differentiated with PMA (0.16 $\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2 $\mu$ g/ml] for 24h. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: LY294002 (100 $\mu$ M), PD98059 (50 $\mu$ M), Curcumin (25 $\mu$ M), SB202190 (10 $\mu$ M), Apigenin (40 $\mu$ M). As a control, differentiated cells were treated with vehicle (DMSO) in the place of ligand. Each ELISA was carried out as described in Section 2.5.7. The level of apoE protein is shown as a value above each bar and results are representative of three separate experiments.

## **4.2.7. EFFECT OF 22(R)-HYDROXYCHOLESTEROL AND CELL SIGNALLING INHIBITORS ON APOE PROTEIN** EXPRESSION IN HUMAN PRIMARY MONOCYTE-DERIVED MACROPHAGES

Although the results from the previous sections have demonstrated clearly that the LXR-ligand 22(R)-hydroxycholesterol induces apoE expression in human THP-1 macrophages and that this inhibition can be abolished by pre-treatment with inhibitors against the JNK/SAPK, PI3K and CK2 pathways. In order for these results to be fully relevant to human atherosclerosis, they must be reproduced in primary cultures of human monocyte-derived macrophages as there are many instances where there are differences in response to stimuli between primary monocytes/macrophages and cell lines. Therefore, the effect of 22(R)-hydroxycholesterol on apoE induction was tested in human primary monocyte-derived macrophages, as was the effect of the inhibitors. Isolation of primary cells was carried out as described in Section 2.3.3 and the monocytes were left to differentiate for 7 days before use in these experiments. Once the cells had differentiated they were treated with inhibitors for 1h before treatment with 22(R)hydroxycholesterol for 24h. For all experiments using primary cultures it is important to note that the concentration of ligand added to the cells was increased to 3µg/ml, as the concentration of cells per sample was much greater than that used in experiments using the THP-1 cell line. Whole cell extracts were prepared as described previously (Section 2.5.1) and western blot analysis was carried out as described in Section 4.2.2 of this chapter.

From Figure 4.14 it can be seen that 22(R)-hydroxycholesterol induced a 3-4-fold induction of apoE protein expression in these cells and this induction was significantly blocked by inhibitors against the JNK/SAPK and PI3K pathways, whilst pre-treatment with the CK2 inhibitor, apigenin had no significant inhibitory effect in these cells. Also as previously shown in the THP-1 cell line, the ERK and p38 MAPK inhibitors had no effect on the ligand-mediated induction of apoE.



#### FIGURE 4.14. EFFECT OF 22R-HYDROXYCHOLESTEROL AND INHIBITORS ON APOE PROTEIN EXPRESSION IN HUMAN PRIMARY MONOCYTE-DERIVED MACROPHAGES

Human monocytes were isolated and cultured for 7days in RPMI supplemented with 5% (v/v) human serum, after-which they were treated with 3µg/ml of 22(R)-hydroxycholesterol (denoted 22R) or 22(S)-hydroxycholesterol (denoted 22S) for 24h. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand. Inhibitors were added 1h before the ligand at various concentrations: Curcumin (35µM), SP600125 (50µM), LY294002 (100µM), Apigenin  $(40\mu M)$ , PD98059 (50 $\mu M$ ), SB202190 (10 $\mu M$ ). Cells were also treated with  $3\mu q/ml$  of the control ligand 22(S)-hydroxycholesterol (denoted 22S). Western blot analysis was carried out using 20µg of whole cell extracts. Blotted membranes were incubated with anti-apoE primary antibody (goat polyclonal IgG) and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. The apoE protein level for each sample was determined by densitometric analysis and plotted on a bar chart. The control represents macrophages treated with vehicle (DMSO) only has been assigned as 1, with the apoE protein level for the remaining samples being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant difference compared to control cells treated with 22(R)-hydroxycholesterol (\*p<0.05).

# **4.2.8.** EFFECT OF 22(R)-HYDROXYCHOLESTEROL AND CELL SIGNALLING INHIBITORS ON ABCA1 AND ABCG1 MRNA EXPRESSION IN THP-1 MACROPHAGES

Given the involvement of the JNK/SAPK, PI3K and CK2 signalling pathways in the regulation of apoE, it was decided to investigate whether these pathways are also involved in the activation of other LXR target genes, such as ABCA1 and ABCG1. To analyse this, THP-1 cells were differentiated in 10% (v/v) delipidated HI-FCS with PMA (0.16 $\mu$ M) for 24h. After the 24h-differentiation period, inhibitors were added for 1h before ligand treatment, and 22(R)-hydroxycholesterol (2 $\mu$ g/ml) was then added to the appropriate flasks and left to incubate for a further 12h. Total RNA was then extracted (Section 2.4.1) and Taqman Real-Time quantitative RT-PCR was carried out using ABCA1-, ABCG1- and  $\beta$ -microtubulin-specific primers (Table 2.12) as described in Section 2.4.5 and reactions were monitored in real time mode using Sequence Detection Systems, version 1.7 software.

As shown in Figure 4.15 and Figure 4.16, 22(R)-hydroxycholesterol induced both ABCA1 and ABCG1 mRNA, to that of levels previously shown in several studies (Costet *et al.*, 2000; Whitney *et al.*, 2001; Venkateswaran *et al.*, 2000a; Venkateswaran *et al.*, 2000b). The results from these experiments also demonstrated that pre-treatment with the p38 MAPK inhibitor, SB202190 had no significant effect on ABCA1 or ABCG1 mRNA induction. In contrast, similar to their effects on apoE mRNA, the JNK/SAPK inhibitors, SP600125 and curcumin, both significantly reduced the ligand-mediated induction for both ABCA1 and ABCG1, as did apigenin (CK2 inhibitor). However, the PI3K inhibitor, LY294002 only had an inhibitory effect on the ligand-mediated induction of ABCG1 mRNA levels. In addition, similar to the result seen for LY294002, the ERK MAPK inhibitor, PD98059, only reduced the mRNA induction of ABCG1 and demonstrated no effect on ABCA1 mRNA levels.

# **4.2.9.** Effect of 22(R)-hydroxycholesterol and cell signalling inhibitors on LXR $\alpha$ mRNA expression in THP-1 macrophages

The LXR $\alpha$  promoter is another important target of the LXRs and it was therefore decided to investigate the effect of these same inhibitors on the 22(R)-hydroxycholesterol-induction of LXR $\alpha$ . Total RNA was extracted from differentiated THP-1 cells treated with ligand and inhibitors for 12h and Taqman Real-Time quantitative RT-PCR was carried out using LXR $\alpha$ - and  $\beta$ -microtubulin-specific primers (Table 2.12) as described in Section 2.4.5.

As shown in Figure 4.17, 22(R)-hydroxycholesterol treatment results in a 5-fold induction in LXR $\alpha$  mRNA expression, which is attenuated by pre-treatment with inhibitors against the p38

MAPK, JNK MAPK, PI3K and CK2 cell signalling pathways. The only inhibitor that had no significant effect in this case was the ERK MAPK inhibitor, PD98059.



#### FIGURE 4.15. EFFECT OF CELL SIGNALLING INHIBITORS ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF ABCA1 MRNA EXPRESSION

THP-1 monocytes were differentiated with PMA (0.16 $\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2 $\mu$ g/ml] for 12h. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: PD98059 (50 $\mu$ M), SB202190 (10 $\mu$ M), Curcumin (25 $\mu$ M), SP600125 (50 $\mu$ M), LY294002 (100 $\mu$ M), Apigenin (40 $\mu$ M). Taqman Real-Time quantitative RT-PCR was carried out using ABCA1- and  $\beta$ -microtubulin-specific primers and reactions were monitored in real time mode using Sequence Detection Systems, version 1.7 software. The ABCA1:  $\beta$ -microtubulin ratio in macrophages treated with vehicle only has been assigned as 1, with the ratio for the remaining samples being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant difference compared to control cells treated with 22(R)-hydroxycholesterol (\*p<0.05).



#### FIGURE 4.16. EFFECT OF CELL SIGNALLING INHIBITORS ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF ABCG1 MRNA EXPRESSION

THP-1 monocytes were differentiated with PMA (0.16 $\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2 $\mu$ g/ml] for 12h. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: PD98059 (50 $\mu$ M), SB202190 (10 $\mu$ M), Curcumin (25 $\mu$ M), SP600125 (50 $\mu$ M), LY294002 (100 $\mu$ M), Apigenin (40 $\mu$ M). Taqman Real-Time quantitative RT-PCR was carried out using ABCG1- and  $\beta$ -microtubulin-specific primers and reactions were monitored in real time mode using Sequence Detection Systems, version 1.7 software. The ABCG1: $\beta$ -microtubulin ratio in macrophages treated with vehicle only has been assigned as 1, with the ratio for the remaining samples being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant difference compared to control cells treated with 22(R)-hydroxycholesterol (\*p<0.05; \*\*p<0.005).



#### FIGURE 4.17. EFFECT OF CELL SIGNALLING INHIBITORS ON THE 22(R)-HYDROXYCHOLESTEROL-MEDIATED INDUCTION OF LXR & MRNA EXPRESSION

THP-1 monocytes were differentiated with PMA (0.16 $\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2 $\mu$ g/ml] for 12h. In addition, differentiated cells were treated with vehicle (DMSO) in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: PD98059 (50 $\mu$ M), SB202190 (10 $\mu$ M), Curcumin (25 $\mu$ M), SP600125 (50 $\mu$ M), LY294002 (100 $\mu$ M), Apigenin (40 $\mu$ M). Taqman Real-Time quantitative RT-PCR was carried out using LXR $\alpha$ - and  $\beta$ -microtubulin-specific primers and reactions were monitored in real time mode using Sequence Detection Systems, version 1.7 software. The LXR $\alpha$ :  $\beta$ -microtubulin ratio in macrophages treated with vehicle only has been assigned as 1, with the ratio for the remaining samples being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant difference compared to control cells treated with 22(R)-hydroxycholesterol (\*p<0.05).

#### 4.3. Discussion

Cell signalling pathways have been implicated in the transactivation function of several nuclear receptors either through direct phosphorylation of the nuclear receptor itself or through modification of coregulator proteins necessary for active transcription. This modification of nuclear receptor function by kinase cascades allows an additional mechanism for regulating the transactivation potential and the specificity of these transcription factors, and allows cross-talk to occur between nuclear receptors and cell signalling pathways. For example, there are several reports indicating that retinoic acid receptors (RARs) and other receptors can be phosphorylated by cyclin-dependant kinases (cdks) and that this phosphorylation is important for ligand-dependant transactivation (Rochette-Egly et al., 1997; Rochette-Egly et al., 1992; Taneja et al., 1997). Furthermore, other nuclear receptors such as the estrogen receptors (ERs) are phosphorylated at serine or threonine residues by the mitogen-activated protein kinase (MAPK) in vitro, and in cells treated with growth factors that stimulate the Ras-MAPK cascade, and this phosphorylation enhances transcriptional activity (Kato et al., 1995; Patrone et al., 1996). PPARa is also modulated by phosphorylation by MAPK, and this phosphorylation enhances transcriptional activity (Juge-Aubry et al., 1999). However, phosphorylation of the AF-1 domain of PPARy by the same kinase negatively regulates its transcriptional functions. Interestingly, this modification reduces ligand binding to the receptor, showing that binding can be regulated by intramolecular communication between the modulatory domain and the C-terminal LBD (Shao et al., 1998). In addition, MAPK-dependent phosphorylation of the RXR can also alter biological actions of a partner receptor (Solomon et al., 1999). Therefore, as no studies have previously shown that cell signalling pathways are involved in the LXRmediated regulation of genes, the overall aim of this chapter was to determine whether any pathways were involved with respect to the LXR-mediated induction of apoE expression.

Work by Laffitte *et al.* (2001b) demonstrated that the nuclear receptors LXR $\alpha$  and LXR $\beta$  and their oxysterol ligands were key regulators of lipid-inducible expression of apoE in both macrophages and adipose tissue. They also showed that the LXR/RXR heterodimer regulated apoE transcription directly, through interaction with a LXRE present in both ME1 and ME2. As this newly revised project was based on the ability of oxysterol ligands, to induce apoE transcription, it was necessary to ensure that the earlier data produced by Laffitte *et al.* (2001b), could be reproduced. It was therefore decided to examine the ability of the LXR ligand, 22(R)-hydroxycholesterol to increase apoE mRNA expression in macrophages (Figure 4.5). In addition to being extensively used in previous LXR studies (Laffitte *et al.*, 2001; Venkateswaran *et al.*, 2000b; Costet *et al.*, 2000; DeBose-Boyd *et al.*, 2001; Chiang *et al.*, 2001), 22(R)-hydroxycholesterol was used in this study due to its ability to specifically activate LXRs without crossreactivity with other known nuclear receptors (Janowski *et al.*, 1996). As shown in Figure 4.5, 22(R)-hydroxycholesterol

induced apoE mRNA expression, and this induction was also observed at the protein level (Figure 4.6). This induction of apoE protein levels by 22(R)-hydroxycholesterol was also shown to be timedependent (Figure 4.7) with a maximum induction occurring at 24h. Ligand treatment of THP-1 macrophages also increased the level of secreted apoE (Figure 4.8), which is of potential therapeutic value as apoE secretion by macrophages is thought to be atheroprotective, not only through its role in the reverse cholesterol transport process but also due to its local antiinflammatory and anti-proliferative effects on the surrounding cells of the vessel wall. Although the study by Laffitte et al. (2001b) did not investigate whether 22(R)-hydroxycholesterol induced apoE secretion, a study by Perez et al. (2003) has since shown, in agreement with this result, that the oxysterol ligand 22(R)-hydroxycholesterol increases apoE secretion in THP-1 macrophages. In addition to treatment with the LXR activator 22(R)-hydroxycholesterol, THP-1 cells were treated with S enantiomer of this ligand, 22(S)-hydroxycholesterol. Though this ligand binds to both subtypes of LXR competitively and with a high affinity in vitro, in cells it is completely inactive (Janowski et al., 1996). As can be seen from Figure 4.5, 4.6 and 4.8, 22(S)-hydroxycholesterol had no effect at all on apoE, confirming that the induction of apoE expression seen with 22(R)hydroxycholesterol, is due to LXR activation.

Confident that the ligand induction of apoE expression was well established, we investigated the action of inhibitors against key cell signalling pathways on this LXR-mediated induction of apoE expression. Several inhibitors were used, and the inhibitors against the JNK/SAPK MAPK, PI3K and CK2 signalling pathways, successfully blocked the ligand-mediated induction of apoE at the mRNA, protein and secreted protein level (Figure 4.10, 4.11, and 4.13). This inhibition was not due to any cytotoxic affect of the compound, as determined by trypan blue exclusion assays, and this inhibition with each compound could be seen to occur dose-dependently (Figure 4.12). Therefore, it can be concluded from these experiments that apoE induction in macrophages by the LXR family of transcription factors occurs through a JNK/CK2/PI3K-dependent manner. This is clearly a novel observation and requires further investigation.

Inhibitors against the ERK MAPK and p38 MAPK were also used, and were shown to have no affect on the induction of apoE expression at the mRNA and protein level, although an affect on apoE secretion (50% reduction) was seen by SB202190 inhibitor (Figure 4.13). This reduction in apoE secretion, could indicate that this LXR-mediated regulation of apoE secretion may have a post-translational component that is affected by the p38 MAPK pathway, although further work would be required to confirm this. The lack of inhibition by these inhibitors at the mRNA and protein level was not due the inhibitor themselves being inactive as the positive action of the SB202190 and PD98059 inhibitors used in this study have been confirmed by other studies in the laboratory. PD98059 (ERK inhibitor) was shown to successfully inhibit the PMA-induced induction of LPL expression in THP-1 monocytes (Mead, 2002; Singh, 2003), whereas SB202190 (p38 inhibitor) has been shown to prevent the TGF-β-mediated induction of apoE mRNA in THP-1 monocytes (Singh, 2003). Thus, in contrast to the effect of inhibitors of the JNK, CK2 and PI3K pathways on the ligand-mediated induction of apoE expression, inhibitors of the ERK MAPK and p38 MAPK pathways had no effect, which implies that these pathways are not involved in the LXR-mediated regulation of apoE.

An important result of this chapter is the ability of the ligand 22(R)-hydroxycholesterol to induce apoE protein expression in primary human macrophages (Figure 4.14). This has not been shown previously and confirms that the LXR regulation of apoE may be of potential therapeutic value in the treatment of human atherosclerosis. In addition, the effect of the cell signalling inhibitors on this ligand-mediated induction of apoE was investigated in primary human macrophages. This confirmation of the results seen in the THP-1 cell line was necessary as there are many instances where there are differences in response to stimuli between primary monocytes/macrophages and cell lines. For example, with respect to the MAPK signalling pathways, LPS stimulation has been shown to activate all three branches of the MAPK cascade in primary human macrophages, yet only the JNK/SAPK branch is activated in the THP-1 cell line (Rao, 2001). Similar differences between cell lines and primary cultures are found in other kinase cascade systems, such as the PI3K pathway (Rao, 2001). Our results demonstrate that a similar mechanism of regulation may be involved in the LXR-mediated regulation of apoE in primary cells as the inhibitors against the JNK/SAPK and PI3K pathways blocked apoE induction by 22(R)hydroxycholesterol in these cells also. Although it is important to note that a difference exists between primary cells and THP-1 macrophages with regards to CK2, as apigenin had no significant affect in primary human cells (Figure 4.14). Whilst this lack of inhibition by apigenin does imply that CK2 does not play an important role in the LXR-mediated induction in human primary monocytederived macrophages, further studies are essential to confirm this as an extensive dose-response experiment was not carried out with this inhibitor in primary cells due to limited amounts of material. Overall the results of these experiments indicates that there are similarities between the LXRmediated regulation of apoE in human primary cells and the THP-1 cell line and indicates that the THP-1 cell line is a suitable model for further investigation into the LXR-mediated regulation of apoE and that the results obtained in this cell line are applicable to primary cell investigations.

A primary aim of this project is to gain a better understanding of LXR-regulated transcription. Although the focus gene of the project is apoE, LXR regulates several other genes of the reverse cholesterol transport pathway. Therefore we decided to investigate the effects of inhibitors on the 22(R)-hydroxycholesterol induction of ABCA1 (Figure 4.15) and ABCG1 (Figure 4.16), in order to see whether certain LXR target genes share a common mode of regulation

through LXR. From Figure 4.15, it can be seen that both the JNK and CK2 pathways are involved in the ligand induction of ABCA1, although the PI3K pathway does not seem to play a role. ABCG1 regulation by LXR seems to be very similar to that of apoE as all three pathways appear to be involved, however the ERK MAPK inhibitor PD98059 also appears to have a significant effect. In addition to ABCA1 and ABCG1, the effect of these inhibitors was investigated on the autoregulation function of LXR $\alpha$ . The LXR $\alpha$  induction, was successfully blocked by the inhibitors against the p38 MAPK, JNK/SAPK MAPK, PI3K and CK2 cell signalling pathways (Figure 4.17). The only inhibitor that had no effect in this case was the ERK MAPK inhibitor, PD98059. These results indicate that a common mode of regulation for these LXR target genes through cell signalling pathways may exist, although a few subtle differences have been identified. These minor differences between the LXRmediated regulation of the various target genes may be of value in devising therapeutic approaches that could exploit these differences, allowing selective up-regulation of atheroprotective target genes without the unfortunate side-effect of triglyceride synthesis. In addition, these findings are consistent with other studies that have demonstrated that key LXR target genes have differential requirements with regards to coregulator recruitment (Wagner *et al.*, 2003).

In conclusion, the results in this chapter have successfully identified three signal transduction pathways (JNK/SAPK MAPK, PI3K and CK2) that may be involved in the LXR-mediated induction of apoE. This is novel data and provides a basis for the further investigation of these pathways in the regulation of apoE by the nuclear receptor LXR. The following chapters will look at each of these pathways in more detail.

# <u>CHAPTER FIVE – ROLE OF JNK/SAPK MAP KINASE</u> <u>SIGNALLING IN THE LXR-MEDIATED REGULATION OF</u> <u>APOE</u>

#### 5.1. INTRODUCTION

The results of the previous chapter have shown that the JNK/SAPK branch of the MAPK signalling cascade is involved in the LXR-mediated activation of apoE gene expression. As the JNK/SAPK pathway has not been described previously in this thesis, a brief description of the major components of this pathway is given below.

As mentioned earlier in Section 4.1, the JNK/SAPK subfamily belongs to the mitogenactivated protein kinase (MAPK) group. MAPKs are a large superfamily of evolutionary conserved enzymes connecting cell-surface receptors to critical regulatory targets within the cell. MAPK signalling pathways have been demonstrated to modulate gene expression through the specific phosphorylation and activation of a number of transcription factors. All MAPKs are "prolinedirected" and phosphorylate Ser/Thr residues only if followed immediately by proline (Roux and Blenis, 2004).



#### FIGURE 5.1. THE MAP KINASE CORE SIGNALLING MODULE

Various extracellular signals feed into the MAPK kinase kinase (MAPKKK)  $\rightarrow$  MAPK kinase  $\rightarrow$  MAPK core through several upstream regulators. Each MAPK is activated by dual Thr/Tyr phosphorylation catalysed by a MAPKK, which is, in turn regulated by Ser/Thr phosphorylation. The downstream targets of this signalling module then mediate the appropriate cellular response. Figure adapted from Kyriakis and Avruch, 2001.
All MAPK pathways are organised into core three-tiered signalling modules (Figure 5.1). Each MAPK is activated by dual phosphorylation of a tripeptide motif (Thr-Xaa-Tyr) located in the activation loop (T-loop) [Kyriakis and Avruch, 2001]. This phosphorylation is catalysed by a family of dual specificity kinases, MAPK kinases (MAPKKs), which are, in turn, regulated by Ser/Thr phosphorylation, catalysed by any of several protein kinase families collectively referred to as MAPK-kinase-kinases (MAPKKK). Upstream of MAPKKKs, MAPK core signalling modules are themselves regulated by a wide variety of upstream activators and inhibitors (Herskowitz, 1995; Kyriakis and Avruch, 2001; Marshall, 1995). In mammals, several MAPK subfamilies have been identified (Schaeffer and Weber, 1999) and each of these MAPK groups are respectively integrated into different signal transduction pathways and show differences in substrate specificity. The ERK and p38 groups of MAPK are related to enzymes found in budding yeast and contain the dual phosphorylation motifs Thr-Glu-Tyr and Thr-Gly-Tyr, respectively, whilst the c-jun N-terminal kinases (JNK), also known as the stress-activated protein kinases (SAPKs), represent a third group of MAPKs and contains the dual phosphorylation motif Thr-Pro-Tyr.

The JNK/SAPKs were initially identified by their ability to bind and phosphorylate the amino-terminal domain of the c-jun transcription factor in response to UV stimulation, and have since been shown to be activated by proinflammatory cytokines such as tumor necrosis factor- $\alpha$  $(TNF-\alpha)$  and interleukin-1 (IL-1), and by environmental stress (Hibi et al., 1993; Dérijard et al., 1994; Kyriakis et al., 1994; Minden et al., 1994). Phosphorylation of the JNK/SAPKs is mediated by the MAPKKs, MKK4/SEK1 and MKK7, both of which are present in the cytoplasm and also the nucleus (Tournier et al., 1999). The upstream components of MAPKK appear to be cell type and stimuli dependent. Several MAPKKKs have been reported to activate the JNK/SAPK signalling pathway. These include members of the MEKK group (MEKK1-4), the mixed-lineage protein kinase group (MLK1, MLK2, MLK3, DLK and LZK), the apoptosis signal-regulating kinase group (ASK1 and ASK2), TGF- $\beta$ -activated kinase (TAK) and TPL-2 (Figure 5.2). In most cases in vitro experimental data has demonstrated that these MAPKKKs phosphorylate and activate MKK4 and/or MKK7, although, it has not been established whether these protein kinases are physiological regulators of the JNK/SAPK pathway in vivo, and it is also unclear which MAPKKKs are relevant to specific physiological stimuli. Further upstream are the Ste-20-like MAPKKKKs such as the p21-activated kinases (PAKs), germinal centre kinase (GCK) and histidine protein kinase (HPK). Just how these kinases are regulated remains to be fully elucidated, but roles have been suggested for adaptor proteins (e.g. Nck, Crk, Shc or TNF-receptor-associated factors) and/or small GTP-binding proteins of the Rho and Rac families (Kyriakis, 1999).

CHAPTER FIVE - ROLE OF JNK/SAPK MAPK SIGNALLING ON THE LXR-MEDIATED REGULATION OF APOE



#### FIGURE 5.2. SCHEMATIC REPRESENTATION OF THE JNK/SAPK MAPK PATHWAY

GCK, germinal centre kinase; PAK, p21-activated kinase; HPK, histidine protein kinase; MEKKs, MAPK/ERK kinase kinase; MLKs, mixed-lineage protein kinases; ASKs, apoptosis signal-regulating kinases; TAK, TGF- $\beta$ -activated kinase; SEK, SAPK/Erk kinase. See text for further details.

JNK/SAPK has several other downstream targets in addition to c-jun, many of which, like c-jun are components of the AP-1 complex. AP-1 is a heterodimer comprised of bZIP transcription factors, typically c-jun and JunD, along with members of the fos (usually c-fos) and ATF (usually ATF-2) families. All bZIP transcription factors contain leucine zippers that enable homo- and heterodimerisation, and AP-1 components are organised into jun-jun, jun-fos, or jun-ATF dimers (Karin *et al.*, 1997). Activation of AP-1 involves both, the direct phosphorylation or dephosphorylation of AP-1 components, as well as the phosphorylation and activation of transcription factors that induce elevated expression of c-jun or c-fos. Phosphorylation of c-jun or ATF-2 within their N-terminal transactivation domains by JNK/SAPK correlates with enhanced transactivating activity (Derijard *et al.*, 1994; Gupta *et al.*, 1995; Kyriakis *et al.*, 1994; Pulverer *et al.*, 1991). The Elk-1 transcription factor is another direct JNK/SAPK target (Cavigelli *et al.*, 1995),

which is involved in the induction of the c-fos gene, thereby indirectly effecting AP-1 activity (Treisman, 1995).

The involvement of cell signalling pathways in the nuclear receptor regulation of target genes has been discussed in the previous chapter. Although we have shown that inhibitors of the JNK/SAPK pathway are able to inhibit the ligand-mediated induction of apoE and other LXR target genes (Chapter 4), it is necessary to investigate how this pathway may potentially effect LXR signalling. The possibility exists that any one of the components of the JNK/SAPK pathway may be affecting the LXR transcription complex. The most straightforward and likely hypothesis behind the involvement of the JNK/SAPK pathway in LXR transcription would be the phosphorylation of the LXR protein or one of the coregulator components of the transcription complex, which would result in an increase in transcriptional activation of the target genes. This mechanism of involvement of cell signalling pathways in nuclear receptor signalling has been described extensively in the previous chapter and the JNK/SAPK pathway has been shown to directly phosphorylate several nuclear receptors including PPAR $\gamma$  (Adams *et al.*, 1997; Hu *et al.*, 1996; Camp *et al.*, 1999), GR (Rogatsky *et al.*, 1998) and RXR (Lee *et al.*, 2000).

Another hypothesis for the involvement of the JNK/SAPK pathway in LXR transcription would be through the activation of one of the various downstream transcription factors, such as the AP-1 complex. Recent work in keratinocytes has demonstrated that LXR activators induce a general increase in expression of AP-1-regulated genes (Schmuth *et al.*, 2004). In addition several other nuclear hormone receptors have been shown to alter transcription in various other cell types through the AP-1 response element (Uht *et al.*, 1997). This involvement of the AP-1 complex may occur through several mechanisms including interaction with the nuclear receptors or coregulators directly, binding to AP-1 elements at the promoters of coregulators or target genes, or even through the binding to AP-1 elements at the promoters of the nuclear receptors. Whether any of these mechanisms are involved in the regulation of apoE transcription by the LXRs remains to be determined and will be discussed further in Section 5.3.

Therefore, the aim of this chapter was to confirm whether the LXR ligand 22(R)hydroxycholesterol activates the JNK/SAPK pathway in THP-1 macrophages and to elucidate which components of the pathway are involved in this LXR-mediated regulation of apoE.

#### 5.2. RESULTS

# 5.2.1. EFFECT OF DOMINANT NEGATIVE EXPRESSION PLASMIDS ON THE LXR-MEDIATED INDUCTION OF APOE IN THP-1 MACROPHAGES

In order to provide further, independent, confirmation of the results obtained from the previous chapter, which demonstrated that the JNK/SAPK pathway was involved in the LXR regulation of apoE, it was decided to use dominant negative expression plasmids for the MKK4/SEK1 MAPKK and the JNK/SAPK MAPK. The cDNA of the enzymatically inactive form of SEK1 was kindly provided by J.R. Woodgett (Ontario Cancer Institute), whilst the dominant negative JNK/SAPK plasmid, SAPK $\alpha$ -VPF, in which the phosphorylation sites Thr-Pro-Tyr were changed to Val-Pro-Phe, was a gift from E.Nishida (Kyoto university).

Transfections were carried out in THP-1 monocytes and the DNA constructs were transfected using the Effectene<sup>™</sup> transfection reagent (Qiagen), as described in Section 2.3.5.2. PMA (0.16µM) was added immediately after the transfection of cells was complete and the cells were left to differentiate for 24h. After the 24h incubation, cells were treated with 2µg/ml of 22(R)-hydroxycholesterol or 22(S)-hydroxycholesterol for a further 24h, after which whole cell protein extracts were prepared (see Section 2.5.1). Of each sample, 20µg of total protein was subjected to SDS-PAGE (Section 2.5.4) and then transferred to a PVDF membrane (Section 2.5.5). The membrane was then incubated with an anti-apoE primary antibody (Biogenesis) and immunodetection was carried out using the appropriate secondary antibody and the ECL detection system (Amersham) as described in Section 2.5.6. In addition to the dominant negative constructs, as a control, cells were transfected with the empty pcDNA<sub>3</sub> vector (Appendix IV) to ensure that the results obtained were due to the expression of the dominant negative and not the plasmid or the transfection reagent.

From Figure 5.3, it can be seen that both the SEK1 and the JNK/SAPK dominant negative plasmids result in a 50% decrease in the induction of apoE by 22(R)-hydroxycholesterol. This reduction can be attributed to the presence of the dominant negative construct as the induction in the apoE protein levels in the control cells transfected with the empty vector remains similar to that seen in previous experiments (Chapter 4). Therefore, this result is consistent with the data obtained with curcumin and SP600125 in the previous chapter and clearly confirms a role for the JNK/SAPK pathway in the LXR-mediated regulation of apoE. Also the reduction of the apoE induction by the SEK1 dominant negative construct, allows us to speculate that this kinase is an upstream component involved in the JNK/SAPK regulation of the LXR-mediated induction of apoE.



FIGURE 5.3. EFFECT OF DOMINANT NEGATIVE EXPRESSION PLASMIDS ON THE LXR-MEDIATED INDUCTION OF APOE IN THP-1 MACROPHAGES

THP-1 monocytes were transfected (see section 2.3.5.2) with the indicated expression plasmids or the control vector and differentiated with PMA (0.16 $\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h. The LXR ligands, 22(R)-hydroxycholesterol (denoted 22R) [2 $\mu$ g/ml] or 22(S)-hydroxycholesterol (denoted 22S) [2 $\mu$ g/ml] were then added to the cells for a further 24h. Western blot analysis was carried out using 20 $\mu$ g of whole cell extracts. Blotted membranes were incubated with anti-apoE primary antibody (goat polyclonal IgG) and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. The apoE protein level for each sample was determined by densitometric analysis and plotted on a bar chart. The control represents macrophages treated with vehicle (DMSO) in the place of ligand and has been assigned as 1 in each case, with the apoE protein level for the remaining samples being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant inhibition compared to pcDNA<sub>3</sub> transfected cells treated with 22(R)-hydroxycholesterol (\*p<0.05).

### 5.2.2. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON JNK/SAPK PHOSPHORYLATION IN THP-1 MACROPHAGES

In order to fully confirm a role for the JNK/SAPK pathway in the regulation of apoE by LXR, it was necessary to investigate whether the ligand, 22(R)-hydroxycholesterol activates this pathway. The JNK/SAPK pathway is a phosphorylation cascade and the various components of the pathway become active once phosphorylated, therefore it was decided to ensure that the main component of this pathway, JNK/SAPK itself is in fact phosphorylated in response to ligand treatment. JNK/SAPK phosphorylation by upstream MAPKKs, such as SEK1 occurs on the dual phosphorylation motif, Thr183-Pro-Tyr185 and recognition of this phosphorylated form of JNK/SAPK can be accomplished through western blot analysis. Antibodies that specifically recognise this phosphorylated form of JNK/SAPK are commercially available and have been extensively used to investigate the role of JNK/SAPK signalling in macrophages (Waetzig *et al.*, 2002; Wang *et al.*, 2002; Dreskin *et al.*, 2001).

Whole cell extracts from differentiated THP-1 cells treated with 22(R)-hydroxycholesterol for various time-points were isolated using the phosphatase-free isolation buffer as described in Section 2.5.1, and western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) was carried out using the anti-phospho-JNK/SAPK antibody with overnight incubation. As shown in Figure 5.4, 22(R)-hydroxycholesterol appears to induce JNK/SAPK phosphorylation. This induction was dependent on ligand treatment and occurred in a time-dependent manner. Ligand-mediated phosphorylation of the p46 isoform peaked with a 4.5-fold induction at 30min and returned to baseline levels after 24h. The following studies presented in this chapter will also demonstrate that this induction was not due to *de novo* protein synthesis as no change in total protein occurred with ligand treatment, and that this induction was dependent on LXR activation as the inactive enantiomer of 22(R)-hydroxycholesterol, 22(S)-hydroxycholesterol did not induce JNK/SAPK activation (Section 5.2.5).



FIGURE 5.4. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON JNK/SAPK PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22(R)-hydroxycholesterol (denoted 22R) [2µg/m]] for the indicated time-points. Western blot analysis was carried out using 80µg of whole cell extracts. Blotted membranes were incubated with phospho-JNK/SAPK (Thr183/Tyr185) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. The control sample represents differentiated THP-1 cells treated with vehicle (DMSO) only, in the place of ligand and was taken at the 15min time-point in this experiment. The p46 phosphorylated protein level for each sample was determined by densitometric analysis and plotted on a bar chart. The control (DMSO-treated cells) has been assigned as 1, with the phosphorylated level for the remaining samples being represented relative to this control. Results are representative of three separate experiments.

#### 5.2.3. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON JNK/SAPK ACTIVITY IN THP-1 MACROPHAGES

Although protein phosphorylation is an excellent marker of activity, another method of measuring JNK/SAPK activity is to determine the kinase activity directly from cell extracts. JNK/SAPK activation can be routinely measured by protein kinase activity towards the transcription factor substrate c-jun. In this study, a non-radioactive method of measuring JNK/SAPK activity was employed based on the JNK/SAPK Assay Kit (NEB). Differentiated THP-1 cells were treated with 22(R)-hydroxycholesterol or DMSO for 30min and whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1). The kinase assay was carried out as described in Section 2.6.3, whereby a c-jun fusion protein linked to agarose beads is used to "pull down" the JNK/SAPK enzyme from whole cell extracts. The isolated JNK/SAPK enzyme, upon addition of a kinase buffer and ATP, phosphorylates the c-jun substrate and this reaction mix was then subjected to SDS-PAGE, whereby the phospho-c-jun (Ser63) antibody used detects c-jun only when catalytically activated by phosphorylation at Ser63.

From Figure 5.5, it can be seen that JNK/SAPK isolated from 22(R)-hydroxycholesterol treated THP-1 macrophages phosphorylates c-jun to a greater extent (2-3-fold) than JNK/SAPK obtained from control cells treated with DMSO. These results are consistent with that seen previously for JNK/SAPK phosphorylation by the ligand and clearly demonstrates the ability of 22(R)-hydroxycholesterol to activate the JNK/SAPK pathway.

CHAPTER FIVE - ROLE OF JNK/SAPK MAPK SIGNALLING ON THE LXR-MEDIATED REGULATION OF APOE



FIGURE 5.5. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON JNK/SAPK ACTIVITY IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] for 30min. The kinase assay was carried out using 250µg of whole cell extracts as described in Section 2.6.3 and blotted membranes were incubated with phospho-c-jun (Ser63) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. The c-jun phospho-protein level for each sample was determined by densitometric analysis and plotted on a bar chart. The control represents macrophages treated with vehicle (DMSO) in the place of ligand and has been assigned as 1, with the phospho-protein level for the ligand-treated sample being represented relative to this control. The data shown is the mean  $\pm$ SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant difference compared to control cells treated with DMSO (\*p<0.05).

#### 5.2.4. EFFECT OF CURCUMIN AND SP600125 ON THE LXR-MEDIATED INDUCTION OF JNK/SAPK ACTIVITY

#### IN THP-1 MACROPHAGES

In order to confirm that the inhibition of the 22(R)-hydroxycholesterol-mediated induction of apoE by the JNK/SAPK inhibitors was due to an inhibition in JNK/SAPK activity, we next investigated the effect of curcumin and SP600125 on the ligand-mediated induction of JNK/SAPK kinase activity.

Whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) from differentiated THP-1 cells that were pre-treated with inhibitors for 1h before the addition of 22(R)-hydroxycholesterol for 30min. The JNK/SAPK kinase assay was then carried out as described in Section 2.6.3. From Figure 5.6a, it can be seen that both curcumin and SP600125

inhibited the induction in JNK/SAPK activity caused by ligand treatment. This inhibition by the JNK/SAPK inhibitors is consistent with the previously observed result, whereby curcumin and SP600125 prevent the ligand-mediated induction of apoE and therefore confirms that activation of JNK pathway is essential in the LXR-mediated regulation of apoE. Also included in this experiment was the inactive ligand, 22(S)-hydroxycholesterol, which did not effect JNK/SAPK activity, indicating that 22(R)-hydroxycholesterol induction of the JNK/SAPK pathway requires LXR activation.

Although unlikely, there is a possibility that the increased activity of JNK/SAPK in 22(R)hydroxycholesterol-treated cells may be the result of *de novo* protein synthesis. Therefore, to rule out this possibility the whole cell lysates used for the kinase assays were subjected to western blot analysis using an antibody which detects the endogenous levels of total JNK/SAPK protein. As can be seen from Figure 5.6b, both the control and 22(R)-hydroxycholesterol-treated cells showed similar steady-state protein levels. Also from these results it can be seen that the effect of the inhibitors was mainly due to an inhibition of JNK/SAPK activity and not due to an effect on the endogenous levels of JNK/SAPK, although a slight reduction in the total protein of the p46 JNK/SAPK isoform was observed in cells pre-treated with curcumin.

### 5.2.5. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED INDUCTION OF JNK/SAPK ACTIVITY IN THP-1 MACROPHAGES

Given that the inhibitors against the PI3K and CK2 pathways also inhibit the apoE induction by 22(R)-hydroxycholesterol, it was decided to investigate whether these inhibitors affected apoE levels indirectly through the JNK/SAPK pathway. Both PI3K and CK2 have been shown to function as upstream components for the JNK/SAPK pathway and inhibitors which target these pathways may, therefore, inhibit JNK/SAPK activity (Go *et al.*, 1998; Assefa *et al.*, 1999; Eom *et al.*, 2001; Fleming *et al.*, 2000).

Whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) from differentiated THP-1 cells that were pre-treated with LY294002 or apigenin for 1h before the addition of ligand for 30min. The JNK/SAPK kinase assay was then carried out as described in Section 2.6.3.

Figure 5.7a shows that 22(R)-hydroxycholesterol induced the JNK/SAPK activity as shown previously and demonstrates that neither LY294002 nor apigenin had a significant effect on this induction of JNK/SAPK activity or on the endogenous JNK/SAPK protein levels (Figure 5.7b). This result clearly indicates that neither the PI3K, nor the CK2 enzymes are acting upstream of JNK/SAPK and that the JNK/SAPK pathway is independent to these pathways in the LXR-mediated regulation of apoE.





#### FIGURE 5.6. EFFECT OF CURCUMIN AND SP600125 ON THE LXR-MEDIATED INDUCTION OF JNK/SAPK ACTIVITY IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA (0.16 $\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22(R)-hydroxycholesterol (denoted 22R){2µg/ml] or 22(S)hydroxycholesterol (denoted 22S) [2µq/ml] for 30min. In addition, cells were treated with the vehicle, DMSO in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: Curcumin (25µM), SP600125 (50µM). A The kinase assay was carried out using 250µg of whole cell extracts as described in Section 2.6.3 and blotted membranes were incubated with phospho-c-jun (Ser63) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. B Western blot analysis was carried out using 80 µg of whole cell extracts. Blotted membranes were incubated with total-JNK/SAPK primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. c The c-jun phospho-protein and total JNK level for each sample was determined by densitometric analysis and the c-jun:total protein for each sample has been represented relative to the control, which has been assigned as 1 and plotted on a bar chart. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \*\* Represents the significant inhibition compared to control cells treated with 22(R)-hydroxycholesterol (\*\*p<0.005).



#### FIGURE 5.7. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED INDUCTION OF JNK/SAPK ACTIVITY IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA (0.16 µM) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22(R)-hydroxycholesterol (denoted 22R){2µg/ml] and 22(S)hydroxycholesterol (denoted 22S) [2µg/ml] for 30min. In addition, cells were treated with the vehicle, DMSO in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: LY294002 (25µM), apigenin (40µM). A The kinase assay was carried out using 250µg of whole cell extracts as described in Section 2.6.3 and blotted membranes were incubated with phospho-c-jun (Ser63) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. B Western blot analysis was carried out using 80µg of whole cell extracts. Blotted membranes were incubated with total-JNK/SAPK primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of three separate experiments. c The c-jun phospho-protein and total JNK level for each sample was determined by densitometric analysis and the c-jun:total protein for each sample has been represented relative to the control, which has been assigned as 1 and plotted on a bar chart. The data shown is the mean ±SD from three independent experiments. The data was analysed by student's t-test.

#### 5.2.6. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON C-JUN PHOSPHORYLATION IN THP-1 MACROPHAGES

The results from the previous sections have clearly demonstrated that the LXR-mediated transcription of apoE requires the activity of JNK/SAPK. Exactly how this cell signalling pathway is involved in the LXR-mediated regulation of apoE is unclear. As described previously cell signalling pathways may effect nuclear receptors regulation of target genes through various methods. Therefore, in order to gain a better understanding of the role of the JNK/SAPK pathway in LXR signalling it was decided to investigate which downstream target, if any was being activated by JNK/SAPK.

One of the best-known substrates of the JNK/SAPK MAPKs is c-jun, the major component of the AP-1 complex (Bohmann *et al.* 1987; Angel *et al.* 1988). c-jun is phosphorylated by the JNK/SAPKs on Ser63 and Ser73 and this phosphorylation enhances its transcriptional activity (Pulverer *et al.* 1991; Smeal *et al.* 1991). Antibodies that specifically recognise the phosphorylated form of c-jun are commercially available therefore c-jun phosphorylation can be investigated by western blot analysis.

For these experiments whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) from differentiated THP-1 cells that were treated with 22(R)-hydroxycholesterol for 30min. Total c-jun proteins were immunoprecipitated from whole cell lysates (Section 2.5.3) and were subjected to western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) using the anti-phospho-c-jun (Ser63) antibody with overnight incubation. As shown in Figure 5.8, 22(R)-hydroxycholesterol induced c-jun phosphorylation after only 30min. This induction was ligand specific as the following studies presented in this chapter demonstrate that the inactive enantiomer of 22(R)-hydroxycholesterol, 22(S)-hydroxycholesterol did not induce c-jun phosphorylation (Section 5.2.8).

CHAPTER FIVE - ROLE OF JNK/SAPK MAPK SIGNALLING ON THE LXR-MEDIATED REGULATION OF APOE



#### FIGURE 5.8. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON C-JUN PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA (0.16µM) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] for 30min. In addition, cells were treated with the vehicle, DMSO in the place of ligand (denoted control). Immunoprecipitation of whole cell extracts (250µg) was carried out using a total-c-jun antibody as described in Section 2.5.3 and the isolated c-jun proteins were subjected to western blot analysis. Blotted membranes were incubated with a phospho-c-jun (Ser63) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of three separate experiments.

# 5.2.7. Effect of curcumin and SP600125 on the LXR-mediated phosphorylation of c-jun in THP-1 macrophages

In order to confirm that the inhibition of the 22(R)-hydroxycholesterol-mediated induction of apoE by the JNK/SAPK inhibitors was due to an inhibition in c-jun activity, we next investigated the effect of curcumin and SP600125 on the ligand-mediated phosphorylation of c-jun. Whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) from differentiated THP-1 cells that were pre-treated with inhibitors for 1h before the addition of 22(R)hydroxycholesterol for 30min. Total c-jun proteins were immunoprecipitated from whole cell lysates (Section 2.5.3) and were subjected to western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) using the anti-phospho-c-jun antibody with overnight incubation. The results in Figure 5.9, demonstrate, as shown in the previous section, that 22(R)-hydroxycholesterol treatment of THP-1 macrophages results in c-jun phosphorylation, and that both, curcumin and SP600125 are able to inhibit this induction. This inhibition by the JNK/SAPK inhibitors is consistent with the previously observed result, whereby curcumin and SP600125 prevent the ligand-mediated induction of apoE and JNK/SAPK activity and therefore confirms that activation of c-jun is essential in the LXR-mediated regulation of apoE. Also included in this experiment was the inactive ligand, 22(S)hydroxycholesterol, which had no effect on c-jun phosphorylation, which indicates that this phosphorylation of c-jun is mediated through the activation of LXR.

CHAPTER FIVE - ROLE OF JNK/SAPK MAPK SIGNALLING ON THE LXR-MEDIATED REGULATION OF APOE



#### FIGURE 5.9. EFFECT OF CURCUMIN AND SP600125 ON THE LXR-MEDIATED PHOSPHORYLATION OF C-JUN IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] and 22(S)-hydroxycholesterol (denoted 22S) [2µg/ml] for 30min. Inhibitors were added 1h before the ligand at various concentrations: Curcumin ( $25\mu$ M), SP600125 ( $50\mu$ M). Immunoprecipitation of whole cell extracts ( $250\mu$ g) was carried out using a total-c-jun antibody as described in Section 2.5.3 and the isolated c-jun proteins were subjected to western blot analysis. Blotted membranes were incubated with a phospho-c-jun (Ser63) primary antibody (rabbit polyclonal IgG) and detected using an antirabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. The c-jun phospho-protein level for each sample was determined by densitometric analysis and plotted on a bar chart. The control represents macrophages treated with vehicle (DMSO) in the place of ligand and has been assigned as 1, with the phospho-protein level for the ligand-treated sample being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \* Represents the significant inhibition compared to control cells treated with 22(R)-hydroxycholesterol (\*p<0.05).

146

## 5.2.8. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED PHOSPHORYLATION OF C-JUN IN

### THP-1 MACROPHAGES

Cross-talk between different signalling pathways is relatively common, and although the PI3K and CK2 pathway inhibitors did not affect the JNK/SAPK pathway upstream or at the level of JNK/SAPK, it is possible that the PI3K and CK2 pathways may converge with the JNK/SAPK pathway at the level of c-jun as both CK2 and downstream components of the PI3K pathway have been shown to activate c-jun in a JNK-independent manner (Pinna *et al.*, 1995; Lin *et al.*, 1992; Li *et al.*, 2004; Peron *et al.*, 2004). Therefore, in order to rule out the possibility that the effect of LY294002 and apigenin on the ligand-mediated induction of apoE was mediated through the inhibition of c-jun phophorylation, the effect of these inhibitors on c-jun phosphorylation was investigated.

Whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) from differentiated THP-1 cells that were pre-treated with inhibitors for 1h before the addition of 22(R)-hydroxycholesterol for 30min. Total c-jun proteins were immunoprecipitated from whole cell lysates (Section 2.5.3) and were subjected to western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) using the anti-phospho-c-jun antibody with overnight incubation.

From Figure 5.10, it can be seen that neither LY294002 nor apigenin had an affect on c-jun phosphorylation, which is consistent with the results of previous experiments in this chapter, whereby these inhibitors had no affect on JNK activity (Figure 5.7). This lack of inhibition by LY294002 and apigenin at the level of c-jun phosphorylation confirms once again that the inhibition of the ligand-mediated induction of apoE seen by these inhibitors does not occur through the JNK/SAPK pathway and that this pathway, in this instance is independent from the PI3K and CK2 pathways in its involvement in the LXR regulation of apoE.

CHAPTER FIVE - ROLE OF JNK/SAPK MAPK SIGNALLING ON THE LXR-MEDIATED REGULATION OF APOE





THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] and 22(S)-hydroxycholesterol (denoted 22S) [2µg/ml] for 30min. In addition, cells were treated with the vehicle, DMSO in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: LY294002 (25µM), apigenin (40µM). Immunoprecipitation of whole cell extracts (250µg) was carried out using a total-c-jun antibody as described in Section 2.5.3 and the isolated c-jun proteins were subjected to western blot analysis. Blotted membranes were incubated with a phospho-c-jun (Ser63) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of two separate experiments.

**5.2.9.** EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON ATF-2 PHOSPHORYLATION IN THP-1 MACROPHAGES JNK/SAPKs can also phosphorylate ATF-2 at Thr69 and Ser71, which activates the *trans*-activating activity of this protein (Gupta *et al.*, 1995). ATF-2 heterodimerises with c-jun, forming an AP-1 complex and also stimulates the expression of the c-jun gene (van Dam *et al.*, 1995). It was therefore decided to investigate whether 22(R)-hydroxycholesterol treatment also resulted in the phosphorylation of this member of the AP-1 family of transcription factors. Antibodies that specifically recognise the phosphorylated form of ATF-2 are commercially available therefore phosphorylation of this transcription factor can be investigated by western blot analysis.

Differentiated THP-1 cells were treated with the oxysterol ligand, 22(R)-hydroxycholesterol for various time-points and whole cell extracts were isolated using the phosphatase-free isolation buffer as described in Section 2.5.1. Western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) was carried out using the anti-phospho-ATF-2 antibody with overnight incubation. Treatment of THP-1 macrophages with 22(R)-hydroxycholesterol did not change the level of phosphorylation of ATF-2 (Figure 5.11). This result rules out a role for the ATF-2 transcription factor in the LXR-mediated regulation of apoE.

CHAPTER FIVE - ROLE OF JNK/SAPK MAPK SIGNALLING ON THE LXR-MEDIATED REGULATION OF APOE



FIGURE 5.11. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON ATF-2 PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] for the indicated time-points. As a control at each time-point, cells were treated with the vehicle, DMSO in the place of ligand. Western blot analysis was carried out using 80µg of whole cell extracts. Blotted membranes were incubated with phospho-ATF-2 (Thr71) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of two separate experiments.

#### 5.3. DISCUSSION

The results from the previous chapter indicated a role for the JNK/SAPK pathway in the LXRmediated regulation of apoE. Therefore, the aim of this chapter was to investigate this pathway in more detail and confirm its involvement in apoE regulation. Data from Chapter 4 also demonstrated that the JNK/SAPK pathway was involved in the LXR-mediated induction of other target genes, such as ABCA1, ABCG1 and LXR $\alpha$  itself. Therefore, the data presented in this chapter may also have a wide-spread relevance with regards to LXR regulation of a range of genes.

The initial experiments of this chapter confirmed through the use of SEK1 and JNK/SAPK dominant negative constructs that the JNK/SAPK pathway is essential for the induction of apoE by the LXR agonist 22(R)-hydroxycholesterol (Figure 5.3). These experiments also identified that SEK1 MAPKK activity in addition to JNK/SAPK MAPK activity was essential in the oxysterol induction of apoE. The following experiments, whereby JNK/SAPK phosphorylation and activation were investigated, also confirmed the role of the JNK/SAPK pathway in the LXR regulation of apoE, and demonstrated that treatment of macrophages with the oxysterol ligand 22(R)-

hydroxycholesterol, induced both phosphorylation (Figure 5.4) and the activity (Figure 5.5) of this enzyme after only 30min. In addition the results from these experiments confirmed that the induction of the JNK/SAPK pathway was due to LXR activation as 22(S)-hydroxycholesterol, which binds but does not activate the receptor had no affect on JNK/SAPK activity (Figure 5.6a). These results have therefore demonstrated a novel mechanism, whereby 22(R)-hydroxycholesterol is able to rapidly activate JNK phosphorylation and kinase activity in a ligand-specific manner that requires LXR activation and does not involve *de novo* protein synthesis.

The rapid kinetics in which the JNK/SAPK pathway is activated by 22(R)hydroxycholesterol in this study suggests that the activation of this pathway by LXR ligands is not dependent on gene activation. This ability of a nuclear receptor agonist to rapidly activate a cell signalling pathway independent of gene transcription is a recently characterised phenomenon, termed non-genomic or extranuclear signalling. These rapid actions of agonists have been described to generate intracellular second messengers, to activate or inhibit cell signalling pathways and modulate ion channels, and studies have indicated in part that this initiation of intracellular signalling mechanisms occurs through the nuclear receptor agonist binding to a receptor present at the plasma membrane or in the cytoplasm (Schmidt *et al.*, 2000; Boonyaratanakornkit and Edwards, 2004). It may be that a subpopulation of nuclear receptors mediates these actions by somehow associating with the cell membrane and/or signalling complexes in the cytoplasm (Schmidt *et al.*, 2000). Although studies have also suggested the potential existence of separate membrane receptors unrelated to the classical nuclear receptors which are responsible for this non-genomic action of nuclear receptor ligands (Boonyaratanakornkit and Edwards, 2004).

Although no such function has previously been described for the LXR subfamily of nuclear receptors, the non-genomic action for several other members of the nuclear receptor family has been well-characterised. For example, the human progesterone receptor (PR) has been shown to mediate rapid progestin activation of the Src/Ras/Raf/MAPK signalling pathway in mammalian cells, this is thought to occur through a direct interaction between a motif located in the N-terminal domain of the receptor and the Src homology 3 domain of Src tyrosine kinases (Migliaccio *et al.*, 1998; Boonyaratanakornkit *et al.*, 2001). Additionally, a novel membrane protein unrelated to nuclear PR was recently identified that has properties of a G-protein-coupled receptor for progesterone and has been shown to be involved in mediating the non-genomic signalling actions of progesterone that promotes oocyte maturation in fish (Boonyaratanakornkit and Edwards, 2004).

Another nuclear receptor agonist that has several rapid non-genomic signalling actions is estrogen. Estrogen induces an increase in the intracellular cGMP level and the release of nitric oxide within a few minutes of administration in endothelial cells (Caulin-Glaser *et al.*, 1997). Also,

estrogen activates the ERK MAPK pathway within 15min (Singh *et al.*, 1999). These actions appear to be mediated via a novel membrane-bound estrogen receptor. A protein termed modulator of non-genomic activity of ER (MNAR) has been identified that acts as an adaptor between ER and Src, and thus provides a mechanism for coupling of estrogen and ER with rapidestrogen-induced activation of Src and the downstream MAPK signalling cascades. Consistent with these findings, bovine serum albumin-conjugated estrogen, which is impermeable to cell membranes, has been reported to activate the ERK pathway in endothelial cells (Russell *et al.*, 2000) and in a neuroblastoma cell line (Watters *et al.*, 1997). Recent work has also established a non-genomic mechanism of gene regulation for PPAR $\gamma$ , whereby its ligand, 15-deoxy- $\Delta^{12,14}$ prostaglandin J<sub>2</sub> (15-d-PGJ2) resulted in enhanced AP-1 binding activity in VSMCs (Takeda *et al.*, 2001). This induction of AP-1 activity occurred within 30min and was shown to be due to activation of the ERK pathway, which occurred in the presence of actinomycin D and was partially dependent on PI3K.

Therefore from the results presented here and from the recent advancements in the understanding of nuclear receptor signalling it is possible to conclude that the LXR-mediated regulation of apoE requires the non-genomic activation of the JNK/SAPK pathway by 22(R)-hydroxycholesterol in addition to the 22(R)-hydroxycholesterol-mediated activation of the LXR heterodimer at the apoE promoter and enhancer sequences (Laffitte *et al.*, 2001b). As no previous work has investigated a non-genomic role for LXR agonists, it is unclear how the ligand is activating the JNK/SAPK pathway. It may be possible that a membrane-bound LXR receptor exists or perhaps 22(R)-hydroxycholesterol is behaving independently of LXR as a signalling molecule, either suggestion requires further investigation, although a membrane-bound receptor is the most probable mechanism as our studies have demonstrated that JNK/SAPK activation is ligand-specific and requires LXR activation as 22(S)-hydroxycholesterol did not activate the JNK/SAPK pathway (Figure 5.6a).

Having concluded that oxysterol ligand treatment of THP-1 macrophages results in an increase in JNK/SAPK phosphorylation and kinase activity, experiments were carried out to investigate whether any downstream targets of this pathway are involved in LXR-mediated regulation of apoE. As described earlier in this chapter, c-jun is a major downstream target of the JNK/SAPK pathway and heterodimerises with other transcription factors, such as c-fos and ATF-2 to form the AP-1 complex. Oxysterols and other nuclear receptor agonists have been shown to affect AP-1 activity in several cell lines and a recent study demonstrated that oxysterols induce specific AP-1 proteins during keratinocyte differentiation, which is due to LXR activation (Schmuth *et al.*, 2004). Therefore, c-jun phosphorylation was investigated and shown to be induced by 22(R)-hydroxycholesterol treatment (Figure 5.8).

CHAPTER FIVE - ROLE OF JNK/SAPK MAPK SIGNALLING ON THE LXR-MEDIATED REGULATION OF APOE



FIGURE 5.12. SCHEMATIC REPRESENTATION OF THE INVOLVEMENT OF THE JNK/SAPK PATHWAY IN THE LXR-mediated induction of apoE. See text for further details

Although the original hypothesis on the involvement of cell signalling pathways was that the LXRs or coregulators were being targeted for phosphorylation, the activation of a transcription factor by 22(R)-hydroxycholesterol suggests the existence of an alternative mechanism in the regulation of apoE by the LXRs (Figure 5.12). It is possible that c-jun may be interacting with the LXR transcriptional complex at the apoE promoter and this has been shown previously to occur in estrogen signalling, whereby ER $\alpha$ , c-jun, and the p160 coactivator GRIP1 form a multiprotein complex in vitro and in intact cells, and the ERa-c-Jun interaction is crucial for the stability of this complex (Teyssier et al., 2001). Another potential mechanism for the role of c-jun in LXR signalling is that this transcription factor, as part of the AP-1 complex is activating apoE through the AP-1 element of the apoE promoter. This AP-1 element has already been shown to be essential for the induction of apoE expression during macrophage differentiation (Basheeruddin et al., 1994) and may also play a role in the LXR activation of apoE. This requirement for the AP-1 element in the LXR-mediated induction of apoE may be due to the low-affinity of the LXRE present in the proximal promoter of apoE (Laffitte et al., 2001b), and the AP-1 transcription factor may therefore be required for the full stimulatory response of the apoE promoter to oxysterol ligands. Indeed, such a mechanism has recently been demonstrated for the rat SREBP-1c promoter, which requires in the

addition to a LXRE, the binding sites for Sp1, nuclear factor Y (NF-Y) and SREBP itself for full activation of the SREBP-1c promoter by insulin (Cagen *et al.*, 2004).

Perhaps the most likely mechanism is that c-jun is upregulating the expression of genes encoding coactivator proteins necessary for the LXR activation of apoE. Recently a model for the farnesoid X receptor (FXR) regulation of the CYP7A1 gene has been proposed, which also involves the JNK/SAPK pathway (Gupta *et al.*, 2001). In this model, bile acids activate JNK/SAPK and c-jun, in addition to FXR. Both FXR and c-jun bind through different promoter elements, IR-1 and AP-1 respectively, and enhance the transcription of the SHP-1 corepressor, which in turn represses *CYP7A1* transcription. This model could potentially be applied to the LXR regulation of apoE, with the JNK/SAPK pathway inducing a coactivator protein. This mechanism may also explain the delay between the apoE induction at 24h (Chapter 4) and the induction of c-jun at 30min.

Therefore, in summary, the results of this chapter have shown conclusively that c-jun is being phosphorylated by the JNK/SAPK pathway in response to ligand treatment and that inhibition of this phosphorylation of c-jun, eliminates the induction of apoE gene expression by 22(R)-hydroxycholesterol. We have also shown that this induction of the JNK/SAPK pathway occurs after only 30min of treatment, which may be due to a non-genomic action of the ligand. Although this rapid induction of the c-jun transcription factor may have one of several roles in the LXR-mediated regulation of apoE (Figure 5.12), it is clear that the overall function of this pathway is to facilitate the LXR-mediated transcription of apoE.

# <u>CHAPTER SIX- ROLE OF PI3KINASE AND CK2</u> <u>SIGNALLING IN THE LXR-MEDIATED</u> <u>REGULATION OF APOE</u>

#### 6.1. INTRODUCTION

In addition to the JNK pathway, the data presented in Chapter 4 clearly demonstrates a role for the PI3K and CK2 pathways in the LXR-mediated regulation of apoE. As with the JNK/SAPK pathway neither of these signalling pathways has been described previously in this thesis, therefore a brief description of the main components of these pathways is given below.

Phosphoinositide 3-kinases (PI3Ks) are a large family of intracellular signal transducers that catalyse the formation of specific inositol lipid mediators in response to external stimuli (Auger *et al.*, 1989; Rameh and Cantley, 1999). PI3Ks phosphorylate the 3'-OH position of the inositol ring in inositol phospholipids, generating 3'-phosphoinositides and *in vivo* PI3Ks produce three lipid products namely PtdIns(3)P, PtdIns(3,4)P<sub>2</sub>, and PtdIns(3,4,5)P<sub>3</sub>, all of which are able to initiate signalling cascades (Toker, 2000). Through the initiation of signalling pathways PI3Ks play a central role in numerous cellular processes including cell growth, differentiation, motility, insulin action, and cell survival to name a few (Toker, 2000). Furthermore, PI3Ks control more specialised enzyme systems such as NADPH oxidase in leukocytes and nitric oxide synthase in endothelial cells (Wymann *et al.*, 2003). Thus, PI3Ks modulate several physiological processes and are involved in the pathology of various diseases including inflammatory and cardiovascular disease (Cantley, 2002).

Members of the PI3K family are divided into three classes according to their structure and *in vivo* substrate specificities and the predominant form of PI3K is a member of the class I subfamily of PI3K enzymes, and comprises of the p85 $\alpha$  regulatory subunit, and the p110 catalytic subunit (Carpenter and Cantley, 1990). Class I PI3Ks can be activated by either receptor tyrosine kinase (RTK) or G-protein-coupled receptors (GPCR) and once activated these PI3Ks generate PtdIns(3,4,5)P<sub>3</sub> leading to the recruitment and activation of protein kinases that contain specialised lipid-binding domains such as the pleckstrin homology (PH) domain. Conversely, PI3K production of phosphoinositides is antagonised by the protein phosphatase and tensin homologue deleted on chromosome 10 (PTEN) tumour suppressor. PTEN possesses 3'-phosphoinositide phosphatase activity, and hydrolyses PtdIns(3,4,5)P<sub>3</sub> to PtdIns(4,5)P<sub>2</sub> *in vitro* and *in vivo*, which results in down-regulation of PI3K signalling pathways (Maehama and Dixon, 1998; Myers *et al.*, 1998; Stambolic *et al.*, 1998; Leslie and Downes, 2002).

One of the best characterised targets of the PI3K lipid products is Akt, a 60kDa Ser/Thr kinase with a PH domain that preferentially binds PtdIns(3,4)P<sub>2</sub>, and PtdIns(3,4,5)P<sub>3</sub> over other PIs (James *et al.*, 1996; Stephens *et al.*, 1998). In unstimulated cells Akt resides in the cytosol, in a low-activity conformation and upon PI3K stimulation, becomes activated through recruitment to the cellular membranes by PI3K generated lipid products and subsequent phosphorylation by the Ser/Thr kinase, PDK1 (Vanhaesebroeck and Alessi, 2000). Although the exact mechanism of Akt

#### CHAPTER SIX - ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE

activation is still not completely known, the widely accepted model is that the interaction of the PH domain of Akt with the PI3K generated lipids at the inner leaflet of the plasma membrane alters the conformation of this protein kinase allowing PDK1 access to previously hidden residues, which become phosphorylated, resulting in Akt activation (Vanhaesebroeck and Alessi, 2000). Once activated Akt translocates through the cytosol to the nucleus (Figure 6.1), where it is able to phosphorylate a range of downstream targets, the majority of which have been implicated in insulin signalling or in cell survival pathways, including glycogen synthase kinase-3β (GSK-3β), mammalian target of rapamycin (mTOR), the forkhead in rhabdomyosarcoma (FKHR) transcription factor, endothelial nitric oxide synthase (eNOS) and several anti-apoptotic effectors (Rameh and Cantley, 1999; Wurmser *et al.*, 1999; Ellson *et al.*, 2002).



#### FIGURE 6.1. SCHEMATIC REPRESENTATION OF THE PI3K/AKT PATHWAY

*P*, phosphate residue; PIP2, phosphatidylinositol(4,5)*P*<sub>2</sub>; PIP3, phosphatidylinositol(3,4,5)*P*<sub>3</sub>; PI3K, phosphatidylinositol-3-kinase; PDK1, phosphoinositide dependent kinase-1; PTEN, protein phosphate and tensin homologue deleted on chromosome 10; GSK, glycogen synthase kinase; BAD, Bcl-2/BclL-antagonist, causing cell death; FKHR, forkhead in rhabdomyosarcoma; mTOR, mammalian target of rapamycin; eNOS, endothelial nitric oxide synthase; R, receptor. See text for further details.

Protein kinase CK2 is an extremely conserved pleiotropic Ser/Thr protein kinase that is ubiquitously distributed in eukaryotic organisms. CK2 has many functions and appears to be implicated in a wide variety of cellular functions such as control of metabolism, gene expression, protein synthesis, cell division and proliferation (Allende and Allende, 1995; Pinna and Meggio, 1997). CK2 usually exists in tetrameric complexes consisting of two catalytic subunits ( $\alpha \alpha$ ,  $\alpha' \alpha'$ , or  $\alpha \alpha'$ ), which are essential for cell viability (Padmanabha *et al.*, 1990), and a dimer of two non-catalytic  $\beta$  subunits that play an important regulatory role as they are responsible for increasing enzyme activity, determining substrate specificity and stabilising the holoenzyme (Faust and Montenarh, 2000; Lebrin *et al.*, 2001). In addition several studies have also demonstrated that CK2 $\alpha$  may also exist free of CK2 $\beta$  as the  $\beta$  subunits can undergo reversible dissociation under physiological conditions (Litchfield *et al.*, 2003).

The mechanism of regulation of CK2 is poorly understood, although it has been postulated that this enzyme is constitutively active (Niefind *et al.*, 1998) and independent of either second messengers or phosphorylation events (Meggio and Pinna, 2003). However, several recent studies have indicated that this may not be the case as CK2 activity appears to be regulated by variety of signalling molecules (Solyakov *et al.*, 2004; Hilgard *et al.*, 2004; Cavin *et al.*, 2003). Some of the mechanisms that may contribute to the regulation of CK2 in cells include regulated expression and assembly, regulation by covalent modification, and regulatory interactions with protein and /or non-protein molecules (Litchfield *et al.*, 2003).

Although the upstream components of CK2 are still unclear, downstream, CK2 has a growing list of more than 300 substrates, all of which share typical phospho-acceptor sites specified by multiple acidic residues located mainly downstream of the phosphor-acceptor residue (Meggio and Pinna, 2003). These substrates are involved in a wide variety of cellular functions including the regulation of transcription, signal transduction and other nuclear functions (Meggio and Pinna, 2003).

Therefore in light of the results in Chapter 4 and the diverse array of signalling components of these pathways, exactly how these signalling molecules are involved in the LXR regulation of apoE forms the focus of this chapter. The aim is to confirm the activation of these pathways by ligand treatment and to attempt to delineate a potential mechanism for the involvement of these signalling cascades in the LXR-mediated regulation of apoE.

#### 6.2. RESULTS

# **6.2.1.** EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT SER473 PHOSPHORYLATION IN THP-1 MACROPHAGES

As mentioned earlier in this chapter the protein kinase Akt is a well-established target of PI3K activation. Akt plays an important role in cell survival, whereby growth factors and matrix-contact signals utilise the PI3K/Akt pathway to counteract apoptosis (Downward, 1998; Wymann *et al.*, 2003). In addition, Akt appears to be common to signalling pathways that mediate the metabolic effects of insulin in several physiologically important target tissues (Whiteman *et al.*, 2002).

Stimulation of PI3K by receptor tyrosine kinases results in Akt becoming activated through recruitment to the cellular membrane by PI3K generated lipid products and subsequent phosphorylation by PDK1 (Vanhaesbroeck and Alessi, 2000). Two specific sites, one in the kinase domain (Thr308) and the other in the C-terminal regulatory region (Ser473), need to be phosphorylated for full Akt activation (Alessi *et al.*, 1996) and once activated this protein kinase detaches from the membrane and translocates through the cytoplasm to the nucleus (Andjelkovic *et al.*, 1997; Meier *et al.*, 1997).

In order to confirm activation of the PI3K pathway by 22(R)-hydroxycholesterol, it was decided to investigate whether Akt becomes phosphorylated in response to ligand treatment. Akt phosphorylation is an effective marker for PI3K activity and several studies have used this approach in investigating the PI3K pathway (Qiu *et al.*, 2004; Jaiswal *et al.*, 2004; Hassan *et al.*, 2004). Whole cell extracts from differentiated THP-1 cells treated with 22(R)-hydroxycholesterol or DMSO for various time-points, were isolated using the phosphatase-free isolation buffer as described in Section 2.5.1, and western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) was carried out using an anti-phospho-Akt antibody that was specific for Akt phosphorylation at Ser473.

From Figure 6.2a, it can be seen that ligand treatment induced a two-fold induction of Akt phosphorylation at the Ser473 residue after 1h of treatment. This induction in Akt phosphorylation was not due to *de novo* protein synthesis as western blot analysis using an antibody that detects endogenous levels of total Akt protein showed similar steady-state protein levels (Figure 6.2b). Also this induction was specific to 22(R)-hydroxycholesterol as the following studies presented in this chapter will demonstrate that treatment with 22(S)-hydroxycholesterol showed no change in Akt phosphorylation (Section 6.2.2).

CHAPTER SIX - ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE



FIGURE 6.2. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] for the indicated time-points. As a control at each time-point, cells were treated with DMSO in the place of ligand. Western blot analysis was carried out using 80µg of whole cell extracts. A Blotted membranes were incubated with phospho-Akt (Ser473) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. B Blotted membranes were incubated with total-Akt primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody complexes were detected using the ECL detection system. C The Akt phospho-protein and total Akt level for each sample was determined by densitometric analysis and the Akt:total protein for each sample has been represented relative to its control, which has been assigned as 1. Results are representative of two separate experiments and the data presented in the form of a histogram is the mean value from these experiments.

# 6.2.2. EFFECT OF LY294002 and Apigenin on the LXR-mediated phosphorylation of Akt Ser473 in THP-1 macrophages

In order to confirm that the inhibition of the 22(R)-hydroxycholesterol-mediated induction of apoE by LY294002 was due to an inhibition of the PI3K pathway, we next investigated the effect of LY294002 on Akt phosphorylation. Whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) from differentiated THP-1 cells that were pre-treated with inhibitors for 1h before the addition of 22(R)-hydroxycholesterol for 1h. Total Akt proteins were immunoprecipitated from whole cell lysates (Section 2.5.3) and were subjected to western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) using the anti-phospho-Akt (ser473) antibody with overnight incubation.

As shown in Figure 6.3, LY294002 inhibits the ligand-mediated induction of Akt Ser473 phosphorylation at 1h, which demonstrates that the activation of this pathway is necessary for the LXR-mediated regulation of apoE. Also, from this experiment it can be seen that the increase in Akt phosphorylation is mediated by LXR activation as the control ligand, 22(S)-hydroxycholesterol had no effect on Akt phosphorylation. In addition to LY294002, cells were pre-treated with the CK2 inhibitor apigenin. It was decided to include this inhibitor as previous data (see Chapter 4) has shown that the apoE induction caused by 22(R)-hydroxycholesterol can be prevented by pre-treatment with apigenin, and it is therefore important to establish whether this inhibition is independent of the PI3K pathway. In addition, several studies have demonstrated that apigenin can directly inhibit PI3K activity and Akt kinase activity in various cellular systems (Way *et al.*, 2004; Liorens *et al.*, 2002; Agullo *et al.*, 1997). From Figure 6.3, it is clear that apigenin is having an inhibitory effect on the ligand-mediated phosphorylation of Akt. Whether this inhibition is due apigenin directly affecting Akt, or whether CK2 converges with the PI3K pathway upstream of Akt and therefore apigenin is indirectly affecting Akt through CK2, is unclear from these studies and requires further investigation.

CHAPTER SIX – ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE



FIGURE 6.3. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED PHOSPHORYLATION OF AKT IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] and 22(S)-hydroxycholesterol (denoted 22S) [2µg/ml] for 1h. In addition cells were treated with vehicle, DMSO in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: LY294002 (25µM), apigenin (40µM). Immunoprecipitation of whole cell extracts (250µg) was carried out using a total-Akt antibody as described in Section 2.5.3 and the isolated Akt proteins were subjected to western blot analysis. Blotted membranes were incubated with a phospho-Akt (Ser473) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of three separate experiments.

### 6.2.3. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT THR308 PHOSPHORYLATION IN THP-1 MACROPHAGES

In addition to Ser473 phosphorylation, full activation of Akt also requires the phosphorylation of a threonine residue located in the activation loop of Akt. This phosphorylation is thought to be necessary for substrate recognition and for the transfer of a phosphate group (Huse and Kuriyan, 2002). In order to fully investigate the mechanism whereby 22(R)-hydroxycholesterol activates the PI3K pathway, the levels of Akt phosphorylation at Thr308 were also assessed in response to ligand treatment.

Whole cell extracts from differentiated THP-1 cells treated with 22(R)-hydroxycholesterol for various time-points were isolated using the phosphatase-free isolation buffer as described in Section 2.5.1, and western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) was carried out using an anti-phospho-Akt (Thr308) antibody with overnight incubation. From Figure 6.4, it is clear that a basal level of phosphorylation of Akt at Thr308 exists in macrophages. However, no additional phosphorylation of Akt at this residue in response to 22(R)-hydroxycholesterol could be seen.



#### FIGURE 6.4. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT THR308 PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] for the indicated time-points. As a control at each time-point, cells were also treated with DMSO in the place of ligand. Western blot analysis was carried out using 80µg of whole cell extracts. Blotted membranes were incubated with phospho-Akt (Thr308) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of two separate experiments.

#### 6.2.4. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT ACTIVITY IN THP-1 MACROPHAGES

Athough we have shown that ligand treatment of THP-1 macrophages results in the phosphorylation of Akt Ser473 (Figure 6.2), the inability of 22(R)-hydroxycholesterol to increase Thr308 phosphorylation indicates that the kinase activity of Akt may not be activated by ligand treatment. Therefore, it was necessary to investigate the effect of 22(R)-hydroxycholesterol on Akt kinase activity. For these experiments, a non-radioactive method of measuring Akt activity was employed based on the Akt Assay Kit (NEB), whereby the ability of Akt to phosphorylate one of its many downstream targets was measured. Differentiated THP-1 cells were treated with 22(R)-hydroxycholesterol for 1h and whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1). The kinase assay was carried out as described in Section 2.6.2, whereby an immobilized Akt Monoclonal Antibody was used to immunoprecipitate Akt from whole cell extracts. Then, an *in vitro* kinase assay was performed using GSK-3 Fusion Protein as a substrate and phosphorylation of GSK-3 was measured by western blot analysis (Section 2.5.5 and 2.5.6) using a phospho-GSK-3 $\alpha$ / $\beta$ (Ser21/9) antibody. From Figure 6.5, it is clear that ligand treatment of differentiated THP-1 cells does in fact result in an increase in Akt kinase activity, which

#### CHAPTER SIX – ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE

demonstrates that although no induced phosphorylation occurred at Thr308, the ligand-mediated phosphorylation of Ser473 is sufficient for an increase in Akt activity in this LXR-mediated mechanism.



**FIGURE 6.5.** EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON AKT ACTIVITY IN THP-1 MACROPHAGES THP-1 monocytes were differentiated with PMA (0.16 $\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22(R)-hydroxycholesterol (denoted 22R) [2 $\mu$ g/ml] for 1h. In addition, cells were treated with vehicle, DMSO in the place of ligand (denoted control). The kinase assay was carried out using 250 $\mu$ g of whole cell extracts as described in Section 2.6.2 and blotted membranes were incubated with phospho-GSK3 $\alpha/\beta$  (Ser21/9) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigenantibody complexes were detected using the ECL detection system. Results are representative of three separate experiments.

### 6.2.5. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED INDUCTION OF AKT ACTIVITY IN THP-1 MACROPHAGES

The effect of the LY294002 and apigenin inhibitors on the ligand-mediated induction of Akt kinase activity was also investigated. Whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) from differentiated THP-1 cells that were pre-treated with inhibitors for 1h, before the addition of 22(R)-hydroxycholesterol for 1h. The kinase assay was carried out as described in Section 2.6.2 and phosphorylation of GSK-3 was measured by western blot analysis (Section 2.5.5 and 2.5.6) using a phospho-GSK-3 $\alpha$ / $\beta$ (Ser21/9) antibody. As shown in Figure 6.6, both inhibitors eliminated the 22(R)-hydroxycholesterol-mediated induction of Akt kinase activity. This data is consistent with the previous result, whereby LY294002 and apigenin inhibited the ligand-induced phosphorylation of Akt Ser473 (Figure 6.3) and demonstrates that although no induced phosphorylation of Akt occurred at Thr308, the ligand-mediated phosphorylation of Ser473 is sufficient for an increase in Akt activity in this LXR-mediated mechanism and that inhibition of this Ser473 phosphorylation by LY294002 and apigenin results in the inhibition of Akt activity.

CHAPTER SIX – ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE



FIGURE 6.6. EFFECT OF LY294002 AND APIGENIN ON THE LXR-MEDIATED INDUCTION IN AKT ACTIVITY IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] and 22(S)-hydroxycholesterol (denoted 22S) [2µg/ml] for 1h. In addition, cells were treated with vehicle, DMSO in the place of ligand (denoted control). Inhibitors were added 1h before the ligand at various concentrations: LY294002 (25µM), apigenin (40µM). The kinase assay was carried out using 250µg of whole cell extracts as described in Section 2.6.2 and blotted membranes were incubated with phospho-GSK-3α/β (Ser21/9) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. The GSK-3 phospho-protein level for each sample was determined by densitometric analysis and plotted on a bar chart. The control represents macrophages treated with vehicle (DMSO) in the place of ligand and has been assigned as 1, with the phospho-protein level for the ligand-treated sample being represented relative to this control. The data shown is the mean ±SD from three independent experiments.

#### 6.2.6. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON PDK1 PHOSPHORYLATION IN THP-1 MACROPHAGES

Akt phosphorylation at Thr308 requires the protein kinase PDK1. PDK1, like Akt possesses a PH domain that can bind to PI(3,4)P<sub>2</sub> or PI(3,4,5)P<sub>3</sub> and requires translocation to the plasma membrane for full activation (Andjelkovic *et al.*, 1997; Anderson *et al.*, 1998). PDK1 exists in an active, phosphorylated configuration under basal conditions and appears to be susceptible to additional activation and phosphorylation upon cell stimulation with agonists which activate PI3K (Alessi *et al.*, 1997; Casamayor *et al.*, 1999). Recent work has also suggested that PDK1, in complex with other proteins, may also mediate the phosphorylation Akt on Ser473 (Balendran *et* 

#### CHAPTER SIX - ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE

*al.*, 1999). Although the main requirement for PDK1 activation is PI(3,4,5)P<sub>3</sub> binding, phosphorylation of this kinase is also equally critical. PDK-1 is phosphorylated at Ser241, which is present in the activation loop, and this phosphorylation occurs through autophosphorylation rather than by an upstream kinase (Casamayor *et al.*, 1999). Several other phosphorylation sites have been identified, although phosphorylation at Ser241 is necessary for PDK1 activity (Casamayor *et al.*, 1999).

Therefore, as PDK1 is the kinase responsible for Akt activation it was next decided to investigate whether this kinase becomes phosphorylated in response to ligand treatment. Whole cell extracts from differentiated THP-1 cells treated with 22(R)-hydroxycholesterol for various time-points were isolated using the phosphatase-free isolation buffer as described in Section 2.5.1, and western blot analysis (Section 2.5.4, 2.5.5 and 2.5.6) was carried out using an anti-phospho-PDK1 (Ser241) antibody with overnight incubation. From Figure 6.7 it can be seen that similar to the result seen for Akt Thr308 phosphorylation, a basal level of PDK-1 phosphorylation exists, but no additional phosphorylation of this protein kinase occurs in response to ligand treatment.



#### FIGURE 6.7. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON PDK1 PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] for the indicated time-points. As a control at each time-point, cells were also treated with DMSO in the place of ligand. Western blot analysis was carried out using 80µg of whole cell extracts. Blotted membranes were incubated with phospho-PDK1 (ser241) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of two separate experiments.

**6.2.7.** EFFECT OF **22(R)**-HYDROXYCHOLESTEROL ON GSK-3 PHOSPHORYLATION IN THP-1 MACROPHAGES The data presented in this chapter has so far demonstrated that the PI3K-Akt pathway is activated by the LXR ligand 22(R)-hydroxycholesterol and that the activation of this pathway is necessary for the LXR-mediated induction of apoE gene expression. In order to identify exactly what role this pathway may play in the apoE induction it is necessary to identify any downstream targets that may also be activated by the oxysterol ligand.

One of the first identified and major substrates of Akt is glycogen synthase kinase-3 (GSK-3) [Burgering and Coffer, 1995]. GSK-3 belongs to a family of conserved Ser/Thr kinases present in all eukaryotic organisms and one of its primary functions is to phosphorylate and inactivate glycogen synthase in response to insulin stimulation (Cross *et al.*, 1995). Two GSK-3 isoforms exist in humans (GSK3 $\alpha$  and  $\beta$ ) and both have a phosphorylation site in the amino-terminal region (Ser21 and Ser9, respectively), where they are phosphorylated and inactivated by Akt in a PI3K-dependent manner. GSK-3 $\beta$  is a ubiquitously expressed, constitutively active serine/threonine kinase that phosphorylates cellular substrates and negatively regulates downstream signalling mechanisms. Akt phosphorylation of GSK-3 $\beta$  results in the inactivation of this kinase and stimulates many cellular functions by removing the negative constraint imposed by GSK-3 $\beta$ .

GSK-3 $\beta$  has also been found to play a role in nuclear receptor signalling, whereby phosphorylation of nuclear receptors such as GR and AR by GSK-3 $\beta$  reduces the transcriptional activity of the receptor (Rogatsky *et al.*, 1998; Salas *et al.*, 2004; Liao *et al.*, 2004). Akt phosphorylation of GSk-3 $\beta$  therefore results in the activation of these nuclear receptors by inhibiting GSK-3 $\beta$ . It was therefore decided to investigate the phosphorylation state of this protein kinase as GSK-3 $\beta$  is a potential candidate that may link the PI3K pathway with LXR transcriptional regulation of genes.

For these experiments, THP-1 cells were differentiated and treated with 22(R)hydroxycholesterol for various time-points. Whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) and western blot analysis was carried out as described in Section 2.5.5 and 2.5.6, using an anti-phospho-GSK-3 $\beta$  (Ser9) antibody with overnight incubation. Figure 6.8 shows clearly that no phosphorylation of GSK-3 $\beta$  occurs in response to oxysterol treatment and rules out the involvement of this kinase in the LXR regulation of apoE. CHAPTER SIX - ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE



#### FIGURE 6.8. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON GSK-3 $\beta$ PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] for the indicated time-points. As a control at each time-point, cells were also treated with DMSO in the place of ligand. Western blot analysis was carried out using 80µg of whole cell extracts. Blotted membranes were incubated with phospho-GSK-3 $\beta$  (Ser9) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of two separate experiments.

#### 6.2.8. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON FKHR PHOSPHORYLATION IN THP-1 MACROPHAGES

Another downstream target of the PI3K-Akt pathway is the FKHR transcription factor, which is responsible for the regulation of pro-apoptotic genes (Brunet *et al.*, 1999; Rena *et al.*, 1999). Akt phosphorylates FKHR at several sites and represses its transactivation function by promoting redistribution of the transcription factor from the nucleus to the cytoplasm (Brunet *et al.*, 1999; Medema *et al.*, 2000). In addition to its original function as a DNA binding transcription factor, several studies have demonstrated a role for FKHR as a nuclear receptor intermediary protein (Zhao *et al.*, 2001; Hirota *et al.*, 2003). FKHR has been shown to interact directly with several nuclear receptors, which enables this transcription factor to affect the transactivation function of these receptors (Zhao *et al.*, 2001; Schuur *et al.*, 2001; Li *et al.*, 2003; Dowell *et al.*, 2003). FKHR has also been shown to be phosphorylated in the non-genomic function of the AR receptor (Baron *et al.*, 2004). Therefore the phosphorylation state of this transcription factor was investigated in response to ligand treatment.

THP-1 cells were differentiated and treated with 22(R)-hydroxycholesterol for different time-points. Whole cell extracts were isolated using the phosphatase-free isolation buffer (Section 2.5.1) and western blot analysis was carried (Section 2.5.5 and 2.5.6) using an anti-phospho-FKHR
#### CHAPTER SIX – ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE

(Ser256) antibody with overnight incubation. Two separate experiments demonstrated that no phosphorylation of FKHR occurs in response to oxysterol treatment (Figure 6.9).



# FIGURE 6.9. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON FKHR PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22(R)-hydroxycholesterol (denoted 22R) [2µg/ml] for the indicated time-points. As a control at each time-point, cells were also treated with DMSO in the place of ligand. Western blot analysis was carried out using 80µg of whole cell extracts. Blotted membranes were incubated with phospho-FKHR (Ser256) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of two separate experiments.

# 6.2.9. EFFECT OF CK2 DOMINANT NEGATIVE EXPRESSION PLASMIDS ON THE LXR-MEDIATED INDUCTION OF

#### APOE IN THP-1 MACROPHAGES

The results form Chapter 4 demonstrated that apigenin is able to repress the effect of the LXR ligand, 22(R)-hydroxycholesterol, on apoE gene expression. This effect was originally thought to occur through the inhibition of the CK2 enzyme as apigenin has been well-established as an inhibitor of this protein kinase and has extensively been used in CK2 studies (Critchfield *et al.*, 1997; Ford *et al.*, 2000; Tsai and Seto, 2002; Mead *et al.*, 2003). Unfortunately, the data in Section 6.2.2 demonstrates that apigenin is affecting the PI3K pathway, which indicates that CK2 may not be involved in the LXR-mediated induction of apoE. Therefore, in order to confirm the involvement of CK2, it was decided to use a dominant negative expression plasmid specific for the CK2 enzyme. The pSG- $\alpha$  (K68A) plasmid encoding the kinase-inactive mutant of CK2 $\alpha$  catalytic subunit (with the substitution Lys68 to Ala) was a kind gift from Dr E. M. Chambaz and Dr C. Cochet (INSERM, Grenoble, France).

Transfections were carried out in THP-1 monocytes and the DNA constructs were transfected using the Effectene<sup>TM</sup> transfection reagent (Qiagen), as described in Section 2.3.5.2. PMA (0.16 $\mu$ M) was added immediately after the transfection of cells was complete and the cells were left to differentiate for 24h. After the 24h incubation, cells were treated with 2 $\mu$ g/ml of 22(R)-hydroxycholesterol or 22(S)-hydroxycholesterol for a further 24h, after which whole cell protein extracts were prepared (see Section 2.5.1). Of each sample 20 $\mu$ g of total protein was subjected to SDS-PAGE (Section 2.5.4) and then transferred to a nitrocellulose membrane (Section 2.5.5). The membrane was then incubated with an anti-apoE primary antibody (Biogenesis) and immunodetection was carried out using the appropriate secondary antibody and the ECL detection system (Amersham) as described in Section 2.5.6. In addition to the dominant negative construct, as a control, cells were transfected with the empty pSG5- $\alpha$  vector (Appendix IV) to ensure that response seen was due to the dominant negative CK2 and not the plasmid vector or the transfection reagent. From Figure 6.10, it can be seen that the CK2 dominant negative plasmid caused a 50% decrease in the induction of apoE by 22(R)-hydroxycholesterol, which confirms that CK2 is indeed involved in the LXR-mediated induction of apoE.



## FIGURE 6.10. EFFECT OF DOMINANT NEGATIVE EXPRESSION PLASMIDS ON THE LXR-MEDIATED INDUCTION OF APOE IN THP-1 MACROPHAGES

THP-1 monocytes were transfected (see section 2.3.5.2) with the indicated expression plasmids or the control vector and differentiated with PMA (0.16 $\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h. The LXR ligands, 22(R)-hydroxycholesterol (denoted 22R) [2 $\mu$ g/ml] and 22(S)-hydroxycholesterol (denoted 22S) [2 $\mu$ g/ml] were then added to the cells for a further 24h. Western blot analysis was carried out using 20 $\mu$ g of whole cell extracts. Blotted membranes were incubated with anti-apoE primary antibody (goat polyclonal IgG) and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of three separate experiments. The apoE protein level for each sample was determined by densitometric analysis and plotted on a bar chart. The control represents macrophages treated with vehicle (DMSO) in the place of ligand and has been assigned as 1 in each case, with the apoE protein level for the remaining samples being represented relative to this control. The data shown is the mean ±SD from three independent experiments. The data was analysed by Student's t-test. \*\* Represents the significant difference compared to pSG5 $\alpha$ transfected cells treated with 22(R)-hydroxycholesterol (\*\*p<0.005).

#### 6.2.10. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON CK2 ACTIVITY IN THP-1 MACROPHAGES

To our knowledge, no studies have been published with respect to the 22(R)-hydroxycholesterol ligand and increased CK2 activity. Therefore, it was important to prove that the two are linked, which was carried out by performing an *in vitro* assay for CK2 activity. For these experiments, THP-1 cells were differentiated for 24h with PMA and then treated with 22(R)-hydroxycholesterol for 30 min in the absence or presence of apigenin. This time-point was chosen as a sample time-point based on previous studies in our laboratory on CK2 activity. After preparation of whole cell extracts using the phosphatase-free isolation buffer (Section 2.5.1), CK2 was immunoprecipitated using an antibody against the  $\alpha$ -subunit and subjected to an assay where its ability to phosphorylate  $\beta$ -casein was measured. This assay utilised  $\gamma$ -<sup>32</sup>P-labelled ATP, which CK2 uses *in vivo* as a phosphate donor (although it may also use GTP), in the substrate mix. After allowing the reaction to proceed for 15 min at 37°C, the contents were sized-fractionated on a SDS-PAGE gel, and the phosphorylation of  $\beta$ -casein, a 24kDa protein, by the immunoprecipitated CK2 was assessed by autoradiography. The results of this experiment are shown in Figure 6.11.

From Figure 6.11a, it can be seen that β-casein was phosphorylated to a greater extent by CK2 immunoprecipitated from 22(R)-hydroxycholesterol-treated cells, than by CK2 isolated from control cells and that this activation was prevented by pre-treatment with the CK2 inhibitor, apigenin. These results, therefore demonstrates the ability of 22(R)-hydroxycholesterol to activate CK2 in THP-1 cells, and clearly indicates that CK2 activation is essential in the LXR-mediated regulation of apoE. The results in Figure 6.11b also show that the increase in CK2 activity is LXR mediated, as the inactive ligand 22(S)-hydroxycholesterol had no effect on CK2 β-casein phosphorylation. The LY294002 inhibitor was also used in these experiments to ensure that CK2 activation was independent of the PI3K pathway. This was confirmed as LY294002 had no effect of CK2 activity (Figure 6.11b). Also included in Figure 6.11a are positive and negative controls. The positive control reaction was performed using purified CK2 enzyme, and the negative control was carried out with substrate alone. The identical signals produced by the purified enzyme and immunoprecipitated CK2 confirm that the assay is specific for CK2, and the lack of any signal in the negative control shows that the signals obtained in experimental lanes are not spuriously created by  $^{32}$ P-labelled ATP in the substrate mix.

The possibility existed that the increased activity of CK2 in 22(R)-hydroxycholesteroltreated cells was the result of *de novo* protein synthesis. This was discounted by the observation that western blot analysis with whole cell lysates obtained from control and ligand treated cells all showed similar steady-state protein levels (Figure 6.11c). CHAPTER SIX – ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE

A

В

С



# FIGURE 6.11. EFFECT OF 22(R)-HYDROXYCHOLESTEROL AND THE INHIBITORS APIGENIN AND LY294002 ON CK2 ACTIVITY AND PROTEIN EXPRESSION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA (0.16µM) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligands, 22(R)-hydroxycholesterol (denoted 22R){2µq/ml] and 22(S)hydroxycholesterol (denoted 22S) [2µg/ml] for 30min. As a control, cells were also treated with DMSO in the place of ligand. Inhibitors were added 1h before the ligand at various concentrations: LY294002 (25 $\mu$ M), apigenin (40 $\mu$ M). A and B The kinase assay was carried out using 200 $\mu$ g of whole cell extracts as described in Section 2.6.1. The contents of the kinase reaction were sizedfractionated on a SDS-PAGE gel [15% (v/v)], and the phosphorylation of  $\beta$ -casein, by the immunoprecipitated CK2 was assessed by autoradiography. Also shown are a positive control with purified CK2 enzyme (1U; Promega) and a negative control where only the substrate mix was size fractionated on the gel. The figure is representative of a gel that was fixed, dried and subsequently exposed to film for 1h. c Immunoprecipitation of whole cell extracts  $(250\mu q)$  was carried out using an anti-CK2 $\alpha$  antibody (goat polyclonal IgG) as described in Section 2.5.3 and the isolated CK2 $\alpha$ proteins were subjected to western blot analysis. Blotted membranes were incubated with the  $CK2\alpha$  primary antibody and detected using an anti-goat horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of three separate experiments.

#### CHAPTER SIX – ROLE OF PI3KINASE AND CK2 SIGNALLING IN THE LXR-MEDIATED REGULATION OF APOE

**6.2.11.** EFFECT OF **22(R)**-HYDROXYCHOLESTEROL ON PTENPHOSPHORYLATION IN THP-1 MACROPHAGES The results from the previous sections confirm that CK2 is in fact being activated by 22(R)hydroxycholesterol. Although the CK2 inhibitor, apigenin has been shown in other studies to affect the PI3K pathway and was shown to be inhibiting Akt phosphorylation in Figure 6.3, it may be that this inhibitor is affecting the PI3K pathway indirectly through CK2 inhibition. Several studies have shown that PTEN, a 3'-phosphoinositide phosphatase that antagonises PI3K signalling, is a substrate of CK2 (Miller *et al.*, 2002; Torres and Pulido, 2001). CK2 phosphorylation of PTEN is inhibitory, and therefore any inhibition of CK2 activity leads to a decrease in Akt activation (Miller *et al.*, 2002; Torres and Pulido, 2001). Therefore we next investigated whether CK2 activation by 22(R)-hydroxycholesterol resulted in the phosphorylation of PTEN.

Whole cell extracts from differentiated THP-1 cells treated with 22(R)-hydroxycholesterol for various time-points were isolated using the phosphatase-free isolation buffer as described in Section 2.5.1, and western blot analysis (Section 2.5.5 and 2.5.6) was carried out using an anti-phospho-PTEN (Ser380) antibody with overnight incubation. From Figure 6.12, it is clear that a basal level of PTEN phosphorylation exists in macrophages, but no additional phosphorylation occurs in response to ligand treatment, which indicates that CK2 is not affecting the PI3K pathway through PTEN phosphorylation.



## FIGURE 6.12. EFFECT OF 22(R)-HYDROXYCHOLESTEROL ON PTEN PHOSPHORYLATION IN THP-1 MACROPHAGES

THP-1 monocytes were differentiated with PMA ( $0.16\mu$ M) in 10% (v/v) delipidated HI-FCS for 24h and then exposed to the LXR ligand, 22R-hydroxycholesterol (denoted 22R) [2µg/ml] for the indicated time-points. As a control at each time-point, cells were also treated with DMSO in the place of ligand. Western blot analysis was carried out using 80µg of whole cell extracts. Blotted membranes were incubated with phospho-PTEN (Ser380) primary antibody (rabbit polyclonal IgG) and detected using an anti-rabbit horseradish peroxidase secondary antibody. Antigen-antibody complexes were detected using the ECL detection system. Results are representative of two separate experiments.

#### 6.3. Discussion

As no studies have been published regarding the involvement of PI3K and CK2 in LXR regulation of target genes, the initial aim of this chapter was to confirm the activation of these pathways in response to ligand treatment.

Firstly, we investigated PI3K activation by 22(R)-hydroxycholesterol. The PI3K pathway has been shown to play a vital role in nuclear receptor signalling with various components of the pathway being involved in regulating the transactivation functions of these transcription factors. The primary role for this pathway seems to be in the phosphorylation of nuclear receptors with receptors such as the androgen receptor (AR) and the estrogen receptor (ER) being phosphorylated in response to a variety of signals (Lin et al., 2001b; Campbell et al., 2001). In order to investigate the role of this pathway in the LXR-mediated regulation of apoE it was necessary to firstly identify whether ligand treatment increased PI3K activity in THP-1 macrophages. Therefore it was decided to measure phosphorylation of Akt, a well-established downstream signalling component of PI3K. From the results presented in this chapter it is clear that ligand treatment of THP-1 macrophages resulted in Akt phosphorylation at Ser473 (Figure 6.2) and kinase activation (Figure 6.5) after only 1h. This induction of Akt was ligand-specific and required LXR activation as 22(S)-hydroxycholesterol had no effect on Akt phosphorylation or activity. Also through the use of inhibitors we are able to conclude that Akt activity is essential for the LXRmediated regulation of apoE as inhibition of this pathway by LY294002 is consistent with the inhibition of apoE with this inhibitor. We also demonstrated that the induction in Akt kinase activity did not require an increased level of Akt Thr308 phosphorylation (Figure 6.4) or an increase in PDK1 phopshorylation (Figure 6.7) although a basal level of phosphorylation of both these proteins does exist in THP-1 macrophages.

This ability of the LXR ligand, 22(R)-hydroxycholesterol to induce Akt activity after only 1h may be attributed to a non-genomic function of the ligand. As described in Chapter 5 nuclear receptors are able to activate cell signalling pathways through non-transcriptional mechanisms in addition to their ligand-inducible transcription factor function. These non-genomic or extranuclear actions of nuclear receptors are rapid and defy the time-course required for transcriptional activation of genes. From our results, which demonstrate that 22(R)-hydroxycholesterol is able to induce PI3K activity after 1h it is possible to conclude that PI3K activation, similar to the JNK/SAPK activation of the previous chapter occurs through a non-genomic action of the ligand, and this rapid activation of PI3K is necessary for the LXR-mediated transactivation of the apoE gene.

Although it is clear that 22(R)-hydroxycholesterol is activating the PI3K pathway through a non-genomic mechanism that does not require transcriptional activation, the precise mechanism of this action requires further investigation. One way in which cell signalling pathways have been

shown to be rapidly activated by nuclear receptors is through protein interactions. Indeed, several studies have previously shown that this is the case for the non-genomic activation of the PI3K pathway by the ER receptor. ER $\alpha$  has been shown to bind and activate the p85 regulatory subunit of PI3K which results in Akt Ser473 phosphorylation and kinase activation (Simoncini et al., 2000; Pozo-Guisado et al., 2004; Haynes et al., 2000; Hisamoto et al., 2001). This has also been shown to be the case for the non-genomic action of androgens, as the AR receptor interacts directly with the p85 regulatory subunit of PI3K and promotes the accumulation of PI3K generated lipid products, which in turn results in an increase in Akt activity (Baron et al., 2004). This mechanism of activation of the PI3K pathway could potentially be applied to an LXR model, whereby a membrane-bound LXR receptor binds, in a ligand-dependent manner to PI3K, which would result in Akt recruitment and activation of this pathway, which ultimately increases the LXR-mediated transcription of apoE. In support of the potential mechanism, whereby LXR binds to a component of the PI3K pathway, resulting in Akt activation, a scan for motifs present in the amino acid sequence of the LXRs was carried out using the Scansite database (Obenauer et al., 2003). The results of this scan indicated several PI3K pathway motifs, including PI3K p85 binding groups and PDK1 binding motifs (AppendixV). These results demonstrate that the necessary sequences for PI3K binding exist in the LXR proteins although further work is required to confirm this hypothesis.

Once Akt activation was established, it was decided to investigate whether any downstream targets of Akt were phosphorylated in response to 22(R)-hydroxycholesterol treatment. The Akt targets investigated were GSK-3 $\beta$  and FKHR, both of which are negatively regulated by phosphorylation by Akt and have been shown to be involved in nuclear receptor signalling (Rogatsky et al., 1998; Salas et al., 2004; Zhao et al., 2001; Schuur et al., 2001; Hirota et al., 2003). Although neither one of these proteins showed any change in phosphorylation in response to ligand treatment (Figure 6.8 and 6.9), the possibility that the PI3K pathway is mediating its effects on the LXR regulation of apoE through a downstream component of Akt, can not be ruled out. More than 50 downstream targets have been identified so far for Akt and several others, in addition to GSK-3B and FKHR, have been shown to play a role in nuclear receptor signalling. For example MDM2, a ubiquitin ligase for p53 (Haupt et al., 1997) has been shown to be a positive regulator of ERa in MCF-7 cells (Saji et al., 2001). Also the TSC2 protein has been shown to bind and selectively modulate transcription mediated by GR, PPAR $\alpha$  and VDR (Henry et al., 1998). Therefore, although it is possible that Akt may be the last step in the PI3K pathway involved in the LXR regulation of apoE and may be directly phosphorylating LXR or one of the coregulator proteins required for transcriptional activation, it is impossible to rule out the activation of another downstream target at this stage. In support of this potential mechanism, whereby the PI3K pathway results in phosphorylation of LXR either through Akt, or a downstream target of Akt, a scan for kinase motifs present in the amino acid sequence of these nuclear receptors was carried out using the Scansite database (Obenauer *et al.*, 2003). The results of this scan indicated several phosphorylation motifs, including Akt and GSK- $3\alpha$  phosphorylation sequences (AppendixV). The identification of these potential phosphorylation sites is of great interest and future work should aim to identify the importance of these residues in the involvement of the PI3K pathway in the LXRmediated regulation of apoE.

In addition to the PI3K pathway, the aim of this chapter was to confirm and investigate the activation of CK2. Although the work presented in Chapter 4 demonstrated conclusively that apigenin inhibited the effect of the LXR ligand, 22(R)-hydroxycholesterol, on apoE gene expression, the data in Figure 6.3 indicated that this may be due to inhibition of the PI3K pathway rather than CK2. Therefore, the involvement of CK2 in the LXR-mediated regulation of apoE was confirmed through the use of dominant negative constructs (Figure 6.10). Also a CK2 kinase assay was carried out and clearly demonstrated that ligand treatment results in CK2 activation, which is blocked by pre-treatment with apigenin (Figure 6.11). Therefore, although apigenin may also affect the PI3K pathway, the results from this chapter prove conclusively that CK2 is activated by ligand treatment. It is also possible that apigenin is inhibiting the PI3K pathway indirectly through CK2 inhibition. This may be due to phosphorylation of PTEN by CK2. Although we have shown that no change in PTEN phosphorylation at Ser380 occurs in response to ligand treatment (Figure 6.12), it does not rule out CK2 phosphorylation of this phosphatase, as recent studies have demonstrated that CK2 phophorylates PTEN at several other critical residues (Torres and Pullido, 2001).

Although the exact role of CK2 in the LXR-mediated regulation of apoE has not been identified in this study, it is possible that CK2 may play one of several roles in LXR signalling as the list of CK2 substrates is extensive and ever-growing. CK2 may directly phosphorylate the LXR receptor, as has been shown for several other nuclear receptors (Jurutka *et al.*, 1996; Zhang *et al.*, 1994; Katz *et al.*, 1995). In fact, a Scansite search of the LXR protein sequences has demonstrated the existence of a CK2 phosphorylation motif in both LXR $\alpha$  and LXR $\beta$  (Appendix V). Other substrates of CK2 that may be involved in LXR signalling include coregulators (Zhou *et al.*, 2001; Meggio and Pinna, 2003), transcription factors (Armstrong *et al.*, 1997; Meggio and Pinna, 2003), and various members of the basal transcriptional machinery (Pinna, 1990; Maldonado and Allende, 1999; Ghavdel and Schultz, 1997).

Therefore, in conclusion the studies in this chapter have conclusively demonstrated that both the PI3K pathway and CK2 are activated in response to ligand treatment. These observations are novel and may also be extended to other LXR target genes. We have also shown that these pathways are activated rapidly, suggesting once again that 22(R)-hydroxycholesterol may have a non-genomic function which supports the LXR-mediated transcription of apoE.

# **CHAPTER SEVEN – GENERAL DISCUSSION**

#### 7.1 OVERVIEW OF THE RESULTS PRESENTED IN THIS THESIS

The role of apoE as an atheroprotective protein has been well established, with multiple studies demonstrating its ability to retard the development of atherosclerosis (Schaefer *et al.*, 1986; Ghiselli *et al.*, 1981; Plump *et al.*, 1992; Zhang *et al.*, 1992; Nakashima *et al.*, 1994; Reddick *et al.*, 1994; Piedrahita *et al.*, 1992). Macrophage-derived apoE is potentially of great importance due to its direct role in preventing or delaying foam cell formation (Fazio *et al.*, 1997). In addition to its role in the regulation of macrophage cholesterol homeostasis, macrophage apoE exerts its anti-atherogenic properties through various alternative mechanisms including its ability to modify T-lymphocyte- and smooth muscle cell-mediated inflammatory responses (Avila *et al.*, 1982; Hui and Harmony, 1980a; Ishigami *et al.*, 1998). Therefore, apoE may be considered as a potential therapeutic target that merits further investigation. Indeed, therapeutic approaches that would modulate macrophage apoE production may be useful in the prevention or treatment of atherosclerosis. It is therefore essential that a detailed understanding is obtained of the regulation of macrophage apoE expression, particularly the mechanisms that are responsible for increasing production.

Unfortunately, there is a general paucity of relevant information on the regulation of macrophage apoE gene expression and, additionally, several previous studies have produced incomplete and often conflicting results (Basheeruddin *et al.*, 1994; Duan *et al.*, 1995; Zuckerman *et al.*, 1992; Menju *et al.*, 1989). Therefore, the overall objective of this thesis was to gain a better understanding of the regulation of apoE expression in macrophages.

This objective was initially approached using transfection based studies, whereby the aim of the project was to elucidate, in detail, the mechanism(s) by which macrophage differentiation induced apoE expression, and to determine the *cis*-acting elements in the apoE promoter that were responsible for this upregulation. A macrophage-based transfection system was therefore systematically optimised to enable the analysis of several DNA constructs containing regions of the apoE promoter and the enhancer regions, ME1 and ME2 of the apoE gene. However, whilst this system was able to demonstrate the previously reported TNF- $\alpha$  induction of the apoE promoter and also the differentiation-dependent induction of the LPL promoter, no such induction was observed for the apoE promoter or enhancer constructs during PMA-induced differentiation, despite the variation of several parameters. There are several possible reasons for this, but the most likely is that the constructs used did not contain the necessary elements required to modulate the effect. To fully overcome this problem, the cloning of greater regions of the apoE promoter is likely to be required, also constructs containing additional sequences such as that of the intergenic region would be of great use.

#### **CHAPTER SEVEN – GENERAL DISCUSSION**

It was therefore decided, in line with our original aims, to investigate the regulation of apoE by the LXR subfamily of nuclear receptors. As described in Chapter 1, members of the LXR subfamily regulate multiple aspects of lipid metabolism and recent work analysing the effect of a synthetic LXR ligand in murine models of atherosclerosis has provided direct evidence for the atheroprotective effect of LXR agonists (Joseph *et al.*, 2002b). Work by Laffitte *et al.* (2001b) has also demonstrated the direct regulation of apoE by LXRs. This ability of LXR agonists to upregulate apoE gene expression and to regulate other key genes of the cholesterol efflux pathway, such as ABCA1 and ABCG1, is of potential therapeutic value. A clearer understanding of LXR-regulated transcription of these genes would enable the development of promising agents for intervention in human cardiovascular disease. As cell signalling pathways appear to be essential in the transactivation function of several nuclear receptors it was decided to determine the signalling pathways involved in the LXR regulation of apoE in macrophages through the use of commercially available inhibitors.

We demonstrated that the oxysterol ligand, 22(R)-hydroxycholesterol, was able to induce apoE mRNA levels, protein levels and secretion and that pre-treatment with inhibitors against the JNK/SAPK, PI3K and CK2 pathways prevented this induction. This inhibition of apoE induction was also shown in human primary-monocyte-derived macrophages. Further investigation also demonstrated that these inhibitors were able to prevent the ligand-mediated induction of ABCG1 and LXRa mRNA. However, in the case of ABCA1, only the JNK and CK2 inhibitors had an affect on the ligand-mediated induction of this gene. Also, in addition to the JNK, PI3K and CK2 inhibitors, the ligand-mediated induction of LXRa was also shown to be eliminated by the p38 inhibitor, SB202190 and the ligand-mediated induction of ABCG1 was inhibited by the ERK MAPK inhibitor.



FIGURE 7.1. SCHEMATIC SUMMARY OF THE CELL SIGNALLING PATHWAYS IDENTIFIED THROUGH INHIBITOR STUDIES TO PARTICIPATE IN THE LXR-MEDIATED REGULATION OF TARGET GENES The results of these experiments are novel findings, demonstrating that the LXR regulation of target genes is more complicated than initially thought. We have demonstrated that cell signalling pathways play a vital role in the LXR-regulation of apoE expression and several other LXR-regulated genes, and have shown that although a common mode of regulation may exist for these genes, there are some gene-specific differences (Figure 7.1). These results, therefore provided a basis for further investigation of the JNK/SAPK, PI3K and CK2 pathways in the LXR-mediated regulation of apoE.

We next investigated the ability of the LXR ligand, 22(R)-hydroxycholesterol to activate the JNK pathway. It was subsequently confirmed that the JNK/SAPK MAPK was indeed phosphorylated in response to ligand treatment and that the kinase activity of this enzyme was also induced. This induction of the JNK pathway was rapid with an increase in phosphorylation occurring after only 30min. The rapid kinetics in which this pathway was activated by 22(R)-hydroxycholesterol indicated that this activation was not dependent on gene transcription and that this ligand may be able to exert non-genomic or extranuclear actions in addition to its transcription-based function. This is a novel observation and suggests that similar to other nuclear receptor agonists such as estrogens and progestins, that have well-characterised non-genomic actions, the activation of these cell signaling pathways may be due to the existence of a receptor outside the nucleus. Further investigation also revealed that the c-jun transcription factor, a downstream target of JNK, becomes phosphorylated in response to ligand treatment, indicating, that similar to the LXR-mediated differentiation of keratinocytes, whereby oxysterols induce AP-1 proteins (Schmuth *et al.*, 2004), the AP-1 transcription complex is important in the LXR-mediated regulation of apoE.

In addition to the JNK pathway, the involvement of the PI3K and CK2 pathways in the LXR-mediated regulation of apoE was also investigated. These studies demonstrated that treatment of THP-1 macrophages with the LXR ligand, 22(R)-hydroxycholesterol resulted, in Akt, a downstream component of PI3K signalling, being phosphorylated and its kinase activity being induced. Akt activates several nuclear factors and we decided to investigate whether any of these factors, with known roles in nuclear receptor signalling were phosphorylated in response to ligand treatment. Both GSK-3β and FKHR were eliminated from involvement in this regulation of apoE, as neither demonstrated phosphorylation in response to ligand treatment.

CK2 involvement in the LXR-mediated regulation of apoE was also demonstrated through dominant negative studies and the kinase activity of this enzyme was investigated, and was shown to be activated in response to ligand treatment.



FIGURE 7.2. SCHEMATIC SUMMARY OF THE SIGNALLING PATHWAYS IDENTIFIED IN THE LXR-MEDIATED INDUCTION OF APOE IN THP-1 MACROPHAGES

Therefore, in summary, the work presented in this thesis has successfully addressed the role of cell signalling in the LXR-mediated regulation of apoE and has led to the following important findings:

- Induction of apoE mRNA, protein and secretion by the oxysterol ligand 22(R)hydroxycholesterol.
- Identification of a novel role of JNK, PI3K and CK2 pathways in the LXR-mediated induction of apoE gene expression in THP-1 macrophages (Figure 7.2).
- A potential role of cell signalling pathways in ABCA1, ABCG1 and LXRα expression with some common components and potential gene-specific differences.

Furthermore, the data presented in this thesis can be approached with a good degree of confidence given that we have performed positive controls, accurately produced previously published work and verified novel findings through the use of several experimental techniques where possible. In addition, the key findings of this study were also confirmed in primary cultures, indicating that the results presented here may be relevant to human atherosclerosis. Consequently, the following sections will attempt to discuss the greater significance of these findings and the future studies that could be performed in order to advance towards the ultimate goal of a therapeutic treatment of atherosclerosis.

#### 7.2. WIDER PERSPECTIVES OF THE NOVEL FINDINGS IN THIS THESIS

Nuclear receptors represent excellent molecular targets for drug therapy and some of the currently marketed drugs are well-established agonists of these transcription factors. Although several animal studies demonstrate support for the overall beneficial role of LXR activation in the control of atherosclerosis, it is still unclear whether or not the biological actions of LXRs make these receptors potential therapeutic targets. Indeed, recent findings have indicated that the treatment of db/db diabetic mice with the LXR ligand T-0901317 causes more severe lipogenic pathology due the LXR transactivation of fatty acid biosynthesis genes, demonstrating that special caution should be taken whilst evaluating LXR ligands as therapeutic agents to lower plasma cholesterol levels and to prevent atherosclerosis (Chisholm *et al.*, 2003). Therefore, the development of selective agonists that dissociate the antiatherosclerotic effects from the triglyceride raising effects seems to be a promising direction for future LXR study from a therapeutic perspective and an understanding of the gene-specific differences in signalling components may help in designing such therapy.

Recently, it has been reported that, in the skeletal muscle of mice and in a myogenic cell line, LXR agonists specifically induce cholesterol metabolism gene expression without affecting genes involved in fatty acid synthesis (Muscat et al., 2002). Also, YT-32 [(22E)-ergost-22-ene-1a, 3β-diol], a synthetic derivative of phytosterol, has been demonstrated to act as a selective LXR agonist to specifically activate intestinal ABC transporters in mice without increasing plasma triglyceride levels (Kaneko et al., 2003). Moreover, gene-selective modulation by the synthetic oxysterol ligand DMHCA (N, N-dimethyl-3β-hydroxycholenamide), whereby ABCA1 transcriptional activity is stimulated with minimal effects on SREBP-1c, has been shown to occur in human THP-1 macrophages (Quinet et al., 2004). Also, the generation of selective LXR modulators that disrupt the binding of LXR to corepressors without leading to coactivator recruitment may have the potential to selectively increase cholesterol efflux genes and thus be used for anti-atherogenic purposes without having a side-effect on lipogenesis. Recent studies have demonstrated NCoR, one of the corepressors recruited by the unliganded LXR heterodimer, is not recruited to LXR target genes in LXR-/- macrophages, which is sufficient to allow increased expression of the ABCA1 gene and enhanced cholesterol efflux, but does not result in derepression of SREBP-1c or increased fatty acid biosynthesis (Hu et al., 2003).

In addition, to selective agonists, the identification of cell signalling pathway involvement in the regulation of nuclear receptor signalling provides another potential mechanism for therapeutic intervention. Not only could research into these pathways enable the differentiation between various target genes of these receptors and provide a better understanding of the precise mechanisms involved in nuclear receptor-activated transcription, which would allow the desired therapeutic outcomes to be achieved without unwanted side-effects, but recent developments have also demonstrated that the selective targeting of protein kinases is becoming a therapeutic reality (Heldin *et al.*, 2001) and the clinical use of small molecule inhibitors may perhaps become a common procedure.

## **7.3. FUTURE STUDIES**

As a result of the work presented in this thesis, many avenues for further experimental studies have become apparent. Although potential mechanisms for the involvement of the JNK, PI3K and CK2 have been suggested, further work in elucidating the major upstream and downstream components of these pathways and their the exact roles in the LXR-mediated regulation of apoE expression must be undertaken. Firstly, it is important to determine how these signalling pathways are being activated by the LXR ligand, 22(R)-hydroxycholesterol. It may be, as has been shown for several other nuclear receptors, that a membrane- or cytoplasm-based LXR receptor exists and ligand-binding by the receptor may then result in LXR activation of various upstream components of these pathways. The identification of an LXR receptor outside the nucleus could possibly be undertaken through investigations using fluorescent microscopy to identify the location of LXR receptors within the cell or perhaps western blot analysis using extracts for the various cellular compartments and a specific LXR antibody. How this receptor is activating these pathways could then be analysed by co-immuniprecipitation studies using an antibody against LXR and the signalling components of this pathway such as JNK, CK2 or the p85 regulatory subunit of PI3K. The results of these investigations would identify whether LXR is binding to the signalling molecules and would provide a potential mechanism for the activation of cell signalling pathways by LXR. Unfortunately, these experiments were not carried out during this study due to time restraints and the lack of a specific, commercial LXR antibody, however, a monoclonal antibody against the human LXR $\alpha$  has recently been established (Watanabe et al., 2003). In addition to identifying the mechanism behind the non-genomic activation of these pathways by 22(R)hydroxycholesterol, these experiments would further confirm that the activation of the JNK, PI3K and CK2 signalling pathways was LXR-dependent.

With respect to the JNK pathway, additional experiments identifying how c-jun is involved in the LXR-mediated upregulation of apoE would be worthwhile. One suggested mechanism is that c-jun is activating the transcription of a coregulator protein necessary for the LXR activation of apoE. Therefore, before these experiments are carried out it may firstly be useful to identify potential coregulators involved in the LXR-mediated transcription of apoE. Transfection based analysis using expression plasmids for coregulator proteins represents one way in which we could carryout these studies, whilst chromatin immuniprecipitation (CHIP) assays would confirm the interaction *in vivo*, in a normal chromatin context. Identification of coregulators through CHIP analysis would entail THP-1 cells being treated with ligand for the requisite time and then incubated with formaldehyde to cross-link the coregulators and DNA-binding proteins to the DNA *in vivo*. Cell nuclei are then prepared and, following lysis, the chromatin is fragmented by sonication and subjected to immunoprecipitation using antibodies specific to the coregulator of interest. The cross-links are then reversed and the precipitated DNA purified and subjected to PCR using primers against the LXRE sequences of the apoE promoter in order to identify whether these sequences were precipitated with the coregulator protein. Once potential coregulators have been identified it would then be possible to investigate whether c-jun is involved in the transcription of these proteins. It may also be possible that c-jun is affecting the apoE promoter directly through the AP-1 element (Basheeruddin *et al.*, 1994), which could be tested by EMSA analysis or chromatin immunoprecipitation. It would also be of great interest to identify whether the promoters of other LXR target genes such as ABCA1, ABCG1 and LXR $\alpha$  contain an AP-1 element, as the induction of both these genes was inhibited by pre-treatment with curcumin.

As detailed in Chapter 6, we have also demonstrated the involvement of the PI3K and CK2 pathways in the LXR-mediated induction of apoE. Although the studies presented in this thesis have demonstrated the activation of Akt, further studies are necessary to identify whether any Akt targets are activated by LXR treatment, this could be carried out through western blot analysis as was done for GSK-3 $\beta$  and FKHR in this study, or possibly through the use of inhibitors.

Akt has been shown to phosphorylate certain nuclear receptors such as ER $\alpha$  and AR (Campbell *et al.*, 2001; Lin *et al.*, 2001b) and we have also identified Akt phosphorylation sites in the LXR proteins, therefore investigation of LXR phosphorylation would be important to progress these studies. This could be undertaken by an *in vitro* phosphorylation assay, whereby a recombinant LXR protein could be used as a substrate for a radioactive Akt kinase assay. This assay would entail immunoprecipitating Akt from cell extracts, which would then be used in a kinase reaction using a specific kinase buffer, LXR as the substrate and  $\gamma$ -<sup>32</sup>P-labelled ATP as the phosphate donor. Once the kinase reaction is complete the contents of this reaction could be size-fractionated on a SDS-PAGE gel and phosphorylation of LXR by Akt could be assessed by autoradiography. This kinase assay could also be used to investigate the ability of CK2 to phosphorylate LXR as we have also identified a potential CK2 phosphorylation motif in the LXR proteins.

With respect to the involvement of cell signalling pathways in the LXR regulation of genes, it would be of great interest to establish whether the activation of the pathways identified in this study are ligand-specific. The data in this thesis could be further extended to synthetic compounds as several synthetic LXR ligands have been shown to demonstrate atheroprotective properties in mouse models and a greater understanding of the mechanism and signalling pathways involved in

this LXR-mediated effect would be of great benefit. Also the potential involvement of other cell signalling pathways should be further investigated with regards to the ligand, 22(R)-hydroxycholesterol, in order to gain a better understanding of the proposed mechanisms. Cell signalling pathways very rarely act in isolation and although we have ruled out the ERK MAPK and p38 MAPK pathways in the LXR-mediated up-regulation of apoE, several other pathways which have been shown to be extensively involved in nuclear receptor signalling, such as protein kinase A (PKA) and the cyclin-dependent kinases (cdks) also require investigation.

It is also important, with regards to the wider implications of this work to determine the involvement of cell signalling pathways in the LXR-mediated regulation of genes involved in lipogenesis such as FAS and SREBP-1c. Again, these experiments could initially be carried out using inhibitors against key signalling pathways, as was carried out in this study. The results of these investigations would provide further information, which could aid the development of therapeutic strategies to selectively activate the expression of atheroprotective genes such as apoE and ABCA1, without the unwanted side-effects of lipogenesis.

#### 7.4. CONCLUDING REMARKS

From the work presented in this thesis, it can be seen that the LXR-mediated regulation of the apoE gene in macrophages is dependent on several cell signalling pathways. The atheroprotective ability of apoE and the ability of LXR to regulate genes of the reverse cholesterol transport pathway and limit foam cell formation, demonstrates the importance of these proteins in the pathogenesis of atherosclerosis and also their potential as therapeutic targets. Therefore, it is crucial to gain as complete an understanding as possible of all aspects of their molecular mechanisms. By pursuing the future studies suggested above, it is hoped that further progress regarding this objective will be made.

185

# **REFERENCES**

ABE Y, HASHIMOTO S, HORIE T, (1999). Curcumin inhibition of inflammatory cytokine production by human peripheral blood monocytes and alveolar macrophages. *Pharmacol Res* 39(1): 41-7.

ABRINK M, GOBL AE, HUANG R, NILSSON K, HELLMAN L, (1994). Human cell lines U-937, THP-1 and Mono Mac 6 represent relatively immature cells of the monocyte-macrophage cell lineage. *Leukemia* 8(9): 1579-84.

ACTON S, RIGOTTI A, LANDSCHULZ KT, XU S, HOBBS HH, KRIEGER M, (1996). Identification of scavenger receptor SR-BI as a high density lipoprotein receptor. Science 271(5248): 518-520.

ADAMS M, REGINATO MJ, SHAO D, LAZAR MA, CHATTERJEE VK, (1997). Transcriptional activation by peroxisome proliferator-activated receptor gamma is inhibited by phosphorylation at a consensus mitogen-activated protein kinase site. *J Biol Chem* 272(8): 5128-32.

ADAM-STITAH S, PENNA L, CHAMBON P, ROCHETTE-EGLY C, (1999). Hyperphosphorylation of the retinoid X receptor alpha by activated c-Jun NH2-terminal kinases. *J Biol Chem* 274(27): 18932-41.

ADLER V, POLOTSKAYA A, WAGNER F, KRAFT AS, (1992). Affinity-purified c-Jun amino-terminal protein kinase requires serine/threonine phosphorylation for activity. *J Biol Chem* 267(24): 17001-5.

AGELLON LB, DROVER VAB, CHEEMA SK, GBAGUIDI GF, WALSH A, (2002). Dietary cholesterol fails to stimulate the human cholesterol 7α-hydroxylase gene (CYP7A1) in transgenic mice. *J Biol Chem* 277(23): 20131-20134.

AGGERBECK LP, WETTERAU JR, WEISGRABER KH, WU CS, LINDGREN FT, (1988). Human apolipoprotein E3 in aqueous solution. II. Properties of the amino- and carboxy-terminal domains. *J Biol Chem* 263(13): 6249-6258.

AGULLO G, GAMET-PAYRASTRE L, MANENTI S, VIALA C, REMESY C, CHAP H, PAYRASTRE B, (1997). Relationship between flavonoid structure and inhibition of phosphatidylinositol 3-kinase: a comparison with tyrosine kinase and protein kinase C inhibition. *Biochem Pharmacol* 53(11):1649-57.

AIELLO RJ, BREES D, BOURASSA PA, ROYER L, LINDSEY S, COSKRAN T, HAGHPASSAND M, FRANCONE OL, (2002). Increased atherosclerosis in hyperlipidemic mice with inactivation of ABCA1 in macrophages. *Arterioscler Thromb Vasc Biol* 22: 630-637.

AIT-SI-ALI S, RAMIREZ S, BARRE FX, DKHISSI F, MAGNAGHI-JAULIN L, GIRAULT JA, ROBIN P, KNIBIEHLER M, PRITCHARD LL, DUCOMMUN B, ET AL., (1998). Histone acetyltransferase activity of CBP is controlled by cycle-dependent kinases and oncoprotein E1A. *Nature* 396: 184-186.

AKIYAMA TE, SAKAI S, LAMBERT G, NICOL CJ, MATSUSUE K, PIMPARLE S, LEE YH, RICOTE M, GLASS CK, BREWER HB JR, GONZALEZ FJ, (2002). Conditional disruption of the peroxisome proliferatoractivated receptor gamma gene in mice results in lowered expression of ABCA1, ABCG1, and apoE in macrophages and reduced cholesterol efflux. *Mol Cell Biol* 22 (8): 2607-2619.

ALBERTI S, SCHUSTER G, PARINI P, FELTKAMP D, DICZFALUSKY U, RUDLING M, ANGELIN B, BJORKHEM I, PETTERSSON S, GUSTAFSSON JA, (2001). Hepatic cholesterol metabolism and resistance to dietary cholesterol in LXRbeta-deficient mice. *J Clin Invest* 107(5): 565-573.

ALESSI DR, ANDJELKOVIC M, CAUDWELL B, CRON P, MORRICE N, COHEN P, HEMMINGS BA, (1996). Mechanism of activation of protein kinase B by insulin and IGF-1. *EMBO J* 15(23): 6541-51.

ALESSI DR, DEAK M, CASAMAYOR A, CAUDWELL FB, MORRICE N, NORMAN DG, GAFFNEY P, REESE CB, MACDOUGALL CN, HARBISON D, ASHWORTH A, BOWNES M, (1997). 3-Phosphoinositide-dependent protein kinase-1 (PDK1): structural and functional homology with the Drosophila DSTPK61 kinase. *Curr Biol* 7(10): 776-89.

AL-HAIDERI M, GOLDBERG IJ, GALEANO NF, GLEESON A, VOGEL T, GORECKI M, STURLEY SL, DECKELBAUM RJ, (1997). Heparan sulfate proteoglycan-mediated uptake of apoliporotein E-triglyceride-rich lipoprotein particles: a major pathway at physiological particle concentrations. *Biochemistry* 36: 12766-12772.

ALLAN CM, WALKER D, SEGREST JP, TAYLOR JM, (1995a). Identification of a new human gene (APOC4) in the apolipoprotein E, C-I, and C-II gene locus. *Genomics* 28(2): 291-300.

ALLAN CM, WALKER D, TAYLOR JM, (1995b). Evolutionary duplication of a hepatic control region in the human apolipoprotein E gene locus. Identification of a second region that confers high level and liver-specific expression of the human apolipoprotein E gene in transgenic mice. *J Biol Chem* 270(44): 26278-81.

ALLENDE JE, ALLENDE CC, (1995). Protein kinases. 4. Protein kinase CK2: an enzyme with multiple substrates and a puzzling regulation. FASEB J 9(5): 313-23.

ANDERSON KE, COADWELL J, STEPHENS LR, HAWKINS PT, (1998). Translocation of PDK-1 to the plasma membrane is important in allowing PDK-1 to activate protein kinase B. *Curr Biol* 8(12): 684-91.

ANDJELKOVIC M, ALESSI DR, MEIER R, FERNANDEZ A, LAMB NJ, FRECH M, CRON P, COHEN P, LUCOCQ JM, HEMMINGS BA, (1997). Role of translocation in the activation and function of protein kinase B. J Biol Chem 272(50): 31515-24.

ANDREANI-MANGENEY M, VANDENBROUCK Y, JANVIER B, GIRLICH D, BEREZIAT G, (1996). Transcriptional regulation of apolipoprotein E expression by cyclic AMP. *FEBS Lett* 397(2-3): 155-158.

ANGEL P, ALLEGRETTO EA, OKINO ST, HATTORI K, BOYLE WJ, HUNTER T, KARIN M, (1988). Oncogene jun encodes a sequence-specific trans-activator similar to AP-1. *Nature* 332(6160): 166-71.

APFEL R, BENBROOK D, LERNHARDT E, ORTIZ MA, SALBERT G, PFAHL M, (1994). A novel orphan receptor specific for a subset of thyroid-hormone responsive elements and its interaction with the retinoid/thyroid hormone receptor subfamily. *Mol Cell Biol* 14(10): 7025-7035.

ARAI T, RINNINGER F, VARBAN L, FAIRCHILD-HUNTRESS V, LIANG CP, CHEN W, SEO T, DECKELBAUM R, HUSZAR D, TALL AR, (1999). Decreased selective uptake of high density lipoprotein cholesteryl esters in apolipoprotein E knock-out mice. *Proc Natl Acad Sci USA* 96(21): 12050-55.

ARANDA A, PASCUAL A, (2001). Nuclear hormone receptors and gene expression. *Physiol Rev* 81(3): 1269-304.

ARMSTRONG SA, BARRY DA, LEGGETT RW, MUELLER CR, (1997). Casein kinase II-mediated phosphorylation of the C terminus of Sp1 decreases its DNA binding activity. *J Biol Chem* 272(21): 13489-95.

ARNOLD SF, VOROJEIKINA DP, NOTIDES AC, (1995). Phosphorylation of tyrosine 537 on the human estrogen receptor is required for binding to an estrogen response element. *J Biol Chem* 270(50): 30205-12.

ASSEFA Z, VALIUS M, VANTUS T, AGOSTINIS P, MERLEVEDE W, VANDENHEEDE JR, (1999). JNK/SAPK activation by platelet-derived growth factor in A431 cells requires both the phospholipase C-gamma and the phosphatidylinositol 3-kinase signaling pathways of the receptor. *Biochem Biophys Res Commun* 261(3): 641-5.

AUBOEUF D, RIEUSSET J, FAJAS L, VALLIER P, FRERING V, RIOU JP, STAELS P, AUWERX J, LAVILLE M, VIDAL H, (1997). Tissue distribution and quantification of the expression of mRNAs of peroxisome proliferator-activated receptors and liver X receptor- $\alpha$  in humans – no alteration in adipose tissue of obese and NIDDM patients. *Diabetes* 46(8): 1319-1327.

AUGER KR, SERUNIAN LA, SOLTOFF SP, LIBBY P, CANTLEY LC, (1989). PDGF-dependent tyrosine phosphorylation stimulates production of novel polyphosphoinositides in intact cells. *Cell* 57(1): 167-75.

AURICCHIO F, MIGLIACCIO A, CASTORIA G, DI DOMENICO M, BILANCIO A, ROTONDI A, (1996). Protein tyrosine phosphorylation and estradiol action. *Ann N Y Acad Sci* 784:149-72.

AUWERX JH, DEEB S, BRUNZELL JD, PENG R, CHAIT A, (1988). Transcriptional activation of the lipoprotein lipase and apolipoprotein E genes accompanies differentiation in some human macrophage-like cell lines. *Biochemistry* 27(8): 2651-2655.

AUWERX J, (1991). The human leukemia cell line, THP-1: a multifacetted model for the study of monocyte-macrophage differentiation. Experientia 47(1): 22-31.

AUWERX J, STAELS B, VAN VAECK F, CEUPPENS JL, (1992). Changes in IgG Fc receptor expression induced by phorbol 12-myristate 13-acetate treatment of THP-1 monocytic leukemia cells. *Leuk Res* 16(3): 317-27.

AVILA EM, HOLDSWORTH G, SASAKI N, JACKSON RL, HARMONY JA, (1982). Apoprotein E suppresses phytohemagglutinin-activated phospholipid turnover in peripheral blood mononuclear cells. *J Biol Chem* 257(10): 5900-5909.

AVIRAM M, BROOK JG, (1983). Characterization of the effect of plasma lipoproteins on platelet function *in vitro*. Haemostasis 13(6): 344-350.

BABAEV VR, FAZIO S, GLEAVES LA, CARTER KJ, SEMENKOVICH CF, LINTON MF, (1999). Macrophage lipoprotein lipase promotes foam cell formation and atherosclerosis *in vivo*. *J Clin Invest* 103(12): 1697-1705.

BABAEV VR, PATEL MB, SEMENKOVICH CF, FAZIO S, LINTON MF, (2000). Macrophage lipoprotein lipase promotes foam cell formation and atherosclerosis in low density lipoprotein receptor-deficient mice. *J Biol Chem* 275: 26293-26299.

BABIKER A, ANDERSSON O, LUND E, XIU RJ, DEEB S, RESHEF A, LEITERSDORF E, DICZFALUSY U, BJORKHEM I, (1997). Elimination of cholesterol in macrophages and endothelial cells by the sterol 27-hydroxylase mechanism. Comparison with high density lipoprotein-mediated reverse cholesterol transport. *J Biol Chem* 272(42): 26253-26261.

BALENDRAN A, CASAMAYOR A, DEAK M, PATERSON A, GAFFNEY P, CURRIE R, DOWNES CP, ALESSI DR, (1999). PDK1 acquires PDK2 activity in the presence of a synthetic peptide derived from the carboxyl terminus of PRK2. *Curr Biol* 9(8): 393-404.

BARGER PM, BROWNING AC, GARNER AN, KELLY DP, (2001). p38 mitogen-activated protein kinase activates peroxisome proliferator-activated receptor alpha: a potential role in the cardiac metabolic stress response. *J Biol Chem* 276(48): 44495-501.

BARETTINO D, VIVANCO RUIZ MM, STUNNENBERG HG, (1994). Characterization of the liganddependent transactivation domain of thyroid hormone receptor. *EMBO J* 13(13): 3039-49.

BARON S, MANIN M, BEAUDOIN C, LEOTOING L, COMMUNAL Y, VEYSSIERE G, MOREL L, (2004). Androgen receptor mediates non-genomic activation of phosphatidylinositol 3-OH kinase in androgen-sensitive epithelial cells. *J Biol Chem* 279(15): 14579-86.

BARTER P, (2000). CETP and atherosclerosis. Arterioscler Thromb Vasc Biol 20(9): 2029-2031.

BASHEERUDDIN K, RECHTORIS C, MAZZONE T, (1992). Transcriptional and post-transcriptional control of apolipopprotein E expression in differentiating human monocytes. *J Biol Chem* 267(2): 1219-1224.

BASHEERUDDIN K, RECHTORIS C, MAZZONE T, (1994). Evaluation of the role of Ap-1 like proteins in the enhanced apolipoprotein E gene transcription accompanying phorbol ester induced macrophage differentiation. *Biochim Biophys Acta* 1218(2): 235-241.

BASU SK, HO YK, BROWN MS, BILHEIMER DW, ANDERSON RG, GOLDSTEIN JL, (1982). Biochemical and genetic studies of the apoprotein E secreted by mouse macrophages and human monocytes. *J Biol Chem* 257(16): 9788-9795.

BASU S, GOLDSTEIN J, BROWN M, (1983). Independent pathways for secretion of cholesterol and apolipoprotein E by macrophages. *Science* 219(4586): 871-873.

BEHR-ROUSSEL D, RUPIN A, SIMONET S, BONHOMME E, COUMAILLEAU S, CORDI A, SERKIZ B, FABIANI JN, VERBEUREN TJ, (2000). Effect of chronic treatment with the inducible nitric oxide synthase inhibitor N-iminoethyl-L-lysine or with L-arginine on progression of coronary and aortic atherosclerosis in hypercholesterolemic rabbits. *Circulation* 102(9): 1033-1038.

BELLOSTA S, MAHLEY RW, SANAN DA, MURATA J, NEWLAND DL, TAYLOR JM, PITAS RE, (1995). Macrophage-specific expression of human apolipoprotein E reduces atherosclerosis in hypercholesterolemic apolipoprotein E-null mice. *J Clin Invest* 96(5): 2170-2179.

BELT AR, BALDASSARE JJ, MOLNAR M, ROMERO R, HERTELENDY F, (1999). The nuclear transcripton factor NF-kappaB mediates interleukin-1beta-induced expression of cyclooxygenase-2 in human myometrial cells. *Am J Obstet Gynecol* 181(2): 359-366.

BENNETT BL, SASAKI DT, MURRAY BW, O'LEARY EC, SAKATA ST, XU W, LEISTEN JC, MOTIWALA A, PIERCE S, SATOH Y, BHAGWAT SS, MANNING AM, ANDERSON DW, (2001). SP600125, an anthrapyrazolone inhibitor of Jun N-terminal kinase. *Proc Natl Acad Sci USA* 98(24): 13681-6.

BERG DT, CALNEK DS, GRINNELL BW, (1995). The human apolipoprotein E gene is negatively regulated in human liver HepG2 cells by the transcription factor BEF-1. *J Biol Chem* 270(26):15447-50.

BERG D, CALNEK D, GRINNELL B, (1996). Trans-repressor BEF-1 phosphorylation. J Biol Chem 271(9): 4589-4592.

BERGE KE, TIAN H, GRAF GA, YU L, GRISHIN NV, SCHULTZ J, KWITEROVICH P, SHAN B, BARNES R, HOBBS HH, (2000). Accumulation of dietary cholesterol in sitosterolemia caused by mutations in adjacent ABC transporters. *Science* 290(5497): 1771-1775.

BHAT MK, ASHIZAWA K, CHENG SY, (1994). Phosphorylation enhances the target gene sequencedependent dimerization of thyroid hormone receptor with retinoid X receptor. *Proc Natl Acad Sci* USA 91(17): 7927-31.

BIELICKI J, MCCALL M, FORTE T, (1999). Apolipoprotein A-I promotes cholesterol release and apolipoprotein E recruitment from THP-1 macrophage-like foam cells. J Lipid Res 40(1): 85-92.

BJORKHEM I, (1992). Mechanism of degradation of the steroid side chain in the formation of bile acids. *J Lipid Res* 33: 455-471.

BJORKHEM I, ANDERSSON U, ELLIS E, ALVELIUS G, ELLEGARD L, DICZFALULY U, SJOVALL J, EINARSSON C, (2001). From brain to bile. Evidence that conjugation and omega-hydroxylation are important for elimination of 24S-hydroxycholesterol (cerebrosterol) in humans. *J Biol Chem* 276: 37004-37010.

BODZIOCH M, ORSO E, KLUCKEN J ET AL., (1999). The gene encoding ATP-binding cassette transporter 1 is mutated in Tangier disease. *Nat Genet* 22: 347-351.

BOHMANN D, BOS TJ, ADMON A, NISHIMURA T, VOGT PK, TJIAN R, (1987). Human proto-oncogene cjun encodes a DNA binding protein with structural and functional properties of transcription factor AP-1. *Science* 238(4832): 1386-92.

BOISVERT WA, SPANGENBERG J, CURTISS LK, (1995). Treatment of severe hypercholesterolemia in apolipoprotein E-deficient mice by bone marrow transplantation. *J Clin Invest* 96: 1118-1124.

BOISVERT WA, SANTIAGO R, CURTISS LK, TERKELTAUB RA, (1998). A leukocyte homologue of the IL-8 receptor CXCR-2 mediates the accumulation of macrophages in atherosclerotic lesions of LDL receptor-deficient mice. *J Clin Invest* 101(2): 353-363.

BOISVERT WA, BLACK AS, CURTISS LK, (1999). ApoAI reduces free cholesterol accumulation in atherosclerotic lesions of apoE-deficient mice transplanted with apoE-expressing macrophages, *Arterioscler Thromb Vasc Biol* 19(3): 525-530.

BOISVERT WA, CURTISS LK, (1999). Elimination of macrophage-specific apolipoprotein E reduces diet-induced atherosclerosis in C57BL/6J male mice. *J Lipid Res* 40(5): 806-813.

BOONYARATANAKORNKIT V, SCOTT MP, RIBON V, SHERMAN L, ANDERSON SM, MALLER JL, MILLER WT, EDWARDS DP, (2001). Progesterone receptor conatins a proline-rich motif that directly interacts with SH3 domains and activates c-Src family tyrosine kinases. *Mol Cell* 8(2): 269-280.

BOONYARATANAKORNKIT V, EDWARDS DP, (2004). Receptor mechanisms of rapid extranuclear signalling initiated by steroid hormones. *Essays Biochem* 40:105-20.

BOREN J, OLIN K, LEE I, CHAIT A, WIGHT TN, INNERARITY TL, (1998). Identification of the principal proteoglycan-binding site in LDL. A single-point mutation in apo-B100 severly affects proteoglycan interaction without affecting LDL receptor binding. *J Clin Invest* 101(12): 2658-2664.

BORING L, GOSLING J, CLEARY M, CHARO IF, (1998). Decreased lesion formation in CCR2-/- mice reveals a role for chemokines in the initiation of atherosclerosis. *Nature* 394(6696): 894-897.

BORTNICK AE, ROTHBLAT GH, STOUDT G, HOPPE KL, ROYER LJ, MCNEISH J, FRANCONE OL, (2000). The correlation of ATP-binding cassette 1 mRNA levels with cholesterol efflux from various cell lines. *J Biol Chem* 275(37): 28634-28640.

BOUCHER P, LIU P, GOTTHARDT M, HIESBERGER T, ANDERSON RG, HERZ J, (2002). Platelet-derived growth factor mediates tyrosine phosphorylation of the cytoplasmic domain of the low density lipoprotein receptor-related protein in caveolae. *J Biol Chem* 277(18): 15507-15513.

BOURGUET W, GERMAIN P, GRONEMEYER H, (2000). Nuclear receptor ligand-binding domains: threedimensional structures, molecular interactions and pharmacological implications. *Trends Pharmacol Sci* 21(10): 381-8.

BRAND K, MACKMAN N, CURTISS L, (1993). Interferon  $\gamma$  inhibits macrophage apolipoprotein E production by posttranslational mechanisms. *J Clin Invest* 91(5): 2031-2039.

BRENDEL C, GELMAN L, AUWERX J, (2002). Multiprotein bridging factor-1 (MBF-1) is a cofactor for nuclear receptors that regulate lipid metabolism. *Mol Endocrinol* 16(6): 1367-77.

BROOKS-WILSON A, MARCIL M, CLEE SM, ET AL., (1999). Mutations in ABC1 in Tangier disease and familial high-density lipoprotein deficiency. *Nat Genet* 22: 336-345.

BROWN MS, HO YK, GOLDSTEIN JL, (1980). The cholesteryl ester cycle in macrophage foam cells. *J* Biol Chem 255(19): 9344-9352.

BROWN DL, HIBBS MS, KEARNEY M, LOUSHIN C, ISNER JM, (1995). Identification of 92-kD gelatinase in human coronary atherosclerotic lesions. Association of active enzyme synthesis with unstable angina. *Circulation* 91(8): 2125-31.

BROWN MS, GOLDSTEIN JL, (1997). The SREBP pathway: regulation of cholesterol metabolism by proteolysis of a membrane-bound transcription factor. *Cell* 89(3): 331-340.

BROWN AJ, JESSUP W, (1999). Oxysterols and atherosclerosis. Atherosclerosis 142(1): 1-28.

BROWN MS, GOLDSTEIN JL, (1999). A proteolytic pathway that controls cholesterol content of membranes, cells, and blood. *Proc Natl Acad Sci USA* 96(20): 11041-11048.

BRUCE C, CHOUINARD RA, TALL AR, (1998). Plasma lipid transfer proteins, high-density lipoproteins and reverse cholesterol transport. *Ann Rev Nutr* 18: 297-330.

BRUNET A, BONNI A, ZIGMOND MJ, LIN MZ, JUO P, HU LS, ANDERSON MJ, ARDEN KC, BLENIS J, GREENBERG ME, (1999). Akt promotes cell survival by phosphorylating and inhibiting a Forkhead transcription factor. *Cell* 96(6): 857-68.

BUNONE G, BRIAND PA, MIKSICEK RJ, PICARD D, (1996). Activation of the unliganded estrogen receptor by EGF involves the MAP kinase pathway and direct phosphorylation 15(9): 2174-83.

BURGERING BM, COFFER PJ, (1995). Protein kinase B (c-Akt) in phosphatidylinositol-3-OH kinase signal transduction. *Nature* 376(6541): 599-602.

CADER A, STEINBERG F, MAZZONE T, CHAIT A, (1997). Mechanisms of enhanced macrophage apoE secretion by oxidised LDL. *J Lipid Res* 38(5): 981-991.

CAGEN LM, DENG X, WILCOX HG, PARK EA, RAGHOW R, ELAM MB, (2004). Insulin activates the rat sterol regulatory element-binding protein 1c (SREBP-1c) promoter through the combinatorial actions of SREBP, LXR, Sp-1, and NF-Y cis-acting elements. *Biochem J* [Epub ahead of print].

CAMP HS, TAFURI SR, LEFF T, (1999). c-Jun N-terminal kinase phosphorylates peroxisome proliferator-activated receptor-gamma1 and negatively regulates its transcriptional activity. *Endocrinology* 140(1): 392-7.

CAMPBELL RA, BHAT-NAKSHATRI P, PATEL NM, CONSTANTINIDOU D, ALI S, NAKSHATRI H, (2001). Phosphatidylinositol 3-kinase/AKT-mediated activation of estrogen receptoralpha: a new model for anti-estrogen resistance. *J Biol Chem* 276(13): 9817-24.

CANTLEY LC, (2002). The phosphoinositide 3-kinase pathway. Science 296(5573): 1655-7.

CAO G, BEYER TP, YANG XP, SCHMIDT RJ, ZHANG Y, BENSCH WR, KAUFFMAN RF, GAO H, RYAN TP, LIANG Y, EACHO PI, JIANG XC, (2002) Phospholipid transfer protein is regulated by liver X receptors in vivo. *J Biol Chem* 277(42): 39561-39565.

CARDIN AD, BOWLIN TL, KRSTENANSKY JL, (1988). Inhibition of lymphocyte proliferation by synthetic peptides homologous to human plasma apolipoproteins B and E. *Biochem Biophys Res Commun* 154(2): 741-745.

CARPENTER CL, CANTLEY LC, (1990). Phosphoinositide kinases. Biochemistry 29(51): 11147-56.

CASAMAYOR A, MORRICE NA, ALESSI DR, (1999). Phosphorylation of Ser-241 is essential for the activity of 3-phosphoinositide-dependent protein kinase-1: identification of five sites of phosphorylation in vivo. *Biochem J* 342(2): 287-92.

CASTANO E, VOROJEIKINA DP, NOTIDES AC, (1997). Phosphorylation of serine-167 on the human oestrogen receptor is important for oestrogen response element binding and transcriptional activation. *Biochem J* 326(1): 149-57.

CASTRILLO A, JOSEPH SB, MARATHE C, MANGELSDORF DJ, TONTONOZ P, (2003a). Liver X receptordependent repression of matrix metalloproteinase-9 expression in macrophages. *J Biol Chem* 278(12): 10443-10449.

CASTRILLO A, JOSEPH SB, VAIDYA SA, HABERLAND M, FOGELMAN AM, CHENG G, TONTONOZ P, (2003b). Crosstalk between LXR and toll-like receptor signaling mediated bacterial and viral antagonism of cholesterol metabolism. *Mol Cell* 12(4): 805-816.

CAULIN-GLASER T, GARCIA-CARDENA G, SARREL P, SESSA WC, BENDER JR, (1997). 17 beta-estradiol regulation of human endothelial cell basal nitric oxide release, independent of cytosolic Ca2+ mobilization. *Circ Res* 81(5): 885-92.

CAVIGELLI M, DOLFI F, CLARET FX, KARIN M, (1995). Induction of c-fos expression through JNKmediated TCF/Elk-1 phosphorylation. EMBO J 14(23): 5957-64. CAVIN LG, ROMIEU-MOUREZ R, PANTA GR, SUN J, FACTOR VM, THORGEIRSSON SS, SONENSHEIN GE, ARSURA M, (2003). Inhibition of CK2 activity by TGF-beta1 promotes IkappaB-alpha protein stabilization and apoptosis of immortalized hepatocytes. *Hepatology* 38(6): 1540-51.

CHAIT A, IVERIUS PH, BRUNZELL JD, (1982). Lipoprotein lipase secretion by human monocyte – derived macrophages. *J Clin Invest* 69: 490-493.

CHAM BE, KNOWLES BR, (1976). A solvent system for delipidation of plasma or serum without protein precipitation. J Lipid Res 17(2): 176-81.

CHANG DJ, PAIK YK, LEREN TP, WALKER DW, HOWLETT GJ, TAYLOR JM, (1990). Characterisation of a human apolipoprotein E gene enhancer element and its associated protein factors. *J Biol Chem* 265(16): 9496-9504.

CHAWLA S, HARDINGHAM GE, QUINN DR, BADING H, (1998). CBP: a signal-regulated transcriptional coactivator controlled by nuclear calcium and CaM kinase IV. Science 281(5382): 1505-9.

CHAWLA A, BOISVERT WA, LEE CH, LAFFITTE BA, BARAK Y, JOSEPH SB, LIAO D, NAGY L, EDWARDS PA, CURTISS LK, EVANS RM, TONTONOZ P, (2001a). A PPAR<sub>γ</sub>-LXR-ABCA1 pathway in macrophages is involved in cholesterol efflux and atherogenesis. *Mol Cell* 7(1): 161-171.

CHAWLA A, BARAK Y, NAGY L, LIAO D, TONTONOZ P, EVANS RM. (2001b). PPAR- $\gamma$  dependent and independent effects on macrophage-gene expression in lipid metabolism and inflammation. *Nat Med* 7(1): 48-52.

CHEN YR, TAN TH, (1998). Inhibition of the c-Jun N-terminal kinase (JNK) signaling pathway by curcumin. Oncogene 17(2):173-8.

CHEN D, PACE PE, COOMBES RC, ALI S, (1999). Phosphorylation of human estrogen receptor alpha by protein kinase A regulates dimerization. *Mol Cell Biol* 19(2): 1002-15.

CHIANG JY, (1998). Regulation of bile acid synthesis. Front Biosci 3: D176-D193.

CHIANG JY, KIMMEL R, STROUP D, (2001). Regulation of cholesterol7 $\alpha$ -hydroxylase gene (CYP7A1) transcription by the liver orphan receptor (LXR $\alpha$ ). *Gene* 262(1-2): 257-265.

CHINETTI G, LESTAVEL S, BOCHER V, REMALEY AT, NEVE B, TORRA IP, TEISSIER E, MINNICH A, JAYE M, DUVERGER N, BREWER HB, FRUCHART JC, CLAVEY V, STAELS B, (2001). PPAR- $\alpha$  and PPAR- $\gamma$  activators induce cholesterol removal from human macrophage foam cells through stimulation of the ABCA1 pathway. *Nat Med* 7(1): 53-58.

CHISHOLM JW, HONG J, MILLS SA, LAWN RM, (2003). The LXR ligand T0901317 induces severe lipogenesis in the db/db diabetic mouse. *J Lipid Res* 44(11): 2039-48.

CLAUDEL T, LEIBOWITZ MD, FIEVET C, TAILLEUX A, WAGNER B, REPA JJ, TORPIER G, LOBACCARO JM, PATERNITI JR, MANGELSDORF DJ, HEYMAN RA, AUWERX J, (2001). Reduction of atherosclerosis in apolipoprotein E knockout mice by activation of the retinoid X receptor. *Proc Natl Acad Sci* 98(5): 2610-2615.

COLLINS RG, VELJI R, GUEVARA NV, HICKS MJ, CHAN L, BEAUDET AL, (2000). P-selectin or intracellular adhesion molecule (I-CAM)-1 deficiency substantially protects against atherosclerosis in apolipoprotein E-deficient mice. *J Exp Med* 191(1): 189-194.

COLLINS JL, FIVUSH AM, WATSON MA, GALARDI CM, LEWIS MC, MOORE LB, PARKS DJ, WILSON JG, TIPPIN TK, BINZ JG, PLUNKET KD, MORGAN DG, BEAUDET EJ, WHITNEY KD, KLIEWER SA, WILLSON TM, (2002). Identification of a nonsteroidal liverX receptor agonist through parallel array synthesis of tertiary amines. *J Med Chem* 45(10): 1963-1966.

COSTET P, LUO Y, WANG N, TALL AR, (2000). Sterol-dependant trans-activation of the ABC1 promoter by the liver X receptor/retinoid X receptor. *J Biol Chem* 275(36): 28240-28245.

CRITCHFIELD JW, COLIGAN JE, FOLKS TM, BUTERA ST, (1997). Casein kinase II is a selective target of HIV-1 transcriptional inhibitors. *Proc Natl Acad Sci USA* 94(12):6110-5.

CROSS DA, ALESSI DR, COHEN P, ANDJELKOVICH M, HEMMINGS BA, (1995). Inhibition of glycogen synthase kinase-3 by insulin mediated by protein kinase B. *Nature* 378(6559): 785-9.

CULLEN P, CIGNARELLA A, BRENNHAUSEN B, MOHR S, ASSMANN G, VON ECKARDSTEIN A, (1998). Phenotype-dependent differentiation in apoE metabolism and in cholesterol homeostasis in human monocyte-derived macrophages. *J Clin Invest* 101(8): 1670-1677.

CYBULSKY MI, GIMBRONE MA, (1991). Endothelial expression of a mononuclear leukocyte adhesion molecule during atherogenesis. *Science* 251(4995): 788-791.

CYBULSKY MI, IIYAMA K, LI H, ZHU S, CHEN M, IIYAMA M, DAVIS V, GUTIERREZ-RAMOS JC, CONNELLY PW, MILSTONE DS, (2001). A major role for VCAM-1, but not ICAM-1, in early atherosclerosis. *J Clin Invest* 107(10): 1255-1262.

CYRUS T, WITZTUM JL, RADER DJ, TANGIRALA R, FAZIO S, LINTON MF, FUNK CD, (1999). Disruption of 12/15-lipoxygenase diminishes atherosclerosis in apoE-deficient mice. *J Clin Invest* 103(11): 1597-1604.

DANG Q, TAYLOR J, (1996). In vivo footprinting analysis of the hepatic control region of the human apolipoprotein E/C-I/C-I/C-II gene locus. *J Biol Chem* 271(45): 28667-76.

DANIELIAN PS, WHITE R, LEES JA, PARKER MG, (1992). Identification of a conserved region required for hormone dependent transcriptional activation by steroid hormone receptors. *EMBO J* 11(3): 1025-33.

DANSKY HM, CHARLTON SA, HARPER MM, SMITH JD, (1997). T and B lymphocytes play a minor role in atherosclerotic plaque formation in the apolipoprotein E-deficient mouse. *Proc Nat Acad Sci USA* 94(9): 4642-4646.

DATTA SR, BRUNET A, GREENBERG ME, (1999). Cellular survival: a play in three Akts. Genes Dev 13(22): 2905-27.

DAVIGNON J, GREGG RE, SING CF, (1988). Apolipoprotein E polymorphism and atherosclerosis. *Arteriosclerosis* 8(1): 1-21.

DAVIS IJ, HAZEL TG, CHEN RH, BLENIS J, LAU LF, (1993). Functional domains and phosphorylation of the orphan receptor Nur77. Mol Endocrinol 7(8): 953-64.

DE ANDRADE M, THANDI I, BROWN S, GOTTO A JR, PATSCH W, BOERWINKLE E, (1995). Relationship of the apolipoprotein E polymorphism with carotid artery atherosclerosis. *Am J Hum Genet* 56(6): 1379-90.

DEAN M, RZHETSKY A, ALLIKMETS R, (2001). The human ATP-binding cassette (ABC) transporter superfamily. *Genome Res* 11(7): 1156-1166.

DEBOSE-BOYD R, OU J, GOLDSTEIN J, BROWN M, (2001). Expression of sterol regulatory elementbinding protein 1c (SREBP-1c) mRNA in rat hepatoma cells requires endogenous LXR ligands. *Proc natl Acad Sci USA* 98(4): 1477-1482.

DELERIVE P, FURMAN C, TEISSIER E, FRUCHART J, DURIEZ P, STAELS B, (2000). Oxidized phospholipids activate PPARalpha in a phospholipase A2-dependent manner. *FEBS Lett* 471(1): 34-38.

DENG J, RUDICK V, DORY L, (1995). Lysosomal degradation and sorting of apolipoprotein E in macrophages. *J Lipid Res* 36(10): 2129-2140.

DE PAUW M, VANLOO B, WEISGRABER KH, ROSSENAU M, (1995). Comparison of lipid-binding and lecithin: cholesterol acyltransferase activation of the amino- and carboxyl-terminal domains of human apolipoprotein E3. *Biochemistry* 34(34): 10953-10966.

DERIJARD B, HIBI M, WU IH, BARRETT T, SU B, DENG T, KARIN M, DAVIS RJ, (1994). JNK1: a protein kinase stimulated by UV light and Ha-Ras that binds and phosphorylates the c-Jun activation domain. *Cell* 76(6): 1025-37.

DESURMONT C, CAILLUND JM, EMMANUEL F, BENOIT P, FRUCHART JC, CASTRO G, BRANELLEC D, HEARD JM, DUVERGER N, (2000). Complete atherosclerosis regression after human apoE gene transfer in apoE-deficient/nude mice. *Arterioscler Thromb Vasc Biol* 20: 435-442.

DETMERS PA, HERNANDEZ M, MUDGETT J, HASSING H, BURTON C, MUNDT S, CHUN S, FLETCHER D, CARD DJ, LISNOCK J, WEIKEL R, BERGSTROM JD, SHEVELL DE, HERMANOWSKI-VOSATKA A, SPARROW CP, CHAO YS, RADER DJ, WRIGHT SD, PURE E, (2000). Deficiency in inducible nitric oxide synthase results in reducedatherosclerosis in apolipoprotein E-deficient mice. *J Immunol* 165(6): 3430-5.

DE WET JR, WOOD KV, DELUCA M, HELSINKI DR, SUBRAMANIUM S, (1987). Firefly luciferase gene: structure and expression in mammalian cells. *Mol Cell Biol* 7(2): 725-737.

DICHTL W, NILSSON L, GONCALVES I, ARES MP, BANFI C, CALARA F, HAMSTEN A, ERIKSSON P, NILSSON J, (1999). Very low-density lipoprotein activates nuclear factor-kappaB in endothelial cells. *Circ Res* 84(9): 1085-1094.

DIETSCHY JM, TURLEY SD, (2002). Control of cholesterol turnover in the mouse. J Biol Chem 277: 3801-3804.

DONG LM, WILSON C, WARDELL MR, SIMMONS T, MAHLEY RW, WEISGRABER KH, AGARD DA, (1994). Human apolipoprotein E. Role of arginine 61 in mediating the lipoprotein preferences of the E3 and E4 isoforms. *J Biol Chem* 269(35): 22358-22365.

DONG LM, PARKIN S, TRAKHANOV SD, RUPP B, SIMMONS T ARNOLD KS, NEWHOUSE YM, INNERARITY TL, WEISGRABER KH, (1996). Novel mechanism for defective receptor binding of apolipoprotein E2 in type III hyperlipoproteinemia. *Nat Struct Biol* 3(8): 718-722.

DONG LM, WEISGRABER KH, (1996). Human apolipoprotein E4 domain interaction. Arginine 61 and glutamic acid 255 interact to direct the preference of very low density lipoproteins. *J Biol Chem* 271(32): 19053-19057.

DONG Z, CHAPMAN S, BROWN A, FRENETTE P, HYNES R, WAGNER D, (1998). The combined role of Pand E-selectins in atherosclerosis. *J Clin Invest* 102(1): 145-152.

DORY L, (1989). Synthesis and secretion of apoE in thioglycolate-elicited mouse peritoneal macrophages: effect of cholesterol efflux. *J Lipid Res* 30(6): 809-816.

DORY L, (1991). Regulation of apolipoprotein E secretion by high-density lipoprotein<sub>3</sub> in mouse macrophages. *J Lipid Res* 32(5): 783-792.

DOWELL P, OTTO TC, ADI S, LANE MD, (2003). Convergence of peroxisome proliferator-activated receptor gamma and Foxo1 signaling pathways. *J Biol Chem* 278(46): 45485-91.

DOWHAN DH, MUSCAT GE, (1996). Characterization of the AB (AF-1) region in the muscle-specific retinoid X receptor-gamma: evidence that the AF-1 region functions in a cell-specific manner. *Nucleic Acids Res* 24(2): 264-71.

DOWNWARD J, (1998). Mechanisms and consequences of activation of protein kinase B/Akt. Curr Opin Cell Biol 10(2): 262-7.

DOYLE GA, SAARIALHO-KERE UK, PARKS WC, (1997). Distinct mechanisms regulate TIMP-1 expression at different stages of phorbol ester-mediated differentiation of U937 cells. *Biochemistry* 36(9): 2492-500.

DRESKIN SC, THOMAS GW, DALE SN, HEASLEY LE, (2001). Isoforms of Jun kinase are differentially expressed and activated in human monocyte/macrophage (THP-1) cells. *J Immunol* 166(9): 5646-5653.

DRIGGERS PH, SEGARS JH, RUBINO DM, (2001). The proto-oncoprotein Brx activates estrogen receptor beta by a p38 mitogen-activated protein kinase pathway. *J Biol Chem* 276(50):46792-7.

DRISCOLL DM, GETZ GS, (1984). Extrahepatic synthesis of apolipoprotein E. J Lipid Res 25(12): 1368-1379.

DUAN H,  $\sqcup$  Z, MAZZONE T, (1995). Tumor necrosis factor- $\alpha$  modulates monocyte/macrophage apoprotein E gene expression. *J Clin Invest* 96(2): 915-922.

DUAN HW, LIN CY, MAZZONE T, (1997). Degradation of macrophage apoE in nonlysosomal compartment. *J Biol Chem* 272(49): 31156-31162.

DUAN H, GU D, MAZZONE T, (2000). Sterols and inhibitors of sterol transport modulates the degradation and secretion of macrophage apoE: Requirement for the C-terminal domain. *Biochim Biophys Acta* 1484(2-3): 142-150.

DUDLEY DT, PANG L, DECKER SJ, BRIDGES AJ, SALTIEL AR, (1995). A synthetic inhibitor of the mitogen-activated protein kinase cascade. *Proc Natl Acad Sci USA* 92(17): 7686-9.

DURAND B, SAUNDERS M, GAUDON C, ROY B, LOSSON R, CHAMBON P, (1994). Activation function 2 (AF-2) of retinoic acid receptor and 9-cis retinoic acid receptor: presence of a conserved autonomous constitutive activating domain and influence of the nature of the response element on AF-2 activity. *EMBO J* 13(22): 5370-82.

DYER CA, CURTISS LK, (1988). Apoprotein E-rich high density lipoproteins inhibit ovarian androgen synthesis. *J Biol Chem* 263(22): 10965-10973.

DYER CA, SMITH RS, CURTISS LK, (1991). Only multimers of a synthetic peptide of human apolipoprotein E are biologically active. *J Biol Chem* 266(23): 15009-15015.

EDWARDS PA, KAST HR, ANISFELD AM, (2002). BAREing it all: the adoption of LXR and FXR and their roles in lipid homeostasis. *J Lipid Res* 43(1): 2-12.

EGGERT M, MOWS CC, TRIPIER D, ARNOLD R, MICHEL J, NICKEL J, SCHMIDT S, BEATO M, RENKAWITZ R, (1995). A fraction enriched in a novel glucocorticoid receptor-interacting protein stimulates receptor-dependent transcription in vitro. *J Biol Chem* 270(51): 30755-9.

EISENBERG S, SEHAYEK E, OLIVECRONA T, VLODAVSKY I, (1992). Lipoprotein lipase enhances binding of lipoproteins to heparan sulfate on cell surfaces and extracellular matrix. *J Clin Invest* 90(5): 2013-2021.

ELLSON CD, ANDREWS S, STEPHENS LR, HAWKINS PT, (2002). The PX domain: a new phosphoinositide-binding module. *J Cell Sci* 115(6): 1099-105.

ELSHOURBAGY NA, LIAO WS, MAHLEY RW, TAYLOR JM, (1985). Apolipoprotein E mRNA is abundant in the brain and arenals, as well as in the liver, and is present in other peripheral tissues of rats and marmosets. *Proc Natl Acad Sci USA* 82(1): 203-207.

ENGEL T, LORKOWSKI S, LUEKEN A, RUST S, SCHLUTER B, BERGER G, CULLEN P, ASSMAN G, (2001). The human abcg4 gene is regulated by oxysterols and retinoids in monocyte-derived macrophages. *Biochem Biophys Res Commun* 288(2): 483-488.

EOM SJ, KIM EY, LEE JE, KANG HJ, SHIM J, KIM SU, GWAG BJ, CHOI EJ, (2001). Zn(2+) induces stimulation of the c-Jun N-terminal kinase signaling pathway through phosphoinositide 3-Kinase. *Mol Pharmacol* 59(5): 981-6.

FAUST M, MONTENARH M, (2000). Subcellular localization of protein kinase CK2. A key to its function? *Cell Tissue Res* 301(3): 329-40.

FAZIO S, BABAEV VR, MURRAY AB, HASTY AH, CARTER KJ, GLEAVES LA, ATKINSON JB, LINTON MF, (1997). Increased atherosclerosis in mice reconstituted with apolipoprotein E null macrophages. *Pro Natl Acad Sci USA* 94(9): 4647-4652.

FAZIO S, LINTON MF, HASTY AH, SWIFT LL, (1999). Recycling of apolipoprotein E in mouse liver. J Biol Chem 274(12): 8247-8253.

FELZIEN LK, FARRELL S, BETTS JC, MOSAVIN R, NABEL GJ, (1999). Specificity of cyclin E-Cdk2, TFIIB, and E1A interactions with a common domain of the p300 coactivator. *Mol Cell Biol* 19(6): 4241-6.

FENG W, WEBB P, NGUYEN P, LIU X, LI J, KARIN M, KUSHNER PJ, (2001). Potentiation of estrogen receptor activation function 1 (AF-1) by Src/JNK through a serine 118-independent pathway. *Mol Endocrinol* 15(1): 32-45.

FLEMING Y, ARMSTRONG CG, MORRICE N, PATERSON A, GOEDERT M, COHEN P, (2000). Synergistic activation of stress-activated protein kinase 1/c-Jun N-terminal kinase (SAPK1/JNK) isoforms by mitogen-activated protein kinase kinase 4 (MKK4) and MKK7. *Biochem J* 352(1): 145-54.

FONT DE MORA J, BROWN M, (2000). AIB1 is a conduit for kinase-mediated growth factor signaling to the estrogen receptor. *Mol Cell Biol* 20: 5041-5047.

FORD HL, LANDESMAN-BOLLAG E, DACWAG CS, STUKENBERG PT, PARDEE AB, SELDIN DC, (2000).Cell cycle-regulated phosphorylation of the human SIX1 homeodomain protein. J Biol Chem 275(29): 22245-54.

FORMAN BM, RUAN BF, CHEN J, SCHROEPFER GJ, EVANS RM, (1997). The orphan nuclear receptor LXR $\alpha$  is positively and negatively regulated by distinct products of mevalonate metabolism. *Proc Natl Acad Sci USA* 94(20): 10588-10593.

FORTE TM, GOTH-GOLDSTEIN R, NORDHAUSEN RW, MCCALL MR, (1993). Apolipoprotein A-I-cell membrane interaction: extracellular assembly of heterogeneous nascent HDL particles. *J Lipid Res* 34 (2): 317-24.

FRANTZ B, KLATT T, PANG M, PARSONS J, ROLANDO A, WILLIAMS H, TOCCI MJ, O'KEEFE SJ, O'NEILL EA, (1998). The activation state of p38 mitogen-activated protein kinase determines the efficiency of ATP competition for pyridinylimidazole inhibitor binding. *Biochemistry* 37(39): 13846-53.

FROSTEGARD J, NILSSON J, HAEGERSTRAND A, HAMSTEN A, WIGZELL H, GIDLUND M, (1990). Oxidized low density lipoprotein induces differentiation and adhesion of human monocytes and the monocytic cell line U937. Proc Natl Acad Sci USA 87(3): 904-8.

FRUCHART JC, DURIEZ P, ZTAELS B, (1999). Peroxisome proliferator-activated receptor-alpha activators regulate genes governing lipoprotein metabolism, vascular inflammation and atherosclerosis. *Curr Opin Lipid* 10: 245-257.

FU X, MENKE JG, CHEN Y, ZHOU G, MACNAUL KL, WRIGHT SD, SPARROW CP, LUND EG, (2001). 27-Hydroxylase is an endogenous ligand for liver X receptor in cholesterol-loaded cells. *J Biol Chem* 276(42): 38378-38387.

FU T, BORENSZTAJN J, (2002). Macrophage uptake of low-density lipoprotein bound to aggregated C-reactive protein: possible mechanism of foam-cell formation in atherosclerotic lesions. *Biochem J* 366(1): 195-201.

GALETTO R, ALBAJAR M, POLANCO JI, ZAKIN MM, RODRIGUEZ-REYJC, (2001). Identification of a peroxisome-proliferator-activated-receptor response element in the apolipoprotein E gene control region. *Biochem J* 357(2): 521-7.

GAN X, KAPLAN R, MENKE JG, MACNAUL K, CHEN Y, SPARROW CP, ZHOU G, WRIGHT SD, CAI TQ, (2001). Dual mechanisms of ABCA1 regulation by geranylgeranyl pyrophosphate. *J Biol Chem* 276(52): 48702-48708.

GARCIA M, VAZQUEZ J, GIMENEZ C, VALDIVIESO F, ZAFRA F, (1996). Transcription factor AP-2 regulate human apolipoprotein E/C-I/C-IV/C-II gene locus. *J Neurosci* 16(23): 7550-7556.

GARCIA-GARCIA E, ROSALES R, ROSALES C, (2002). Phosphatidylinositol 3-kinase and extracellular signal-regulated kinase are recruited for Fc receptor-mediated phagocytosis during monocyte-to-macrophage differentiation. *J Leukoc Biol* 72(1):107-14.

GHAVIDEL A, SCHULTZ MC, (1997). Casein kinase II regulation of yeast TFIIIB is mediated by the TATA-binding protein. *Genes Dev* 11(21): 2780-9.

GHISELLI G, SCHAEFER EJ, GASCON P, BRESER HB, (1981). Type III hyperlipoproteinemia associated with apolipoprotein E deficiency. *Science* 214(4526): 1239-41.

GIANNI M, KOPF E, BASTIEN J, OULAD-ABDELGHANI M, GARATTINI E, CHAMBON P, ROCHETTE-EGLY C, (2002). Down-regulation of the phosphatidylinositol 3-kinase/Akt pathway is involved in retinoic acid-induced phosphorylation, degradation, and transcriptional activity of retinoic acid receptor gamma 2. *J Biol Chem* 277(28): 24859-62.

GIBBONS GF, (1990). Assembly and secretion of hepatic very-low-density lipoprotein. *Biochem J* 268(1): 1-13.

GILL G, PTASHNE M, (1988). Negative effect of the transcriptional activator GAL4. *Nature* 334(6184): 721-4.

GILLOTTE KL, ZAIOU M, LUND-KATZ S, ANANTHARAMAIAH GM, HOLVOET P, DHOEST A, PALGUNACHARI MN, SEGREST JP, WEISGRABER KH, ROTHBLAT GH, PHILLIPS MC, (1999). Apolipoprotein-mediated plasma membrane microsolubilization. Role of lipid affinity and membrane penetration in the efflux of cellular cholesterol and phospholipid. *J Biol Chem* 274 (4): 2021-2028.

GIMBLE JM, DORHEIM MA, CHENG Q, PEKALA P, ENERBECK S, ELLINGSWORTH L, KINCADE PW, WANG CS, (1989). Response of bone marrow stromal cells to adipogenic antagonists. *Mol Cell Biol* 9(11): 4587-4595.

GLASS CK, WITZTUM JL, (2001). Atherosclerosis: the road ahead. Cell 104(4): 503-516.

GLINEUR C, ZENKE M, BEUG H, GHYSDAEL J, (1990). Phosphorylation of the v-erbA protein is required for its function as an oncogene. *Genes Dev* 4(10): 1663-76.

GLOMSET JA, (1968). The plasma lecithins: cholesterol acyltransferase reaction. J Lipid Res 9(2): 155-167.

GO YM, PARK H, MALAND MC, DARLEY-USMAR VM, STOYANOV B, WETZKER R, JO H, (1998). Phosphatidylinositol 3-kinase gamma mediates shear stress-dependent activation of JNK in endothelial cells. *Am J Physiol* 275(5 Pt 2): H1898-904.

GOLDBERG I, (1996). Lipoprotein lipase and lipolysis: central roles in lipoprotein metabolism and atherogenesis. *J Lipid Res* 37(4): 693-707.

GOLDSTEIN JL, BROWN MS, (1977). The low-density lipoprotein pathway and its relation to atherosclerosis. Ann Rev Biochem 46: 897-930.

GORDON DA, FENJVES ES, WILLIAMS DL, TAICHMAN LB, (1989). J Invest Dermatol 92: 96-99.

GOUGH PJ, RAINES EW, (2003). Advances in retroviral transduction of hematopoietic stem cells for the gene therapy of atherosclerosis. *Curr Opin Lipidol* 14(5): 491-7.

GOULD AL, ROSSOUW JE, SANTANELLO NC, HEYSE JF, FURBERG CD, (1998). Cholesterol reduction yields clinical benefit: Impact of statin trials. *Circulation* 97: 946-952.

GRAF GA, LI WP, GERARD RD, GELISSEN I, WHITE A, COHEN JC, HOBBS HH, (2002) Coexpression of ATP-binding cassette proteins ABCG5 and ABCG8 permits their transport to the apical surface. *J Clin Invest* 110: 659-669.

GREGG RE, ZECH LA, SCHAEFER EJ, BREWER HB, (1981). Type III hyperlipoproteinemia: defective metabolism of an abnormal apolipoprotein E. Science 211(4482): 584-586.

GU L, OKADA Y, CLINTON SK, GERARD C, SUKHOVA GK, LIBBY P, ROLLINS BJ, (1998). Absence of monocyte chemoattractant protein-1 reduces atherosclerosis in low-density lipoprotein-deficient mice. *Mol Cell* 2(2): 275-281.

GUPTA S, CAMPBELL D, DERIJARD B, DAVIS RJ, (1995). Transcription factor ATF2 regulation by the JNK signal transduction pathway. *Science* 267(5196): 389-93.

GUPTA S, BARRETT T, WHITMARSH AJ, CAVANAGH J, SLUSS HK, DERIJARD B, DAVIS RJ, (1996). Selective interaction of JNK protein kinase isoforms with transcription factors. *EMBO J* 15(11): 2760-70.

GUPTA S, STRAVITZ RT, DENT P, HYLEMON PB, (2001). Down-regulation of cholesterol 7alphahydroxylase (CYP7A1) gene expression by bile acids in primary rat hepatocytes is mediated by the c-Jun N-terminal kinase pathway. *J Biol Chem* 276(19): 15816-22.

HAGHPASSAND M, BOURASSA PA, FRANCONE OL, AIELLO RJ, (2001). Monocyte/macrophage expression of ABCA1 has minimal contribution to plasma HDL levels, *J Clin Invest* 108: 1315-1320.

HAMMER GD, KRYLOVA I, ZHANG Y, DARIMONT BD, SIMPSON K, WEIGEL NL, INGRAHAM HA, (1999). Phosphorylation of the nuclear receptor SF-1 modulates cofactor recruitment: integration of hormone signaling in reproduction and stress. *Mol Cell* 3(4): 521-6.

HAN KH, CHANG MK, BOULLIER A, GREEN SR, LI A, GLASS CK, QUEHENBERGER O, (2000). Oxidized LDL reduces monocyte CCR2 expression through pathways involving peroxisome proliferator-activated receptor gamma. *J Clin Invest* 106(6): 793-802.

HARA H, YOKOYAMA S, (1991). Interaction of free apolipoproteins with macrophages. Formation of high density lipoprotein-like lipoproteins and reduction of cellular cholesterol. *J Biol Chem* 266: 3080-3066.

HASSAN S, DOBNER PR, CARRAWAY RE, (2004). Involvement of MAP-kinase, PI3-kinase and EGFreceptor in the stimulatory effect of Neurotensin on DNA synthesis in PC3 cells. *Regul Pept* 120(1-3):155-66.

HASTY AH, LINTON MF, BRANDT SJ, BABAEV VR, GLEAVES LA, FAZIO S, (1999). Retroviral gene therapy in apoE-deficient mice: apoE expression in the artery wall reduces early foam cell lesion formation. *Circulation* 99(19): 2571-2576.

HAUPT Y, MAYA R, KAZAZ A, OREN M, (1997). Mdm2 promotes the rapid degradation of p53. *Nature* 387(6630): 296-9.

HAYDEN MR, CLEE SM, BROOKS-WILSON A, ET AL, (2000). Cholesterol efflux regulatory protein, Tangier disease and familial high-density lipoprotein deficiency. *Curr Opin Lipidol* 11: 117-122.

HAYEK TJ, OIKNINE J, BROOK JG, AVIRAM M, (1994). Role of HDL apolipoprotein E in cellular cholesterol efflux. Studies in apoE knockout transgenic mice. *Biochem Biophys Res Commun* 205(2): 1072-1078.

HAYNES MP, SINHA D, RUSSELL KS, COLLINGE M, FULTON D, MORALES-RUIZ M, SESSA WC, BENDER JR, (2000). Membrane estrogen receptor engagement activates endothelial nitric oxide synthase via the PI3-kinase-Akt pathway in human endothelial cells. *Circ Res* 87(8): 677-82.

HEEREN J, WEBER W, BEISIEGEL U, (1999). Intracellular processing of endocytosed triglyceride-rich lipoproteins comprises both recycling and degradation. *J Cell Sci* 112(3): 349-359.

HELDIN CH, (2001). Signal transduction: multiple pathways, multiple options for therapy. Stem Cells 19(4): 295-303.

HENRY KW, YUAN X, KOSZEWSKI NJ, ONDA H, KWIATKOWSKI DJ, NOONAN DJ, (1998). Tuberous sclerosis gene 2 product modulates transcription mediated by steroid hormone receptor family members. *J Biol Chem* 273(32): 20535-9.

HERICHE JK, LEBRIN F, RABILLOUD T, LEROY D, CHAMBAZ EM, GOLDBERG Y, (1997). Regulation of protein phosphatase 2A by direct interaction with casein kinase 2alpha. *Science* 276(5314): 952-5.

HERMANSON O, GLASS CK, ROSENFELD MG, (2002). Nuclear receptor coregulators: multiple modes of modification. *Trends Endocrinol Metab* 13(2): 55-60.

HERSKOWITZ I, (1995). MAP kinase pathways in yeast: for mating and more. Cell 80(2): 187-97.

HERZ J, WILLNOW TE, (1995). Lipoprotein and receptor interactions in vivo. *Curr Opin Lipidol* 6(2): 97-103.

HIBI M, LIN A, SMEAL T, MINDEN A, KARIN M, (1993). Identification of an oncoprotein- and UV-responsive protein kinase that binds and potentiates the c-Jun activation domain. *Genes Dev* 7(11): 2135-48.

HILGARD P, CZAJA MJ, GERKEN G, STOCKERT RJ, (2004). Proapoptotic function of protein kinase CK2alpha is mediated by a JNK signaling cascade. *Am J Physiol Gastrointest Liver Physiol* 287(1): G192-201.

HIPSKIND RA, BACCARINI M, NORDHEIM A, (1994). Transient activation of RAF-1, MEK, and ERK2 coincides kinetically with ternary complex factor phosphorylation and immediate-early gene promoter activity in vivo. *Mol Cell Biol* 14(9): 6219-31.

HIROTA K, DAITOKU H, MATSUZAKI H, ARAYA N, YAMAGATA K, ASADA S, SUGAYA T, FUKAMIZU A, (2003). Hepatocyte nuclear factor-4 is a novel downstream target of insulin via FKHR as a signal-regulated transcriptional inhibitor. *J Biol Chem* 278(15): 13056-60.

HISAMOTO K, OHMICHI M, KURACHI H, HAYAKAWA J, KANDA Y, NISHIO Y, ADACHI K, TASAKA K, MIYOSHI E, FUJIWARA N, TANIGUCHI N, MURATA Y, (2001). Estrogen induces the Akt-dependent activation of endothelial nitric-oxide synthase in vascular endothelial cells. *J Biol Chem* 276(5): 3459-67.

HO YY, AL-HAIDERI M, MAZZONE T, VOGEL T, PRESLEY JF, STURLEY SL, DECKELBAUM RJ, (2000). Endogenous epressed apoliporotein E has different effects on cell lipid. *Biochemistry* 39: 4746-4754.

HONG SH, PRIVALSKY ML, (2000). The SMRT corepressor is regulated by a MEK-1 kinase pathway: inhibition of corepressor function is associated with SMRT phosphorylation and nuclear export. *Mol Cell Biol* 20(17): 6612-25.

HORKKO S, BINDER CJ, SHAW PX, CHANG MK, SILVERMAN G, PALINSKI W, WITZTUM JL, (2000). Immunological responses to oxidized LDL. Free Radic Biol Med 28(12): 1771-1779.

HORTON JD, SHIMOMURA I, (1999). Sterol regulatory element-binding proteins: activators of cholesterol and fatty acid biosynthesis. *Curr Opin Lipidol* 10(2): 143-150.

HU E, KIM JB, SARRAF P, SPIEGELMAN BM, (1996). Inhibition of adipogenesis through MAP kinasemediated phosphorylation of PPARgamma. *Science* 274(5295): 2100-3.

Hu X, LI S, WU J, XIA C, LALA DS, (2003). Liver X receptors interact with corepressors to regulate gene expression. *Mol Endocrinol* 17(6): 1019-26.

HUANG Y, LI XG, RALL SC, TAYLOR JM, VON ECKARDSTEIN A, ASSMANN G, MAHLEY RW, (1998). Overexpression and accumulation of apolipoprotein E as a cause of hypertriglyceridemia. *J Biol Chem* 273(41): 26388-26393.

HUANG ZH, LIN CY, ORAM JF, MAZZONE T, (2001). Sterol efflux mediated by endogenous macrophage apoE expression is independent of ABCA1. *Arterioscler Thromb Vasc Biol* 21(12): 2019-2025.

HUANG ZQ, LI J, SACHS LM, COLE PA, WONG J, (2003). A role for cofactor-cofactor and cofactorhistone interactions in targeting p300, SWI/SNF and Mediator for transcription. *EMBO J* 22(9): 2146-55.

HUANG Y, WEISGRABER KH, MUCKE L, MAHLEY RW, (2004). Apolipoprotein E: diversity of cellular origins, structural and biophysical properties, and effects in Alzheimer's disease. *J Mol Neurosci* 23(3): 189-204.

HUGHES TR, TENGKU-MUHAMMAD TS, IRVINE SA, RAMJI DP, (2002). A novel role of Sp1 and Sp3 in the interferon-gamma -mediated suppression of macrophage lipoprotein lipase gene transcription. *J Biol Chem* 277(13): 11097-106.

HUI DY, HARMONY JAK, (1980a). Inhibition of calcium accumulation in mitogen-activated lymphocytes: Role of membrane-bound plasma lipoproteins. *Proc Natl Acad Sci USA* 77(8): 4764-4768.

HUI DY, HARMONY JAK, (1980b). Phosphatidylinositol turnover in mitogen-activated lymphocytes. *Biochem J* 192(1): 91-98.

HUSE M, KURIYAN J, (2002). The conformational plasticity of protein kinases. Cell 109(3): 275-82.

HUSSAIN MM, MAXFIELD FR, MAS-OLIVA J, TABAS I, JI ZS, INNERARITY TL, MAHLEY RW, (1991). Clearance of chylomicron remanants by the low density lipoprotein receptor-related protein/ $\alpha$ 2-macroglobulin receptor. *J Biol Chem* 266(21): 13936-13940.

HUUSKONEN J, FIELDING PE, FIELDING CJ, (2004). Role of p160 coactivator complex in the activation of liver X receptor. *Arterioscler Thromb Vasc Biol* 24(4): 703-8.

INNERARITY TL, MAHLEY RW, (1978). Enhanced binding by cultured human fibroblasts of apoE contaiming lipoproteins as compared with low density lipoproteins. *Biochemistry* 17(8): 1440-1447.
INNERARITY TL, WEISGREBER KH, ARNOLD KS, RALL SC, MAHLEY RW, (1984). Normalization of receptor binding in apolipoprotein E2. Evidence for modulation of the binding site conformation. *J Biol Chem* 259(11): 7261-7267.

IP YT, DAVIS RJ, (1998). Signal transduction by the c-Jun N-terminal kinase (JNK)--from inflammation to development. *Curr Opin Cell Biol* 10(2): 205-19.

ISHIGAMI M, SWERTFEGER DK, HUI MS, GRANHOLM NA, HUI DY, (1998). Apolipoprotein E inhibits platelet-derived growth factor-induced vascular smooth muscle cell migration and proliferation by suppressing signal transduction and preventing entry to G1 phase. *J Biol Chem* 273(32): 20156-20161.

ISHIGAMI M, SWERTFEGER DK, HUI MS, GRANHOLM NA, HUI DY, (2000). Apolipoprotein E inhibition of vascular SMC proliferation but not the inhibition of migration is mediated through activation of inducible nitric oxide synthase. *Arterioscler Thromb Vasc Biol* 20(4): 1020-1026.

JACQ X, BROU C, LUTZ Y, DAVIDSON I, CHAMBON P, TORA L, (1994). Human TAFII30 is present in a distinct TFIID complex and is required for transcriptional activation by the estrogen receptor. *Cell* 79(1): 107-17.

JAISWAL K, TELLO V, LOPEZ-GUZMAN C, NWARIAKU F, ANTHONY T, SAROSI GA JR (2004). Bile salt exposure causes phosphatidyl-inositol-3-kinase-mediated proliferationb in a Barrett's adenocarcinoma cell line. Surgery 136(2): 160-8.

JAMES SR, DOWNES CP, GIGG R, GROVE SJ, HOLMES AB, ALESSI DR, (1996). Specific binding of the Akt-1 protein kinase to phosphatidylinositol 3,4,5-trisphosphate without subsequent activation. *Biochem J* 315(3): 709-13.

JANOWSKI BA, WILLY PJ, RAMA-DEVI T, FALCK JR, MANGELSDORF DJ, (1996). An oxysterol signalling pathway mediated by the nuclear receptor LXR $\alpha$ . *Nature* 383(6602): 728-731.

JANOWSKI BA, GROGAN MJ, JONES SA, WISLEY GB, KLIEWER SA, COREY EJ, MANGELSDORF DJ, (1999). Structural requirements of ligands for the oxysterol liver X receptors LXRalpha and LXRbeta. *Proc Natl Acad Sci USA* 96(1): 266-271.

JI ZS, BRECHT WJ, MIRANDA RD, HUSSAIN MM, INNERARITY TL, MAHLEY RW, (1993). Role of heparan sulfate proteoglycans in the binding and uptake of apolipoprotein E-enriched remnant lipoproteins by cultured cells. *J Biol Chem* 268(14): 10160-7.

JIZS, FAZIO S, LEE YL, MAHLEY RW, (1994a). Secretion-capture role for apolipoprotein E in remnant lipoprotein metabolism involving cell surface heparan sulfate proteoglycans. *J Biol Chem* 269(4): 2764-2772.

JI ZS, FAZIO S, MAHLEY RW, (1994b). Variable heparan sulfate proteoglycan binding of apolipoprotein E variants may modulate the expression of type III hyperlipoproteinemia. *J Biol Chem* 269(18): 13421-13428.

JI ZS, LAUER SJ, FAZIO S, BENSADOUN A, TAYLOR JM, MAHLEY RW, (1994c). Enhanced binding and uptake of remnant lipoproteins by hepatic-lipase secreting hepatoma cells in culture. *J Biol Chem*: 269(18): 13429-13436.

JI ZS, SANAN DA, MAHLEY RW, (1995). Intravenous heparinase inhibits remnant lipoprotein clearance from the plasma and uptake by the liver: in vivo role of heparan sulfate proteoglycans. *J Lipid Res* 36(3): 583-592.

JI ZS, DICHEK HL, MIRANDA RD, MAHLEY RW, (1997). Heparan sulfate proteoglycans participate in hepatic lipase- and apolipoprotein E-mediated binding and uptake of plasma lipoproteins, including high density lipoproteins. *J Biol Chem* 272(50): 31285-31292.

JIANG G, NEPOMUCENO L, YANG Q, SLADEK FM, (1997). Serine/threonine phosphorylation of orphan receptor hepatocyte nuclear factor 4. *Arch Biochern Biophys* 340(1): 1-9.

JIANG XC, BRUCE C, MAR J, LIN M, JI Y, FRANCONE OL, TALL AR, (1999). Targeted mutation of plasma phospholipid transfer protein gene markedly reduces high density lipoprotein levels. *J Clin Invest* 103(6): 907-914.

JO DW, LEREN TP, YANG ZY, CHUNG YH, TAYLOR JM, PAIK YK, (1995). Characterisation of an upstream regulatory element of the human apolipoprotein E gene, and purification of its binding protein from the human placenta. *J Biol Chem* 117(4): 915-922.

JOBIN C, BRADHAM CA, RUSSO MP, JUMA B, NARULA AS, BRENNER DA, SARTOR RB, (1999). Curcumin blocks cytokine-mediated NF-kappa B activation and proinflammatory gene expression by inhibiting inhibitory factor I-kappa B kinase activity. *J Immunol* 163(6): 3474-83.

JONASSON L, HOLM J, SKALLI O, BONDJERS G, HANSSON G, (1986). Regional accumulations of T-cells, macrophages, and smooth muscle cells in the human atherosclerotic plaque. *Arteriosclerosis* 6(2): 131-8.

JOSEPH SB, LAFFITTE BA, PATEL PH, WATSON MA, MATSUKUMA KE, WALCZAK R, COLLINS JL, OSBORNE TF, TONTONOZ P, (2002a). Direct and indirect mechanisms for regulation of fatty acid synthase gene expression by liver X receptors. *J Biol Chem* 277(13): 11019-11025.

JOSEPH SB, MCKILLIGIN E, PEI L, WATSON MA, COLLINS AR, LAFFITTE BA, CHEN M, NOH G, GOODMAN J, HAGGER GN, TRAN J, TIPPIN TK, WANG X, LUSIS AJ, HSUEH WA, LAW RE, COLLINS JL, WILSON TM, TONTONOZ P, (2002b). Synthetic LXR ligand inhibits the development of atherosclerosis in mice, *Proc Natl Acad Sci USA* 99(11): 7604-7609.

JOSEPH SB, CASTRILLO A, LAFFITTE BA, MANGELSDORF DJ, TONTONOZ P, (2003). Reciprocal regulation of inflammation and lipid metabolism by liver X receptors. *Nat Med* 9(2): 213-219.

JOYCE CW, AMAR MJ, LAMBERT G, VAISMAN BL, PAIGEN B, ET AL., (2002). The ATP-binding cassette transporter A1 (ABCA1) modulates the development of aortic atherosclerosis in C57BL/6 and apoE-knockout mice. *Proc Natl Acad Sci USA* 99(1): 407-412.

JUGE-AUBRY CE, HAMMAR E, SIEGRIST-KAISER C, PERNIN A, TAKESHITA A, CHIN WW, (1999). Regulation of the transcriptional activity of the peroxisome proliferator-activated receptor alpha by phosphorylation of a ligand-independent trans-activating domain. *J Biol Chem* 274(15): 10505-10510.

JURUTKA PW, HSIEH JC, NAKAJIMA S, HAUSSLER CA, WHITFIELD GK, HAUSSLER MR, (1996). Human vitamin D receptor phosphorylation by casein kinase II at Ser-208 potentiates transcriptional activation. *Proc Natl Acad Sci USA* 93(8): 3519-24.

KAGEYAMA R, OHTSUKA T, TOMITA K, (2000). The bHLH gene Hes1 regulates differentiation of multiple cell types. *Mol Cells* 10(1): 1-7.

KAINU T, KONONEN J, ENMARK E, GUSTAFSSON JA, PELTO-HUIKKO M, (1996). Localization and ontogeny of the orphan receptor OR-1 in the rat brain. *J Mol Neurosci* 7(1): 29-39.

KANEKO E, MATSUDA M, YAMADA Y, TACHIBANA Y, SHIMOMURA I, MAKISHIMA M, (2003). Induction of intestinal ATP-binding cassette transporters by a phytosterol-derived liver X receptor agonist. *J Biol Chem* 278(38): 36091-36098.

KARIN M, LIU Z, ZANDI E, (1997). AP-1 function and regulation. Curr Opin Cell Biol 9(2): 240-6.

KATO S, ENDOH H, MASUHIRO Y, KITAMOTO T, UCHIYAMA S, SASAKI H, MASUSHIGE S, GOTOH Y, NISHIDA E, KAWASHIMA H, METZGER D, CHAMBON P, (1995). Activation of the estrogen receptor through phosphorylation by mitogen-activated protein kinase. *Science* 270(5241): 1491-1494.

KATSUYAMA R, MORIOKA A, OKA S, KAWASAKI T, (2001). Expression of macrophage asialoglycoprotein-binding protein is induced through MAPK classical pathway. *Biochem Biophys Res Commun* 280(5): 1269-73.

KATZ D, REGINATO MJ, LAZAR MA, (1995). Functional regulation of thyroid hormone receptor variant TR alpha 2 by phosphorylation. *Mol Cell Biol* 15(5): 2341-8.

KATZENELLENBOGEN BS, MONTANO MM, EDIGER TR, SUN J, EKENA K, LAZENNEC G, MARTINI PG, MCINERNEY EM, DELAGE-MOURROUX R, WEIS K, KATZENELLENBOGEN JA, (2000). Estrogen receptors: selective ligands, partners, and distinctive pharmacology. *Recent Prog Horm Res* 55:163-93

KASHYAP VS, SANTAMARINA-FOJO S, BROWN DR, PARROTT CL, APPLEBAUM-BOWDEN D, MEYN S, TALLEY G, PAIGEN B, MAEDA N, BREWER HB, (1995). Apolipoprotein E deficiency in mice. Gene replacement and prevention of atherosclerosis using adenovirus vectors. *J Clin Invest* 96: 1612-1620.

KELLY ME, CLAY MA, MISTRY MJ, HSIEH-LI HM, HARMONY JA, (1994). Apolipoprotein E inhibition of of mitogen-activated T-lymphocytes: production of interleukin 2 with reduced biological activity. *Cell Immunol* 159(2): 124-139.

KENNEDY MA, VENKATESWARAN A, TARR PT, XENARIOS I, KUDOH J, SHIMIZU N, EDWARDS PA, (2001). Characterization of the human abcg1 gene. Liver X receptor activates an internal promoter that produces a novel transcript encoding an alternative form of the protein. *J Biol Chem* 276(42): 39438-39447.

KHOO JC, MAHONEY EM, STEINBERG D, (1981). Neutral cholesterol esterase activity in macrophages and its enhancement by cAMP-dependant protein kinase. *J Biol Chem* 256(24): 12659-12661.

KIKUCHI T, HAGIWARA K, HONDA Y, GOMI K, KOBAYASHI T, TAKAHASHI H, TOKUE Y, WATANABE A, NUKIWA T, (2002). Clarithromycin suppresses lipopolysaccharide-induced interleukin-8 production by human monocytes through AP-1 and NF-kappa B transcription factors. *J Antimicrob Chemother* 49(5): 745-55.

KIM JB, SPIEGELMAN BM, (1996). ADD1/SREBP1 promotes adipocyte differentiation and gene expression linked to fatty acid metabolism. *Genes Dev* 10(9): 1096-1107.

KIM SW, PARK K, KWAK E, CHOI E, LEE S, HAM J, KANG H, KIM JM, HWANG SY, KONG YY, LEE K, LEE JW, (2003). Activating signal cointegrator 2 required for liver lipid metabolism mediated by liver X receptors in mice. *Mol Cell Biol* 23(10): 3583-92.

KINOSHITA M, ARAI H, FUKASAWA M, WATANABE T, TSUKAMOTO K, HASHIMOTOINOUE K, KUROKAWA K, TERAMOTO T, (1993). Apolipoprotein E enhances lipid exchange between lipoproteins mediated by cholesteryl ester transfer protein. *J Lipid Res* 34(2): 261-268.

KLUCKEN J, BUCHLER C, ORSO E, KAMINSKI WE, PORSCH-OZCURUMEZ M, LIEBISCH G, KAPINSKY M, DIEDRERICH W, DROBNIK W, DEAN M, ALLIKMETS R, SCHMITZ G, (2000). ABCG1 (ABC8), the human homolog of the *Drosophila* white gene, is a regulator of macrophage cholesterol and phospholipid transport. *Proc Natl Acad Sci USA* 97(2): 817-822.

KNOUFF C, HINSDALE ME, MEZDOUR H, ALTENBURG MK, WATANABE M, QUARFORDT SH, ET AL., (1999). ApoE structure determines VLDL clearance and atherosclerosis risk in mice. *J Clin Invest* 103: 1579-1586.

KNUTTI D, KRESSLER D, KRALLI A, (2001). Regulation of the transcriptional coactivator PGC-1 via MAPK-sensitive interaction with a repressor. *Proc Natl Acad Sci USA* 98: 9713-9718.

KOHRO T, NAKAJIMA T, WADA Y, SUGIYAMA A, ISHII M, TSUTSUMI S, ABURATANI H, IMOTO I, INAZAWA J, HAMAKUBO T, KODAMA T, EMI M, (2000). Genomic structure and mapping of human orphan receptor LXR alpha: upregulation of LXRa mRNA during monocyte to macrophage differentiation. *J Atheroscler Thromb* 7(3): 145-51.

KOREN HS, ANDERSON SJ, LARRICK JW, (1979). *In vitro* activation of a human macrophage-like cell line. *Nature* 279(5711): 328-31.

KOWAL RC, HERZ J, GOLDSTEIN JL, ESSER V, BROWN MS, (1989). Low density lipoprotein receptorrelated protein mediates uptake of cholesteryl esters derived from apoprotein E-enriched lipoproteins. *Proc Natl Acad Sci USA* 86(15): 5810-5814.

KRIEGER M, (1999). Charting the fate of the "good cholesterol": identification and characterisation of the high-density lipoprotein receptor SR-BI. *Ann Rev Biochem* 68: 523-558.

KRIMBOU L, DENIS M, HAIDAR B, CARRIER M, MARCIL M, GENEST J, (2004). Molecular interactions between apoE and ABCA1: Impact on apoE lipidation. *J Lip Res* 45(5): 839-48.

KRSTIC MD, ROGATSKY I, YAMAMOTO KR, GARABEDIAN MJ, (1997). Mitogen-activated and cyclindependent protein kinases selectively and differentially modulate transcriptional enhancement by the glucocorticoid receptor. *Mol Cell Biol* 17(7): 3947-54.

KRUTH HS, SKARLATOS SI, GAYNOR PM, GAMBLE W, (1994). Production of cholesterol-enriched nascent high density lipoproteins by human monocyte-derived macrophages is a mechanism that contributes to macrophage cholesterol efflux. *J Biol Chem* 269(39): 24511-24518.

KUIPERS F, JONG MC, LIN Y, ECK M, HAVINGA R, BLOKS V, VERKADE HJ, HOFKER MH, MOSHAGE H, BERKEL TJ, VONK RJ, HAVEKES LM, (1997). Impaired secretion of very low density lipoprotein-triglycerides by apolipoprotein E-deficient mouse hepatocytes. *J Clin Invest* 100(11): 2915-2922.

KWON HJ, KIM DS, (2003). Production of nuclease activity in U937 cells by phorbol 12-myristate 13acetate and lipopolysaccharide. *J Biochem Mol Biol* 36(5): 520-3. KYRIAKIS JM, AVRUCH J, (1990). pp54 microtubule-associated protein 2 kinase. A novel serine/threonine protein kinase regulated by phosphorylation and stimulated by poly-L-lysine. *J Biol Chem* 265(28): 17355-63.

KYRIAKIS JM, BANERJEE P, NIKOLAKAKI E, DAI T, RUBIE EA, AHMAD MF, AVRUCH J, WOODGETT JR, (1994). The stress-activated protein kinase subfamily of c-Jun kinases. *Nature* 369(6476): 156-60.

Kyriakis JM, (1999). Activation of the AP-1 transcription factor by inflammatory cytokines of the TNF family. *Gene Expr* 7(4-6): 217-31.

KYRIAKIS JM, AVRUCH J, (2001). Mammalian mitogen-activated protein kinase signal transduction pathways activated by stress and inflammation. *Physiol Rev* 81(2):807-69.

LAFFITTE BA, JOSEPH SB, WALCZAK R, PEI L, WILPITZ DC, COLLINS JL, TONTONOZ P, (2001a). Autoregulation of the human liver X receptor alpha promoter. *Mol Cell Biol* 21(22): 7558-7568.

LAFFITTE BA, REPA JJ, JOSEPH SB, WILPITZ DC, KAST HR, MANGELSDORF DJ, TONTONOZ P, (2001b). LXRs control lipid-inducible expression of the apolipoprotein E gene in macrophages and adipocytes. *Proc Natl Acad Sci USA* 98(2); 507-512.

LAFFITTE BA, JOSEPH SB, CHEN M, CASTRILLO A, REPA JJ, WILPITZ DC, MANGELSDORF DJ, TONTONOZ P, (2003). The phospholipid transfer protein gene is an LXR target gene expressed by macrophages in atherosclerotic lesions. *Mol Cell Biol* 23(6): 2182-2191.

LALAZAR A, MAHLEY RW, (1989). Human apolipoprotein E. Receptor binding activity of truncated variants with carboxyl-terminal deletions. *J Biol Chem* 264(15): 8447-8450.

LANDIS MS, PATEL HV, CAPONE JP, (2002). Oxysterol activators of liver X receptor and 9-cis-retinoic acid promote sequential steps in the synthesis and secretion of tumor necrosis factor-alpha from human monocytes. *J Biol Chem* 277(7): 4713-4721.

LANGE CA, SHEN T, HORWITZ KB, (2000). Phosphorylation of human progesterone receptors at serine-294 by mitogen-activated protein kinase signals their degradation by the 26S proteasome. *Proc Natl Acad Sci USA* 97(3): 1032-7.

LANGER C, HUANG Y, CULLEN P, WIESENHUTTER B, MAHLEY RW, ASSMANN G, VON ECKARDSTEIN A, (2000). Endogenous apolipoprotein E modulates cholesterol efflux and cholesteryl ester hydrolysis mediated by high-density lipoprotein-3 and lipid-free apolipoproteins in mouse peritoneal macrophages. *J Mol Med* 78(4): 217-227.

LANGMANN T, KLUCKEN J, REIL M, LIEBISCH G, LUCIANI MF, CHIMINI G, KAMINSKI WE, SCHMITZ G, (1999). Molecular cloning of the human ATP binding cassette transporter 1 (hABC1): evidence for sterol-dependant regulation in macrophages. *Biochem Biophys Res Commun* 257(1): 29-33.

LARRICK JW, FISCHER DG, ANDERSON SJ, KOREN HS, (1980). Characterization of a human macrophage-like cell line stimulated in vitro: a model of macrophage functions. *J Immunol* 125(1):6-12.

LEBRIN F, CHAMBAZ EM, BIANCHINI L, (2001). A role for protein kinase CK2 in cell proliferation: evidence using a kinase-inactive mutant of CK2 catalytic subunit alpha. *Oncogene* 20(16): 2010-22.

LEE JC, LAYDON JT, MCDONNELL PC, GALLAGHER TF, KUMAR S, GREEN D, MCNULTY D, BLUMENTHAL MJ, HEYS JR, LANDVATTER SW, ET AL., (1994). A protein kinase involved in the regulation of inflammatory cytokine biosynthesis. *Nature* 372(6508): 739-46.

LEE HY, SUH YA, ROBINSON MJ, CLIFFORD JL, HONG WK, WOODGETT JR, COBB MH, MANGELSDORF DJ, KURIE JM, (2000). Stress pathway activation induces phosphorylation of retinoid X receptor. *J Biol Chem* 275(41): 32193-9.

LEE MH, LU K, PATEL SB, (2001). Genetic basis of sitosterolemia. Curr Opin Lipidol 12(2): 141-149.

LEEVERS SJ, VANHAESEBROECK B, WATERFIELD MD, (1999). Signalling through phosphoinositide 3-kinases: the lipids take centre stage. Curr Opin Cell Biol 11(2): 219-25.

LEHMANN JM, KLIEWER SA, MOORE LB, SMITH-OLIVER TA, OLIVER BB, SU JL, SUNDSETH SS, WINEGAR DA, BLANCHARD DE, SPENCER TA, WILSON TM, (1997). Activation of the nuclear receptor LXR by oxysterols defines a new hormone response pathway. *J Biol Chem* 272(6): 3137-3140.

LESLIE NR, DOWNES CP, (2002). PTEN: The down side of PI 3-kinase signalling. *Cell Signal* 14(4): 285-95.

LI F, HUI DY, (1997). Modified low density lipoprotein enhances the secretion of bile salt-stimulated cholesterol esterase by human monocyte-macrophages. Species-specific difference in macrophage cholesteryl ester hydrolase. *J Biol Chem* 272(45): 28666-71.

LI AC, BROWN KK, SILVESTRE MJ, WILLSON TM, PALINSKI W, GLASS CK, (2000). Peroxisome proliferators-activated receptor γ ligands inhibit development of atherosclerosis in LDL receptordeficient mice. *J Clin Invest* 106: 523-531.

LI AC, GLASS CK, (2002). The macrophage foam cell as a target for therapeutic intervention. Nat Med 8(11): 1235-42.

LI X, KAN HY, LAVRENTIADOU M, KRIEGER ZANNIS V, (2002a). Reconstituted discoidal ApoEphospholipid particles are ligands for the scavenger receptor BI. The amino-terminal 1-165 domain of ApoE suffices for receptor binding. *J Biol Chem* 277(24): 21149-21157.

LI Y, BOLTEN C, BHAT BG, WOODRING-DIETZ J, LI S, PRAYAGA SK, XIA C, LALA DS, (2002b). Induction of human liver X receptor alpha gene expression via an autoregulatory loop mechanism. *Mol Endocrinol* 16(3): 506-514.

LI P, LEE H, GUO S, UNTERMAN TG, JENSTER G, BAI W, (2003). AKT-independent protection of prostate cancer cells from apoptosis mediated through complex formation between the androgen receptor and FKHR. Mol Cell Biol 23(1): 104-18.

LI J, CHEN H, TANG MS, SHI X, AMIN S, DESAI D, COSTA M, HUANG C, (2004). PI-3K and Akt are mediators of AP-1 induction by 5-MCDE in mouse epidermal Cl41 cells. *J Cell Biol* 165(1): 77-86.

LIANG YC, HUANG YT, TSAI SH, LIN-SHIAU SY, CHEN CF, LIN JK, (1999). Suppression of inducible cyclooxygenase and inducible nitric oxide synthase by apigenin and related flavonoids in mouse macrophages. *Carcinogenesis* 20(10): 1945-52.

LIAO H, LANGMANN T, SCHMITZ G, ZHU Y, (2002). Native LDL upregulation of ATP-binding cassette transporter-1 in human vascular endothelial cells. *Arterioscler Thronb Vasc Biol* 22(1): 127-132.

LIAO X, THRASHER JB, HOLZBEIERLEIN J, STANLEY S, LI B, (2004). Glycogen synthase kinase-3beta activity is required for androgen-stimulated gene expression in prostate cancer. Endocrinology 145(6): 2941-9.

LIBEU CP, LUND-KATZ S, PHILLIPS MC, WEHRLI S, HERNAIZ MJ, CAPILA I, LINHARDT RJ, RAFFAI RL, NEWHOUSE YM, ZHOU F, WEISGRABER KH, (2001). New insights into the heparan sulfate proteoglycan-binding activity of apolipoprotein E. *J Biol Chem* 276(42): 39138-39144.

LIN A, FROST J, DENG T, SMEAL T, AL-ALAWI N, KIKKAWA U, HUNTER T, BRENNER D, KARIN M, (1992). Casein kinase II is a negative regulator of c-Jun DNA binding and AP-1 activity. *Cell* 70(5): 777-89.

LIN CY, LUCAS M, MAZZONE T, (1998). Endogenous apoE expression modulates HDL<sub>3</sub> binding to macrophages. *J Lipid Res* 39(2): 293-301.

LIN CY, DUAN H, MAZZONE T, (1999). Apolipoprotein E-dependant cholesterol efflux from macrophages: kinetics study and divergent mechanisms for endogenous versus exogenous apolipoprotein E. *J Lipid Res* 40(9): 1618-1626.

LIN CY, HUANG ZH, MAZZONE T, (2001a). Interaction with proteoglycans enhances the sterol efflux produced by endogenous expression of macrophage apoE. *J Lipid Res* 42(7): 1125-1133.

LIN HK, YEH S, KANG HY, CHANG C, (2001b). Akt suppresses androgen-induced apoptosis by phosphorylating and inhibiting androgen receptor. *Proc Natl Acad Sci USA* 98(13): 7200-5.

LINCOLN J, HOYLE CHV, BURNSTOCK G, (1997). Nitric oxide in health and disease.

LINTON MF, ATKINSON JB, FAZIO S, (1995). Prevention of atherosclerosis in apolipoprotein Edeficient mice by bone marrow transplantation. *Science* 267(5200): 1034-1037.

LINTON MF, HASTY AH, BABAEV VR, FAZIO S, (1998). Hepatic apoE expression is required for remnant lipoprotein clearance in the absence of the low density lipoprotein receptor. *J Clin Invest* 101(8): 1726-1736.

LINTON MF, BABAEV VR, GLEAVES LA, FAZIO S, (1999). A direct role for the macrophage low-density lipoprotein receptor in atherosclerotic lesion formation. *J Biol Chem* 274: 19204-19210.

LLORENS F, GARCIA L, ITARTE E, GOMEZ N, (2002). Apigenin and LY294002 prolong EGF-stimulated ERK1/2 activation in PC12 cells but are unable to induce full differentiation. *FEBS Lett* 510(3): 149-53.

LITCHFIELD DW, (2003). Protein kinase CK2: structure, regulation and role in cellular decisions of life and death. *Biochem J* 369(1): 1-15.

LIU MY, WU MC, (1992). Induction of human monocyte cell line U937 differentiation and CSF-1 production by phorbol ester. *Exp Hematol* 20(8): 974-9.

LIU H, PERLMAN H, PAGLIARI LJ, POPE RM, (2001). Constitutively activated Akt-1 is vital for the survival of human monocyte-differentiated macrophages. Role of McI-1, independent of nuclear factor (NF)-kappaB, Bad, or caspase activation. *J Exp Med* 194(2): 113-26.

LIU L, BORTNICK AE, NICKEL M, DHANASEKARAN P, SUBBAIAH PV, LUND-KATZ S, ROTHBLAT GH, PHILLIPS MC, (2003). Effects of apolipoprotein A-I on ABCA1-mediated efflux of macrophage phospholipid and cholesterol: Formation of nascent HDL particles. *J Biol Chem* 278(44): 42976-84.

LODIE TA, SAVEDRA R JR, GOLENBOCK DT, VAN BEVEREN CP, MAKI RA, FENTON MJ, (1997). Stimulation of macrophages by lipopolysaccharide alters the phosphorylation state, conformation, and function of PU.1 via activation of casein kinase II. *J Immunol* 158(4): 1848-56.

LONARD DM, NAWAZ Z, SMITH CL, O'MALLEY BW, (2000). The 26S proteasome is required for estrogen receptor-alpha and coactivator turnover and for efficient estrogen receptor-alpha transactivation. *Mol Cell* 5(6): 939-48.

LUC G, DUCIMETIERE P, BARD JM, ARVEILER D, EVANS A, CAMBIEN F, FRUCHART JC, FIEVET C, (1997). Distribution of apolipoprotein E between apo B- and non apo B-containing lipoproteins according to apo E phenotype. *Atherosclerosis* 131(2): 257-62.

LUCAS M, MAZZONE T, (1996). Cell surface proteoglycans modulate net synthesis and secretion of macrophage aplipoprotein E. *J Biol Chem* 271(23): 13454-13460.

LUISI BF, XU WX, OTWINOWSKI Z, FREEDMAN LP, YAMAMOTO KR, SIGLER PB, (1991). Crystallographic analysis of the interaction of the glucocorticoid receptor with DNA. *Nature* 352(6335): 497-505.

LUNDHOLM L, MOVERARE S, STEFFENSEN KR, NILSSON M, OTSUKI M, OHLSSON C, GUSTAFSSON JA, DAHLMAN-WRIGHT K, (2004). Gene expression profiling identifies liver X receptor alpha as an estrogen-regulated gene in mouse adipose tissue. *J Mol Endocrinol* 32(3): 879-92.

LUO Y, TALL AR, (2000). Sterol upregulation of human CETP expression in vitro and in transgenic mice by an LXR element. *J Clin Invest* 105(4): 513-520.

LUSIS AJ, (2000). Atherosclerosis. Nature 407(6801): 233-41.

LUTGENS E, GORELIK L, DAEMEN MJ. DE MUINCK ED, GREWAL IS, KOTELIANSKY VE, FLAVELL RA, (1999). *Nat Med* 5: 1313-1316.

MA W, LIM W, GEE K, AUCOIN S, NANDAN D, KOZLOWSKI M, DIAZ-MITOMA F, KUMAR A, (2001). The p38 mitogen-activated kinase pathway regulates the human interleukin-10 promoter via the activation of Sp1 transcription factor in lipopolysaccharide-stimulated human macrophages. *J Biol Chem* 276(17): 13664-74.

MACK KD, WEI R, ELBAGARRI A, ABBEY N, MCGRATH MS, (1998). A novel method for DEAE-dextran mediated transfection of adherent primary cultured human macrophages. *J Immunol Methods* 211(1-2): 79-86.

MAEHAMA T, DIXON JE, (1998). The tumor suppressor, PTEN/MMAC1, dephosphorylates the lipid second messenger, phosphatidylinositol 3,4,5-trisphosphate. *J Biol Chem* 273(22): 13375-8.

MAHLEY R, INNERARITY TL, PITAS RE, WEISGRABER KH, BROWN JH, GROSS E, (1977). Inhibition of lipoprotein binding to cell surface receptors of fibroblasts following selective modification of arginyl residues in arginine-rich and B apoproteins. *J Biol Chem* 252(20): 7279-7287.

MAHLEY R, INNERARITY TL, RALL SC, WEISGRABER KH, (1984). Plasma lipoproteins: apolipoprotein structure and function. *J Lipid Res* 25(12): 1277-1294.

MAHLEY R, (1988). Apolipoprotein E: cholesterol transport protein with expanding role in cell biology. *Science* 240(4852): 622-630.

MAHLEY R, HUANG Y, (1999). Apolipoprotein E: from atherosclerosis to Alzheimers disease and beyond. *Curr Opin Lipidol* 10(3): 207-217.

MAHLEY RW, HUANG Y, RALL SC, (1999). Pathogenesis of type III hyperlipoproteinemia (dysbetalipoproteinemia). Questions, quandaries, and paradoxes. *J Lipid Res* 40(11): 1933-49.

MAHLEY RW, JI ZS, (1999). Remnant lipoprotein metabolism: key pathways involving cell surface heparan sulfate proteoglycans and apolipoprotein E. J Lipid Res 40(1): 1-16.

MAJACK RA, CASTLE CK, GOODMAN LV, WEISGRABER KH, MAHLEY RW, SHOOTER EM, GEBICKE-HAERTER PJ, (1988). J Cell Biol 107: 1207-1213.

Mak PA, LAFFITTE BA, DESRUMAUX C, JOSEPH SB, CURTISS LK, MANGELSDORF DJ, TONTONOZ P, EDWARDS PA, (2002a). Regulated expression of the apolipoprotein E/C-I/C-IV/C-II gene cluster in murine and human macrophages. A critical role for nuclear liver X receptors  $\alpha$  and  $\beta$ . *J Biol Chem* 277(35): 31900-31908.

MAK PA, KAST-WOELBERN HR, ANISFELD AM, EDWARDS PA, (2002b). Identification of PLTP as an LXR target gene and apoE as an FXR target gene reveals overlapping targets for the two nuclear receptors. *J Lipid Res* 43(12): 2037-2041.

MALDONADO E, ALLENDE JE, (1999). Phosphorylation of yeast TBP by protein kinase CK2 reduces its specific binding to DNA. *FEBS Lett* 443(3): 256-60.

MANGELSDORF DJ, EVANS RM, (1995). The RXR heterodimers and orphan receptors. Cell 83(6): 841-850.

MARSHALL CJ, (1995). Specificity of receptor tyrosine kinase signaling: transient versus sustained extracellular signal-regulated kinase activation. *Cell* 80(2): 179-85.

MASON DP, KENAGY RD, HASENSTAB D, BOWEN-POPE DF, SEIFERT RA, COATS S, HAWKINS SM, CLOWES AW, (1999). Matrix metalloproteinase-9 overexpression enhances vascular smooth muscle cell migration and alters remodeling in the injured rat carotid artery. *Circ Res* 85(12): 1179-85.

MAY M, MENGUS G, LAVIGNE AC, CHAMBON P, DAVIDSON I, (1996). Human TAF(II28) promotes transcriptional stimulation by activation function 2 of the retinoid X receptors. *EMBO J* 15(12): 3093-104.

MAZZONE T, PAPAGIANNES E, MAGNER J, (1986). Early kinetics of human macrophage apolipoprotein E synthesis and incorporation of carbohydrate precursors. *Biochim Biophys Acta* 875(2): 393-396.

MAZZONE T, GUMP H, DILLER P, GETZ G, (1987). Macrophage free cholesterol content regulates apolipoprotein E synthesis. *J Biol Chem* 262(24): 11657-11662.

MAZZONE T, BASHEERUDDIN K, POULOS C, (1989). Regulation of macrophage apolipoprotein E gene expression by cholesterol. *J Lipid Res* 30(7): 1055-1064.

MAZZONE T, BASHEERUDDIN K, (1991). Dissociated regulation of macrophage LDL receptor and apoE gene expression by sterol. *J Lipid Res* 32(3): 507-514.

MAZZONE T, PUSTELNIKAS L, REARDON C, (1992). Post-translational regulation of macrophage apoprotein E production. *J Biol Chem* 267(2): 1081-1087.

MAZZONE T, REARDON C, (1994). Expression of heterologous human apolipoprotein E by J774 macrophages enhances cholesterol efflux to HDL<sub>3</sub>. *J Lipid Res* 35(8): 1345-1353.

MAZZONE T, (1996). Apolipoprotein E secretion by macrophages: its potential physiological functions. *Curr Opin Lipidol* 7(5): 303-307.

MEAD JR, (2002). Studies on the regulation of macrophage lipoprotein lipase gene expression.

MEAD JR, IRVINE SA, RAMJI DP, (2002). Lipoprotein lipase: structure, function, regulation, and role in disease. J Mol Med 80(12): 753-69.

MEAD JR, HUGHES TR, IRVINE SA, SINGH NN, RAMJI DP, (2003). Interferon-gamma stimulates the expression of the inducible cAMP early repressor in macrophages through the activation of casein kinase 2. A potentially novel pathway for interferon-gamma-mediated inhibition of gene transcription. *J Biol Chem* 278(20): 17741-51.

MEDEMA RH, KOPS GJ, BOS JL, BURGERING BM, (2000). AFX-like Forkhead transcription factors mediate cell-cycle regulation by Ras and PKB through p27kip1. *Nature* 404(6779): 782-7.

MEGGIO F, PINNA LA, (2003). One-thousand-and-one substrates of protein kinase CK2? FASEB J 17(3): 349-68.

MEIER R, ALESSI DR, CRON P, ANDJELKOVIC M, HEMMINGS BA, (1997). Mitogenic activation, phosphorylation, and nuclear translocation of protein kinase B beta. J Biol Chem 272(48): 30491-7.

MENJU M, TAJIMA S, YAMAMOTO A, (1989). Expression of the apolipoprotein gene in a human macrophage-like cell line, THP-1. *J Biochem* 106(3): 505-510.

MENSENKAMP AR, JONG MC, VAN GOOR H, VAN LUYN MJ, BLOKS V, HAVINGA R, VOSHOL PJ, HOFKER MH, VAN DIJK KW, HAVEKES LM, KUIPERS F, (1999). Apolipoprotein E participates in the regulation of very low density lipoprotein-triglyceride secretion by the liver. *J Biol Chem* 274(50): 35711-35718.

MERKEL M, HEEREN J, DUDECK W, RINNINGER F, RADNER H, BRESLOW JL, GOLDBERG IJ, ZECHNER R, GRETEN H, (2002). Inactive lipoprotein lipase (LPL) alone increases selective cholesterol ester uptake *in vivo*, whereas in the presence of active LPL it also increases triglyceride hydrolysis and whole particle lipoprotein uptake. *J Biol Chem* 277: 7405-7411.

MEYER ME, GRONEMEYER H, TURCOTTE B, BOCQUEL MT, TASSET D, CHAMBON P, (1989). Steroid hormone receptors compete for factors that mediate their enhancer function. *Cell* 57(3): 433-42.

MCINERNEY EM, ROSE DW, FLYNN SE, WESTIN S, MULLEN TM, KRONES A, INOSTROZA J, TORCHIA J, NOLTE RT, ASSA-MUNT N, MILBURN MV, GLASS CK, ROSENFELD MG, (1998). Determinants of coactivator LXXLL motif specificity in nuclear receptor transcriptional activation. *Genes Dev* 12(21): 3357-68.

MCKENNA NJ, O'MALLEY BW, (2002). Combinatorial control of gene expression by nuclear receptors and coregulators. *Cell* 108(4): 465-74.

MCNEISH J, AIELLO RJ, GUYOT D, TURI T, GABEL C, ALDINGER C, HOPPE KL, ROACH ML, ROYER LJ, DE WET J, BROCCARDO C, CHIMINI G, FRANCONE OL, (2000). High density lipoprotein deficiency and foam cell accumulation in mice with targeted disruption of ATP-binding cassette transporter-1. *Proc Natl Acad Sci USA* 97(8): 4245-4250.

MIGLIACCIO A, PICCOLO D, CASTORIA G, DI DOMENICO M, BILANCIO A, LOMBARDI M, GONG W, BEATO M, AURICCHIO F, (1998). Activation of the Src/p21ras/Erk pathway by progesterone receptor via cross-talk with estrogen receptor. *EMBO J* 17(7): 2008-18.

MILLER SJ, LOU DY, SELDIN DC, LANE WS, NEEL BG, (2002). Direct identification of PTEN phosphorylation sites. FEBS Lett 528(1-3): 145-53.

MINAMIKAWA J, TANAKA S, YAMAUCHI M, INOUE D, KOSHIYAMA H, (1998). Potent inhibitory effect of troglitazone on carotid arterial wall thickness in type 2 diabetes. *J Clin Endocrinol Metab* 83(5): 1818-1820.

MINDEN A, LIN A, SMEAL T, DERIJARD B, COBB M, DAVIS R, KARIN M, (1994). c-Jun N-terminal phosphorylation correlates with activation of the JNK subgroup but not the ERK subgroup of mitogen-activated protein kinases. *Mol Cell Biol* 14(10): 6683-8.

MISTRY MJ, CLAY MA, KELLY ME, STEINER MA, HARMONY JAK, (1995). Apolipoprotein E restricts interleukin-dependant T-lymphocyte proliferation at the G1A/G1B boundary. *Cell Immunol* 160(1): 14-23.

MIYATA KS, MCCAW SE, PATEL HV, RACHUBINSKI RA, CAPONE JP, (1996). The orphan nuclear hormone receptor LXR alpha interacts with the peroxisome proliferator-activated receptor and inhibits peroxisome proliferator signaling. *J Biol Chem* 271(16): 9189-92.

MIYATA M, SMITH JD, (1996). Apolipoprotein E allele-specific antioxidant activity and effects on cytotoxicity by oxidative insults and beta-amyloid peptides. *Nat Genet* 14(1): 55-61.

MOORE KJ, ROSEN ED, FITZGERALD ML, RANDOW F, ANDERSSON LP, ALTSHULER D, MILSTONE DS, MORTENSEN RM, SPIEGELMAN BM, FREEMAN MW, (2001). The role of PPAR- $\gamma$  in macrophage differentiation and cholesterol uptake. *Nat Med* 7: 41-47.

MORIGUCHI T, KAWASAKI H, MATSUDA S, GOTOH Y, NISHIDA E, (1995). Evidence for multiple activators for stress-activated protein kinase/c-Jun amino-terminal kinases. Existence of novel activators. *J Biol Chem* 270(22): 12969-72.

MULLICK AE, DECKELBAUM RJ, GOLDBERG IJ, AL-HAIDERI M, RUTLEDGE JC, (2002). Apolipoprotein E and lipoprotein lipase increase triglyceride-rich particle binding but decrease particle penetration in arterial wall. *Arterioscler Thromb Vasc Biol* 22(12): 2080-2085.

MURTHY S, BORN E, MATHUR SN, FIELD FJ, (2002). LXR/RXR activation enhances basolateral efflux of cholesterol in CaCo-2 cells. *J Lipid Res* 43(7): 1054-1064.

MUSCAT GE, WAGNER BL, HOU J, TANGIRALA RK, BISCHOFF ED, ROHDE P, PETROWSKI M, LI J, SHAO G, MACONDRAY G, SCHULMAN IG, (2002). Regulation of cholesterol homeostasis and lipid metabolism in skeletal muscle by liver X receptors. *J Biol Chem* 277(43): 40722-8.

MYERS MP, PASS I, BATTY IH, VAN DER KAAY J, STOLAROV JP, HEMMINGS BA, WIGLER MH, DOWNES CP, TONKS NK, (1998). The lipid phosphatase activity of PTEN is critical for its tumor supressor function. *Proc Natl Acad Sci USA* 95(23): 13513-8.

NAGY L, TONTONOZ P, ALVAREZ JG, CHEN H, EVANS RM, (1998). Oxidized LDL regulates macrophage gene expression through ligand activation of PPARgamma. *Cell* 93(2): 229-240.

NAKASHIMA Y, PLUMP AS, RAINES EW, BRESLOW JL, ROSS R, (1994). ApoE-deficient mice develop lesions of all phases of atherosclerosis throughout the arterial tree. *Arterioscler Thromb* 14(1): 133-140.

NAKASHIMA Y, RAINES EW, PLUMP ES, BRESLOW JL, ROSS, R, (1998). Upregulation of VCAM-I and ICAM-I at atherosclerosis-prone sites on the endothelium in the apoE-deficient mouse. *Arterioscler Thromb Vasc Biol* 18: 842-851.

NAVAB M, BERLINER JA, WATSON AD, HAMA SY, TERRITO MC, LUSIS AJ, SHIH DM, VAN LENTEN BJ, FRANK JS, DEMER DEMER LL, EDWARDS PA, FOGELMAN AM, (1996). The yin and yang of oxidation in the development of the fatty streak. A review based on the 1994 George Lyman Duff Memorial Lecture. *Arterioscler Thromb Vasc Biol* 16(7): 831-842.

NAZARETH LV, WEIGEL NL, (1996). Activation of the human androgen receptor through a protein kinase A signaling pathway. *J Biol Chem* 271(33): 19900-7.

NEWMAN T, DAWSON P, RUDEL L, WILLIAMS D, (1985). Quantitation of apolipoprotein E mRNA in the liver and peripheral tissues of non-human primates. *J Biol Chem* 260(4): 2452-2457.

NIEFIND K, GUERRA B, PINNA LA, ISSINGER OG, SCHOMBURG D, (1998). Crystal structure of the catalytic subunit of protein kinase CK2 from Zea mays at 2.1 A resolution. *EMBO J* 17(9): 2451-62.

NIEMEIER A, GAFVELS M, HEEREN J, MEYER N, ANGELIN B, BEISIEGEL U, (1996). VLDL receptor mediates the uptake of human chylomicron remnants in vitro. *J Lipid Res* 37(8): 1733-42.

NIKOULIN IR, CURTISS LK, (1998). An apolipoprotein synthetic peptide targets to lipoproteins in plasma and mediates both cellular lipoprotein interactions in vitro and acute clearance of cholesterol-rich lipoproteins in vivo. *J Clin Invest* 101: 223-234.

NISHIGUCHI M, MATSUMOTO M, TAKAO T, HOSHINO M, SHIMONISHI Y, TSUJI S, BEGUM NA, TAKEUCHI O, AKIRA S, TOYOSHIMA K, SEYA T, (2001). Mycoplasma fermentans lipoprotein M161Ag-induced cell activation is mediated by Toll-like receptor 2: role of N-terminal hydrophobic portion in its multiple functions. *J Immunol* 166(4): 2610-6.

NOLTE RT, ATKINSON D, (1992). Conformational analysis of apolipoprotein A-I and E-3 based on primary sequence and circular dichroism. *Biophys J* 63(5): 1221-1239.

O'BRIEN KD, DEEB SS, FERGUSON M, MCDONALD TO, ALLEN MD, ALPERS CE, CHAIT A, (1994). Apolipoprotein E localisation in human coronary atherosclerotic plaques by in situ hybridisation and immunohistochemistry and comparison with lipoprotein lipase. *Am J Pathol* 144(3): 538-548.

OBENAUER JC, CANTLEY LC, YAFFE MB, (2003). Scansite 2.0: Proteome-wide prediction of cell signaling interactions using short sequence motifs. *Nucleic Acids Res* 31(13): 3635-41.

OBERKOFLER H, SCHRAML E, KREMPLER F, PATSCH W, (2003). Potentiation of liver X receptor transcriptional activity by peroxisome-proliferator-activated receptor gamma co-activator 1 alpha. *Biochem J* 371(1): 89-96.

OBUNIKE JC, PILLARISETTI S, PAKA L, KAKO Y, BUTTERI MJ, HO YY, WAGNER WD, YAMADA N, MAZZONE T, DECKELBAUM RJ, GOLDBERG IJ, (2000). The heparin-binding proteins apolipoprotein E and lipoprotein lipase enhance cellular proteoglycan production. *Arterioscler Thromb Vasc Biol* 20: 111-118.

OGBONNA G, THERIAULT A, ADELI K, (1993). Hormonal regulation of human apolipoprotein E gene expression in HepG2 cells. *Int J Biochem* 25(5):635-40.

OHH M, SMITH CA, CARPENITO C, TAKEI F, (1994). Regulation of intercellular adhesion molecule-1 gene expression involves multiple mRNA stabilization mechanisms: effects of interferon-gamma and phorbol myristate acetate. Blood 84(8): 2632-9.

OIKAWA T, YAMADA T, (2003). Molecular biology of the Ets family of transcription factors. *Gene* 303: 11-34.

OKANO Y, MACY M, CARDIN AD, HARMONY JA, (1985). Suppression of lymphocyte activation by plasma lipoproteins: modulation by cell number and type. *Exp Cell Biol* 53(4): 199-212.

OLIVEIRA HC, MA L, MILNE R, MARCOVINA SM, INAZU A, MABUCHI H, TALL AR, (1997). Cholesteryl ester transfer protein activity enhances plasma cholesteryl ester formation. Studies in CETP transgenic mice and human genetic CETP deficiency. *Arterioscler Thromb Vasc Biol* 17(6): 1045-1052.

OLSSON I, GULLBERG U, IVHED I, NILSSON K, (1983). Induction of differentiation of the human histiocytic lymphoma cell line U-937 by 1 alpha,25-dihydroxycholecalciferol. Cancer Res 43(12 Pt 1): 5862-7.

ONATE SA, TSAI SY, TSAI MJ, O'MALLEY BW, (1995). Sequence and characterization of a coactivator for the steroid hormone receptor superfamily. *Science* 270(5240): 1354-7.

ONO Y, FUJII T, OGITA K, KIKKAWA U, IGARASHI K, NISHIZUKA Y, (1988). The structure, expression, and properties of additional members of the protein kinase C family. *J Biol Chem* 263(14): 6927-32.

ORAM JF, LAWN RM, GARVER MR, WADE DP, (2000). ABCA1 is the cAMP-inducible receptor that mediates cholesterol secretion from macrophages. *J Biol Chem* 275(44): 34508-34511.

ORAM JF, VAUGHAN AM, (2000). ABC1-mediated transport of cellular cholesterol to HDL and apolipoproteins. *Curr Opin Lipidol* 11(3): 253-260.

ORSO E, BROCCARDO C, KAMINSKI WE, BOTTCHER A, LIEBISCH G, DROBNIK W, GOTZ A, CHAMBENOIT O, DIEDRICH W, LANGMANN T, SPRUSS T, LUCIANI MF, ROTHE G, LACKNER KJ, CHIMINI G, SCHMITZ G, (2000). Transport of lipids from Golgi to plasma membrane is defective in Tangier disease patients and ABC1-deficient mice. *Nat Genet* 24(2): 192-196.

OU J, TU H, SHAN B, LUK A, DEBOSE-BOYD RA, BASHMAKOV Y, GOLDSTEIN JL, BROWN MS, (2001). Unsaturated fatty acids inhibit transcription of the sterol regulatory element-binding protein-1c (SREBP-1c) gene by antagonising ligand-dependent activation of the LXR. *Proc Natl Acad Sci USA* 98(11): 6027-6032.

PADMANABHA R, CHEN-WU JL, HANNA DE, GLOVER CV, (1990). Isolation, sequencing, and disruption of the yeast CKA2 gene: casein kinase II is essential for viability in Saccharomyces cerevisiae. *Mol Cell Biol* 10(8): 4089-99.

PAIK YK, CHANG D, REARDON C, WALKER M, TAXMAN E, TAYLOR J, (1988). Identification and characterisation of transcriptional regulatory regions associated with expression of the human apolipoprotein E gene. *J Biol Chem* 263(26): 13340-13349.

PAIK YK, REARDON CA, TAYLOR JM, CHOI BK, (1995). Characterization of an upstream regulatory sequence and its binding protein in the mouse apolipoprotein E gene. *Biochim Biophys Acta* 1262(2-3): 124-32.

PAKA L, KAKO Y, OBUNIKE J, PILLARISETTI S, (1988). Apolipoprotein E containing high density lipoprotein stimulates endothelial cell production of heparan sulfate rich in biologically active heparin-like domains. *J Biol Chem* 274(8): 4816-4823.

PALINSKI W, ORD VA, PLUMP AS, BRESLOW JL, STEINBERG D, WITZUM JL, (1994). ApoE-deficient mice are a model of lipoprotein oxidation in atherogenesis: demonstration of oxidation-specific epitopes in lesions and high titers of autoantibodies to malondialdehyde-lysine in serum. *Arterioscler Thromb* 14(4): 605-616.

PARK JW, JANG MA, LEE YH, PASSANITI A, KWON TK, (2001). p53-independent elevation of p21 expression by PMA results from PKC-mediated mRNA stabilization. *Biochem Biophys Res Commun* 280(1): 244-8.

PARKER PJ, KOUR G, MARAIS RM, MITCHELL F, PEARS C, SCHAAP D, STABEL S, WEBSTER C, (1989). Protein kinase C-a family affair. *Mol Cell Endocrinol* 65(1-2): 1-11.

PATRONE C, MA ZQ, POLLIO G, AGRATI P, PARKER MG, MAGGI A, (1996). Cross-coupling between insulin and estrogen receptor in human neuroblastoma cells. *Mol Endocrinol* 10(5): 499-507.

PEET DJ, TURLEY SD, MA WZ, JANOWSKI BA, LOBACCARO JMA, HAMMER RE, MANGELSDORF DJ, (1998). Cholesterol and bile acid metabolism are imapaired in mice lacking the nuclear oxysterol receptor LXR $\alpha$ . *Cell* 93(5): 693-704.

PENTIKAINEN MO, OKSJOKI R, OORNI K, KOVANEN PT, (2002). Lipoprotein lipase in the arterial wall. Arterioscler Thromb Vasc Biol 22(2): 211-217.

PEPE MG, CURTISS LK, (1986). Apolipoprotein E is a biologically active constituent of the normal immunoregulatory lipoprotein, LDL-In. *J Immunol* 136(10): 3716-3723.

PEREZ A, THUILLARD JL, BENTZEN CL, NIESOR EJ, (2003). Expression of nuclear receptors and apoE secretion during the differentiation of monocytic THP-1 cells into macrophages. *Cell Biol Toxicol* 19(2): 95-105.

PERON P, RAHMANI M, ZAGAR Y, DURAND-SCHNEIDER AM, LARDEUX B, BERNUAU D, (2001). Potentiation of Smad transactivation by Jun proteins during a combined treatment with epidermal growth factor and transforming growth factor-beta inrat hepatocytes. Role of phosphatidylinositol 3-kinase-induced AP-1 activation. *J Biol Chem* 276(13):10524-31.

PIEDRAHITA JA, ZHANG SH, HAGAMAN JR, OLIVER PM, MAEDA N, (1992). Generation of mice carrying a mutant apolipoprotein E gene inactivated by gene targeting in embryonic stem cells. *Proc Natl Acad Sci USA* 89(10): 4471-4475.

PINNA LA, (1990). Casein kinase 2: an 'eminence grise' in cellular regulation? *Biochim Biophys Acta* 1054(3): 267-84.

PINNA LA, MEGGIO F, (1997). Protein kinase CK2 ("casein kinase-2") and its implication in cell division and proliferation. *Prog Cell Cycle Res* 3: 77-97.

PINNA LA, MEGGIO F, SARNO S, (1997). Protein kinase CK2 ("casein kinase-2") and cell signalling. *Biochemistry of cell membranes*: 15-27. PLOSCH T, KOK T, BLOKS VW, SMIT MJ, HAVINGA R, CHIMINI G, GROEN AK, KUIPERS F, (2002). Increased hepatobiliary and fecal cholesterol excretion upon activation of the liver X receptor is independent of ABCA1. *J Biol Chem* 277(37): 33870-33877.

PLUMP AS, SMITH JD, HAYEK T, AALTO-SETALA K, WALSH A, VERSTUYFT JG, RUBIN EM, BRESLOW JL, (1992). Severe hypercholesterolemia and atherosclerosis in apolipoprotein E-deficient mice created by homologous recombination in ES cells. *Cell* 71(2): 343-53.

PLUMP AS, MASUCCI-MAGOULAS L, BRUCE C, BISGAIER CL, BRESLOW JL, TALL AR, (1999). Increased atherosclerosis in ApoE and LDL receptor gene knock-out mice as a result of human cholesteryl ester transfer protein transgene expression. *Arterioscler Thromb Vasc Biol* 19: 1105-1110.

Pozo-GUISADO E, LORENZO-BENAYAS MJ, FERNANDEZ-SALGUERO PM, (2004). Resveratrol modulates the phosphoinositide 3-kinase pathway through an estrogen receptor alpha-dependent mechanism: relevance in cell proliferation. *Int J Cancer* 109(2): 167-73.

PRESCOTT MF, SAWYER WK, VON LINDEN-REED J, JEUNE M, CHOU M, CAPLAN SL, JENG AY, (1999). Effect of matrix metalloproteinase inhibition on progression of atherosclerosis and aneurysm in LDL receptor-deficient mice overexpressing MMP-3, MMP-12, and MMP-13 and on restenosis in rats after balloon injury. *Ann N Y Acad Sci* 878: 179-90.

PUIGSERVER P, WU Z, PARK CW, GRAVES R, WRIGHT M, SPIEGELMAN BM, (1998). A cold-inducible coactivator of nuclear receptors linked to adaptive thermogenesis. *Cell* 92(6): 829-39.

PULVERER BJ, KYRIAKIS JM, AVRUCH J, NIKOLAKAKI E, WOODGETT JR, (1991). Phosphorylation of cjun mediated by MAP kinases. *Nature* 353(6345): 670-4.

QIAO JH, TRIPATHI J, MISHRA NK, CAI Y, TRIPATHI S, WANG XP, IMES S, FISHBEIN MC, CLINTON SK, LIBBY P, LUIS AJ, RAJAVASHISTH TB, (1997). Role of macrophage colony-stimulating factor in atherosclerosis: studies of osteopetrotic mice. *Am J Pathol* 150(5): 1687-1699.

QIU Q, YANG M, TSANG BK, GRUSLIN A, (2004). EGF-induced trophoblast secretion of MMP-9 and TIMP-1 involves activation of both PI3K and MAPK signalling pathways. *Reproduction* 128(3): 355-63.

QUINET EM, SAVIO DA, HALPERN AR, CHEN L, MILLER CP, NAMBI P, (2004). Gene-selective modulation by a synthetic oxysterol ligand of the liver X receptor. *J Lipid Res* 45(10):1929-42.

RALL, SC, WEISGRABER KH, (1982). Structural basis for receptor binding heterogeneity of apolipoprotein E from type III hyperlipoproteinemic subjects. *Proc Natl Acad Sci USA* 79(15): 4696-4700.

RALL SC, WEISGRABER KH, MAHLEY R, (1982). Human apolipoprotein E. The complete amino acid sequence. *J Biol Chem* 257(8): 4171-4178.

RAMEH LE, CANTLEY LC, (1999). The role of phosphoinositide 3-kinase lipid products in cell function. *J Biol Chem* 274(13): 8347-50.

RAO KM, (2001). MAP kinase activation in macrophages. J Leukoc Biol 69(1): 3-10.

RAUSSENS V, FISHER CA, GOORMAGHTIGH E, RYAN RO, RUYSSCHAERT JM, (1998). The low density lipoprotein receptor active conformation of apolipoprotein E. Helix organisation in N-terminal domain-phospholipid disc particles. *J Biol Chem* 273(40): 25825-25830.

RAY A, PREFONTAINE KE, (1994). Physical association and function antagonism between the p65 subunit of transcription factor NF-kB and the glucocorticoid receptor. *Proc Natl Acad Sci USA* 91(2): 752-756.

REDDICK RL, ZHANG SH, MAEDA N, (1994). Atherosclerosis in mice lacking apoE: evaluation of lesional development and progression. *Arterioscler Thromb* 14(1): 141-147.

REES D, SLOANE T, JESSUP W, DEAN R, KRITHARIDES L, (1999). Apolipoprotein A-1 stimulates secretion of apolipoprotein E by foam cell macrophages. *J Biol Chem* 274(39): 27925-27933.

RENA G, GUO S, CICHY SC, UNTERMAN TG, COHEN P, (1999). Phosphorylation of the transcription factor forkhead family member FKHR by protein kinase B. *J Biol Chem* 274(24): 17179-83.

REPA JJ, LIANG G, OU J, BASHMAKOV Y, LOBACCARO JM, SHIMOMURA I, SHAN B, BROWN MS, GOLDSTEIN JL, MANGELSDORF DJ, (2000a). Regulation of mouse and sterol regulatory element-binding protein-1c (SREBP-1c) by oxysterol receptors LXR $\alpha$  and LXR $\beta$ . *Genes Dev* 14(22): 2819-2830.

REPA JJ, MANGELSDORF DJ, (2000). The role of orphan nuclear receptors in the regulation of cholesterol homeostasis. *Annu Rev Cell Dev Biol* 16: 459-481.

REPA JJ, TURLEY SD, LOBACCARO JA, MEDINA J, LI L, LUSTIG K, SHAN B, HEYMAN RA, DIETSCHY JM, MANGELSDORF DJ, (2000b). Regulation of absorption and ABC1-mediated efflux of cholesterol by RXR heterodimers. *Science* 289(5484): 1524-1529.

REPA JJ, BERGE KE, POMAJZL C, RICHARDSON JA, HOBBS H, MANGELSDORF DJ, (2002). Regulation of ATP-binding cassette sterol transporters ABCG5 and ABCG8 by the liver X receptors  $\alpha$  and  $\beta$ . *J Biol Chem* 277(21): 18793-18800.

REYLAND ME, RALL SC, (1991). Suppression of cAMP-mediated signal transduction in mouse adrenocortical cells which express apolipoprotein E. *J Biol Chem* 266: 21099-21104.

RICOTE M, LI AC, WILLSON TM, KELLY CJ, GLASS CK, (1998). The peroxisome proliferator-activated receptor-γ is a negative regulator of macrophage activation. *Nature* 391(6662): 79-82.

RIDDELL DR, SHEIKH S, JAMES RW, OWEN JS, (1996). Native immunoaffinity-isolated apolipoprotein E-containing high-density lipoprotein particles inhibit platelet aggregation. *Biochem Soc Trans* 24(3): 454S.

RIDDELL DR, GRAHAM A, OWEN JS, (1997). Apolipoprotein E inhibits platelet aggregation through the L-arginine: nitric oxide pathway. Implications for vascular disease. *J Biol Chem* 272(1): 89-95.

RIDDELL DR, VINOGRADOV DV, STANNARD AK, CHADWICK N, OWEN JS, (1999). Identification and characterisation of LRP8 (apoER2) in human blood platelets. *J Lipid Res* 40(10): 1925-1930.

RIDDELL DR, SUN XM, STANNARD AK, SOUTAR AK, OWEN JS, (2001). Localization of apolipoprotein E receptor 2 to caveolae in the plasma membrane. *J Lipid Res* 42(6): 998-1002.

RIGOTTI A, TRIGATTI BL, PENMAN M, RAYBURN H, HERZ J, KRIEGER M, (1997). A targeted mutation in the murine gene encoding the high density lipoprotein (HDL) receptor scavenger receptor class B type I reveals its key role in HDL metabolism. *Proc Natl Acad Sci USA* 94 (23): 12610-12615.

RINNINGER F, KAISER T, MANN WA, MEYER N, GRETEN H, BEISIEGEL U, (1998). Lipoprotein lipase mediates an increase in the selective uptake of high density lipoprotein-associated cholesteryl esters by hepatic cells in culture. *J Lipid Res* 39(7): 1335-1348.

RIVAL Y, BENETEAU N, CHAPUIS V, TAILLANDIER T, LESTIENNE F, DUPONT-PASSELAIGUE E, PATOISEAU JF, COLPAERT FC, JUNQUERO D, (2004). Cardiovascular drugs inhibit MMP-9 activity from human THP-1 macrophages. DNA Cell Biol 23(5): 283-92.

ROCHETTE-EGLY C, GAUB MP, LUTZ Y, ALI S, SCHEUER I, CHAMBON P, (1992). Retinoic acid receptorbeta: immunodetection and phosphorylation on tyrosine residues. *Mol Endocrinol* 6(12): 2197-2209.

ROCHETTE-EGLY C, OULAD-ABDELGHANI M, STAUB A, PFISTER V, SCHEUER I, CHAMBON P, GAUB MP, (1995). Phosphorylation of the retinoic acid receptor-alpha by protein kinase A. *Mol Endocrinol* 9(7): 860-71.

ROCHETTE-EGLY C, ADAM S, ROSSIGNOL M, EGLY JM, CHAMBON P, (1997). Stimulation of RAR alpha activation function AF-1 through binding to the general transcription factor TFIIH and phosphorylation by CDK7. *Cell* 90(1): 97-107.

ROCHETTE-EGLY C, (2003). Nuclear receptors: integration of multiple signalling pathways through phosphorylation. *Cell Signal* 15(4):355-66.

ROGATSKY I, WAASE CL, GARABEDIAN MJ, (1998). Phosphorylation and inhibition of rat glucocorticoid receptor transcriptional activation by glycogen synthase kinase-3 (GSK-3). Species-specific differences between human and rat glucocorticoid receptor signaling as revealed through GSK-3 phosphorylation. *J Biol Chem* 273(23): 14315-21.

Ross R, (1999). Atherosclerosis-An anti-inflammatory disease. N Eng J Med 340(2): 115-126.

ROSENFELD ME, BUTLER S, ORD VA, LIPTON BA, DYER CA, CURTISS LK, PALINSKI W, WITZUM JL, (1993). Abundant expression of apoprotein E by macrophages in human and rabbit atherosclerotic lesions. *Arterioscler Thromb* 13(9): 1382-1389.

ROTHBLAT GH, DE LA LLERA-MOYA M, ATGER V, KELLNER-WEIBEL G, WILLIAMS DL, PHILLIPS MC, (1999). Cell cholesterol efflux: integration of old and new observations provides new insights. *J Lipid Res* 40(5): 781-96.

ROUX PP, BLENIS J, 2004. ERK and p38 MAPK-activated protein kinases: a family of protein kinases with diverse biological functions. *Microbiol Mol Biol Rev* 68(2): 320-44.

ROWAN BG, O'MALLEY BW, (2000). Progesterone receptor coactivators. Steroids 65(10-11): 545-9.

ROWAN BG, WEIGEL NL, O'MALLEY BW, (2000). Phosphorylation of steroid receptor coactivator-1. Identification of the phosphorylation sites and phosphorylation through the mitogen-activated protein kinase pathway. *J Biol Chem* 275(6): 4475-83.

RUPPRECHT AP, COLEMAN DL, (1991). Transfection of adherent murine peritoneal macrophages with a reporter gene using DEAE-dextran. J Immunol Methods 144(2): 157-63.

RUSSELL DW, SETCHELL KDR, (1992). Bile acid biosynthesis. Biochemistry 31(20): 4737-4749.

RUSSELL KS, HAYNES MP, SINHA D, CLERISME E, BENDER JR, (2000). Human vascular endothelial cells contain membrane binding sites for estradiol, which mediate rapid intracellular signaling. *Proc Natl Acad Sci USA* 97(11): 5930-5.

RUST S, ROSIER M, FUNKE H, ET AL., (1999). Tangier disease is caused by mutations in the gene encoding ATP-binding cassette transporter 1. *Nat Genet* 22: 352-355.

SACRE SM, STANNARD AK, OWEN JS, (2003). Apolipoprotein E isoforms differentially induce nitric oxide production in endothelial cells. *FEBS Letters* 540(1-3): 181-187.

SAJI S, OKUMURA N, EGUCHI H, NAKASHIMA S, SUZUKI A, TOI M, NOZAWA Y, SAJI S, HAYASHI S, (2001). MDM2 enhances the function of estrogen receptor alpha in human breast cancer cells. *Biochem Biophys Res Commun* 281(1): 259-65.

SALAS TR, KIM J, VAKAR-LOPEZ F, SABICHI AL, TRONCOSO P, JENSTER G, KIKUCHI A, CHEN SY, SHEMSHEDINI L, SURAOKAR M, LOGOTHETIS CJ, DIGIOVANNI J, LIPPMAN SM, MENTER DG, (2004). Glycogen synthase kinase-3 beta is involved in the phosphorylation and suppression of androgen receptor activity. *J Biol Chem* 279(18): 19191-200.

SALERO E, PEREZ-SEN R, ARUGA J, GIMENEZ C, ZAFRA F, (2001). Transcription factors Zic1 and Zic2 bind and transactivate the apolipoprotein E gene promoter. *J Biol Chem* 276(3): 1881-1888.

SALOMONSSON L, PETTERSSON S, ENGLUND MC, WIKLUND O, OHLSSON BG, (2002). Posttranscriptional regulation of VEGF expression by oxidised LDL in human macrophages. *Eur J Clin Invest* 32(10): 767-74.

SANCHEZ I, HUGHES RT, MAYER BJ, YEE K, WOODGETT JR, AVRUCH J, KYRIAKIS JM, ZON LI, (1994). Role of SAPK/ERK kinase-1 in the stress-activated pathway regulating transcription factor c-Jun. *Nature* 372(6508): 794-8.

SAXENA U, KLEIN MG, VANNI TM, GOLDBERG IJ, (1992). Lipoprotein lipase increases low-density lipoprotein retention by subendothelial cell matrix. *J Clin Invest* 89(2): 373-380.

SCHAEFER E, GREGG R, GHISELLI G, FORTE T, ORDOVAS J, ZECH L, BREWER B, (1986). Familial apolipoprotein E deficiency. *J Clin Invest* 78(5): 1206-1219.

SCHAEFFER HJ, WEBER MJ, (1999). Mitogen-activated protein kinases: specific messages from ubiquitous messengers. *Mol Cell Biol* 19(4): 2435-44.

SCHMIDT BM, GERDES D, FEURING M, FALKENSTEIN E, CHRIST M, WEHLING M, (2000). Rapid, nongenomic steroid actions: A new age? Front Neuroendocrinol 21(1): 57-94.

SCHMUTH M, ELIAS PM, HANLEY K, LAU P, MOSER A, WILLSON TM, BIKLE DD, FEINGOLD KR, (2004). The effect of LXR activators on AP-1 proteins in keratinocytes. *J Invest Dermatol* 123(1): 41-8.

SCHULE R, RANGARAJAN P, YANG N, KLIEWER S, RANSONE LJ, BOLADO J, VERMA IM, EVANS RM, (1991). Retinoic acid is a negative regulator of AP-1 responsive genes. *Proc Natl Acad Sci USA* 88(14): 6092-6096.

SCHULTZ JR, TU H, LUK A, REPA JJ, MEDINA JC, LI L, SCHWENDER S, WANG S, THOOLEN M, MANGELSDORF DJ, LUSTIG KD, SHAN B, (2000). Role of LXRs in control of lipogenesis. *Genes Dev* 14(22): 2831-2838.

SCHUSTER GU, PARINI P, WANG L, ALBERTI S, STEFFENSEN KR, HANSSON GK, ANGELIN B, GUSTAFSSON JA, (2002) Accumulation of foam cells in liver X receptor-deficient mice. *Circulation* 106(9): 1147-1153.

SCHUUR ER, LOKTEV AV, SHARMA M, SUN Z, ROTH RA, WEIGEL RJ, (2001). Ligand-dependent interaction of estrogen receptor-alpha with members of the forkhead transcription factor family. *J Biol Chem* 276(36): 33554-60.

SCHWABE JW, CHAPMAN L, FINCH JT, RHODES D, (1993). The crystal structure of the estrogen receptor DNA-binding domain bound to DNA: how receptors discriminate between their response elements. *Cell* 75(3):567-78.

SCHWARTZ K, LAWN RM, WADE DP, (2000). ABC1 gene expression and ApoA-I-mediated cholesterol efflux are regulated by LXR. *Biochem Biophys Res Commun* 274(3): 794-802.

SCHWARZ M, MURPHY PM, (2001). Kaposi's sarcoma-associated herpesvirus G protein-coupled receptor constitutively activates NF-kappa B and induces proinflammatory cytokine and chemokine production via a C-terminal signaling determinant. *J Immunol* 167(1): 505-13.

SCHWENKE DC, CAREW TE, (1989). Initiation of atherosclerotic lesions in cholesterol-fed rabbits. I. Focal increases in arterial LDL concentration precede development of fatty streak lesions. *Arteriosclerosis* 9(6): 895-907.

SCHWENDE H, FITZKE E, AMBS P, DIETER P, (1996). Differences in the state of differentiation of THP-1 cells induced by phorbol ester and 1,25-dihydroxyvitamin D3. *J Leukoc Biol* 59(4): 555-61.

SEMENKOVICH CF, COLEMAN T, DAUGHERTY A, (1998). Effects of heterozygous lipoprotein lipase deficiency on diet-induced atherosclerosis in mice. J Lipid Res 39: 1141-1151.

SEO HS, LOMBARDI DM, POLINSKY P, POWELL-BRAXTON L, BUNTING S, SCHWARTZ SM, ROSENFELD ME. (1997). Peripheral vascular stenosis in apolipoprotein E-deficient mice: potential roles of lipid deposition, medial atrophy, and adventitial inflammation. *Arterioscler Thromb Vasc Biol* 17: 3593-3601.

SEO JB, MOON HM, KIM WS, LEE YS, JEONG HW, YOO EJ, HAM J, KANG H, PARK MG, STEFFENSEN KR, STULNIG TM, GUSTAFSSON JA, PARK SD, KIM JB, (2004). Activated liver X receptors stimulate adipocyte differentiation through induction of peroxisome proliferator-activated receptor gamma expression. *Mol Cell Biol* 24(8):3430-44.

SHALEV A, SIEGRIST-KAISER CA, YEN PM, WAHLI W, BURGER AG, CHIN WW, MEIER CA, (1996). The peroxisome proliferator-activated receptor alpha is a phosphoprotein: regulation by insulin. *Endocrinology* 137: 4499-4502.

SHANMUGAM N, REDDY MA, GUHA M, NATARAJAN R, (2003). High glucose-induced expression of proinflammatory cytokine and chemokine genes in monocytic cells. *Diabetes* 52(5): 1256-64.

SHAO D, RANGWALA SM, BAILEY ST, KRAKOW SL, REGINATO MJ, LAZAR MA, (1998). Interdomain communication regulating ligand binding by PPAR-gamma. *Nature* 396(6709): 377-380.

SHAO D, LAZAR MA, (1999). Modulating nuclear receptor function: may the phos be with you. J Clin Invest 103(12): 1617-8.

SHEN T, HORWITZ KB, LANGE CA, (2001). Transcriptional hyperactivity of human progesterone receptors is coupled to their ligand-dependent down-regulation by mitogen-activated protein kinase-dependent phosphorylation of serine 294. *Mol Cell Biol* 21(18): 6122-31.

SHIFFMAN D, MIKITA T, TAI JT, WADE DP, PORTER JG, SEILHAMER JJ, SOMOGYI R, LIANG S, LAWN RM, (2000). Large scale gene expression analysis of cholesterol-loaded macrophages. *J Biol Chem* 275(48): 37324-37332.

SHIH SJ, ALLAN C, GREHAN S, TSE E, MORAN C, TAYLOR J, (2000). Duplicated downstream enhancers control expression of the human apolipoprotein E gene in macrophages and adipose tissue. *J Biol Chem* 275(41): 31567-31572.

SHIMADA M, ISHIBASHI S, INABA T, YAGYU H, HARADA K, OSUGA JI, OHASHI K, YAZAKI Y, YAMADA N, (1996). Suppression of diet-induced atherosclerosis in low density lipoprotein receptor knockout mice overexpressing lipoprotein lipase. *Proc Natl Acad Sci USA* 93(14): 7242-7246.

SHIMANO H, OHSUGA J, SHIMANDA M, NAMBA Y, GOTODA T, HARADA K, KATSUKI M, YAZAKI Y, YAMADA N, (1995). Inhibition of diet-induced atheroma formation in transgenic mice expressing apolipoprotein E in the arterial wall. *J Clin Invest* 95(2): 469-476.

SHIMANO H, YAHAGI N, AMEMIYA-KUDO M, HASTY AH, OSUGA J, TAMURA Y, SHIONOIRI F, IIZUKA Y, OHASHI K, HARADA K, GOTODA T, ISHIBASHI S, YAMADA N, (1999). Sterol regulatory element-binding protein-1 as a key transcription factor for nutritional induction of lipogenic enzyme genes. *J Biol Chem* 274(50): 35832-35839.

SHORE B, SHORE VG, (1973). Heterogeneity of human plasma very low density lipoproteins. Separation of species differing in protein components. *Biochemistry* 12(3): 502-507.

SIMONCINI T, HAFEZI-MOGHADAM A, BRAZIL DP, LEY K, CHIN WW, LIAO JK, (2000). Interaction of oestrogen receptor with the regulatory subunit of phosphatidylinositol-3-OH kinase. *Nature* 407 (6803): 538-541.

SIMONET W, BUCAY N, PITAS R, LAUER S, TAYLOR J, (1991). Multiple tissue-specific elements control the apolipoprotein E/C-1 gene locus in transgenic mice. *J Biol Chem* 266(14): 8651-8654.

SIMONET W, BUCAY N, LAUER S, TAYLOR J, (1993). A far-downstream hepatocyte-specific control region directs expression of the linked human apolipoprotein E and C-I genes in transgenic mice. *J Biol Chem* 268(11): 8221-8229.

SINCLAIR AJ, (2003). bZIP proteins of human gamma herpes viruses. J Gen Virol 84(8): 1941-9.

SINGARAJA RR, BOCHER V, JAMES ER, CLEE SM, ZHANG LH, LEAVITT BR, TAN B, BROOKS-WILSON A, KWOK A, BISSADA N, YANG YZ, LIU G, TAFURI SR, FIEVET C, WELLINGTON CL, STAELS B, HAYDEN MR, (2001). Human ABCA1 BAC transgenic mice show increased HDL-C and ApoAI dependant efflux stimulated by an internal promoter containing LXREs in intron 1. *J Biol Chem* 276(36): 33969-33979.

SINGH M, SETALO G JR, GUAN X, WARREN M, TORAN-ALLERAND CD, (1999). Estrogen-induced activation of mitogen-activated protein kinase in cerebral cortical explants: convergence of estrogen and neurotrophin signaling pathways. *J Neurosci* 19(4): 1179-88.

SINGH NN, (2003). Regulation of apolipoprotein E gene expression in THP-1 cells by TGF-β.

SMEAL T, BINETRUY B, MERCOLA DA, BIRRER M, KARIN M, (1991). Oncogenic and transcriptional cooperation with Ha-Ras requires phosphorylation of c-Jun on serines 63 and 73. *Nature* 354(6353): 494-6.

SMITH JD, MELIAN A, LEFF T, BRESLOW JL, (1988). Expression of the human apolipoprotein E gene is regulated by multiple positive and negative elements. *J Biol Chem* 263(17): 8300-8308.

SMITH JD, PLUMP AS, HAYEK T, WALSH A, BRESLOW JL, (1990). Accumulation of human apoliporotein E in the plasma of transgenic mice. *J Biol Chem* 265(25): 14709-14712.

SMITH JD, TROGAN E, GINSBERG M, GRIGAUX C, TIAN J, MIYATA M, (1995a). Decreased atherosclerosis in mice deficient in both macrophage colony stimulating factor (op) and apolipoprotein E. *Proc Natl Acad Sci USA* 92(18): 8264-8268.

SMITH JD, WONG E, GINSBERG M, (1995b). Cytochrome P450 1A1 promoter as a genetic switch for the regulatable and physiological expression of a plasma protein in transgenic mice. *Proc Natl Acad Sci USA* 92: 11926-11930.

SMITH JD, MIYATA M, GINSBERG M, GRIGAUX C, SHMOOKLER E, PLUMP AS, (1996). Cyclic AMP induces apolipoprotein E binding activity and promotes cholesterol efflux from a macrophage cell line to apolipoprotein receptors. *J Biol Chem* 271(48): 30647-30655.

SOLOMON C, WHITE JH, KREMER R, (1999). Mitogen-activated protein kinase inhibits 1,25dihydroxyvitamin D<sub>3</sub>-dependant signal transduction by phosphorylating human retinoid X receptor alpha. *J Clin Invest* 103(12): 1729-1735.

SOLYAKOV L, CAIN K, TRACEY BM, JUKES R, RILEY AM, POTTER BV, TOBIN AB, (2004). Regulation of casein kinase-2 (CK2) activity by inositol phosphates. *J Biol Chem* [Epub ahead of print].

SONG C, HIIPAKKA, RA, KOKONTIS JM, LIAO SS, (1995). Ubiquitous receptor: structures, immunocytochemical localization, and modulation of gene activation by receptors for retinoic acids and thyroid hormones. *Ann NY Acad Sci* 761: 38-49.

SONG C, HIIPAKKA RA, LIAO S, (2000). Selective activation of liver X receptor alpha by 6alphahydroxy bile acids and analogs. *Steroids* 65(8): 38378-38387.

SONG C, HIIPAKKA RA, LIAO S, (2001). Auto-oxidized cholesterol sulfates are antagonistic ligands of liver X receptors: implications for the development and treatment of atherosclerosis. *Steroids* 66(6): 473-479.

SPARROW JT, SPARROW DA, FERNANDO G, CULWELL AR, KOVAR M, GOTTO AM, (1992). Apolipoprotein E: phospholipid binding studies with synthetic peptides from the carboxyl terminus. *Biochemistry* 31(4): 1065-1068.

SPARROW CP, BAFFIC J, LAM MH, LUND EG, ADAMS AD, FU X, HAYES N, JONES AB, MACNAUL KL, ONDEEYKA J, SINGH S, WANG J, ZHOU G, MOLLER DE, WRIGHT SD, MENKE JG, (2002). A potent synthetic LXR agonist is more effective than cholesterol loading at inducing ABCA1 mRNA and stimulating cholesterol efflux. *J Biol Chem* 277(12): 10021-10027.

SPENCER TA, LI D, LI D, RUSSEL JS, COLLINS JL, BLEDSOE RK, CONSLER TG, MOORE LB, GALARDI CM, MCKEE DD, MOORE JT, WATSON MA, PARKS DJ, LAMBERT MH, WILLSON TM, (2001). Pharmacophore analysis of the nuclear oxysterol receptor LXRalpha. *J Med Chem* 44(6): 886-897.

SRIVASTAVA RA, BHASIN N, SRIVASTAVA N, (1996). Apolipoprotein E gene expression in various tissues of mouse and regulation by estrogen. *Biochem Mol Biol Int* 38(1): 91-101.

STAMBOLIC V, SUZUKI A, DE LA POMPA JL, BROTHERS GM, MIRTSOS C, SASAKI T, RULAND J, PENNINGER JM, SIDEROVSKI DP, MAK TW, (1998). Negative regulation of PKB/Akt-dependent cell survival by the tumor suppressor PTEN. *Cell* 95(1): 29-39.

STANNARD AK, RIDDELL DR, SACRE S, TAGALAKIS AD, LANGER C, VON ECKARDSTEIN A, CULLEN P, ATHANASOPOULOS T, DICKSON G, OWEN JS, (2001). Cell-derived apolipoprotein E (ApoE) particles inhibit vascular cell adhesion molecule-1 (VCAM-1) expression in human endothelial cells. *J Biol Chem* 276(49): 46011-46016.

STARCK M, BERTRAND P, PEPIN S, SCHIELE F, SIEST G, GALTEAU MM, (2000). Effects of proinflammatory cytokines on apolipoprotein E secretion by a human astrocytoma cell line (CCF-STTG1). *Cell Biochem Funct* 18(1): 9-16.

STEFFENSEN KR, HOLTER E, TOBIN KA, LECLERC S, GUSTAFSSON JA, GUERIN SL, ESKILD W, (2001). Members of the nuclear factor 1 family reduce the transcriptional potential of the nuclear receptor LXRalpha promoter. *Biochem Biophys Res Commun* 289(5): 1262-7.

STEFFENSEN KR, SCHUSTER GU, PARINI P, HOLTER E, SADEK CM, CASSEL T, ESKILD W, GUSTAFSSON JA, (2002). Different regulation of the LXRalpha promoter activity by isoforms of CCAAT/enhancerbinding proteins. *Biochem Biophys Res Commun* 293(5): 1333-40.

STEFFENSEN KR, HOLTER E, ALIKHANI N, ESKILD W, GUSTAFSSON JA, (2003). Glucocorticoid response and promoter occupancy of the mouse LXRalpha gene. *Biochem Biophys Res Commun* 312(3): 716-24.

STEINBERG D, PARTHASARATHY S, CAREW TE, KHOO JC, WITZUM JZ, (1989). Beyond cholesterol. Modifications of low-density lipoprotein that increase its atherogenicity. *N Engl J Med* 320(14): 915-924.

STEPHENS L, ANDERSON K, STOKOE D, ERDJUMENT-BROMAGE H, PAINTER GF, HOLMES AB, GAFFNEY PR, REESE CB, MCCORMICK F, TEMPST P, COADWELL J, HAWKINS PT, (1998). Protein kinase B kinases that mediate phosphatidylinositol 3,4,5-trisphosphate-dependent activation of protein kinase B. *Science* 279(5351): 710-4.

STRAUSS JG, FRANK S, KRATKY D, HAMMERLE G, HRZENJAK A, KNIPPING G, VON ECKARDSTEIN, KOSTNER GM, ZECHNER R, (2001). Adenovirus-mediated rescue of lipoprotein lipase-deficient mice. Lipolysis of triglyceride-rich lipopoproteins is essential for high density lipoprotein maturation in mice. *J Biol Chem* 276(39): 36083-36090.

SUNDSTROM C, NILSSON K, (1976). Establishment and characterization of a human histiocytic lymphoma cell line (U-937). Int J Cancer 17(5): 565-77.

SUNG JY, SHIN SW, AHN YS, CHUNG KC, (2001). Basic fibroblast growth factor-induced activation of novel CREB kinase during the differentiation of immortalized hippocampal cells. *J Biol Chem* 276(17):13858-66.

SUZUKI H, KURIHARA Y, TAKEYA M, KAMADA N, KATAOKA M, JISHAGE K, UEDA O, SAKAGUCHI H, HIGASHI T, SUZUKI T, TAKASHIMA Y, KAWABE Y, CYNSHI O, WADA Y, HONDAM, KURIHARA H, ABURATANI H, DOI T, MATSUMOTO A, AZUMA S, NODA T, TOYODA Y, ITAKURA H, YAZAKI Y, KODAMA T, ET AL., (1997). A role for macrophage scavenger receptors in atherosclerosis and susceptibility to infection. *Nature* 386(6622): 292-296.

SVENSSON S, OSTBERG T, JACOBSSON M, NORSTROM C, STEFANSSON K, HALLEN D, JOHANSSON IC, ZACHRISSON K, OGG D, JENDEBERG L, (2003). Crystal structure of the heterodimeric complex of LXRalpha and RXRbeta ligand-binding domains in a fully agonistic conformation. *EMBO J* 22(18): 4625-33.

TAJIMA S, HAYASHI R, TSUCHIYA S, MIYAKE Y, YAMAMOTO A, (1985). Cells of a human monocytic leukemia cell line (THP-1) synthesise and secrete apolipoprotein E and lipoprotein lipase. *Biochem Biophys Res Commun* 126(1): 526-531.

TAKAHASHI S, SUZUKI J, KOHNO M, OIDA K, TAMAI T, MIYABO S, YAMAMOTO T, NAKAI T, (1995). Enhancement of the binding of triglyceride-rich lipoproteins to the very low density lipoprotein receptor by apolipoprotein E and lipoprotein lipase. *J Biol Chem* 270(26): 15747-54.

TAKAHASHI Y, SMITH JD, (1999). Cholesterol efflux to apolipoprotein Al involves endocytosis and resecretion in a calcium-dependant pathway. *Proc Natl Acad Sci USA* 96(20): 11358-11363.

TAKAHASHI Y, MIYATA M, ZHENG P, IMAZATO T, HORWITZ A, SMITH JD, (2000). Identification of cAMP analogue inducible genes in RAW264 macrophages. *Biochim Biophys Acta* 1492(2-3): 385-394.

TAKEDA K, ICHIKI T, TOKUNOU T, IINO N, TAKESHITA A, (2001). 15-Deoxy-delta 12,14-prostaglandin J2 and thiazolidinediones activate the MEK/ERK pathway through phosphatidylinositol 3-kinase in vascular smooth muscle cells. *J Biol Chem* 276(52): 48950-5.

TALL AR, (1993). Plasma cholesteryl ester transfer protein. J Lipid Res 34(8):1255-74.

TAMASAWA N, TAMASAWA A, TAKEBE K, (1993). Higher levels of plasma cholesterol sulfate in patients with liver cirrhosis and hypercholesterolemia. *Lipids* 28(9): 833-836.

TAMMINEN M, MOTTINO G, QIAO JH, BRESLOW JL, FRANK JS, (1999). Ultrastructure of early lipid accumulation in apoE-deficient mice. *Arterioscler Thromb Vasc Biol* 19: 847-853.

TANEJA R, ROCHETTE-EGLY C, PLASSANT JL, PENNA L, GAUB MP, CHAMBON P, (1997). Phosphorylation of activation functions AF-1 and AF-2 of RAR alpha and RAR gamma is indispensable for differentiation of F9 cells upon retinoic acid and cAMP treatment. *EMBO J* 16(21): 6452-6465.

TANGIRALA RK, BISCHOFF ED, JOSEPH SB, WAGNER BL, WALCZAK R, LAFFITTE BA, DAIGE CL, THOMAS D, HEYMAN RA, MANGELSDORF DJ, WANG X, LUSIS AJ, TONTONOZ P, SCHULMAN IG, (2002). Identification of liver X receptors as inhibitors of atherosclerosis. *Proc Natl Acad Sci USA* 99(18): 11896-11901.

TAYLOR JM, LAUER S, ELSHOURBAGY N, REARDON C, TAXMAN E, WALKER D, CHANG D, PAIK YK, (1987). Structure and evolution of human apolipoprotein genes: identification of regulatory elements of the human apolipoprotein E gene. *Ciba Found Symp* 130: 70-86.

TEBOUL M, ENMARK E, LI Q, WIKSTROM AC, PELTO-HUIKKO M, GUSTAFSSON JA, (1995). OR-1, a member of the nuclear receptor superfamily that interacts with the 9-cis-retinoic acid receptor. *Proc Natl Acad Sci USA* 92(6): 2096-2100.

TERASAKA N, HIROSHIMA A, KOIEYAMA T, UBUKATA N, MORIKAWA Y, NAKAI D, INABA T, (2003). T-0901317, a synthetic liver X receptor ligand, inhibits development of atherosclerosis in LDL receptor-deficient mice. *FEBS Lett* 536: 6-11. TERKELTAUB R, BOISVERT WA, CURTISS LK, (1998). Chemokines and atherosclerosis. Curr Opin Lipidol 9: 397-405.

TEYSSIER C, BELGUISE K, GALTIER F, CHALBOS D, (2001). Characterization of the physical interaction between estrogen receptor alpha and JUN proteins. *J Biol Chem* 276(39): 36361-9.

THORNGATE FE, RUDEL LL, WALZEM RL, WILLIAMS DL, (2000). Low levels of extrahepatic nonmacrophage apoE inhibit atherosclerosis without correcting hypercholesterolemia in apoE-deficient mice. *Arterio Thromb Vasc Biol* 20(8): 1939-1945.

THORNGATE FE, YANCEY PG, KELLNER-WEIBEL G, RUDEL LL, ROTHBLAT GH, WILLIAMS DL, (2003). Testing the role of apoA-I, HDL, and cholesterol efflux in the atheroprotective action of low-level apoE expression. *J Lipid Res* 44(12): 2331-2338.

THUAHNAI ST, LUND-KATZ S, ANANTHARAMAIAH GM, WILLIAMS DL, PHILLIPS MC, (2003). A quantitative analysis of apolipoprotein binding to SR-BI: multiple binding sites for lipid-free and lipid-associated apolipoproteins. *J Lipid Res* 44(6): 1132-1142.

THUREN T, WEISGRABER KH, SISSON P, WAITE M, (1992). Role of apolipoprotein E in in hepatic lipase catalysed hydrolysis of phospholipid in high density lipoproteins. *Biochemistry* 31(8): 2332-2338.

TING YT, BHAT MK, WONG R, CHENG S, (1997). Tissue-specific stabilization of the thyroid hormone beta1 nuclear receptor by phosphorylation. *J Biol Chem* 272(7): 4129-34.

TOBIN KA, STEINEGER HH, ALBERTI S, SPYDEVOLD O, AUWERX J, GUSTAFFSON JA, NEBB HI, (2000). Cross-talk between fatty acid and cholesterol metabolism mediated by liver X receptor- $\alpha$ . *Mol Endocrinol* 14(5): 741-752.

TOBIN KA, ULVEN SM, SCHUSTER GU, STEINEGER HH, ANDRESEN SM, GUSTAFSSON JA, NEBB HI, (2002). Liver X receptors as insulin-mediating factors in fatty acid and cholesterol biosynthesis. *J Biol Chem* 277(12): 10691-10697.

TOKER A, (2000). Protein kinases as mediators of phosphoinositide 3-kinase signaling. *Mol Pharmacol* 57(4): 652-8.

TONE Y, COLLINGWOOD TN, ADAMS M, CHATTERJEE VK, (1994). Functional analysis of a transactivation domain in the thyroid hormone beta receptor. *J Biol Chem* 269(49): 31157-61.

TOURNIER C, WHITMARSH AJ, CAVANAGH J, BARRETT T, DAVIS RJ, (1999). The MKK7 gene encodes a group of c-Jun NH2-terminal kinase kinases. *Mol Cell Biol* 19(2): 1569-81.

TORRES J, PULIDO R, (2001). The tumor suppressor PTEN is phosphorylated by the protein kinase CK2 at its C terminus. Implications for PTEN stability to proteasome-mediated degradation. *J Biol Chem* 276(2): 993-8.

TREISMAN R, (1995). Journey to the surface of the cell: Fos regulation and the SRE. *EMBO J* 14(20): 4905-13.

TREMBLAY A, TREMBLAY GB, LABRIE F, GIGUERE V, (1999). Ligand-independent recruitment of SRC-1 to estrogen receptor beta through phosphorylation of activation function AF-1. *Mol Cell* 3(4): 513-519.

TSAI SC, SETO E, (2002). Regulation of histone deacetylase 2 by protein kinase CK2. *J Biol Chem* 277(35): 31826-33.

TSUCHIYA S, YAMABE M, YAMAGUCHI Y, KOBAYASHI Y, KONNO T, TADA K, (1980). Establishment and characterization of a human acute monocytic leukemia cell line (THP-1). Int J Cancer 26(2): 171-6.

TSUCHIYA S, KOBAYASHI Y, GOTO Y, OKUMURA H, NAKAE S, KONNO T, TADA K, (1982). Induction of maturation in cultured human monocytic leukemia cells by a phorbol diester. *Cancer Res* 42(4): 1530-6.

TZAGARAKIS-FOSTER C, PRIVALSKY ML, (1998). Phosphorylation of thyroid hormone receptors by protein kinase A regulates DNA recognition by specific inhibition of receptor monomer binding. *J* Biol Chem 273(18): 10926-32.

UHT RM, ANDERSON CM, WEBB P, KUSHNER PJ, (1997). Transcriptional activities of estrogen and glucocorticoid receptors are functionally integrated at the AP-1 response element. *Endocrinology* 138(7):2900-8.

VAISMAN BL, (2001). ABCA1 overexpression leads to hyperalphalipoproteinemia and increased biliary cholesterol excretion in transgenic mice. *J Clin Invest* 108: 303-309.

VANDENBROUCK Y, JANVIER B, LORIETTE C, BEREZIAT G, ANDREANI-MANGENEY M, (1994). The modulation of apolipoprotein E expression by 3, 3'-5-triiodothyronine in HepG2 cells occurs at transcriptional and post-transcriptional levels. *Eur J Biochem* 224(2): 463-471.

VAN DAM H, WILHELM D, HERR I, STEFFEN A, HERRLICH P, ANGEL P, (1995). ATF-2 is preferentially activated by stress-activated protein kinases to mediate c-jun induction in response to genotoxic agents. *EMBO J* 14(8): 1798-811.

VAN ECK M, HERIJGERS N, YATES J, PEARCE NJ, HOOGERBRUGGE PM, GROOT PH, VAN BERKEL TJ, (1997). Bone Marrow transplantation in apolipoprotein E-deficient mice: Effect of apoE gene dosage on serum lipid concentrations, (beta) VLDL catabolism, and atherosclerosis. *Arterioscler Vasc Biol* 17: 3117-3126.

VAN ECK M, (1999). Bone marrow transplantation as a novel strategy to study the role of macrophage-derived apolipoprotein E in lipoprotein metabolism an atherogenesis.

VAN ECK M, HERIJGERS N, VIDGEON-HART M, PEARCE NJ, HOOGERBRUGGE PM, GROOT PH, VAN BERKEL TJ, (2000). Accelerated atherosclerosis in C57BI/6 mice transplanted with ApoE-deficient bone marrow. *Atherosclerosis* 150(1): 71-80.

VAN ECK M, BOS IS, KAMINSKI WE, ORSO E, ROTHE G, TWISK J, BOTTCHER A, VAN AMERSFOOT ES, CHRISTIANSEN-WEBER TA, FUNG-LEUNG WP, ET AL., (2002). Leukocyte ABCA1 controls susceptibility to atherosclerosis and macrophage recruitment into tissues. *Proc Natl Acad Sci USA* 99: 6298-6303.

VAN FURTH R, (1989). Origin and turnover of monocytes and macrophages. *Curr Topics Pathol* 79: 125-150.

VANHAESEBROECK B, ALESSI DR, (2000). The PI3K-PDK1 connection: more than just a road to PKB. *Biochem J* 346(3): 561-76.

VAN TOL A, (2002). Phospholipid transfer protein. Curr Opin Lipidol 13(2): 135-139.

VENKATESWARAN A, LAFFITTE B, JOSEPH SB, MAK PA, WILPITZ DC, EDWARDS PA, TONTONOZ P, (2000a). Control of cellular cholesterol efflux by the nuclear oxysterol receptor LXR alpha. *Proc Natl Acad Sci USA* 97(22): 12097-12102.

VENKATESWARAN A, REPA JJ, LOBACCARO JMA, BRONSON A, MANGELSDORF DJ, EDWARDS PA, (2000b). Human white/murine ABC8 mRNA levels are highly induced in lipid-loaded macrophages. *J Biol Chem* 275(19): 14700-14707.

VLAHOS CJ, MATTER WF, HUI KY, BROWN RF, (1994). A specific inhibitor of phosphatidylinositol 3kinase, 2-(4-morpholinyl)-8-phenyl-4H-1-benzopyran-4-one (LY294002). *J Biol Chem* 269(7): 5241-8.

VOGEL T, GUO N, GUY R DREZLICH N, KRUTZSCH HC, BLAKE DA, PANET A, ROBERTS DD, (1994). Apolipoprotein E: A potent inhibitor of endothelial and tumor cell proliferation. *J Cell Biochem* 54(3): 299-308.

VON ECKARDSTEIN A, LANGER C, ENGEL T, SCHAUKAL I, CIGNARELLA A, REINHARDT J, LORKOWSKI S, LI Z, ZHOU X, CULLEN P, ASSMAN G, (2001). ATP binding cassette transporter ABCA1 modulates the secretion of apoE from human monocyte-derived macrophages. *FASEB J* 15(9): 1555-1561.

WAETZIG GH, SEEGERT D, ROSENSTIEL P, NIKOLAUS S, SCHREIBER S, (2002). p38 mitogen-activated protein kinase is activated and linked to TNF-alpha signalling in inflammatory bowel disease. *J Immunol* 168(10): 5342-5351.

WAGER RE, ASSOIAN RK, (1990). A phorbol ester-regulated ribonuclease system controlling transforming growth factor beta 1 gene expression in hematopoietic cells. *Mol Cell Biol* 10(11): 5983-90.

WAGNER BL, VALLEDOR AF, SHAO G, DAIGE CL, BISCHOFF ED, PETROWSKI M, JEPSEN K, BAEK SH, HEYMAN RA, ROSENFELD MG, SCHULMAN IG, GLASS CK, (2003). Promoter-specific roles for liver X receptor/corepressor complexes in the regulation of ABCA1 and SREBP1 gene expression. *Mol Cell Biol* 23(16): 5780-5789.

WALCZAK R, JOSEPH SB, LAFFITTE BA, CASTRILLO A, PEI L, TONTONOZ P, (2004). Transcription of the vascular endothelial growth factor gene in macrophages is regulated by liver X receptors. *J Biol Chem* 279(11):9905-11.

WALFISH PG, YOGANATHAN T, YANG YF, HONG H, BUTT TR, STALLCUP MR, (1997). Yeast hormone response element assays detect and characterize GRIP1 coactivator-dependent activation of transcription by thyroid and retinoid nuclear receptors. *Proc Natl Acad Sci USA* 94(8): 3697-702.

WANG AM, DOYLE MV, MARK DF, (1989). Quantitation of mRNA by the polymerase chain reaction. *Proc Natl Acad Sci USA* 86(24): 9717-21.

WANG N, ARAI T, JI Y, RINNINGER F, TALL AR, (1998). Liver-specific overexpression of scavenger receptor BI decreases levels of very low density lipoprotein apoB, low density lipoprotein apoB and high density lipoprotein in transgenic mice. *J Biol Chem* 273(49): 32920-32926.

WANG H, CHEN J, HOLLISTER K, SOWERS LC, FORMAN BM, (1999). Endogenous bile acids are ligands for the nuclear receptor FXR/BAR. *Mol Cell* 3(5): 543-553.

WANG N, SILVER DL, COSTET P, TALL AR, (2000). Specific binding of apoA-I, enhanced cholesterol efflux and altered plasma membrane morphology in cells expressing ABC1. *J Biol Chem* 275(42): 33053-33058.

WANG H, HUANG ZQ, XIA L, FENG Q, ERDJUMENT-BROMAGE H, STRAHL BD, BRIGGS SD, ALLIS CD, WONG J, TEMPST P, ZHANG Y, (2001). Methylation of histone H4 at arginine 3 facilitating transcriptional activation by nuclear hormone receptor. *Science* 293(5531): 853-7.

WANG Y, LI C, WANG X, ZHANG J, CHANG Z, (2002). Heat shock response inhibits IL-18 expression through the JNK pathway in murine peritoneal macrophages. *Biochem Biophys Res Commun* 296(3): 742-748.

WARDELL MR, WEISGRABER KH, HAVEKES LM, RALL SC JR, (1989). Apolipoprotein E3-Leiden contains a seven-amino acid insertion that is a tandem repeat of residues 121-127. *J Biol Chem* 264 (35): 21205-10.

WATANABE Y, TANAKA T, UCHIYAMA Y, TAKENO T, IZUMI A, YAMASHITA H, KUMAKURA J, IWANARI H, SHU-YING J, NAITO M, MANGELSDORF DJ, HAMAKUBO T, KODAMA T, (2003). Establishment of a monoclonal antibody for human LXRalpha: Detection of LXRalpha protein expression in human macrophages. *Nucl Recept* 1(1): 1.

WATTERS JJ, CAMPBELL JS, CUNNINGHAM MJ, KREBS EG, DORSA DM, (1997). Rapid membrane effects of steroids in neuroblastoma cells: effects of estrogen on mitogen activated protein kinase signalling cascade and c-fos immediate early gene transcription. *Endocrinology* 138(9): 4030-3.

WAY TD, KAO MC, LIN JK, (2004). Apigenin induces apoptosis through proteasomal degradation of HER2/neu in HER2/neu-overexpressing breast cancer cells via the phosphatidylinositol 3-kinase/Akt-dependent pathway. *J Biol Chem* 279(6): 4479-89.

WAYS DK, DODD RC, EARP HS, (1987). Dissimilar effects of phorbol ester and diacylglycerol derivative on protein kinase activity in the monoblastoid U937 cell. *Cancer Res* 47(13): 3344-50.

WEISGRABER KH, INNERARITY TL, MAHLEY RW, (1982). Abnormal lipoprotein receptor-binding activity of the human E apoprotein due to cysteine-arginine interchange at a single site. *J Biol Chem* 257(5): 2518-2521.

WEISGRABER KH, RALL SC, MAHLEY RW, MARCEL YL, SPARROW JT, (1986). Human apolipoprotein E. Determination of the heparin binding sites of apolipoprotein E3. *J Biol Chem* 261(5): 2068-2076.

WELLINGTON CL, Walker EK, Suarez A, Kwok A, Bissada N, Singaraja R, Yang YZ,Zhang LH, James E, Wilson JE, Francone O, McManus BM, Hayden MR, (2002). ABCA1 mRNA and protein distribution patterns predict multiple different roles and levels of regulation. *Lab Invest* 82(3): 273-283.

WENNER C, LORKOWSKI S, ENGEL T, CULLEN P, (2001). Apolipoprotein E in macrophages and hepatocytes is degraded via the proteasomal pathway. *Biochem Biophys Res Commun* 282(2): 608-14.

WERB Z, CHIN J, (1983). Endotoxin suppresses expression of apoprotein E by mouse macrophages *in vivo* and in culture. *J Biol Chem* 258(17): 10642-10648.

WERNETTE-HAMMOND ME, LAUER SJ, CORSINI A, WALKER D, TAYLOR JM, RALL SC, (1989). Glycosylation of human apolipoprotein E: The carbohydrate attachment site is threonine 194. *J Biol Chem* 264(15): 9094-9101.

WESTERLUND JA, WEISGRABER KH, (1993). Discrete carboxy-terminal segments of apolipoprotein E mediate lipoprotein association and protein oligomerisation. *J Biol Chem* 268(21): 15745-15750.

WHITE AL, GUERRA B, WANG J, LANFORD RE, (1999). Presecretory degradation of apolipoprotein(a) is mediated by the proteasome pathway. *J Lipid Res* 40(2): 275-286.

WHITE R, SJOBERG M, KALKHOVEN E, PARKER MG, (1997). Ligand-independent activation of the oestrogen receptor by mutation of a conserved tyrosine. *EMBO J* 16(6): 1427-1435.

WHITEMAN EL, CHO H, BIRNBAUM MJ, (2003). Role of Akt/protein kinase B in metabolism. *Trends Endocrinol Metab* 13(10): 444-51.

WHITNEY KD, WATSON MA, GOODWIN B, GALARDI CM, MAGLICH JM, WILSON JG, WILLSON TM, COLLINS JL, KLIEWER SA, (2001). Liver X receptor (LXR) regulation of the LXRalpha gene in human macrophages. *J Biol Chem* 276(47): 43509-43515.

WIEBEL FF, GUSTAFSSON JA, (1997). Heterodimeric interaction between retinoid X receptor  $\alpha$  and orphan nuclear receptor OR1 reveals dimerisation-induced activation as a novel mechanism of nuclear receptor activation. *Mol Cell Biol* 17(7): 3977-3986.

WIEBEL FF, STEFFENSEN KR, TREUTER E, FELTKAMP D, GUSTAFFSON JA, (1999). Ligand-independent coregulator recruitment by the triply activable OR1/retinoid X receptor-alpha nuclear receptor heterodimer. *Mol Endocrinol* 13: 1105-1118.

WILLNOW TE, SHENG Z, ISHIBASHI S, HERZ J, (1994). Inhibition of hepatic chylomicron remnant uptake by gene transfer of a receptor antagonist. *Science* 264(5164): 1471-1474.

WILLY PJ, UMESONO K, ONG ES, EVANS RM, HEYMAN RA, MANGELSDORF DJ, (1995). LXR, a nuclear receptor that defines a distinct retinoid response pathway. *Genes Dev* 9(9): 1033-1045.

WILLY PJ, MANGELSDORF DJ, (1997). Unique requirements for retinoid dependant transcriptional activation by the orphan receptor LXR. *Genes Dev* 11(3): 289-298.

WILSON C, WARDELL MR, WEISGRABER KH, MAHLEY RW, AGARD DA, (1991). Three-dimensional structure of the LDL receptor-binding domain of human apolipoprotein E. *Science* 252(5014): 1817-1822.

WILSON C, MAU T, WEISGRABER KH, WARDELL MR, MAHLEY RW, AGARD DA, (1994). Salt bridge relay triggers defective LDL receptor binding by a mutant apolipoprotein. *Structure* 2(8): 713-718.

WILSON K, FRY GL, CHAPPELL DA, SIGMUND CD, MEDH JD, (2001). Macrophage-specific expression of human lipoprotein lipase accelerates atherosclerosis in transgenic apolipoprotein E knockout mice but not in C57BL/6 mice. *Arterioscler Thromb Vasc Biol* 21(11): 1809-1815.

WU LX, ANARAKI F, MORAHAN PS, LEARY K, (1990). Transient expression of virus-specific promoters in murine resident peritoneal macrophages. *J Leukoc Biol* 48(3): 229-36.

WURMSER AE, GARY JD, EMR SD, (1999). Phosphoinositide 3-kinases and their FYVE domaincontaining effectors as regulators of vacuolar/lysosomal membrane trafficking pathways. *J Biol Chem* 274(14): 9129-32.

WURTZ JM, BOURGUET W, RENAUD JP, VIVAT V, CHAMBON P, MORAS D, GRONEMEYER H, (1996). A canonical structure for the ligand-binding domain of nuclear receptors. *Nat Struct Biol* 3(2):206.

WYMANN MP, ZVELEBIL M, LAFFARGUE M, (2003). Phosphoinositide 3-kinase signalling-which way to target? *Trends Pharmacol Sci* 24(7): 366-76.

YAGYU H, ISHIBASHI S, CHEN Z, OSUGA J, OKAZAKI M, PERREY S, KITAMINE T, SHIMADA M, OHASHI K, HARADA K, SHIONOIRI F, YAHAGI N, GOTODA T, YAZAKI Y, YAMADA N, (1999). Overexpressed lipoprotein lipase protects against atherosclerosis in apolipoprotein E knockout mice. *J Lipid Res* 40(9): 1677-1685.

YAMADA N, INOUE I, KAWAMURA M, HARADA K, WATANABE Y, SHIMANO H, GOTODA T, SHIMADA M, KOHZAKI K, TSUKADA T, ET AL., (1992). Apolipoprotein E prevents the progression of atherosclerosis in Watanabe heritable hyperlipidemic rabbits. *J Clin Invest* 89: 706-711.

YAMADA Y, DOI T, HAMAKUBO T, KODAMA T, (1998). Scavenger receptor family proteins: roles for atherosclerosis, host defence and disorders of the central nervous system. *Cell Mol Life Sci* 54(7): 628-640.

YANCEY PG, BIELICKI JK, JOHNSON WJ, LUND-KATZ S, PALGUNACHARI MN, ANANTHARAMAIAH GM, SEGREST JP, PHILLIPS MC, ROTHBLAT GH, (1995). Efflux of cellular cholesterol and phospholipid to lipid-free apolipoproteins and class A amphipathic peptides. *Biochemistry* 34 (24): 7955-65.

YEH S, LIN HK, KANG HY, THIN TH, LIN MF, CHANG C, (1999). From HER2/Neu signal cascade to androgen receptor and its coactivators: a novel pathway by induction of androgen target genes through MAP kinase in prostate cancer cells. *Proc Natl Acad Sci USA* 96(10): 5458-63.

YEN FT, MANN CJ, GUERMANI LM, ET AL. Identification of a lipolysis-stimulated receptor that is distinct from the LDL receptor and the LDL receptor-related protein. *Biochemistry* 33: 1172-80.

YEUNG SJ, CHEN SH, CHAN L, (1996). Ubiquitin-proteasome pathway mediates intarcellular degradation of apolipoprotin B. *Biochemistry* 35(43): 13843-13848.

YOSHIKAWA T, SHIMANO H, AMEMIYA-KUDO M, YAHAGI N, HASTY AH, MATSUZAKA T, OKAZAKI H, TAMURA Y, IIZUKA Y, OHASHI K, OSUGA J, HARADA K, GOTODA T, KIMURA S, ISHIBASHI S, YAMADA N, (2001). Identification of liver X receptor-retinoid X receptor as an activator of the sterol regulatory element-binding protein 1c gene promoter. *Mol Cell Biol* 21(9): 2991-3000.

YOSHIKAWA T, SHIMANO H, YAHAGI N, IDE T, AMEMIYA-KUDO M, MATSUZAKA T, NAKAKUKI M, TOMITA S, OKAZAKI H, TAMURA Y, IIZUKA Y, OHASHI K, TAKAHASHI A, SONE H, OSUGA JI J, GOTODA T, ISHIBASHI S, YAMADA N, (2002). Polyunsaturated fatty acids suppress sterol regulatory element-binding protein 1c promoter activity by inhibition of liver X receptor (LXR) binding to LXR response elements. *J Biol Chem* 277(3):1705-1711.

YU L, HAMMER RE, LI-HAWKINS J, VON BERGMANN K, LUTJOHANN D, COHEN JC, HOBBS HH, (2002a). Disruption of ABCG5 and ABCG8 in mice reveals their crucial role in biliary cholesterol secretion. *Proc Natl Acad Sci USA* 99(25): 16237-16242.

YU L, LI-HAWKINS J, HAMMER RE, BERGE KE, HORTON JD, COHEN JC, HOBBS HH, (2002b). Overexpression of ABCG5 and ABCG8 promotes biliary cholesterol secretion and reduces fractional absorption of dietary cholesterol. *J Clin Invest* 110(5): 671-680.

ZANNI EE, KOUVATSI A, HADZOPOULOU-CLADRAS M, KRIEGER M, ZANNIS I, (1989). Expression of apoE gene in chinese hamster cells with a reversible defect in O-glycosylation. Glycosylation is not required for apoE secretion. *J Biol Chem* 264: 9137-9140.

ZANNIS VI, BRESLOW JL, (1981). Human very low density lipoprotein apolipoprotein E isoprotein polymorphism is explained by genetic variation and posttranslational modification. *Biochemistry* 20(4): 1033-1041.

ZANNIS VI, KURNIT DM, BRESLOW JL, (1982). Hepatic apo-A-I and apo-E and intestinal apo-A-I are synthesized in precursor isoprotein forms by organ cultures of human fetal tissues. *J Biol Chem* 257(1): 536-544.

ZANNIS V, MCPHERSON J, GOLDBERGER G, KARATHANASIS S, BRESLOW J, (1984). Synthesis, intracellular processing, and signal peptide of human apolipoprotein E. *J Biol Chem* 259(9): 5495-5499.

ZHANG SH, REDDICK RL, PIEDRAHITA JA, MAEDA N, (1992). Spontaneous hypercholesterolemia and arterial lesions in mice lacking apolipoprotein E. *Science* 258 (5081): 468-472.

ZHANG SH, REDDICK RL, BURKEY B, MAEDA N, (1994a). Diet-induced atherosclerosis in mice heterozygous and homozygous for apolipoprotein E gene disruption. *J Clin Invest* 94(3): 937-945.

ZHANG Y, BECK CA, POLETTI A, EDWARDS DP, WEIGEL NL, (1994b). Identification of phosphorylation sites unique to the B form of human progesterone receptor. In vitro phosphorylation by casein kinase II. *J Biol Chem* 269(49): 31034-40.

ZHANG WY, GAYNOR PM, KRUTH HS, (1996). Apolipoprotein E produced by human monocytederived macrophages mediates cholesterol efflux that occurs in the absence of added cholesterol acceptors. *J Biol Chem* 271(45): 28641-28646.

ZHANG Y, BECK CA, POLETTI A, CLEMENT JP 4TH, PRENDERGAST P, YIP TT, HUTCHENS TW, EDWARDS DP, WEIGEL NL, (1997). Phosphorylation of human progesterone receptor by cyclin-dependent kinase 2 on three sites that are authentic basal phosphorylation sites in vivo. *Mol Endocrinol* 11(6): 823-32.

ZHANG Y, REPA JJ, GAUTHIER K, MANGELSDORF DJ, (2001). Regulation of lipoprotein lipase by the oxysterol receptors, LXR $\alpha$  and LXR $\beta$ . *J Biol Chem* 276(46): 43018-43024.

ZHANG Y, MANGELSDORF DJ, (2002). Luxuries of lipid homeostasis: The unity of nuclear hormone receptors, transcription regulation, and cholesterol sensing. *Mol Intervent* 2(2): 78-87.

ZHAO Y, MAZZONE T, (2000). Transport and processing of endogenously synthesised apoE on the macrophage cell surface. *J Biol Chem* 275(7): 4759-4765.

ZHAO HH, HERRERA RE, CORONADO-HEINSOHN E, YANG MC, LUDES-MEYERS JH, SEYBOLD-TILSON KJ, NAWAZ Z, YEE D, BARR FG, DIAB SG, BROWN PH, FUQUA SA, OSBORNE CK, (2001). Forkhead homologue in rhabdomyosarcoma functions as a bifunctional nuclear receptor-interacting protein with both coactivator and corepressor functions. *J Biol Chem* 276(30): 27907-12.

ZHAO M, LIU Y, WANG X, NEW L, HAN J, BRUNK UT, (2002). Activation of the p38 MAP kinase pathway is required for foam cell formation from macrophages exposed to oxidized LDL. *APMIS* 110(6): 458-68.

ZHOU Y, GROSS W, HONG SH, PRIVALSKY ML, (2001). The SMRT corepressor is a target of phosphorylation by protein kinase CK2 (casein kinase II). *Mol Cell Biochem* 220(1-2): 1-13.

ZHU YH, BELLOSTA S, LANGER C, BERNINI F, PITAS RE, MAHLEY RW, ASSMAN G, VON ECKARDSTEIN A, (1998). Low-dose expression of a human apolipoprotein E transgene in macrophages restores cholesterol efflux capacity of apolipoprotein E-deficient mouse plasma. *Proc Natl Acad Sci USA* 95(13): 7585-7590.

ZSIGMOND E, KOBAYASHI K, TZUNG KW, LI L, FUKE Y, CHAN L, (1997). Adenovirus-mediated gene transfer of human lipoprotein lipase ameliorates the hyperlipidemias associated with apolipoprotein E and LDL receptor deficiencies in mice. *Hum Gene Ther* 8(16): 1921-1933.

ZUCKERMAN S, EVANS G, O'NEAL L, (1992). Cytokine regulation of macrophage apoE secretion: opposing effects of GM-CSF and TGF-β. *Atherosclerosis* 96(2-3): 203-214.

ZUCKERMAN S, EVANS G, O'NEAL L, (1993). Exogenous glucocorticoids increasemacrophage secretion of apoE by cholesterol independent pathways. *Atherosclerosis* 103(1): 43-54.

ZUCKERMAN S, O'NEAL L, (1994). Endotoxin and GM-CSF mediated down-regulation of macrophage apoE secretion is inhibited by a TNF-  $\alpha$  specific monoclonal antibody. *J Leukoc Biol* 55(6): 743-748.

## **APPENDICES**

## APPENDIX I - MAPS OF DNA MOLECULAR WEIGHT MARKERS



100BP DNA MARKER (NEB)

1KB DNA MARKER (GIBCO)

Full Range		High Range	
KDa		KDa	
250 160 105 75 50 35 30 25 15 10		250 97 66 45 30 20.1 14.3	

APPENDIX II - MAP OF THE RAINBOW PROTEIN SIZE MARKER

ADAPTED FROM THE AMERSHAM-PHARMACIA PRODUCT MANUAL

APPENDIX III - MAP OF THE PCMV-BGAL PLASMID



 $PCMV-\beta Gal$  (Clonetech)- Plasmid used to express  $\beta$ -galactosidase in mammalian cells from the human CMV immediate early gene promoter



APPENDIX IV - MAPS FOR PLASMID VECTORS USED IN THIS STUDY

## APPENDIX V - SCANSITE MOTIF SCAN PROFILE FOR THE HUMAN LXRA AND LXRB PROTEIN SEQUENCES

				A REAL PROPERTY OF
	Phosp	hoserine/thre	conine binding group (pST	_bind)
	14-3-3 N	lode 1	Gene Card Y	WHAZ
Site	Score	Percentile	Sequence	SA
S244	0.5958	3.368 %	<u>QCNKRSFSDQPKVTP</u>	1.594
15		Tyrosine	e kinase group (Y_kin)	Children of State
Literal 1	Fgr K	inase	Gene Card	FGR
Site	Score	Percentile	Sequence	SA
Y370	0.5180	3.490 %	<b>LGLDDAEYALLIAIN</b>	0.432
1	Lek K	inase	Gene Card	LCK
Site	Score	Percentile	Sequence	SA
Y370	0.5871	3.768 %	LGLDDAEYALLIAIN	0.432
C. Control	Sec. Sta	Sre hon	nology 2 group (SH2)	
PI	LCg N-ter	minal SH2		LCGI
Site	Score	Percentile	Sequence	SA
Y399	0.4943	3.899 %	VEALOOPYVEALLSY	1.249
PI	LCg C-ter	minal SH2	Gene Card P	LCG1
Site	Score	Percentile	Sequence	SA
Y399	0.5265	3.505 %	VEALQQPYVEALLSY	1.249
PI	Cg C-ter	minal SH2	Gene Card P	LCG1
Site	Score	Percentile	Sequence	SA
Y370	0.5282	3.611 %	<b>LGLDDAEYALLIAIN</b>	0.432
Fyn SH2			Gene Card	FYN
Site	Score	Percentile	Sequence	SA
Y406	0.5297	2.154 %	<b>YVEALLSYTRIKRPO</b>	0.786
	She S	5H2	Gene Card	SHC1
Site	Score	Percentile	Sequence	SA
Y406	0.5810	4.373 %	<b>YVEALLSYTRIKRPO</b>	0.786
1125	a galler	Sre hon	nology 3 group (SH3)	
And the second	Grb2	STATISTICS CONTRACTOR OF STATISTICS	Gene Card	
Site	Score	Percentile	Sequence	<u>SA</u>
P44	0.4986	0.447 %	<b>PWPGGPDPDVPGTDE</b>	1.754
and the second s	Abl S	SH3	Gene Card A	BL1
Site	Score	Percentile	Sequence	SA
P44	0.5274	1.570 %	<b>PWPGGPDPDVPGTDE</b>	1.754
	Grb2	SH3	Gene Card	RB2
Site	Score	Percentile	Sequence	SA

1

LXRa (Accession NUMBER NP\_005684) - SCANNED FOR ALL MOTIFS AT LOW STRIGENCY
	Grb2	SH3	Gene Card	GRB2
Site	Score	Percentile	Sequence	SA
P14	0.5835	2.135 %	SSLDTPLPGNGPPOP	0.739
	Cortacti	in SH3	Gene Card H	CLS1
Site	Score	Percentile	Sequence	<u>SA</u>
P21	<u>0.5837</u>	0.538 %	PGNGPPOPGAPSSSP	1.467
122000	Grb2	SH3	Gene Card	GRB2
Site	Score	Percentile	Sequence	<u>SA</u>
P18	0.5945	2.586 %	TPLPGNGPPOPGAPS	2.425
121	Src S	SH3	Gene Card	<u>SRC</u>
Site	Score	Percentile	Sequence	<u>SA</u>
P44	<u>0.6089</u>	2.992 %	PWPGGPDPDVPGTDE	1.754
	Crk	SH3	Gene Card	CRK
Site		Percentile	Sequence	
P21	0.6105	3.862 %	PGNGPPQPGAPSSSP	
	Nek 2n	d SH3	Gene Card	ICK1
Site	Score	Percentile	Sequence	SA
P185	0.6114	2.665 %	SQSQSQSPVGPQGSS	0.934
10-31	Crk	SH3	Gene Card	CRK
Site	Score	Percentile	Sequence	<u>SA</u>
P185	0.6277	4.822 %	<u>SQSQSQSPVGPQGSS</u>	0.934
4.4.1	Sre S	SH3	Gene Card	SRC
Site	Score	Percentile	Sequence	<u>SA</u>
P21		4.583 %	PGNGPPQPGAPSSSP	
	Nek 2n		Gene Card N	ICK1
Site		Percentile	Sequence	<u>SA</u>
P77		4.481 %	ERKRKKGPAPKMLGH	
	PLCg		Gene Card P	LCG1
Site		Percentile	Sequence	<u>SA</u>
P251		2.188 %	SDOPKVTPWPLGADP	
1	PLCg		Gene Card P	
Site		Percentile	Sequence	<u>SA</u>
P24	<u>0.6637</u>		<u>GPPQPGAPSSSPTVK</u>	
Cbl-A		protein C-SH.		
~	Score	Percentile	Sequence	SA
Site				
<u>Site</u> P188	0.6907		<u>QSQSPVGPQGSSSSA</u>	
P188	<u>0.6907</u> PLCg	SH3	Gene Card P	LCG1
P188 Site	0.6907 PLCg <u>Score</u>	SH3 Percentile	Gene Card P Sequence	LCG1 SA
P188	0.6907 PLCg Score 0.6953	SH3 Percentile 4.085 %	Gene Card P Sequence QSQSPVGPQGSSSSA	LCG1 <u>SA</u> 0.598
P188 <u>Site</u> P188	0.6907 PLCg Score 0.6953 Itk S	SH3 Percentile 4.085 % SH3	Gene Card P Sequence QSQSPVGPQGSSSSA Gene Card	LCG1 <u>SA</u> 0.598 <u>ITK</u>
P188 Site P188 Site	0.6907 PLCg Score 0.6953 Itk S Score	SH3 Percentile 4.085 % SH3 Percentile	Gene Card P Sequence QSQSPVGPQGSSSSA Gene Card Sequence	LCG1 SA 0.598 ITTK SA
P188 <u>Site</u> P188	0.6907 PLCg Score 0.6953 Itk S	SH3 Percentile 4.085 % SH3 Percentile	Gene Card P Sequence QSQSPVGPQGSSSSA Gene Card	LCG1 SA 0.598 ITTK SA

-	p85 SH3	mode2	Gene Card Pl	K3R1
Site		Percentile	Sequence	SA
P24		3.704 %		
_		protein C-SH.		
Site		Percentile	Sequence	SA
P21	0.7729	1.290 %	PGNGPPQPGAPSSSP	
	Sre S	5H3	Gene Card	
Site	Score	Percentile	Sequence	<u>SA</u>
P185	0.6449	4.583 %	SOSOSOSPVGPOGSS	0.934
	Basophi	lic serine/thre	eonine kinase group (Baso_	ST_kin)
	РКС	mu	Gene Card Pk	RKCM
Site	Score	Percentile	Sequence	<u>SA</u>
T430	0.4419	0.308 %	MKLVSLRTLSSVHSE	0.791
	PKC e	psilon	Gene Card Pl	RKCE
Site	Score	Percentile	Sequence	<u>SA</u>
T29	0.4830	1.979 %	<b>GAPSSSPTVKEEGPE</b>	1.762
	PKC (	delta	Gene Card Pl	RKCD
Site	Score	Percentile	Sequence	<u>SA</u>
S114	<u>0.4886</u>	1.893 %	<b>CKGFFRRSVVRGGAR</b>	1.271
	РКС	delta	Gene Card Pl	RKCD
Site	Score	Percentile	Sequence	<u>SA</u>
Г29	<u>0.5023</u>	2.571 %	<b>GAPSSSPTVKEEGPE</b>	1.762
PK	C alpha/b	eta/gamma	Gene Card Pl	RKCA
Site	Score	Percentile	Sequence	<u>SA</u>
S427	0.5029	1.683 %	RMLMKLVSLRTLSSV	0.438
	РКС	mu	Gene Card PF	RKCM
Site	Score	Percentile	Sequence	<u>SA</u>
Т308	0.5074	1.410 %	<b>IALLKASTIEIMLLE</b>	0.381
PK	C alpha/b	eta/gamma	Gene Card Pl	RKCA
Site	Score	Percentile	Sequence	<u>SA</u>
S114	0.5136	2.064 %	CKGFFRRSVVRGGAR	1.271
	PKC e	psilon	Gene Card Pl	RKCE
Site	Score	Percentile	Sequence	<u>SA</u>
T334	0.5223	4.789 %	ITFLKDFTYSKDDFH	2.008
Calmoo	dulin depe	endent Kinase	2 Gene Card <u>CA</u>	MK2G
Site	Score	Percentile	Sequence	<u>SA</u>
T334		1.251 %	ITFLKDFTYSKDDFH	
			2 Gene Card CA	MK2G
Site		Percentile	Sequence	<u>SA</u>
S244		1.289 %	<u>QCNKRSFSDQPKVTP</u>	
-		Cinase A	Gene Card PR	KACG
		Percentile	Sequence	<u>SA</u>
S114	0.5547	2.140 %	CKGFFRRSVVRGGAR	1.271

	РКС	mu	Gene Card PR	RKCM
Site	Score	Percentile	Sequence	SA
S436	0.5685	4.650 %	<b>RTLSSVHSEQVFALR</b>	0.690
Calmod	lulin depe	ndent Kinase	2 Gene Card CA	MK2G
Site	Score	Percentile	Sequence	SA
S178	0.6050	4.109 %	RKQQQQESQSQSQSP	3.702
Calmod	lulin depe	endent Kinase	2 Gene Card <u>CA</u>	MK2G
Site	Score	Percentile	Sequence	SA
T308	0.6090	4.361 %	IALLKASTIEIMLLE	0.381
	PKC	zeta	Gene Card Pl	RKCZ
Site	Score	Percentile	Sequence	SA
S95	0.6148	4.338 %	<b>RVCGDKASGFHYNVL</b>	0.723
and and	Protein K	linase A	Gene Card PR	KACG
Site	Score	Percentile	Sequence	SA
T334	0.6156	4.455 %	<b>ITFLKDFTYSKDDFH</b>	2.008
	РКС	zeta	Gene Card Pl	RKCZ
Site	Score	Percentile	Sequence	SA
S114	0.6198	4.643 %	CKGFFRRSVVRGGAR	1.271
	РКС	zeta	Gene Card Pl	RKCZ
Site	Score	Percentile	Sequence	SA
T328	0.6198	4.643 %	NHETECITFLKDFTY	0.177
	Protein K	linase A	Gene Card PR	KACG
Site	Score	Percentile	Sequence	SA
T308	0.6204	4.699 %	IALLKASTIEIMLLE	0.381
	Protein K	linase A	Gene Card PR	KACG
Site	Score	Percentile	Sequence	SA
S427	0.6254	4.952 %	RMLMKLVSLRTLSSV	0.438
	Akt Ki	inase	Gene Card	<u>KTI</u>
Site	Score	Percentile	Sequence	SA
S244	0.6469	1.853 %	QCNKRSFSDQPKVTP	1.594
	D	NA damage ki	nase group (DNA_dam_k	in)
	DNA	РК	Gene Card Pl	RKDC
Site	Score	Percentile	Sequence	SA
S8	0.5082	0.602 %	<b>MSSPTTSSLDTPLPG</b>	1.181
Statement Statement	DNA	РК	Gene Card Pl	RKDC
		Percentile	Sequence	SA
<u>Site</u>	Score	rereentine		DIL
<u>Site</u> S209	<u>Score</u> 0.5227	0.758 %	PGGSEAGSQGSGEGE	0.705
<u>Site</u> S209		0.758 %		0.705
	0.5227	0.758 %	PGGSEAGSQGSGEGE	0.705
S209	0.5227 ATM K	0.758 % Kinase <u>Percentile</u>	PGGSEAGSQGSGEGE Gene Card	0.705 <u>ATM</u> <u>SA</u>
S209 <u>Site</u>	0.5227 ATM K Score	0.758 % Kinase <u>Percentile</u> 0.848 %	PGGSEAGSQGSGEGE Gene Card	0.705 <u>XTM</u> <u>SA</u> 3.702
S209 <u>Site</u>	0.5227 ATM K Score 0.5266	0.758 % Kinase <u>Percentile</u> 0.848 % PK	PGGSEAGSQGSGEGE Gene Card Sequence RKOOQQESQSQSQSP	0.705 <u>XTM</u> <u>SA</u> 3.702
S209 <u>Site</u> S178	0.5227 ATM K Score 0.5266 DNA	0.758 % Kinase <u>Percentile</u> 0.848 % PK	PGGSEAGSQGSGEGE Gene Card Sequence RKOQQQESQSQSQSP Gene Card P1	0.705 <u>ATM</u> <u>SA</u> 3.702 <u>RKDC</u>

	DNA	РК	Gene Card PI	RKDC
Site	Score	Percentile	Sequence	SA
S180	0.5464	1.099 %	OOQQESQSQSQSPVG	2.865
and the second	DNA	РК	Gene Card PI	RKDC
Site	Score	Percentile	Sequence	SA
S182	0.5678	1.542 %	<b>OOESQSQSQSPVGPQ</b>	2.558
	ATM k	linase	Gene Card	TM
Site	Score	Percentile	Sequence	SA
S161	0.5696	1.769 %	MREQCVLSEEQIRKK	0.976
Contraction of the	ATM k	linase	Gene Card	<u>TM</u>
Site	Score	Percentile	Sequence	SA
S180	0.5901	2.482 %	<b>QQQQESQSQSQSPVG</b>	2.865
	ATM k	linase	Gene Card	<u>TM</u>
Site	Score	Percentile	Sequence	<u>SA</u>
T49	<u>0.5970</u>	2.761 %	<b>PDPDVPGTDEASSAC</b>	1.479
	ATM k	Cinase	Gene Card	<u>NTM</u>
Site	Score	Percentile	Sequence	SA
S182	<u>0.6097</u>	3.368 %	<u>QQESQSQSQSPVGPQ</u>	2.558
	ATM k	Cinase	Gene Card	<u>TM</u>
Site	Score	Percentile	Sequence	SA
S336	0.6146	3.625 %	<b>FLKDFTYSKDDFHRA</b>	3.873
	ATM k	Cinase	Gene Card	<u>VTM</u>
Site	Score	Percentile	Sequence	<u>SA</u>
S103	0.6245	4.213 %	<u>GFHYNVLSCEGCKGF</u>	0.173
King I.	Acidoph	ilic serine/thr	eonine kinase group (Acid	_ST_kin)
	Casein K	linase 1	Gene Card CS	NKIG2
Site	Score		Sequence	<u>SA</u>
S196	0.3823	0.219 %	<u>QGSSSSASGPGASPG</u>	0.630
	Casein K	linase 1	Gene Card CS	NK1G2
Site	Score	Percentile	Sequence	<u>SA</u>
S8	0.4365	0.912 %	MSSPTTSSLDTPLPG	1.181
	Casein K	linase 1	Gene Card CS	NK1G2
Cita.			~	C 4
Site		Percentile	Sequence	<u>SA</u>
<u>Site</u> S209		Percentile 1.077 %	<u>Sequence</u> <u>PGGSEAGSQGSGEGE</u>	a second s
S209		1.077 %		0.705
S209 Site	<u>0.4431</u> Casein K	1.077 % Cinase 1 Percentile	PGGSEAGSQGSGEGE	0.705
S209	0.4431 Casein K <u>Score</u> 0.4752	1.077 % Cinase 1 Percentile 2.063 %	PGGSEAGSQGSGEGE Gene Card CS	0.705 NK1G2
S209 Site T29	0.4431 Casein K <u>Score</u> 0.4752	1.077 % Cinase 1 Percentile	PGGSEAGSQGSGEGE Gene Card CS Sequence	0.705 NK1G2 SA 1.762
S209 <u>Site</u> T29 <u>Site</u>	0.4431 Casein K Score 0.4752 Casein K Score	1.077 % Cinase 1 Percentile 2.063 % Cinase 1 Percentile	PGGSEAGSQGSGEGE Gene Card <u>CS</u> Sequence GAPSSSPTVKEEGPE	0.705 NK1G2 SA 1.762
S209 Site T29	0.4431 Casein K Score 0.4752 Casein K Score 0.5268	1.077 %         Cinase 1         Percentile         2.063 %         Cinase 1         Percentile         4.939 %	PGGSEAGSQGSGEGE         Gene Card CS         Sequence         GAPSSSPTVKEEGPE         Gene Card CS	0.705 <u>NK1G2</u> <u>SA</u> 1.762 <u>NK1G2</u> <u>SA</u>
S209 <u>Site</u> T29 <u>Site</u>	0.4431 Casein K Score 0.4752 Casein K Score	1.077 %         Cinase 1         Percentile         2.063 %         Cinase 1         Percentile         4.939 %	PGGSEAGSQGSGEGE         Gene Card CS         Sequence         GAPSSSPTVKEEGPE         Gene Card CS         Sequence         Sequence	0.705 NK1G2 SA 1.762 NK1G2 SA 0.302
S209 Site T29 Site S54 Site Site	0.4431 Casein K Score 0.4752 Casein K Score 0.5268 GSK3 F	1.077 %         Cinase 1         Percentile         2.063 %         Cinase 1         Percentile         4.939 %	PGGSEAGSQGSGEGE         Gene Card CS         Sequence         GAPSSSPTVKEEGPE         Gene Card CS         Sequence         PGTDEASSACSTDWV	0.705 NK1G2 SA 1.762 NK1G2 SA 0.302
S209 Site T29 Site S54	0.4431 Casein K Score 0.4752 Casein K Score 0.5268 GSK3 F	1.077 %Cinase 1Percentile2.063 %Cinase 1Percentile4.939 %CinasePercentile	PGGSEAGSQGSGEGE         Gene Card CS         Sequence         GAPSSSPTVKEEGPE         Gene Card CS         Sequence         PGTDEASSACSTDWV         Gene Card C	0.705 <u>NK1G2</u> <u>SA</u> 1.762 <u>NK1G2</u> <u>SA</u> 0.302 <u>SK3A</u>

	Casein K	linase 2	Gene Card CS	NK2B
Site	Score	Percentile	Sequence	SA
S53	0.5651	2.439 %	VPGTDEASSACSTDW	0.390
	GSK3 k	Cinase	Gene Card G	SK3A
Site	Score	Percentile	Sequence	<u>SA</u>
S201	0.5843	1.343 %	<b>SASGPGASPGGSEAG</b>	0.465
Section of the sectio	Casein K	linase 2	Gene Card CS	NK2B
Site	Score	Percentile	Sequence	SA
T29	0.5877	3.449 %	<b>GAPSSSPTVKEEGPE</b>	1.762
	GSK3 k	Kinase	Gene Card G	SK3A
Site	Score	Percentile	Sequence	SA
S205	0.5927	1.549 %	<b>PGASPGGSEAGSQGS</b>	0.521
	Casein K	linase 2	Gene Card CS	NK2B
Site	Score	Percentile	Sequence	SA
T49	0.6051	4.553 %	<b>PDPDVPGTDEASSAC</b>	1.479
	GSK3 F	Kinase	Gene Card G	SK3A
Site	Score	Percentile	Sequence	SA
S178	0.6169	2.329 %	RKQQQQESQSQSQSP	3.702
	GSK3 k	Kinase	Gene Card G	SK3A
Site	Score	Percentile	Sequence	<u>SA</u>
S180	0.6335	2.990 %	<b>OOOOESOSOSOSPVG</b>	2.865
Contraction of the	GSK3 k	Kinase	Gene Card G	SK3A
		Ellineide	Cone curd o	
		Percentile		
Site	Score	Percentile		SA
<u>Site</u> S454	<u>Score</u> 0.6491	Percentile 3.876 %	Sequence	<u>SA</u> 0.267
<u>Site</u> S454 Pi	<u>Score</u> <u>0.6491</u> roline-dep	Percentile 3.876 % endent serine	Sequence KKLPPLLSEIWDVHE	<u>SA</u> 0.267 Pro_ST_kin)
Site S454 Pi	<u>Score</u> <u>0.6491</u> roline-dep p38 M	Percentile 3.876 % endent serine	Sequence KKLPPLLSEIWDVHE /threonine kinase group (F Gene Card MA	<u>SA</u> 0.267 Pro_ST_kin)
Site S454 Pi Site	<u>Score</u> <u>0.6491</u> roline-dep p38 M	Percentile 3.876 % endent serine APK	Sequence KKLPPLLSEIWDVHE /threonine kinase group (F Gene Card MA	<u>SA</u> 0.267 Pro_ST_kin) APK14 <u>SA</u>
<u>Site</u> S454 Pi	<u>Score</u> 0.6491 roline-dep p38 M <u>Score</u>	Percentile 3.876 % endent serine APK Percentile 4.766 %	Sequence KKLPPLLSEIWDVHE /threonine kinase group (F Gene Card MA Sequence	<u>SA</u> 0.267 Pro_ST_kin) APK14 <u>SA</u> 0.810
<u>Site</u> S454 Pi <u>Site</u> S184	<u>Score</u> 0.6491 roline-dep p38 M <u>Score</u> 0.4495	Percentile 3.876 % endent serine APK Percentile 4.766 %	Sequence KKLPPLLSEIWDVHE /threonine kinase group (P Gene Card MA Sequence ESOSOSOSPVGPQGS	<u>SA</u> 0.267 Pro_ST_kin) APK14 <u>SA</u> 0.810
Site S454 Pt Site S184 Site	<u>Score</u> 0.6491 roline-dep p38 M <u>Score</u> 0.4495 Cdk5 K	Percentile 3.876 % endent serine APK Percentile 4.766 % (inase Percentile	Sequence KKLPPLLSEIWDVHE /threonine kinase group (P Gene Card M Sequence ESQSQSQSPVGPQGS Gene Card (	<u>SA</u> 0.267 Pro_ST_kin) APK14 <u>SA</u> 0.810 CDK5
Site S454 Pt Site S184 Site	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score	Percentile 3.876 % endent serine APK Percentile 4.766 % Cinase Percentile 1.058 %	Sequence KKLPPLLSEIWDVHE /threonine kinase group (F Gene Card M/ Sequence ESQSQSQSPVGPQGS Gene Card C Sequence	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         CDK5         SA         0.914
Site S454 Pr Site S184 Site S27	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119	Percentile 3.876 % endent serine APK Percentile 4.766 % Cinase Percentile 1.058 %	Sequence         KKLPPLLSEIWDVHE         /threonine kinase group (F         Gene Card M/         Sequence         ESQSQSQSPVGPQGS         Gene Card C         Sequence         QPGAPSSSPTVKEEG	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         CDK5         SA         0.914
Site S454 Pi Site S184 Site S27 Site	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 K Score 0.5119 Erk1 K	Percentile 3.876 % endent serine APK Percentile 4.766 % (inase Percentile 1.058 % (inase Percentile	Sequence KKLPPLLSEIWDVHE /threonine kinase group (F Gene Card M Sequence ESQSQSQSPVGPQGS Gene Card C Sequence QPGAPSSSPTVKEEG Gene Card E	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         CDK5         SA         0.914         PHB2
Site S454 Pi Site S184 Site S27 Site	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119 Erk1 K Score	Percentile 3.876 % endent serine APK Percentile 4.766 % Cinase Percentile 1.058 % Cinase Percentile 0.828 %	Sequence         KKLPPLLSEIWDVHE         /threonine kinase group (F         Gene Card M         Sequence         ESOSOSOSPVGPOGS         Gene Card C         Sequence         OPGAPSSSPTVKEEG         Gene Card E         Sequence         OPGAPSSSPTVKEEG         Gene Card E	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         CDK5         SA         0.914         PHB2         SA         0.898
Site S454 Site S184 Site S27 Site T11	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119 Erk1 k Score 0.5330	Percentile 3.876 % endent serine APK Percentile 4.766 % Cinase Percentile 1.058 % Cinase Percentile 0.828 %	Sequence         KKLPPLLSEIWDVHE         /threonine kinase group (F         Gene Card M         Sequence         ESQSQSQSPVGPQGS         Gene Card C         Sequence         OPGAPSSSPTVKEEG         Gene Card E         Sequence         OPGAPSSSPTVKEEG         Gene Card E         PTTSSLDTPLPGNGP	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         CDK5         SA         0.914         PHB2         SA         0.898
Site S454 Pr Site S184 Site S27 Site T11 Site	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119 Erk1 K Score 0.5330 Cdc2 k	Percentile 3.876 % endent serine APK Percentile 4.766 % Cinase Percentile 1.058 % Cinase Percentile 0.828 % Cinase Percentile	Sequence KKLPPLLSEIWDVHE /threonine kinase group (F Gene Card M Sequence ESQSQSQSPVGPQGS Gene Card C Sequence QPGAPSSSPTVKEEG Gene Card F Sequence PTTSSLDTPLPGNGP Gene Card C	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         CDK5         SA         0.914         PHB2         SA         0.898         CDC2
Site S454 Pi Site	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119 Erk1 k Score 0.5330 Cdc2 k Score	Percentile 3.876 % endent serine APK Percentile 4.766 % (inase Percentile 1.058 % (inase Percentile 0.828 % (inase Percentile 0.927 %	Sequence   KKLPPLLSEIWDVHE     (threonine kinase group (F)   Gene Card M   Sequence   ESQSQSQSPVGPQGS   Gene Card C   Sequence   OPGAPSSSPTVKEEG   Gene Card E   Sequence   PTTSSLDTPLPGNGP   Gene Card C   Sequence	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         CDK5         SA         0.914         PHB2         SA         0.898         CDC2         SA         0.914
Site S454 Pr Site S184 Site S27 Site T11 Site	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 K Score 0.5119 Erk1 K Score 0.5330 Cdc2 K Score 0.5379	Percentile 3.876 % endent serine APK Percentile 4.766 % (inase Percentile 1.058 % (inase Percentile 0.828 % (inase Percentile 0.927 %	Sequence         KKLPPLLSEIWDVHE         /threonine kinase group (F         Gene Card M         Sequence         ESQSQSQSPVGPQGS         Gene Card C         Sequence         QPGAPSSSPTVKEEG         Gene Card F         Sequence         PTTSSLDTPLPGNGP         Gene Card C         Sequence         PTTSSLDTPLPGNGP         Gene Card C         Sequence         PTTSSLDTPLPGNGP         Gene Card C         Sequence         PTTSSLDTPLPGNGP	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         CDK5         SA         0.914         PHB2         SA         0.898         CDC2         SA         0.914
Site S454 S454 Site S184 Site S27 Site S27 Site S27 Site S27	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119 Erk1 k Score 0.5330 Cdc2 k Score 0.5379 Cdc2 k	Percentile 3.876 % endent serine APK Percentile 4.766 % (inase Percentile 1.058 % (inase Percentile 0.828 % (inase Percentile 0.927 % (inase Percentile	Sequence         KKLPPLLSEIWDVHE         /threonine kinase group (F         Gene Card M         Sequence         ESQSQSQSPVGPQGS         Gene Card G         Sequence         OPGAPSSSPTVKEEG         Sequence         PTTSSLDTPLPGNGP         Gene Card G         Sequence         OPGAPSSSPTVKEEG         OPGAPSSSPTVKEEG         Gene Card G         Sequence         OPGAPSSSPTVKEEG	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         DK5         SA         0.914         PHB2         SA         0.898         CDC2         SA         0.914
Site S454 S454 Site S184 Site S27 Site S27 Site S27 Site S27	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119 Erk1 k Score 0.5330 Cdc2 k Score 0.5379 Cdc2 k	Percentile 3.876 % endent serine APK Percentile 4.766 % Cinase Percentile 1.058 % Cinase Percentile 0.828 % Cinase Percentile 0.927 % Cinase Percentile 1.703 %	Sequence KKLPPLLSEIWDVHE /threonine kinase group (F Gene Card M Sequence ESQSQSQSPVGPQGS ESQSQSQSPVGPQGS Gene Card C Sequence OPGAPSSSPTVKEEG Sequence OPGAPSSSPTVKEEG Gene Card C Sequence OPGAPSSSPTVKEEG	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         DK5         SA         0.914         PHB2         SA         0.898         CDC2         SA         0.914         CDC2         SA         0.914
Site S454 Pr Site S184 Site S27 Site T11 Site S27	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119 Erk1 k Score 0.5330 Cdc2 k Score 0.5379 Cdc2 k Score 0.5379	Percentile 3.876 % endent serine APK Percentile 4.766 % Cinase Percentile 1.058 % Cinase Percentile 0.828 % Cinase Percentile 0.927 % Cinase Percentile 1.703 %	Sequence         KKLPPLLSEIWDVHE         /threonine kinase group (F         Gene Card M         Sequence         ESQSQSQSPVGPQGS         Gene Card G         Sequence         OPGAPSSSPTVKEEG         Sequence         PTTSSLDTPLPGNGP         Gene Card G         Sequence         OPGAPSSSPTVKEEG         Gene Card G         Sequence         OPGAPSSSPTVKEEG         Gene Card G         Sequence         OPGAPSSSPTVKEEG         Sequence          KK	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         DK5         SA         0.914         PHB2         SA         0.898         CDC2         SA         0.914         CDC2         SA         0.914
Site S454 Pr Site S184 Site S27 Site S27 Site S27 Site S27 Site S27	Score 0.6491 roline-dep p38 M Score 0.4495 Cdk5 k Score 0.5119 Erk1 k Score 0.5330 Cdc2 k Score 0.5379 Cdc2 k Score 0.5379 Cdc2 k Score 0.5379 Cdc2 k	Percentile 3.876 % endent serine APK Percentile 4.766 % Cinase Percentile 1.058 % Cinase Percentile 0.828 % Cinase Percentile 0.927 % Cinase Percentile 1.703 %	Sequence         KKLPPLLSEIWDVHE         /threonine kinase group (F         Gene Card M         Sequence         ESQSQSQSPVGPQGS         Gene Card C         Sequence         QPGAPSSSPTVKEEG         Gene Card E         Sequence         PTTSSLDTPLPGNGP         Gene Card C         Sequence         OPGAPSSSPTVKEEG         Gene Card C         Sequence         PTTSSLDTPLPGNGP         Gene Card C         Sequence         OPGAPSSSPTVKEEG         Sequence         SASGPGASPGGSEAG         Gene Card E	SA         0.267         Pro_ST_kin)         APK14         SA         0.810         DK5         SA         0.914         PHB2         SA         0.898         CDC2         SA         0.914         CDC2         SA         0.914         CDC2         SA         0.914         CDC2         SA         0.914         CDC2         SA         0.465         PHB2         SA         0.465

	Cdk5 k	linase	Gene Card C	DK5
Site		Percentile	Sequence	SA
<u>Site</u> S201		3.238 %	SASGPGASPGGSEAG	No. of Concession, Name of
5201				
<b>C</b> 1	Erk1 K		Gene Card El	
Site		Percentile	Sequence	SA
T250		2.649 %	FSDQPKVTPWPLGAD	a second second second second second
and the state	Cdk5 k		Gene Card C	
Site		Percentile	Sequence	<u>SA</u>
T11	0.6312	3.910 %	PTTSSLDTPLPGNGP	0.898
	Cdk5 k	Cinase	Gene Card C	DK5
Site	Score	Percentile	Sequence	SA
S184	0.6369	4.108 %	<b>ESOSOSOSPVGPOGS</b>	0.810
	Erk1 K	linase	Gene Card El	PHB2
Site	Score	Percentile	Sequence	SA
S27	0.6490	3.732 %	<b>QPGAPSSSPTVKEEG</b>	0.914
THE OWNER WHEN	Erk1 K		Gene Card El	
Site		Percentile	Sequence	SA
S196		3.908 %	OGSSSSASGPGASPG	and the second
5170	Erk1 K		Gene Card El	
Site		Percentile	Sequence	SA
S201		4.274 %	SASGPGASPGGSEAG	A REAL PROPERTY OF A REAL PROPER
5201	Cdc2 K		Gene Card C	
and the second second	Cuca n	inase	Uche Calu C	DCA
Sito	Saara	Porcontilo	Coquence	SA.
Site		Percentile	Sequence	<u>SA</u>
<u>Site</u> T11	0.6650	4.049 %	PTTSSLDTPLPGNGP	0.898
T11	<u>0.6650</u> Cdc2 K	4.049 % Cinase	PTTSSLDTPLPGNGP Gene Card	0.898 DC2
T11 Site	0.6650 Cdc2 K Score	4.049 % Cinase Percentile	PTTSSLDTPLPGNGP Gene Card C Sequence	0.898 DC2 SA
T11	0.6650 Cdc2 K Score 0.6701	4.049 % Kinase Percentile 4.227 %	PTTSSLDTPLPGNGP Gene Card C Sequence ESOSOSOSPVGPOGS	0.898 DC2 SA 0.810
T11 <u>Site</u> S184	0.6650 Cdc2 K Score 0.6701 Cdc2 K	4.049 % Cinase Percentile 4.227 % Cinase	PTTSSLDTPLPGNGP Gene Card C Sequence ESQSQSQSPVGPQGS Gene Card C	0.898 DC2 SA 0.810 DC2
T11 <u>Site</u> S184 <u>Site</u>	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score	4.049 % Cinase Percentile 4.227 % Cinase Percentile	PTTSSLDTPLPGNGP         Gene Card         Sequence         ESOSOSOSPVGPQGS         Gene Card         Sequence         Sequence	0.898 DC2 SA 0.810 DC2 SA
T11 <u>Site</u> S184	0.6650 Cdc2 K Score 0.6701 Cdc2 K	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 %	PTTSSLDTPLPGNGP         Gene Card O         Sequence         ESQSQSQSPVGPQGS         Gene Card O         Sequence         Sequence         FSDQPKVTPWPLGAD	0.898 DC2 SA 0.810 DC2 SA
T11 Site S184 Site	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi	PTTSSLDTPLPGNGP Gene Card C Sequence ESOSOSOSPVGPQGS Gene Card C Sequence FSDQPKVTPWPLGAD ng site group (Kin_bind)	0.898 DC2 SA 0.810 DC2 SA 1.234
T11 Site S184 Site T250	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score	4.049 % Kinase Percentile 4.227 % Kinase Percentile 4.701 % Kinase bindi Kinase bindi	PTTSSLDTPLPGNGP         Gene Card O         Sequence         ESQSQSQSPVGPQGS         Gene Card O         Sequence         Sequence         FSDQPKVTPWPLGAD	0.898 DC2 SA 0.810 DC2 SA 1.234
T11 Site S184 Site	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi	PTTSSLDTPLPGNGP Gene Card C Sequence ESOSOSOSPVGPQGS Gene Card C Sequence FSDQPKVTPWPLGAD ng site group (Kin_bind)	0.898 DC2 SA 0.810 DC2 SA 1.234
T11 Site S184 Site T250	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B	4.049 % Kinase Percentile 4.227 % Kinase Percentile 4.701 % Kinase bindi Kinase bindi	PTTSSLDTPLPGNGP         Gene Card G         Sequence         ESQSQSQSPVGPQGS         Gene Card G         Sequence         FSDQPKVTPWPLGAD         ng site group (Kin_bind)         Gene Card P	0.898 DC2 <u>SA</u> 0.810 DC2 <u>SA</u> 1.234
T11 Site S184 Site T250 Site Site	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score	4.049 % Kinase Percentile 4.227 % Kinase Percentile 4.701 % Kinase bindi Sinding Percentile 0.602 %	PTTSSLDTPLPGNGP Gene Card C Sequence ESQSQSQSPVGPQGS Gene Card C Sequence FSDQPKVTPWPLGAD ng site group (Kin_bind) Gene Card P Sequence	0.898 DC2 SA 0.810 DC2 SA 1.234 DPK1 SA 2.008
T11 Site S184 Site T250 Site Site	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score 0.5942	4.049 % Kinase Percentile 4.227 % Kinase Percentile 4.701 % Kinase bindi Sinding Percentile 0.602 %	PTTSSLDTPLPGNGP Gene Card C Sequence ESOSOSOSPVGPOGS Gene Card C Sequence FSDQPKVTPWPLGAD ng site group (Kin_bind) Gene Card P Sequence ITFLKDFTYSKDDFH	0.898 DC2 SA 0.810 DC2 SA 1.234 DPK1 SA 2.008
T11 Site S184 Site T250 Site T334	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score 0.5942 PDK1 B	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi Sinding Percentile 0.602 % Cinding Percentile	PTTSSLDTPLPGNGP         Gene Card G         Sequence         ESQSQSQSPVGPQGS         Gene Card G         Sequence         FSDQPKVTPWPLGAD         ng site group (Kin_bind)         Gene Card P         Sequence         ITFLKDFTYSKDDFH         Gene Card P	0.898 DC2 SA 0.810 DC2 SA 1.234 DPK1 SA 2.008 DPK1 SA
T11 Site S184 Site T250 Site T334 Site	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score 0.5942 PDK1 B	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi Cinding Percentile 0.602 % Cinding Percentile 0.982 %	PTTSSLDTPLPGNGP         Gene Card G         Sequence         ESOSOSOSPVGPQGS         Gene Card G         Sequence         FSDQPKVTPWPLGAD         ng site group (Kin_bind)         Gene Card P         Sequence         ITFLKDFTYSKDDFH         Gene Card P         Sequence         Sequence         Sequence         Sequence         Sequence         Gene Card P         Sequence	0.898 DC2 SA 0.810 DC2 SA 1.234 DPK1 SA 2.008 DPK1 SA 2.950
T11 Site S184 Site T250 Site T334 Site	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score 0.5942 PDK1 B Score 0.6249	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi Cinding Percentile 0.602 % Cinding Percentile 0.982 %	PTTSSLDTPLPGNGP         Gene Card G         Sequence         ESQSQSQSPVGPQGS         Gene Card G         Sequence         FSDQPKVTPWPLGAD         Ig site group (Kin_bind)         Gene Card P         Sequence         ITFLKDFTYSKDDFH         Gene Card P         Sequence         ITFLKDFTYSKDDFH         Gene Card P	0.898 DC2 SA 0.810 DC2 SA 1.234 DPK1 SA 2.008 DPK1 SA 2.950
T11 Site S184 Site T250 Site T334 Site D339	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score 0.5942 PDK1 B Score 0.5942 PDK1 B	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi Cinding Percentile 0.602 % Cinding Percentile 0.982 % Comain Percentile	PTTSSLDTPLPGNGP         Gene Card G         Sequence         ESOSOSOSPVGPQGS         Gene Card G         Sequence         FSDQPKVTPWPLGAD         Ig site group (Kin_bind)         Gene Card P         Sequence         ITFLKDFTYSKDDFH         Gene Card P         Sequence         DFTYSKDDFHRAGLQ         Gene Card M	0.898 DC2 SA 0.810 DC2 SA 1.234 DPK1 SA 2.008 DPK1 SA 2.950 APK1 SA
T11 Site S184 Site T250 Site T334 Site D339 Site	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score 0.5942 PDK1 B Score 0.6249 Erk D-d Score	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi Cinding Percentile 0.602 % Cinding Percentile 0.982 % Comain Percentile 0.416 %	PTTSSLDTPLPGNGP Gene Card C Sequence ESQSQSQSPVGPQGS Gene Card C Sequence FSDQPKVTPWPLGAD Ig site group (Kin_bind) Gene Card P Sequence ITFLKDFTYSKDDFH Gene Card P Sequence DFTYSKDDFHRAGLQ Gene Card M	0.898 DC2 <u>SA</u> 0.810 DC2 <u>SA</u> 1.234 DPK1 <u>SA</u> 2.008 DPK1 <u>SA</u> 2.950 APK1 <u>SA</u> 0.607
T11 Site S184 Site T250 Site T334 Site D339 Site V426	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score 0.5942 PDK1 B Score 0.6249 Erk D-d Score 0.6299	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi Cinding Percentile 0.602 % Cinding Percentile 0.982 % Comain Percentile 0.416 % Cinding	PTTSSLDTPLPGNGP         Gene Card G         Sequence         ESOSOSOSPVGPOGS         Gene Card G         Sequence         FSDQPKVTPWPLGAD         g site group (Kin_bind)         Gene Card P         Sequence         ITFLKDFTYSKDDFH         Gene Card P         Sequence         DFTYSKDDFHRAGLQ         Gene Card M         Sequence         PRMLMKLVSLRTLSS         Gene Card P	0.898 DC2 SA 0.810 DC2 SA 1.234 DPK1 SA 2.008 DPK1 SA 2.950 APK1 SA 0.607 DPK1
T11 Site S184 Site T250 Site T334 Site D339 Site	0.6650 Cdc2 K Score 0.6701 Cdc2 K Score 0.6829 PDK1 B Score 0.5942 PDK1 B Score 0.6249 Erk D-d Score 0.6299 PDK1 B	4.049 % Cinase Percentile 4.227 % Cinase Percentile 4.701 % Kinase bindi Cinding Percentile 0.602 % Cinding Percentile 0.982 % Comain Percentile 0.416 %	PTTSSLDTPLPGNGP         Gene Card G         Sequence         ESQSQSQSPVGPQGS         Gene Card G         Sequence         FSDQPKVTPWPLGAD         Ig site group (Kin_bind)         Gene Card P         Sequence         ITFLKDFTYSKDDFH         Gene Card P         Sequence         DFTYSKDDFHRAGLQ         Gene Card M         Sequence         PRMLMKLVSLRTLSS	0.898 DC2 <u>SA</u> 0.810 DC2 <u>SA</u> 1.234 DPK1 <u>SA</u> 2.008 DPK1 <u>SA</u> 2.950 APK1 <u>SA</u> 0.607

and and	- Carrier 1	Section 1.		
	PDK1 B	inding	Gene Card P	DPK1
Site	Score	Percentile	Sequence	<u>SA</u>
E355	0.6566	1.309 %	<b>EFINPIFEFSRAMRR</b>	0.548
The state	Erk D-d	omain	Gene Card M	IAPK1
Site	Score	Percentile	Sequence	<u>SA</u>
I311	0.6568	0.634 %	<b>LKASTIEIMLLETAR</b>	0.131
No.	Erk D-d	lomain	Gene Card M	IAPK1
Site	Score	Percentile	Sequence	<u>SA</u>
L274	0.6996	1.218 %	<b>RFAHFTELAIISVQE</b>	0.234
1 Section	Erk D-d	lomain	Gene Card M	IAPK1
Site	Score	Percentile	Sequence	SA
L293	<u>0.7171</u>	1.580 %	AKQVPGFLQLGREDC	0.229
and a	Erk D-d	lomain	Gene Card M	IAPK1
Site	Score	Percentile	Sequence	SA
L449	0.7304	1.883 %	<u>LRLQDKKLPPLLSEI</u>	1.490
	Erk D-d	lomain	Gene Card M	IAPK1
Site	Score	Percentile	Sequence	<u>SA</u>
L254	0.6461	0.533 %	PKVTPWPLGADPQSR	0.513

## LXR $\beta$ (Accession number NP\_005684) - Scanned for all motifs at low strigency

	Phosp	hoserine/thre	eonine binding group (pST	_bind)		
	14-3-3 N	Iode 1	Gene Card	Gene Card YWHAZ		
Site	Score	Percentile	Sequence	SA		
S198	<u>0.3941</u>	0.371 %	<b>SLPPRASSPPQILPQ</b>	1.722		
-	14-3-3 N	lode 1	Gene Card Y	WHAZ		
Site	Score	Percentile	Sequence	<u>SA</u>		
<b>S191</b>	<u>0.5218</u>	1.637 %	<b>EEQAHATSLPPRASS</b>	0.883		
14-3-3 Mode 1			Gene Card Y	WHAZ		
Site	Score	Percentile	Sequence	<u>SA</u>		
S230	<u>0.5551</u>	2.277 %	<u>QCNRRSFSDRLRVTP</u>	0.961		
		Sre hor	nology 2 group (SH2)			
	p85 S	SH2	Gene Card P	IK3R1		
Site	Score	Percentile	Sequence	<u>SA</u>		
Y145	<u>0.7008</u>	4.296 %	GHCPMDTYMRRKCO	E 3.285		
S. Sela	NAN STAT	Src hor	nology 3 group (SH3)			
	Grb2	SH3	Gene Card	Gene Card <u>GRB2</u>		
Site	Score	Percentile	Sequence	<u>SA</u>		
P10	<u>0.4948</u>	0.416 %	<u>LWLGAPVPDIPPDSA</u>	0.736		
	Nek 2n	d SH3	Gene Card N	ICK1		
<u>Site</u>	Score	Percentile	Sequence	<u>SA</u>		

-				
A succession with	Crk S		Gene Card C	
Site		Percentile	Sequence	<u>SA</u>
P10		1.597 %	LWLGAPVPDIPPDSA	and the second second
	Sre S		Gene Card	
Site		Percentile	Sequence	
P10		1.385 %	LWLGAPVPDIPPDSA	
and the second s	Amphiphy		Gene Card A	
Site		Percentile	Sequence	
239		1.034 %	<u>RLRVTPWPMAPDPHS</u>	
and an	Abl S		Gene Card A	BL1
Site	Score	Percentile	Sequence	<u>SA</u>
P193	<u>0.5963</u>	3.923 %	<u>OAHATSLPPRASSPP</u>	1.198
	Abl S	SH3	Gene Card A	BL1
Site	Score	Percentile	Sequence	<u>SA</u>
P10	0.5982	4.021 %	LWLGAPVPDIPPDSA	0.736
	Intersecti	n SH3A	Gene Card I	TSN
Site	Score	Percentile	Sequence	<u>SA</u>
213	0.6119	0.312 %	GAPVPDIPPDSAVEL	1.435
1000	Src S	H3	Gene Card	SRC
Site	Score	Percentile	Sequence	<u>SA</u>
193	0.6243	3.619 %	<b>QAHATSLPPRASSPP</b>	1.198
	Grb2	SH3	Gene Card G	RB2
<u>Site</u>			Gene Card G	
	Score	SH3	<u>Sequence</u> <u>OAHATSLPPRASSPP</u>	<u>SA</u> 1.198
P193	Score	SH3 Percentile 4.044 %	Sequence	<u>SA</u> 1.198
P193	<u>Score</u> 0.6255 Nck 2nd	SH3 Percentile 4.044 %	<u>Sequence</u> <u>OAHATSLPPRASSPP</u>	<u>SA</u> 1.198 CK1
9193 Site	Score 0.6255 Nck 2nd Score 0.6434	SH3Percentile4.044 %1 SH3Percentile4.325 %	Sequence OAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR	<u>SA</u> 1.198 CK1 <u>SA</u> 3.405
P193 Site	Score 0.6255 Nck 2nd Score	SH3Percentile4.044 %1 SH3Percentile4.325 %	Sequence OAHATSLPPRASSPP Gene Card N Sequence	<u>SA</u> 1.198 CK1 <u>SA</u> 3.405
Dite Dite Dite Dite	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score	SH3       Percentile       4.044 %       1 SH3       Percentile       4.325 %       SH3       Percentile	Sequence OAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR	<u>SA</u> 1.198 CK1 <u>SA</u> 3.405
P193 Site P73 Site	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509	SH3         Percentile         4.044 %         1 SH3         Percentile         4.325 %         SH3         Percentile         1.942 %	Sequence QAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR Gene Card P	<u>SA</u> 1.198 CK1 <u>SA</u> 3.405 LCG1 <u>SA</u>
P193 Site P73 Site	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score	SH3         Percentile         4.044 %         1 SH3         Percentile         4.325 %         SH3         Percentile         1.942 %	Sequence OAHATSLPPRASSPP Gene Card N Sequence LLTRAEPPSEPTEIR Gene Card PI Sequence	SA         1.198         CK1         SA         3.405         LCG1         SA         1.245
2193 <u>Site</u> 273 <u>Site</u> 2242	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509	SH3         Percentile         4.044 %         1 SH3         Percentile         4.325 %         SH3         Percentile         1.942 %	Sequence QAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA	SA         1.198         CK1         SA         3.405         LCG1         SA         1.245
P193 Site P73 Site P242 Site	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti	SH3Percentile4.044 %1 SH3Percentile4.325 %SH3Percentile1.942 %n SH3	Sequence QAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR Gene Card PI Sequence VTPWPMAPDPHSREA Gene Card H Sequence	SA         1.198         CK1         SA         3.405         LCG1         SA         1.245         CLS1
P193       Site       P73       Site       P242       Site       P193	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658	SH3Percentile4.044 %I SH3Percentile4.325 %SH3Percentile1.942 %In SH3PercentilePercentile	Sequence OAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA Gene Card H Sequence OAHATSLPPRASSPP	SA         1.198         CK1         SA         3.405         LCG1         SA         1.245         CLS1         SA         1.198
P193 Site P73 Site P242 Site P193 Cbl-As	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658	SH3Percentile4.044 %1 SH3Percentile4.325 %SH3Percentile1.942 %n SH3Percentile2.001 %	Sequence OAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA Gene Card H Sequence OAHATSLPPRASSPP	SA         1.198         CK1         SA         3.405         LCG1         SA         1.245         CLS1         SA         1.198
2193 Site 273 Site 2242 Site 2193 Cbl-As Site	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658	SH3 Percentile 4.044 % 1 SH3 Percentile 4.325 % SH3 Percentile 1.942 % n SH3 Percentile 2.001 % percentile 2.001 %	Sequence QAHATSLPPRASSPP Gene Card N Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA Gene Card H Sequence QAHATSLPPRASSPP Gene Card	SA         1.198         CK1         SA         3.405         LCG1         SA         1.245         CLS1         SA         1.198         N/A         SA
P193 Site P73 Site P242 Site P193 Cbl-As Site	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658 ssociated Score	SH3 Percentile 4.044 % SH3 Percentile 4.325 % SH3 Percentile 1.942 % n SH3 Percentile 2.001 % Percentile 0.360 %	Sequence OAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA Gene Card H Sequence OAHATSLPPRASSPP Gene Card	SA         1.198         CK1         SA         3.405         CG1         SA         1.245         CLS1         SA         1.198         N/A         SA         1.198
P193 Site P73 Site P242 Site P193 Cb1-As Site P193	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658 ssociated Score 0.6658	SH3 Percentile 4.044 % SH3 Percentile 4.325 % SH3 Percentile 1.942 % n SH3 Percentile 2.001 % Percentile 0.360 %	Sequence QAHATSLPPRASSPP Gene Card N Sequence LLTRAEPPSEPTEIR Gene Card Pl Sequence VTPWPMAPDPHSREA Gene Card H Sequence QAHATSLPPRASSPP ABAR SEPP	SA         1.198         CK1         SA         3.405         CG1         SA         1.245         CLS1         SA         1.198         N/A         SA         1.198
P193 Site P73 Site P242 Site P193 Cbl-As Site P193 Site P193	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658 ssociated Score 0.6743 Ittk S	SH3 Percentile 4.044 % 1 SH3 Percentile 4.325 % SH3 Percentile 1.942 % in SH3 Percentile 2.001 % Percentile 0.360 % H3	Sequence QAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA Gene Card M Sequence QAHATSLPPRASSPP Gene Card	SA         1.198         CK1         SA         3.405         CCG1         SA         1.245         CLS1         SA         1.198         N/A         SA         1.198         M/A         SA         1.198         SA         1.198         SA         1.198         SA         SA         SA
P193 Site P73 Site P242 Site P193 Cbl-As Site P193 Site P193	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658 ssociated Score 0.6743 Ittk S	SH3         Percentile         4.044 %         4 SH3         Percentile         4.325 %         SH3         Percentile         1.942 %         n SH3         Percentile         2.001 %         protein C-SH3         Percentile         0.360 %         H3         Percentile         2.064 %	Sequence QAHATSLPPRASSPP Gene Card N Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA Gene Card M Sequence QAHATSLPPRASSPP Gene Card Sequence QAHATSLPPRASSPP	SA         1.198         CK1         SA         3.405         CCG1         SA         1.245         CLS1         SA         1.198         N/A         SA         1.198         M/A         SA         1.198         CLS1         SA         1.198         SA         1.198         SA         1.198         CLS1
Site P73 Site P242 Site P193	Score 0.6255 Nek 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658 sociated Score 0.6658 sociated Score 0.6743 Itk S	SH3         Percentile         4.044 %         4 SH3         Percentile         4.325 %         SH3         Percentile         1.942 %         n SH3         Percentile         2.001 %         protein C-SH3         Percentile         0.360 %         H3         Percentile         2.064 %	Sequence QAHATSLPPRASSPP Gene Card M Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA Gene Card H Sequence QAHATSLPPRASSPP Gene Card Sequence QAHATSLPPRASSPP Gene Card	SA         1.198         CK1         SA         3.405         CCG1         SA         1.245         CLS1         SA         1.198         N/A         SA         1.198         N/A         SA         1.198         CLS1         SA         1.198         SA         1.198         SA         2.191
P193         Site         P73         Site         P242         Site         P193         Cbl-As         Site         P193	Score 0.6255 Nck 2nd Score 0.6434 PLCg Score 0.6509 Cortacti Score 0.6658 Sociated Score 0.6743 Itk S Score 0.6743 Itk S Score 0.6922	SH3         Percentile         4.044 %         1 SH3         Percentile         4.325 %         SH3         Percentile         1.942 %         n SH3         Percentile         2.001 %         Portein C-SH3         Percentile         0.360 %         H3         Percentile         2.064 %         SH3	Sequence QAHATSLPPRASSPP Gene Card N Sequence LLTRAEPPSEPTEIR Gene Card P Sequence VTPWPMAPDPHSREA Gene Card H Sequence OAHATSLPPRASSPP Gene Card Sequence QAHATSLPPRASSPP Gene Card Sequence PQKRKKGPAPKMLGN	SA 1.198 CK1 SA 3.405 CCS1 SA 1.245 CLS1 SA 1.198 N/A SA 1.198 TK SA 2.191 CCG1

	p85 SH3	mode1	Gene Card PI	K3R1
Site		Percentile	Sequence	SA
P10		3.315 %	LWLGAPVPDIPPDSA	the second second second second
P10				
C'	PLCg		Gene Card Pl	
Site		Percentile	Sequence	<u>SA</u>
P14		4.659 %	APVPDIPPDSAVELW	
Same a	Cortact	in SH3	Gene Card H	CLS1
Site	Score	Percentile	Sequence	<u>SA</u>
P73	<u>0.7084</u>	3.711 %	<b>LLTRAEPPSEPTEIR</b>	3.405
11000	Cortact	in SH3	Gene Card H	CLS1
Site	Score	Percentile	Sequence	SA
P10	0.7213	4.437 %	LWLGAPVPDIPPDSA	0.736
-	Intersecti		Gene Card I	TSN
Site		Percentile	Sequence	SA
	0.7245		OAHATSL <b>P</b> PRASSPP	and the second se
	p85 SH3		Gene Card PI	
Site		Percentile	Sequence	SA
				and a second sec
-	0.7282		RLRVTPWPMAPDPHS	
		protein C-SH3	and the second se	
Site		Percentile	Sequence	<u>SA</u>
_	<u>0.8671</u>		VTPWPMAPDPHSREA	
Cbl-A	ssociated	protein C-SH3	Gene Card	N/A
Site	Score	Percentile	Sequence	<u>SA</u>
P13		and the second se	<u>GAPVPDIPPDSAVEL</u>	1.435
P13	0.8708	3.828 %	<u>GAPVPDIPPDSAVEL</u> onine kinase group (Baso_	
P13	0.8708	3.828 % ilic serine/three		ST_kin)
P13	0.8708 Basophi PKC	3.828 % ilic serine/three	onine kinase group (Baso_ Gene Card PR	ST_kin)
	0.8708 Basophi PKC	3.828 % ilic serine/three mu <u>Percentile</u>	onine kinase group (Baso_ Gene Card PR	ST_kin) RKCM
<u>Site</u> T416	0.8708 Basophi PKC <u>Score</u> 0.4419	3.828 % ilic serine/three mu <u>Percentile</u> 0.308 %	onine kinase group (Baso_ Gene Card PR Sequence MKLVSLRTLSSVHSE	ST_kin) RKCM SA 0.791
Site T416 PK	0.8708 Basophi PKC Score 0.4419 C alpha/b	3.828 % ilic serine/three mu <u>Percentile</u> 0.308 % beta/gamma	onine kinase group (Baso Gene Card PR Sequence MKLVSLRTLSSVHSE Gene Card PH	ST_kin)           RKCM           SA           0.791           RKCA
Site T416 PK Site	0.8708 Basophi PKC Score 0.4419 C alpha/b Score	3.828 % ilic serine/three mu <u>Percentile</u> 0.308 % beta/gamma <u>Percentile</u>	onine kinase group (Baso Gene Card PR Sequence MKLVSLRTLSSVHSE Gene Card PR Sequence	ST_kin)           RKCM           SA           0.791           RKCA           SA
Site T416 PK	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732	3.828 % ilic serine/three mu <u>Percentile</u> 0.308 % ecta/gamma <u>Percentile</u> 0.927 %	onine kinase group (Baso Gene Card PR Sequence MKLVSLRTLSSVHSE Gene Card PH Sequence CKGFFRRSVIKGAHY	ST_kin)         SKCM         SA         0.791         RKCA         SA         1.226
Site T416 PK Site S125	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC 0	3.828 % ilic serine/three mu Percentile 0.308 % ecta/gamma Percentile 0.927 %	onine kinase group (Baso Gene Card PR Sequence MKLVSLRTLSSVHSE Gene Card PH Sequence CKGFFRRSVIKGAHY Gene Card PH	ST_kin)           SA           0.791           RKCA           SA           1.226           RKCD
Site T416 PK Site S125 Site	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC 0 Score	3.828 % ilic serine/three mu Percentile 0.308 % ecta/gamma Percentile 0.927 % delta Percentile	onine kinase group (Baso_Gene Card PR         Sequence         MKLVSLRTLSSVHSE         Gene Card PR         Sequence         CKGFFRRSVIKGAHY         Gene Card PR         Sequence         CKGFFRRSVIKGAHY         Gene Card PR         Sequence         CKGFFRRSVIKGAHY         Gene Card PR	ST_kin)         SA         0.791         RKCA         SA         1.226         RKCD         SA
Site T416 PK Site S125	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC 0 Score 0.4801	3.828 % ilic serine/three mu Percentile 0.308 % eta/gamma Percentile 0.927 % delta Percentile 1.585 %	onine kinase group (Baso         Gene Card PR         Sequence         MKLVSLRTLSSVHSE         Gene Card PR         Sequence         CKGFFRRSVIKGAHY         Gene Card PR         Sequence         FALLIAISIFSADRP	ST_kin)         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177
Site T416 PK Site S125 Site S363	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC 0 Score 0.4801 PKC e	3.828 % ilic serine/three mu Percentile 0.308 % ecta/gamma Percentile 0.927 % delta Percentile 1.585 %	onine kinase group (Baso_Gene Card PR         Sequence         MKLVSLRTLSSVHSE         Gene Card PR         Sequence         CKGFFRRSVIKGAHY         Gene Card PR         Sequence         FALLIAISIFSADRP         Gene Card PR	ST_kin)         KCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         RKCE
Site           T416           PK           Site           S125           Site           S363           Site           Site	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC of Score 0.4801 PKC of Score	3.828 % ilic serine/three mu Percentile 0.308 % beta/gamma Percentile 0.927 % delta Percentile 1.585 % psilon Percentile	onine kinase group (Baso         Gene Card PR         Sequence         MKLVSLRTLSSVHSE         Gene Card PR         Sequence         CKGFFRRSVIKGAHY         Gene Card PR         Sequence         FALLIAISIFSADRP         Gene Card PR         Sequence         FALLIAISIFSADRP         Gene Card PR         Sequence	ST_kin)         KCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         RKCE         SA
Site         T416         PK         Site         S125         Site         S363         Site         S363         Site         S320	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC of Score 0.4801 PKC of Score 0.4993	3.828 % ilic serine/three mu Percentile 0.308 % reta/gamma Percentile 0.927 % delta Percentile 1.585 % psilon Percentile 3.027 %	Image: sequence         MKLVSLRTLSSVHSE         Gene Card PH         Sequence         CKGFFRRSVIKGAHY         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         ITFLKDFSYNREDFA	ST_kin)         KCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         RKCE         SA         2.276
Site         T416         PK         Site         S125         Site         S363         Site         S363         Site         S320	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC of Score 0.4801 PKC of Score 0.4993	3.828 % ilic serine/three mu Percentile 0.308 % beta/gamma Percentile 0.927 % delta Percentile 1.585 % psilon Percentile	onine kinase group (Baso         Gene Card PR         Sequence         MKLVSLRTLSSVHSE         Gene Card PR         Sequence         CKGFFRRSVIKGAHY         Gene Card PR         Sequence         FALLIAISIFSADRP         Gene Card PR         Sequence         FALLIAISIFSADRP         Gene Card PR         Sequence	ST_kin)         KCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         RKCE         SA         2.276
Site         T416         PK         Site         S125         Site         S363         Site         S363         Site         S320	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC of Score 0.4801 PKC of Score 0.4993	3.828 % ilic serine/three mu Percentile 0.308 % reta/gamma Percentile 0.927 % delta Percentile 1.585 % psilon Percentile 3.027 %	Image: sequence         MKLVSLRTLSSVHSE         Gene Card PH         Sequence         CKGFFRRSVIKGAHY         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         ITFLKDFSYNREDFA	ST_kin)         KCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         RKCE         SA         2.276
Site         T416         PK         Site         S125         Site         S363         Site         S363         Site         S320         PK	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC of Score 0.4801 PKC of Score 0.4993 C alpha/b	3.828 % ilic serine/three mu Percentile 0.308 % beta/gamma Percentile 0.927 % delta Percentile 1.585 % psilon Percentile 3.027 %	Image: sequence         MKLVSLRTLSSVHSE         Gene Card PH         Sequence         CKGFFRRSVIKGAHY         Gene Card PH         Sequence         CKGFFRRSVIKGAHY         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         ITFLKDFSYNREDFA         Gene Card PH	ST_kin)         KCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         KCE         SA         2.276         RKCA         SA
Site         T416         PK         Site         S125         Site         S363         Site         S363         Site         S320         PK         Site         Site         S320         PK         Site	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC of Score 0.4801 PKC of Score 0.4993 C alpha/b	3.828 % ilic serine/three mu Percentile 0.308 % beta/gamma Percentile 0.927 % delta Percentile 1.585 % psilon Percentile 3.027 % beta/gamma Percentile 1.683 %	Image: sequence         MKLVSLRTLSSVHSE         Gene Card PF         MKLVSLRTLSSVHSE         Gene Card PF         Sequence         CKGFFRRSVIKGAHY         Gene Card PF         Sequence         FALLIAISIFSADRP         Gene Card PF         Sequence         FALLIAISIFSADRP         Gene Card PF         Sequence         ITFLKDFSYNREDFA         Gene Card PF         Sequence         ISequence         ISequence         Sequence         ISequence         ISequence         Sequence         ISequence	ST_kin)         SKCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         RKCE         SA         0.1276         RKCA         SA         0.438
Site         T416         PK         Site         S125         Site         S363         Site         S363         Site         S320         PK         Site         Site         S320         PK         Site	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC e Score 0.4801 PKC e Score 0.4993 C alpha/b Score 0.4993	3.828 % ilic serine/three mu Percentile 0.308 % beta/gamma Percentile 0.927 % delta Percentile 1.585 % psilon Percentile 3.027 % beta/gamma Percentile 1.683 %	Image: sequence         MKLVSLRTLSSVHSE         Gene Card PH         Sequence         CKGFFRRSVIKGAHY         Gene Card PH         Sequence         CKGFFRRSVIKGAHY         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         FALLIAISIFSADRP         Gene Card PH         Sequence         ITFLKDFSYNREDFA         Gene Card PH         Sequence         IMKLVSLRTLSSV	ST_kin)         KCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         RKCE         SA         2.276         RKCA         SA         0.438
Site         T416         PK         Site         S125         Site         S363         Site         S320         PK         Site         S320         PK         Site         S413	0.8708 Basophi PKC Score 0.4419 C alpha/b Score 0.4732 PKC e Score 0.4801 PKC e Score 0.4993 C alpha/b Score 0.4993 C alpha/b	3.828 % ilic serine/three mu Percentile 0.308 % reta/gamma Percentile 0.927 % delta Percentile 1.585 % psilon Percentile 3.027 % reta/gamma Percentile 1.683 %	Image: sequence         MKLVSLRTLSSVHSE         Gene Card PI         MKLVSLRTLSSVHSE         Gene Card PI         Sequence         CKGFFRRSVIKGAHY         Gene Card PI         Sequence         FALLIAISIFSADRP         Gene Card PI         Sequence         FALLIAISIFSADRP         Gene Card PI         Sequence         ITFLKDFSYNREDFA         Gene Card PI         Sequence         IMKLVSLRTLSSV         Gene Card PI	ST_kin)         KCM         SA         0.791         RKCA         SA         1.226         RKCD         SA         0.177         RKCE         SA         0.177         RKCE         SA         0.138         RKCE

	Protein K	inase A	Gene Card PR	KACG
Site		Percentile	Sequence	SA
	0.5357	the second s	CKGFFRRSVIKGAHY	
-		eta/gamma	Gene Card Ph	_
Site		Percentile	Sequence	
	0.5525		FALLIAISIFSADRP	
5505	<u>0.3325</u> PKC		Gene Card PH	
Site		Percentile	Sequence	SA
S125		1.449 %	CKGFFRRSVIKGAHY	COLUMN SEC.
5125		n.449 70	Gene Card PR	
Site		Percentile	Sequence	
		Contraction of the local sector of the local s		
S293	<u>0.5651</u> PKC	4.365 %	QIALLKTSAIEVMLL	
			Gene Card <u>PR</u> Sequence	
Site		Percentile		100000
-	0.5685		RTLSSVHSEQVFALR	
and the second second		Cinase A	Gene Card PR	
Site		Percentile		<u>SA</u>
and the second	0.6096		<u>QCNRRSFSDRLRVTP</u>	
			2 Gene Card CA	
Site		Percentile		
			LPGFLQLSREDQIAL	
Calmoo	dulin depe	indent Kinase	2 Gene Card <u>CA</u>	MK2G
~.	~	-	~	~ .
Site		Percentile	Sequence	
S230	0.6131	4.635 %	<u>OCNRRSFSDRLRVTP</u>	0.961
S230	<u>0.6131</u> Protein K	4.635 % Cinase A	OCNRRSFSDRLRVTP Gene Card PR	0.961 KACG
S230 Site	0.6131 Protein K Score	4.635 % (inase A Percentile	OCNRRSFSDRLRVTP Gene Card PR Sequence	0.961 KACG SA
S230	0.6131 Protein K Score 0.6134	4.635 % <b>Cinase A</b> <u>Percentile</u> 4.348 %	OCNRRSFSDRLRVTP Gene Card PR Sequence OOOCNRRSFSDRLRV	0.961 <b>KACG</b> <u>SA</u> 2.283
S230 Site S228	0.6131 Protein K Score 0.6134 Protein K	4.635 %Cinase APercentile4.348 %Cinase A	OCNRRSFSDRLRVTP Gene Card PR Sequence OOOCNRRSFSDRLRV Gene Card PR	0.961 KACG <u>SA</u> 2.283 KACG
S230 Site S228 Site	0.6131 Protein K Score 0.6134 Protein K Score	4.635 % Cinase A Percentile 4.348 % Cinase A Percentile	OCNRRSFSDRLRVTP Gene Card PR Sequence OOOCNRRSFSDRLRV Gene Card PR Sequence	0.961 KACG SA 2.283 KACG SA
S230 Site S228	0.6131 Protein K Score 0.6134 Protein K Score 0.6184	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %	OCNRRSFSDRLRVTP         Gene Card PR         Sequence         OOOCNRRSFSDRLRV         Gene Card PR         Sequence         Sequence         ITFLKDFSYNREDFA	0.961 KACG SA 2.283 KACG SA 2.276
S230 <u>Site</u> S228 <u>Site</u> S320	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase A	OCNRRSFSDRLRVTP         Gene Card PR         Sequence         OOOCNRRSFSDRLRV         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR	0.961 KACG SA 2.283 KACG SA 2.276 KACG
S230 Site S228 Site S320 Site Site	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile	OCNRRSFSDRLRVTP         Gene Card PR         Sequence         OOOCNRRSFSDRLRV         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA
S230 <u>Site</u> S228 <u>Site</u> S320	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %	OCNRRSFSDRLRVTP         Gene Card PR         Sequence         OQOCNRRSFSDRLRV         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         IMMKLVSLRTLSSV	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438
S230 Site S228 Site S320 Site S413	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %inase	OCNRRSFSDRLRVTP         Gene Card PR         Sequence         OOOCNRRSFSDRLRV         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         IMLMKLVSLRTLSSV         Gene Card A	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1
S230 Site S228 Site S320 Site S413 Site Site	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki Score	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %inasePercentile	OCNRRSFSDRLRVTP         Gene Card PR         Sequence         OQOCNRRSFSDRLRV         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Sequence         RMLMKLVSLRTLSSV         Gene Card A         Sequence	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1 SA
S230 Site S228 Site S320 Site S413	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki Score 0.6784	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %inasePercentile2.682 %	OCNRRSFSDRLRVTP         Gene Card PR         Sequence         OOOCNRRSFSDRLRV         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         RMLMKLVSLRTLSSV         Gene Card A         Sequence         SLPPRASSPPOILPQ	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1 SA 1.722
S230 Site S228 Site S320 Site S413 Site S413 Site S198	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki Score 0.6784 Akt Ki	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %inasePercentile2.682 %inase	OCNRRSFSDRLRVTP         Gene Card PR         Sequence         OQOCNRRSFSDRLRV         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Gene Card PR         Sequence         ITFLKDFSYNREDFA         Sequence         RMLMKLVSLRTLSSV         Gene Card A         Sequence	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1 SA 1.722 KT1
S230 Site S228 Site S320 Site S413 Site S198 Site S198	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki Score 0.6784 Akt Ki	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %inasePercentile2.682 %inasePercentile2.682 %	OCNRRSFSDRLRVTPGene Card PRSequenceOOOCNRRSFSDRLRVGene Card PRSequenceITFLKDFSYNREDFAGene Card PRSequenceRMLMKLVSLRTLSSVGene Card ASequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSLPPRASSPPOILPOGene Card ASequenceSequence	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1 SA 1.722 KT1 SA
S230 Site S228 Site S320 Site S413 Site S413 Site S198	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki Score 0.6784 Akt Ki Score 0.6784	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %inasePercentile2.682 %inasePercentile3.020 %	OCNRRSFSDRLRVTPGene Card PRSequenceOQOCNRRSFSDRLRVGene Card PRSequenceITFLKDFSYNREDFAGene Card PRSequenceRMLMKLVSLRTLSSVGene Card ASequenceSLPPRASSPPQILPQGene Card ASequenceSLPPRASSPPQILPQGene Card ASequenceSLPRASSPRAIPGene Card ASequenceSLPRASSPRAIPGene Card ASequenceOCNRRSFSDRLRVTP	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1 SA 1.722 KT1 SA 0.961
S230 Site S228 Site S320 Site S413 Site S198 Site S198	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki Score 0.6784 Akt Ki	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %inasePercentile2.682 %inasePercentile3.020 %	OCNRRSFSDRLRVTPGene Card PRSequenceOOOCNRRSFSDRLRVGene Card PRSequenceITFLKDFSYNREDFAGene Card PRSequenceRMLMKLVSLRTLSSVGene Card ASequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSequenceSLPPRASSPPOILPOGene Card ASequenceSequence	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1 SA 1.722 KT1 SA 0.961
S230 Site S228 Site S320 Site S413 Site S198 Site Site	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki Score 0.6784 Akt Ki Score 0.6784	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %inasePercentile2.682 %inasePercentile3.020 %	OCNRRSFSDRLRVTPGene Card PRSequenceOQOCNRRSFSDRLRVGene Card PRSequenceITFLKDFSYNREDFAGene Card PRSequenceRMLMKLVSLRTLSSVGene Card ASequenceSLPPRASSPPQILPQGene Card ASequenceSLPPRASSPPQILPQGene Card ASequenceSLPRASSPRAIPGene Card ASequenceSLPRASSPRAIPGene Card ASequenceOCNRRSFSDRLRVTP	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1 SA 1.722 KT1 SA 0.961 LK2 SA
S230 Site S228 Site S320 Site S413 Site S198 Site S198 Site S230	0.6131 Protein K Score 0.6134 Protein K Score 0.6184 Protein K Score 0.6254 Akt Ki Score 0.6784 Akt Ki Score 0.6784	4.635 %Cinase APercentile4.348 %Cinase APercentile4.594 %Cinase APercentile4.952 %cinasePercentile2.682 %cinasePercentile3.020 %CinasePercentile3.020 %CinasePercentile3.020 %	OCNRRSFSDRLRVTPGene Card PRSequenceOOOCNRRSFSDRLRVGene Card PRSequenceITFLKDFSYNREDFAGene Card PRSequenceRMLMKLVSLRTLSSVGene Card ASequenceSLPPRASSPPOILPOGene Card ASequenceSLPPRASSPPOILPOGene Card ASequenceOCNRRSFSDRLRVTPGene Card ASequenceSequenceSLPPRASSPPOILPOGene Card ASequenceOCNRRSFSDRLRVTPGene Card A	0.961 KACG SA 2.283 KACG SA 2.276 KACG SA 0.438 KT1 SA 1.722 KT1 SA 0.961 LK2

12.12				
	Clk2 K		Gene Card C	
Site		Percentile	Sequence	SA
S418	and the second second second		LVSLRTLSSVHSEQV	
1. 1. 1. 1. 1.	and the second second		kinase group (DNA_dam_ki	the second se
and the second	ATM k		Gene Card	
Site		Percentile		<u>SA</u>
S207		0.247 %	POILPOLSPEQLGMI	
1999	ATM k	NAME OF TAXABLE PARTY.	Gene Card	
Site	and the second	Percentile	Sequence	<u>SA</u>
S170		1.404 %	MREECVLSEEQIRLK	0.976
	ATM k	linase	Gene Card	ATM
Site	Score	Percentile	Sequence	<u>SA</u>
S282	0.5814	2.154 %	<b>LPGFLOLSREDQIAL</b>	2.485
-	ATM k	(inase	Gene Card	TM
Site	Score	Percentile	Sequence	SA
S114	0.6245	4.213 %	<b>GFHYNVLSCEGCKGF</b>	0.173
	ATM k	Cinase	Gene Card	ATM
Site	Score	Percentile	Sequence	SA
S30	0.6345	4.876 %	<b>PGAQDASSQAQGGSS</b>	1.260
	DNA	PK	Gene Card PI	RKDC
Site	Score	Percentile	Sequence	SA
S30	0.6467	4.798 %	PGAODASSOAQGGSS	1.260
	DNA	РК	Gene Card PI	RKDC
Site	Score	Percentile	Sequence	SA
S312	0.6467	4.798 %	RYNPGSESITFLKDF	0.624
Sec. 19	Acidoph	ilic serine/th	reonine kinase group (Acid	ST kin)
Total.	and the second se	inase 2		
Site	Score	Percentile	Sequence	SA
S74		1.440 %	LTRAEPPSEPTEIRP	
134	GSK3 I	Kinase	Gene Card G	and the second se
Site		Percentile	Sequence	SA
T68		1.004 %	AEPTALLTRAEPPSE	
1	GSK3 H		Gene Card G	
Site		Percentile	Sequence	SA
S198	0.6103		SLPPRASSPPQILPQ	CONTRACTOR OF THE PARTY OF
HARSING ST	GSK3 H		Gene Card G	
Site		Percentile	Sequence	SA
S191	A CONTRACTOR OF A	2.927 %	EEQAHATSLPPRASS	Contraction of the second second
	GSK3 H		Gene Card G	
Site	Score		Sequence	SA
				2.838
S74	0.6423	3.461 %	<b>LTRAEPPSEPTEIRP</b>	2.050

Louis and	GSK3 I	Kinase	Gene Card G	<u>SK3Å</u>
Site		Percentile	Sequence	<u>SA</u>
S440	<u>0.6491</u>	3.876 %	KKLPPLLSEIWDVHE	0.267
Р	roline-dep	oendent serine	/threonine kinase group (F	Pro_ST_kin)
	Erk1 k	linase	Gene Card E	PHB2
Site	Score	Percentile	Sequence	<u>SA</u>
S198	0.5495	1.051 %	<b>SLPPRASSPPQILPQ</b>	1.722
a summer	Cdk5 F	Cinase	Gene Card	CDK5
Site	Score	Percentile	Sequence	<u>SA</u>
S198	0.6163	3.412 %	<b>SLPPRASSPPQILPQ</b>	1.722
Ser.	Erk1 k	linase	Gene Card E	PHB2
Site	Score	Percentile	Sequence	SA
T236	0.6284	2.944 %	<b>FSDRLRVTPWPMAPD</b>	1.209
Sec. 21	Cde2 k	Cinase	Gene Card	DC2
Site	Score	Percentile	Sequence	SA
S198	0.6514	3.592 %	<b>SLPPRASSPPOILPO</b>	1.722
	Cdk5 k	Kinase	Gene Card	DK5
Site	Score	Percentile	Sequence	SA
T236	0.6555	4.789 %	FSDRLRVTPWPMAPD	1.209
	Cde2 k	Cinase	Gene Card	CDC2
Site	Score	Percentile	Sequence	SA
T236	0.6871	4.854 %	FSDRLRVTPWPMAPD	1.209
		Kinase bindi	ing site group (Kin_bind)	THE BALL
	Erk D-d	lomain	Gene Card M	APKI
Site	Score	Percentile	Sequence	<u>SA</u>
L233	0.5212	0.068 %	RRSFSDRLRVTPWPM	1.297
	PDK1 B	linding	Gene Card P	DPK1
Site	Score	Percentile	Sequence	SA
E341	0.6206	0.929 %	EFINPIFEFSRAMNE	0.548
	PDK1 B	linding	Gene Card P	DPK1
Site	Score	Percentile	Sequence	SA
D325	0.6249	0.982 %	DFSYNREDFAKAGLQ	2.271
	PDK1 B	linding	Gene Card P	DPK1
Site	Score	Percentile	Sequence	SA
S320	0.6270	1.007 %	ITFLKDFSYNREDFA	2.276
Erk D-domain			Gene Card M	
Site	Score	Percentile	Sequence	SA
V412	0.6299		PRMLMKLVSLRTLSS	0.607
	PDK1 B		Gene Card P	
Site	Score	Percentile	Sequence	SA
T258	0.6374	1.139 %	<b>OORFAHFTELAIVSV</b>	0.562
	0.0011		YYMM TELMTOT	0.002

	Erk D-d	lomain	Gene Card MAPK1		
Site	Score	Percentile	Sequence	SA	
V297	0.6487	0.555 %	<b>LKTSAIEVMLLETSR</b>	0.139	
Erk D-domain			Gene Card MAPK1		
Site	Score	Percentile	Sequence	<u>SA</u>	
L260	0.6996	1.218 %	<b>RFAHFTELAIVSVQE</b>	0.248	
NY S	Erk D-d	lomain	Gene Card MAPK1		
Site	Score	Percentile	Sequence	SA	
L279	0.7043	1.314 %	AKQLPGFLQLSREDQ	0.310	
	Erk D-domain		Gene Card MAPK1		
Site	Score	Percentile	Sequence	<u>SA</u>	
1202	0.7223	1.678 %	RASSPPOILPQLSPE	0.950	
Erk D-domain			Gene Card MAPK1		
Site	Score	Percentile	Sequence	<u>SA</u>	
V386	0.7565	2.652 %	ERLOHTYVEALHAYV	0.555	
Erk D-domain			Gene Card M	APK1	
Site	Score	Percentile	Sequence	<u>SA</u>	
L435	<u>0.7304</u>	1.883 %	<b>LRLODKKLPPLLSEI</b>	1.490	
	S 1	Lipid bin	iding group (Lip_bind)		
PIP3-binding PH			Gene Card PIP3-E		
Site	Score	Percentile	Sequence	<u>SA</u>	
F121	0.7529	1.070 %	<b>SCEGCKGFFRRSVIK</b>	1.305	

## **PUBLICATIONS ARISING AS A RESULT OF THIS THESIS**

GREENOW K., PEARCE N., RAMJI D.P., (2004). The key role of apolipoprotein E in atherosclerosis. Journal of Molecular Medicine (in revision).

GREENOW K., PEARCE N., RAMJI D.P., (2004). LXRs, from basic biochemistry to atherosclerosis. (in preparation).

GREENOW K., PEARCE N., RAMJI D.P., (2004). A critical role for the JNK/SAPK signal transduction pathway in the activation of apolipoprotein E gene expression in macrophages. *(in preparation)*.

GREENOW K., PEARCE N., RAMJI D.P., (2004). A critical role for the phosphoinositide-3-kinase signal transduction pathway in the activation of apolipoprotein E gene expression. *Atherosclerosis* (*Supplements*) 5(1): 25.

RAMJI D.P., FOKA P., IRVINE S., HUGHES T., ROGERS S., SINGH N., GREENOW K., EVANS S., HARVEY E., ALI S., (2004). Signal transduction pathways and transcriptional control mechanisms involved in the cytokine-mediated regulation of key genes in macrophages implicated in foam cell formation and atherosclerosis. *Atherosclerosis (Supplements)* 5(1): 29.

